# SECURITIES AND EXCHANGE COMMISSION

# FORM 6-K

Current report of foreign issuer pursuant to Rules 13a-16 and 15d-16 Amendments

Filing Date: 2022-05-06 | Period of Report: 2022-05-05 SEC Accession No. 0001213900-22-024348

(HTML Version on secdatabase.com)

# **FILER**

# SAI.TECH Global Corp

CIK:1847075| IRS No.: 981584130 | Fiscal Year End: 1231 Type: 6-K | Act: 34 | File No.: 001-40368 | Film No.: 22898615 SIC: 7374 Computer processing & data preparation Mailing Address 437 MADISON AVENUE 27TH FLOOR NEW YORK NY 10022 Business Address 437 MADISON AVENUE 27TH FLOOR NEW YORK NY 10022 (732) 910-9692

# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# FORM 6-K

# REPORT OF FOREIGN PRIVATE ISSUER PURSUANT TO SECTION 13A-16 OR 15D-16 UNDER THE SECURITIES EXCHANGE ACT OF 1934

For the month of May 2022

Commission File Number: 001-40368

# **SAI.TECH Global Corporation**

(Exact Name of Registrant as Specified in Its Charter)

#01-05 Pearl's Hill Terrace Singapore, 168976 (Address of principal executive offices)

Indicate by check mark whether the registrant files or will file annual reports under cover of Form 20-F or Form 40-F.

Form 20-F  $\blacksquare$  Form 40-F  $\square$ 

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(1):

Indicate by check mark if the registrant is submitting the Form 6-K in paper as permitted by Regulation S-T Rule 101(b)(7):

# CONTENTS

# **Business Combination**

On April 29, 2022 (the "<u>Closing Date</u>"), SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability (the "<u>Company</u>" or "SAI") consummated the previously announced business combination pursuant to the Business Combination Agreement, dated September 27, 2021, (as amended, the "<u>Business Combination Agreement</u>"), by and among the Company, TradeUP Global Corporation, a Cayman Islands exempted company ("<u>TradeUP</u>"), and TGC Merger Sub, a Cayman Islands exempted company ("<u>Merger Sub</u>"). Pursuant to the Business Combination Agreement, Merger Sub merged with and into the Company, with the Company surviving the merger as a wholly owned subsidiary of TradeUP. In connection with the business combination, TradeUP changed its corporate name to "SAI.TECH Global Corporation" ("<u>New SAI</u>").

On April 29, 2022, when Merger Sub merged with and into the Company (the "Merger"), said Merger resulted in the following:

. the separate corporate existence of Merger Sub ceased, and the Company continued as the surviving entity, and as a whollyowned subsidiary of TradeUP, which was subsequently renamed SAI.TECH Global Corporation;

each outstanding unit of TradeUP immediately prior to the effective time of the Business Combination (the "Merger Effective Time") was automatically separated, and the holder thereof was deemed to hold one share of TradeUP Class A

- ii. ordinary share, par value \$0.0001 per share (each, a "<u>TradeUP Class A Ordinary Share</u>"), and one-half of one redeemable warrant of TradeUP (each, a "<u>TradeUP Warrant</u>"). Each whole TradeUP Warrant entitled the holder to purchase one TradeUP Class A Ordinary Share at the price of \$11.50 per share.
- each issued and outstanding TradeUP Class A Ordinary Share, immediately prior to the Merger Effective Time, was iii. automatically cancelled and exchanged for the right of the holder thereof to receive one Class A ordinary share of New SAI, par value \$0.0001 per share (each, a "<u>New SAI Class A Ordinary Share</u>");

each issued and outstanding Class B ordinary share of TradeUP, par value \$0.0001 per share (each, a "<u>TradeUP Class B</u> <u>Ordinary Share</u>"), immediately prior to the Merger Effective Time, was transferred and exchanged for the right of the holder

iv. thereof to receive one TradeUP Class A Ordinary Share, each of which subsequently was transferred and exchanged for a New SAI Class A Ordinary Share;

each issued and outstanding public warrant of TradeUP, immediately prior to the Merger Effective Time, were automatically converted into the right of the holder thereof to receive one warrant of New SAI entitling the holder thereof to purchase one New SAI Class A Ordinary Share on substantially the same terms and conditions with respect to the TradeUP Warrants (each, a "New SAI Warrant");

following completion of the Merger, holders of 3,492,031 TradeUP Class A Ordinary Shares, remaining after the redemption of 2,071,735 shares, received 3,492,031 New SAI Class A Ordinary Shares, holders of 272,247 TradeUP Class B Ordinary Shares received 272,247 New SAI Class A Ordinary Shares, and holders of 2,244,493 TradeUP Warrants

2

Immediately prior to the Merger Effective Date, (1) each Class A ordinary share, par value \$0.0001 per share, of the Company outstanding as of immediately prior to the Merger Effective Date was converted into a right to receive a number of New SAI Class A Ordinary Shares, par value \$0.0001 per share, determined on the basis of an exchange ratio derived from an implied equity value for the Company of \$188.0 million and \$10.00 per share (the "exchange ratio"); and (2) each Class B ordinary share, par value \$0.0001 per share, of the Company outstanding as of immediately prior to the Merger Effective Date was converted into a right to receive a number of New SAI Class B Ordinary Shares, par value \$0.0001 per share, determined on the basis of the company outstanding as of immediately prior to the Merger Effective Date was converted into a right to receive a number of New SAI Class B Ordinary Shares, par value \$0.0001 per share, determined on the basis of the exchange ratio. As of the Closing Date the exchange ratio was approximately 0.13376.

• the exchange ratio. As of the Closing Date, the exchange ratio was approximately 0.13376.

received New SAI Warrants to purchase 2,244,493 New SAI Class A Ordinary Shares.

Following the Merger, TradeUP acquired all of the issued and outstanding shares of the Company from the shareholders of the Company in exchange for the payment, issuance and delivery of the New SAI Class A Ordinary Shares and New SAI Class B Ordinary Shares to the shareholders of the Company (the "<u>Share Acquisition</u>" and together with the Merger, the "<u>Business</u> <u>Combination</u>"). As a result of the Share Acquisition, the Company became a wholly owned subsidiary of New SAI, and shareholders of the Company became shareholders of New SAI.

• Following the completion of the Business Combination, there are an aggregate of 12,933,653 New SAI Class A Ordinary Shares issued and outstanding.

The New SAI Class A Ordinary Shares and the New SAI Warrants began trading on The Nasdaq Stock Market LLC on May 2, 2022, under the symbols "SAI" and "SAITW," respectively.

The foregoing description of the Business Combination is qualified in its entirety by reference to the Business Combination Agreement, a copy of which is filed as Exhibit 2.1 to this Report of a Foreign Private Issuer on Form 6-K and is incorporated by reference herein.

On April 29, 2022, TradeUP adopted the Amended and Restated Memorandum and Articles of Association, a copy of which is filed as Exhibit 1.1 to this Report of a Foreign Private Issuer on Form 6-K and is incorporated by reference herein.

On the Closing Date, the Company and TradeUP issued a press release announcing the closing of the Business Combination. A copy of the press release is being furnished as Exhibit 99.1 to this Report of a Foreign Private Issuer on Form 6-K.

#### Ancillary Agreements

#### Support Agreements

In connection with the execution of the Business Combination Agreement, TradeUP, the TradeUP initial shareholders, SAI, the SAI Founder and certain shareholders of SAI entered into transaction support agreements, pursuant to which, among other things, each agreed to vote at any meeting of TradeUP (in the case of the TradeUP initial shareholders) or SAI (in the case of SAI shareholders) in favor of the Business Combination Agreement and the consummation of the transactions contemplated thereby.

#### TradeUP Support Agreement

In connection with the execution of the Business Combination Agreement, TradeUP, the TradeUP initial shareholders and SAI entered into the TradeUP Support Agreement. Under the TradeUP Support Agreement, the TradeUP initial shareholders, among other things, agreed to: (1) appear at any meeting of TradeUP shareholders for purposes of determining a quorum; (2) vote their respective TradeUP Class A Ordinary Shares in favor of the Business Combination Proposal, the Articles Amendment Proposal and the Share Issuance Proposal; (3) not to transfer their respective TradeUP Class A Ordinary Shares prior to the termination of the TradeUP Support Agreement; (4) waive anti-dilution rights under the current memorandum and articles of association (as holders of TradeUP Class B Ordinary Shares); (5) waive dissenter's rights under Section 238 of the Companies Act; and (6) not to redeem any of their respective TradeUP Class A Ordinary Shares.

#### SAI Support Agreement

In connection with the execution of the Business Combination Agreement, TradeUP, TradeUP Global Sponsor LLC (the "Sponsor"), SAI and certain SAI shareholders entered into the SAI Support Agreement. Under the SAI Support Agreement, the SAI shareholders, among other things, agreed to: (1) appear at any meeting of SAI shareholders for purposes of determining a quorum; (2) vote their respective SAI ordinary shares in favor of the Business Combination Agreement and the transactions contemplated thereby, including the merger; (3) not to transfer their respective TradeUP Class A Ordinary Shares prior to the termination of the SAI Support Agreement; and (4) waive dissenter's rights under Section 238 of the Companies Act.

#### New Registration Rights Agreement

In connection with the Business Combination, concurrently with Closing, New SAI, the Sponsor and certain New SAI shareholders entered into the New Registration Rights Agreement. Pursuant to the New Registration Rights Agreement, among other things, subject to certain requirements and customary conditions, including with regard to the number of demand rights that may be exercised, New SAI is required, as soon as practicable, but in any event within 30 days after Closing, to file a registration statement to permit the public resale of all the registrable securities held by any party to the New Registration Rights Agreement from time to time and holders of registrable securities under the New Registration Rights Agreement may demand New SAI facilitate a registered offering of such securities. The New Registration Rights Agreement also (1) provided the holders of registrable securities with "piggy-back" registration rights, subject to certain requirements and customary conditions and (2) terminated the Registration Rights Agreement.

#### Lock-Up Agreements

At Closing, (1) the TradeUP initial shareholders entered into the TradeUP Lock-Up Agreement, (2) the SAI Founder and management of SAI, and certain other SAI shareholders, entered into the SAI Affiliate Lock-Up Agreement and (3) the other SAI shareholders entered into the SAI Shareholder Lock-Up Agreement. Immediately following the consummation of the Business Combination, approximately 24,147,035 New SAI ordinary shares, or approximately 84.3% of the outstanding New SAI ordinary shares (assuming no redemptions), will be subject to the lock-up arrangements described below.

#### TradeUP Lock-Up Agreement

The TradeUP Lock-Up Agreement contains certain restrictions on transfer with respect any New SAI ordinary shares held by the TradeUP initial shareholders immediately after Closing. Such restrictions began at Closing and end on the first anniversary of Closing, with such New SAI ordinary shares being subject to earlier release on the date on which the volume weighted average trading price of New SAI Class A Ordinary Shares exceeds \$14.00 per share (with respect to 50% of such New SAI ordinary shares) and \$17.50 per share (with respect to the remaining 50% of such New SAI ordinary shares) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after Closing.

#### SAI Affiliate Lock-Up Agreement

The SAI Affiliate Lock-Up Agreement contains certain restrictions on transfer with respect any New SAI ordinary shares received pursuant to the Business Combination Agreement. Such restrictions began at Closing and end on the first anniversary of the closing, with such New SAI ordinary shares being subject to earlier release on the date on which the volume weighted average trading price of New SAI Class A Ordinary Shares exceeds \$14.00 per share (with respect to 50% of such New SAI ordinary shares) and \$17.50 per share (with respect to the remaining 50% of such new SAI ordinary shares) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after Closing.

#### SAI Shareholder Lock-Up Agreement

The SAI Shareholder Lock-Up Agreement contains certain restrictions on transfer with respect to the New SAI ordinary shares received by SAI shareholders pursuant to the Business Combination Agreement other than SAI shareholders subject to the SAI Affiliate Lock-Up Agreement. Such restrictions began at Closing and end on the six-month anniversary of Closing.

4

#### **Board of Directors**

In connection with the Business Combination, directors were elected to the board of directors the ("Board") of New SAI such that the entire Board is:

Risheng Li	
Jianwei Li	
Yao Shi <sup>123</sup>	
Yusen Chen <sup>123</sup>	
Jinlong Zhu <sup>3</sup>	

<sup>1</sup> Chairperson and Member of the Audit Committee.

<sup>2</sup> Member of the Compensation Committee.

<sup>3</sup> Member of the Nominating and Governance Committee.

Information regarding the directors and executive officers of New SAI following the completion of the Business Combination is included in the Registration Statement under the section entitled "*Management of New SAI Following the Business Combination*" and is incorporated herein by reference.

#### Indemnity Agreements

On the Closing Date, the Company entered into indemnity agreements with each of its directors and officers (the "Indemnity <u>Agreements</u>"), undertaking to indemnify them to the fullest extent permitted by law on the terms set forth therein. This indemnification is limited to events where the director or officer acted in good faith and in a manner the relevant director or officer reasonably believed to be in and/or not opposed to the best interests of New SAI, and the relevant director or officer had no reasonable cause to believe that his or her conduct was unlawful.

The foregoing summary of the Indemnity Agreements does not purport to be complete and is subject to and qualified in its entirety by reference to the form of Indemnity Agreement, a copy of which is filed as Exhibit 10.1 to this Report of a Foreign Private Issuer on Form 6-K and is incorporated herein by reference.

5

# EXHIBIT INDEX

Exhibit No.	Description
1.1	Amended and Restated Memorandum and Articles of Association
2.1	Business Combination Agreement, dated September 27, 2021, by and among the Company, Merger Sub, and TradeUP (incorporated by reference to Exhibit 2.1 to the Registration on Form F-4 (File No. 333-260418), as amended, filed with the
	SEC on October 22, 2021).
10.1	Form of Indemnity Agreement.
99.1	Press release dated April 29, 2022.

6

#### SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

# **SAI.TECH Global Corporation**

By: /s/ Risheng Li

Risheng Li Chief Executive Officer

7

Copyright © 2022 www.secdatabase.com. All Rights Reserved.

Please Consider the Environment Before Printing This Document

Date: May 5, 2022

# THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES AMENDED AND RESTATED MEMORANDUM AND ARTICLES OF ASSOCIATION OF SAI.TECH Global Corporation

# (ADOPTED BY SPECIAL RESOLUTION DATED APRIL 22, 2022 AND EFFECTIVE ON APRIL 29, 2022 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

# THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES AMENDED AND RESTATED MEMORANDUM OF ASSOCIATION OF SAI.TECH Global Corporation

# (ADOPTED BY SPECIAL RESOLUTION DATED APRIL 22, 2022 AND EFFECTIVE ON APRIL 29, 2022 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

1 The name of the Company is **SAI.TECH Global Corporation**.

6

The Registered Office of the Company shall be at the offices of ICS Corporate Services (Cayman) Limited, 3-212 Governors Square, 23 Lime Tree Bay Avenue, P.O. Box 30746, Seven Mile Beach, Grand Cayman KY1-1203, Cayman Islands, or at such other place within the Cayman Islands as the Directors may decide.

The objects for which the Company is established are unrestricted and except as prohibited or limited by the laws of the Cayman Islands, the Company shall have full power and authority to carry out any object and shall have and be capable of from time to time and at all times exercising any and all of the powers at any time or from time to time exercisable by a natural person or body corporate in any part of the world whether as principal, agent, contractor or otherwise.

The Company will not trade in the Cayman Islands with any Person, firm or corporation except in furtherance of the business of the Company carried on outside the Cayman Islands; provided, that nothing in this section shall be construed as to prevent the Company effecting and concluding contracts in the Cayman Islands, and exercising in the Cayman Islands all of its powers necessary for the carrying on of its business outside the Cayman Islands

5 The liability of each Member is limited to the amount unpaid on such Member's shares.

The authorized share capital of the Company is US\$35,000 divided into 350,000,000 shares of a par value of US\$0.0001 each, of which 330,369,366 shall be designated as Class A ordinary shares, 9,630,634 shall be designated as convertible Class B ordinary shares, and 10,000,000 preference shares. Subject to the provisions of the Statue and the Articles, the Company has the power to redeem or purchase any of its shares and to increase or reduce the share capital and to issue any part of its capital, whether original, redeemed or increased with or without any preference, priority or special privilege or subject to any postponement of rights or to any conditions or restrictions and so that unless the conditions of issue shall otherwise expressly declare every issue of shares whether declared to be preference or otherwise shall be subject to the powers hereinbefore contained.

- 7 The Company has power to register by way of continuation as a body corporate limited by shares under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.
- 8 Capitalised terms that are not defined in this Memorandum bear the respective meanings given to them in the Articles.

#### THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES

#### AMENDED AND RESTATED ARTICLES OF ASSOCIATION OF

#### **SAI.TECH Global Corporation**

# (ADOPTED BY SPECIAL RESOLUTION DATED APRIL 22, 2022 AND EFFECTIVE ON APRIL 29, 2022 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

## 1 Interpretation

1.1 In the Articles Table A in the First Schedule to the Statute does not apply and, unless there is something in the subject or context inconsistent therewith, the following words shall have the following meaning:

"Affiliate"	in respect of a Person, means any other Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person, and (a) in the case of a natural Person, shall include, without limitation, such Person's spouse, parents, children, siblings, mother-in-law and father-in-law and brothers and sisters-in-law, whether by blood, marriage or adoption or anyone residing in such Person's home, a trust for the benefit of any of the foregoing, a company, partnership or any natural Person or entity wholly or jointly owned by any of the foregoing and (b) in the case of an entity, shall include a partnership, a corporation or any natural Person or entity which directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such entity.
"Applicable Law"	means, with respect to any Person, all provisions of laws, statutes, ordinances, rules, regulations, permits, certificates, judgments, decisions, decrees or orders of any governmental authority applicable to such Person.
"Articles"	means these amended and restated articles of association of the Company.
"Audit Committee"	means the audit committee of the board of directors of the Company established pursuant to the Articles, or any successor committee.
"Auditor"	means the Person for the time being performing the duties of auditor of the Company (if any).

3

"Business Combination Agreement" means that certain Business Combination Agreement among the Company (f/k/a TradeUp Global Corporation), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability, and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability, dated as of September 27, 2021.

"Cause" has the meaning ascribed to such term in Founder's employment agreement with the Company as in effect from time to time.

"Chairman"	means the chairman of the board of Directors, if any, appointed by the Directors pursuant to Article 31.7.
"Clearing House"	means a clearing house recognised by the laws of the jurisdiction in which the Shares (or depositary receipts therefor) are listed or quoted on a stock exchange or interdealer quotation system in such jurisdiction.
"Class A Share"	means a class A ordinary share of a par value of US\$0.0001 in the share capital of the Company.
"Class B Share"	means a convertible class B ordinary share of a par value of US\$0.0001 in the share capital of the Company.
"Class B Ordinary Shareholder"	means a holder of Class B Shares and any Permitted Transferee of such holder so long as such Permitted Transferee is a holder of any Class B Shares.
"Company"	means the above named company.
"Company's Website"	means the website of the Company and/or its web-address or domain name (if any).
"Compensation Committee"	means the compensation committee of the board of directors of the Company established pursuant to the Articles, or any successor committee.
"Designated Stock Exchange"	means any United States national securities exchange on which the securities of the Company are listed for trading, including the Nasdaq Capital Market.
"Directors"	means the directors for the time being of the Company.
"Dividend"	means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the Articles.
"Electronic Communication"	means a communication sent by electronic means, including electronic posting to the Company's Website, transmission to any number, address or internet website (including the website of the Securities and Exchange Commission) or other electronic delivery methods as otherwise decided and approved by the Directors.

4

"Electronic Record"	has the same meaning as in the Electronic Transactions Act.
"Electronic Transactions Act"	means the Electronic Transactions Act (As Revised) of the Cayman Islands.
"Exchange Act"	means the United States Securities Exchange Act of 1934, as amended, or any similar U.S. federal statute and the rules and regulations of the Securities and Exchange Commission thereunder, all as the same shall be in effect at the time.
"Family Members"	means and includes only the following individuals:
	(a) the applicable individual;
	(b) the spouse of the applicable individual (including former spouses);
	(c) the parents of the applicable individual;

	(d) the lineal descendants of the applicable individual, the siblings of the applicable individual; and
	(e) the lineal descendants of a sibling of the applicable individual.
	For purposes of the (a) to (e), the descendants of any individual shall include adopted individuals and their issue but only if the adopted individual was adopted prior to attaining age eighteen.
"Founder"	means Li Risheng.
"Incapacity"	means, with respect to an individual, the permanent and total disability of such individual so that such individual is unable to engage in any substantial gainful activity by reason of any medically determinable mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months as determined by a licensed medical practitioner. In the event of a dispute regarding whether an individual has suffered an Incapacity, no Incapacity of such individual will be deemed to have occurred unless and until an affirmative ruling regarding such Incapacity has been made by a court or arbitral panel of competent jurisdiction, and such ruling has become final and non- appealable.
"Independent Director"	has the same meaning as in the rules and regulations of the Designated Stock Exchange or in Rule 10A-3 under the Exchange Act, as the case may be.
"Member"	has the same meaning as in the Statute.
"Memorandum"	means the amended and restated memorandum of association of the Company.

5

"Merger"	has the meaning ascribed to such term in the Business Combination Agreement.
"Merger Effective Date"	has the meaning ascribed to such term in the Business Combination Agreement.
"Nominating and Corporate Governance Committee"	means the nominating and corporate governance committee of the board of directors of the Company established pursuant to the Articles, or any successor committee.
"Officer"	means a Person appointed to hold an office in the Company.
"Ordinary Resolution"	means a resolution passed by a simple majority of the Members as, being entitled to do so, vote in Person or, where proxies are allowed, by proxy at a general meeting, and includes a unanimous written resolution. In computing the majority when a poll is demanded regard shall be had to the number of votes to which each Member is entitled by the Articles.
"Ordinary Shares"	means, collectively, the Class A Shares and the Class B Shares.
"Permitted Entity"	means with respect to Founder:
	(a) any Person in respect of which Founder has, directly or indirectly:
	(i) control with respect to the voting of all the Class B Shares held or to be transferred to such Person;

the ability to direct or cause the direction of management and policies of such Person or any other(ii) Person having the authority referred to in the proceeding clause (a)(i) (whether by contract, as executor, trustee, trust protector or otherwise); or

- (iii) the operational or practical control of such Person, including the right to appoint, designate, remove or replace the Person having the authority referred to in the preceding clauses (a)(i) or (ii) above;
- (b) any trust the beneficiaries of which consist primarily of Founder, his Family Members, and/or any Persons controlled directly or indirectly by such a trust;
- (c) any Person Controlled by a trust described in the immediately preceding clause (b) above; and
- (d) for the avoidance of doubt, Energy Science Artist Holding Limited.

1
v

"Permitted Transferee"	means, with respect to the Class B Ordinary Shareholders, any or all of the following:
	(a) Founder;
	(b) any Founder Permitted Entities;
	(c) any Family Members of Founder;
	(d) any other relative of Founder approved by the Board;
	(e) any transferee by operation of law, including in connection with divorce proceedings;
	(f) any charitable organization, foundation or similar entity;
	(g) in connection with a transfer as a result of, or in connection with the death or Incapacity of Founder, any Family Member of Founder or other designee, in each case, approved by the Directors; and
	(h) the Company;
	provided that, (1) as a condition of any transfer to a Permitted Transferee under clauses (c) through (f) above, such Permitted Transferee shall have executed an original, a counterpart of, or a deed of adherence with respect to, a Proxy and Voting Deed; and (ii) in the case of any transfer of Class B Shares pursuant to clauses (b) through (f) above to a Person who at a later time ceases to qualify as a Permitted Transferee under the relevant clause, the Company shall be able to refuse registration of any subsequent transfer of such Class B Shares except back to the Founder or his Permitted Entities, and in the absence of such transfer back to the Founder or his Permitted Entities B Shares shall automatically convert in accordance with Article 17.4 applied <i>mutatis mutandis</i> .
"Person"	means any individual, corporation, partnership, joint venture, trust, unincorporated organization, limited liability company, estate, association, joint stock company, unincorporated organization, company or other form of business or legal entity or government authority.
<b>"Preference</b> Share"	means a preference share of a par value of US\$0.0001 in the share capital of the Company.
"Proxy and Voting Deed"	means a deed or agreement conveying to Founder voting rights of outstanding Class B Shares.

"Register of means the register of Members maintained in accordance with the Statute and includes (except where otherwise stated) any branch or duplicate register of Members.

"Registered Office"	means the registered office for the time being of the Company.
"Seal"	means the common seal of the Company and includes every duplicate seal.
"Securities Act"	means the U.S. Securities Act of 1933.
"Securities and Exchange Commission"	means the United States Securities and Exchange Commission.
"Share"	means a Class A Share, a Class B Share, or a Preference Share and includes a fraction of a share in the Company.
"Special Resolution"	subject to Article 29.4, has the same meaning as in the Statute, and includes a unanimous written resolution.
"Statute"	means the Companies Act (As Revised) of the Cayman Islands.
"Treasury Share'	' means a Share held in the name of the Company as a treasury share in accordance with the Statute.
In the Articles:	
(a) words impor	ing the singular number include the plural number and vice versa;
(b) words import	ing the masculine gender include the feminine gender;
	any agreement, deed or other instrument (including to these Articles and the Memorandum) is a reference to nt or instrument as amended, restated, or novated from time to time;
(d) "written" and an Electronic	I "in writing" include all modes of representing or reproducing words in visible form, including in the form of Record;
(e) "shall" shall	be construed as imperative and "may" shall be construed as permissive;
(f) references to re-enacted or	provisions of any law or regulation shall be construed as references to those provisions as amended, modified, replaced;
	atroduced by the terms "including", "include", "in particular" or any similar expression shall be construed as and shall not limit the sense of the words preceding those terms;
the term "and	d/or" is used herein to mean both "and" as well as "or." The use of "and/or" in certain contexts in no respects

1.2

(h) qualifies or modifies the use of the terms "and" or "or" in others. The term "or" shall not be interpreted to be exclusive and the term "and" shall not be interpreted to require the conjunctive (in each case, unless the context otherwise requires);

- (i) headings are inserted for reference only and shall be ignored in construing the Articles;
- (j) any requirements as to delivery under the Articles include delivery in the form of an Electronic Record;
- (k) any requirements as to execution or signature under the Articles including the execution of the Articles themselves can be satisfied in the form of an electronic signature as defined in the Electronic Transactions Act;
- (l) sections 8 and 19(3) of the Electronic Transactions Act shall not apply;
- (m) the term "clear days" in relation to the period of a notice means that period excluding the day when the notice is received or deemed to be received and the day for which it is given or on which it is to take effect;
- (n) a reference to "\$", "US\$", "USD", or dollar(s) is a reference to the lawful currency of the United States of America;
- (o) "directly or indirectly" means, except where the context requires otherwise, directly or indirectly through one or more intermediate Persons or through contractual or other arrangements, and "direct or indirect" has the correlative meaning; and
- (p) the term "holder" in relation to a Share means a Person whose name is entered in the Register of Members as the holder of such Share.

#### 2 Commencement of Business

- 2.1 The business of the Company may be commenced as soon after incorporation of the Company as the Directors shall see fit.
- 2.2 The Directors may pay, out of the capital or any other monies of the Company, all expenses incurred in or about the formation and establishment of the Company, including the expenses of registration.

# **3** Issue of Shares and other Securities

emptive rights.

3.1

Subject to the provisions, if any, in the Memorandum (and to any direction that may be given by the Company in general meeting) and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, and without prejudice to any rights attached to any existing Shares, the Directors may allot, issue, grant options over or otherwise dispose of Shares (including fractions of a Share) with or without preferred, deferred or other rights or restrictions, whether in regard to Dividends or other distributions, voting, return of capital or otherwise and to such Persons, at such times and on such other terms as they think proper, and may also (subject to the Statute and the Articles) vary such rights, save that the Directors shall not allot, issue, grant options over or otherwise dispose of Shares (including fractions of a Share) to the extent that it may affect the ability of the Company to carry out a conversion of the Class B Shares set out in the Articles. No holder of Ordinary Shares shall have pre-

9

The Company may issue rights, options, warrants or convertible securities or securities of similar nature conferring the right
 upon the holders thereof to subscribe for, purchase or receive any class of Shares or other securities in the Company on such terms as the Directors may from time to time determine.

Except as set forth otherwise in Article 10(c)(iv), the Directors may provide, out of the unissued shares, for a series of Preference
 Shares. Before any Preference Shares of any such series are issued, the Directors shall fix, by resolution or resolutions, the following provisions of the Preference Shares thereof, if applicable:

- (a) the designation of such series, the number of Preference Shares to constitute such series and the subscription price thereof if different from the par value thereof;
- (b) whether the shares of such series shall have voting rights, in addition to any voting rights provided by law, and, if so, the terms of such voting rights, which may be general or limited;

the dividends, if any, payable on such series, whether any such dividends shall be cumulative, and, if so, from what dates,

- (c) the conditions and dates upon which such dividends shall be payable, the preference or relation which such dividends shall bear to the dividends payable on any shares of any other class or any other series of Preference Shares;
- (d) whether the Preference Shares of such series shall be subject to redemption by the Company, and, if so, the times, prices and other conditions of such redemption;
- (e) the amount or amounts payable upon Preference Shares of such series upon, and the rights of the holders of such series in, a voluntary or involuntary liquidation, dissolution or winding up, or upon any distribution of the assets, of the Company;

whether the Preference Shares of such series shall be subject to the operation of a retirement or sinking fund and, if so, the extent to and manner in which any such retirement or sinking fund shall be applied to the purchase or redemption of

 (f) the extent to and manner in which any such retirement or sinking rule share or applied to the purchase or redemption or the Preference Shares of such series for retirement or other corporate purposes and the terms and provisions relative to the operation thereof;

whether the Preference Shares of such series shall be convertible into, or exchangeable for, shares of any other class or any(g) other series of Preference Shares or any other securities and, if so, the price or prices or the rate or rates of conversion or exchange and the method, if any, of adjusting the same, and any other terms and conditions of conversion or exchange;

the limitations and restrictions, if any, to be effective while any Preference Shares of such series are outstanding upon the (h) payment of dividends or the making of other distributions on, and upon the purchase, redemption or other acquisition by the

Company of, the existing shares or shares of any other class of shares or any other series of Preference Shares;

10

the conditions or restrictions, if any, upon the creation of indebtedness of the Company or upon the issue of any additionalshares, including additional shares of such series or of any other class of shares or any other series of Preference Shares; and

(j) any other powers, preferences and relative, participating, optional and other special rights, and any qualifications, limitations and restrictions thereof.

Without limiting the foregoing and subject to <u>Articles 17.3 and 23.1</u>, the voting powers of any series of Preference Shares may include the right, in the circumstances specified in the resolution or resolutions providing for the issuance of such Preference Shares, to elect one or more Directors who shall serve for such term and have such voting powers as shall be stated in the resolution or resolutions providing for the issuance of such Preference Shares. The term of office and voting powers of any Director elected in the manner provided in the immediately preceding sentence of this Article 3.3 may be greater than or less than those of any other Director or class of Directors.

3.4 The powers, preferences and relative, participating, optional and other special rights of each series of Preference Shares, and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. All shares of any one series of Preference Shares shall be identical in all respects with all other shares of such series, except that shares of any one series issued at different times may differ as to the dates from which dividends thereon shall be cumulative.

3.5 The Company shall not issue Shares to bearer.

#### 4 Register of Members

4.1 The Company shall maintain or cause to be maintained the Register of Members in accordance with the Statute.

The Directors may determine that the Company shall maintain one or more branch Registers of Members in accordance with
 the Statute. The Directors may also determine which Register of Members shall constitute the principal register and which shall constitute the branch register or registers, and to vary such determination from time to time.

# 5 Closing Register of Members or Fixing Record Date

5.1

6.1

For the purpose of determining Members entitled to notice of, or to vote at any meeting of Members or any adjournment thereof, or Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose, the Directors may, after notice has been given by advertisement in an appointed newspaper or any other newspaper or by any other means in accordance with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, provide that the Register of Members shall be closed for transfers for a stated period which shall not in any case exceed forty days.

11

5.2 In lieu of, or apart from, closing the Register of Members, the Directors may fix in advance or arrears a date as the record date for any such determination of Members entitled to notice of, or to vote at any meeting of the Members or any adjournment thereof, or for the purpose of determining the Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose.

5.3 If the Register of Members is not so closed and no record date is fixed for the determination of Members entitled to notice of, or to vote at, a meeting of Members or Members entitled to receive payment of a Dividend or other distribution, the date on which notice of the meeting is sent or the date on which the resolution of the Directors resolving to pay such Dividend or other distribution is passed, as the case may be, shall be the record date for such determination of Members. When a determination of Members entitled to vote at any meeting of Members has been made as provided in this Article, such determination shall apply to any adjournment thereof.

# 6 Certificates for Shares

A Member shall only be entitled to a share certificate if the Directors resolve that share certificates shall be issued. Share certificates representing Shares, if any, shall be in such form as the Directors may determine. Share certificates shall be signed by one or more Directors or other Person authorised by the Directors. The Directors may authorise certificates to be issued with the authorised signature(s) affixed by mechanical process. All certificates for Shares shall be consecutively numbered or otherwise identified and shall specify the Shares to which they relate. All share certificates shall be cancelled and, subject to the Articles, no new certificate shall be issued until the former certificate representing a like number of relevant Shares shall have been surrendered and cancelled.

- 6.2 The Company shall not be bound to issue more than one certificate for Shares held jointly by more than one Person and delivery of a certificate to one joint holder shall be a sufficient delivery to all of them.
- If a share certificate is defaced, worn out, lost or destroyed, it may be renewed on such terms (if any) as to evidence and
   indemnity and on the payment of such expenses reasonably incurred by the Company in investigating evidence, as the Directors may prescribe, and (in the case of defacement or wearing out) upon delivery of the old certificate.
- 6.4 Every share certificate sent in accordance with the Articles will be sent at the risk of the Member or other Person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

Share certificates shall be issued within the relevant time limit as prescribed by the Statute, if applicable, or as the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law may from time to time determine, whichever is shorter, after the allotment or, except in the case of a Share transfer which the Company is for the time being entitled to refuse to register and does not register, after lodgement of a Share transfer with the Company.

# 7 Transfer of Shares

Subject to the terms of the Articles, any Member may transfer all or any of his Class A Shares by an instrument of transfer 7.1 provided that such transfer complies with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law. If the Class A Shares in question were issued in conjunction with rights, options, warrants or units issued pursuant to the Articles on terms that one cannot be transferred without the other, the Directors shall refuse to register the transfer of any such Share without evidence satisfactory to them of the like transfer of such right, option, warrant or unit.

The instrument of transfer of any Class A Share shall be in writing in the usual or common form or in a form prescribed by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law or in any other form approved by the Directors and shall be executed by or on behalf of the transferor (and if the Directors so require, signed by or on behalf of the transferee) and may be under hand or, if the transferor or transferee is a Clearing House or its nominee(s), by hand or by machine imprinted signature or by such other manner of execution as the Directors may approve from time to time. The transferor shall be deemed to remain the holder of a Share until the name of the transferee is entered in the Register of Members.

#### 8 Redemption, Repurchase and Surrender of Shares

Subject to the provisions of the Statute, and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, the Company may issue Shares that are to be redeemed or are liable to be redeemed at the option of the Member or the Company. The redemption of such Shares shall be effected in such manner and upon such other terms as the Company may, by Special Resolution, determine before the issue of such Shares.

Subject to the provisions of the Statute, and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law,
8.2 the Company may purchase its own Shares (including any redeemable Shares) in such manner and on such other terms as the Directors may agree with the relevant Member. For the avoidance of doubt, redemptions, repurchases and surrenders of Shares in the circumstances described in the Article above shall not require further approval of the Members.

- 8.3 The Company may make a payment in respect of the redemption or purchase of its own Shares in any manner permitted by the Statute, including out of capital.
- 8.4 The Directors may accept the surrender for no consideration of any fully paid Share.

# 9 Treasury Shares

7.2

- 9.1 The Directors may, prior to the purchase, redemption or surrender of any Share, determine that such Share shall be held as a Treasury Share.
- 9.2 The Directors may determine to cancel a Treasury Share or transfer a Treasury Share on such terms as they think proper (including, without limitation, for nil consideration).

#### 10 Variation of Rights of Shares

Subject to Articles 3.1 and 17.3, if at any time the share capital of the Company is divided into different classes of Shares, all or any of the rights attached to any class (unless otherwise provided by the terms of issue of the Shares of that class) may, whether or not the Company is being wound up, be varied without the consent of the holders of the issued Shares of that class where such variation is considered by the Directors not to have a material adverse effect upon such rights; otherwise, any such variation shall be made only with the consent in writing of the holders of not less than two thirds of the issued Shares of that class or with

10.1 the approval of a resolution passed by a majority of not less than two thirds of the votes cast at a separate meeting of the holders of the Shares of that class. For the avoidance of doubt, the Directors reserve the right, notwithstanding that any such variation may not have a material adverse effect, to obtain consent from the holders of Shares of the relevant class. To any such meeting all the provisions of the Articles relating to general meetings shall apply *mutatis mutandis*, except that the necessary quorum shall be one Person holding or representing by proxy at least one third of the issued Shares of the class and that any holder of Shares of the class present in Person or by proxy may demand a poll.

- 10.2 The provisions of these Articles relating to general meetings shall apply to every such general meeting of the holders of one class or series of shares, except the following:
  - (a) separate general meetings of the holders of a call or series of shares may be called only by:
    - (i) the Chairman, if any;
    - (ii) a majority of all of the Directors (unless otherwise specifically provided by the terms of issue of the shares of such class or series); or
    - (iii) with respect to general meetings of the holders of Class B Shares, the Founder.
  - (b) except as set forth in clause (a) above, nothing in this Article 10 shall be deemed to give any Member the right to call a class or series meeting; and
  - (c) the necessary quorum shall be one or more Persons holding or representing by proxy at least one-third of the issues shares of the class or series and that any holder of shares of the class or series present in person or by proxy may demand a poll.
- The rights conferred upon the holders of the Shares of any class issued with preferred or other rights shall not, unless otherwise expressly provided by the terms of issue of the Shares of that class, be deemed to be varied by the creation or issue of further Shares ranking *pari passu* therewith or Shares issued with preferred or other rights.

#### 11 Commission on Sale of Shares

The Company may, in so far as the Statute permits, pay a commission to any Person in consideration of his subscribing or agreeing to subscribe (whether absolutely or conditionally) or procuring or agreeing to procure subscriptions (whether absolutely or conditionally) for any Shares. Such commissions may be satisfied by the payment of cash and/or the issue of fully or partly paid-up Shares. The Company may also on any issue of Shares pay such brokerage as may be lawful.

#### 12 Non Recognition of Trusts

No Person shall be recognised by the Company as holding any share upon any trust (other than any trust recognized as a Permitted Entity or Permitted Transferee). The Company shall not be bound by or compelled to recognise in any way (even when notified) any equitable, contingent, future or partial interest in any Share, or (except only as is otherwise provided by the Articles or the Statute) any other rights in respect of any Share other than an absolute right to the entirety thereof in the holder. It is understood that a Proxy and Voting Deed shall not be the holding of any share upon a trust.

#### 13 Lien on Shares

The Company shall have a first and paramount lien on all Shares (whether fully paid-up or not) registered in the name of a Member (whether solely or jointly with others) for all debts, liabilities or engagements to or with the Company (whether

- 13.1 presently payable or not) by such Member or his estate, either alone or jointly with any other Person, whether a Member or not, but the Directors may at any time declare any Share to be wholly or in part exempt from the provisions of this Article. The registration of a transfer of any such Share shall operate as a waiver of the Company's lien thereon. The Company's lien on a Share shall also extend to any amount payable in respect of that Share.
- 13.2 The Company may sell, in such manner as the Directors think fit, any Shares on which the Company has a lien, if a sum in respect of which the lien exists is presently payable, and is not paid within fourteen clear days after notice has been received

or deemed to have been received by the holder of the Shares, or to the Person entitled to it in consequence of the death or bankruptcy of the holder, demanding payment and stating that if the notice is not complied with the Shares may be sold.

To give effect to any such sale the Directors may authorise any Person to execute an instrument of transfer of the Shares sold to, or in accordance with the directions of, the purchaser. The purchaser or his nominee shall be registered as the holder of the Shares comprised in any such transfer, and he shall not be bound to see to the application of the purchase money, nor shall his title to the Shares be affected by any irregularity or invalidity in the sale or the exercise of the Company's power of sale under the Articles.

The net proceeds of such sale after payment of costs, shall be applied in payment of such part of the amount in respect of which the lien exists as is presently payable and any balance shall (subject to a like lien for sums not presently payable as existed upon the Shares before the sale) be paid to the Person entitled to the Shares at the date of the sale.

#### 14 Call on Shares

13.3

Subject to the terms of the allotment and issue of any Shares, the Directors may make calls upon the Members in respect of any monies unpaid on their Shares (whether in respect of par value or premium), and each Member shall (subject to receiving at least fourteen clear days' notice specifying the time or times of payment) pay to the Company at the time or times so specified

- 14.1 the amount called on the Shares. A call may be revoked or postponed, in whole or in part, as the Directors may determine. A call may be required to be paid by instalments. A Person upon whom a call is made shall remain liable for calls made upon him notwithstanding the subsequent transfer of the Shares in respect of which the call was made.
- 14.2 A call shall be deemed to have been made at the time when the resolution of the Directors authorising such call was passed.
- 14.3 The joint holders of a Share shall be jointly and severally liable to pay all calls in respect thereof.

14.4 If a call remains unpaid after it has become due and payable, the Person from whom it is due shall pay interest on the amount unpaid from the day it became due and payable until it is paid at such rate as the Directors may determine (and in addition all expenses that have been incurred by the Company by reason of such non-payment), but the Directors may waive payment of the interest or expenses wholly or in part.

- An amount payable in respect of a Share on issue or allotment or at any fixed date, whether on account of the par value of the 14.5 Share or premium or otherwise, shall be deemed to be a call and if it is not paid all the provisions of the Articles shall apply as if that amount had become due and payable by virtue of a call.
- 14.6 The Directors may issue Shares with different terms as to the amount and times of payment of calls, or the interest to be paid.
- The Directors may, if they think fit, receive an amount from any Member willing to advance all or any part of the monies uncalled and unpaid upon any Shares held by him, and may (until the amount would otherwise become payable) pay interest at such rate as may be agreed upon between the Directors and the Member paying such amount in advance.

No such amount paid in advance of calls shall entitle the Member paying such amount to any portion of a Dividend or other distribution payable in respect of any period prior to the date upon which such amount would, but for such payment, become payable.

#### **15** Forfeiture of Shares

If a call or instalment of a call remains unpaid after it has become due and payable the Directors may give to the Person from whom it is due not less than fourteen clear days' notice requiring payment of the amount unpaid together with any interest which

15.1 may have accrued and any expenses incurred by the Company by reason of such non-payment. The notice shall specify where payment is to be made and shall state that if the notice is not complied with the Shares in respect of which the call was made will be liable to be forfeited.

15.2 If the notice is not complied with, any Share in respect of which it was given may, before the payment required by the notice has been made, be forfeited by a resolution of the Directors. Such forfeiture shall include all Dividends, other distributions or other monies payable in respect of the forfeited Share and not paid before the forfeiture.

15.3 A forfeited Share (other than a Class B Share) may be sold, re-allotted or otherwise disposed of on such terms and in such manner as the Directors think fit and at any time before a sale, re-allotment or disposition the forfeiture may be cancelled on such terms as the Directors think fit. Where for the purposes of its disposal a forfeited Share is to be transferred to any Person the Directors may authorise some Person to execute an instrument of transfer of the Share in favour of that Person.

A Person any of whose Shares have been forfeited shall cease to be a Member in respect of them and shall surrender to the Company for cancellation the certificate for the Shares forfeited and shall remain liable to pay to the Company all monies which at the date of forfeiture were payable by him to the Company in respect of those Shares together with interest at such rate as the Directors may determine, but his liability shall cease if and when the Company shall have received payment in full of all monies due and payable by him in respect of those Shares.

16

A certificate in writing under the hand of one Director or Officer that a Share has been forfeited on a specified date shall be conclusive evidence of the facts stated in it as against all Persons claiming to be entitled to the Share. The certificate shall (subject to the execution of an instrument of transfer) constitute a good title to the Share and the Person to whom the Share is sold or otherwise disposed of shall not be bound to see to the application of the purchase money, if any, nor shall his title to the Share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the Share.

The provisions of the Articles as to forfeiture shall apply in the case of non payment of any sum which, by the terms of issue of a Share, becomes payable at a fixed time, whether on account of the par value of the Share or by way of premium as if it had been payable by virtue of a call duly made and notified.

# 16 Transmission of Shares

16.3

If a Member dies, the survivor or survivors (where he was a joint holder), or his legal personal representatives (where he was a sole holder), shall be the only Persons recognised by the Company as having any title to his Shares. The estate of a deceased Member is not thereby released from any liability in respect of any Share, for which he was a joint or sole holder.

Any Person becoming entitled to a Share in consequence of the death or bankruptcy or liquidation or dissolution of a Member (or in any other way than by transfer) may, upon such evidence being produced as may be required by the Directors, elect, by a notice in writing sent by him to the Company, either to become the holder of such Share or to have some Person nominated by him registered as the holder of such Share. If he elects to have another Person registered as the holder of such Share he shall aim on instrument of transfer of that Share to that Person. The Directory shall in either area have the same right to dealing an

16.2 by him registered as the holder of such Share. If he elects to have another Person registered as the holder of such Share he shall sign an instrument of transfer of that Share to that Person. The Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution, as the case may be.

A Person becoming entitled to a Share by reason of the death or bankruptcy or liquidation or dissolution of a Member (or in any other case than by transfer) shall be entitled to the same Dividends, other distributions and other advantages to which he would be entitled if he were the holder of such Share. However, he shall not, before becoming a Member in respect of a Share, be entitled in respect of it to exercise any right conferred by membership in relation to general meetings of the Company and the Directors may at any time give notice requiring any such Person to elect either to be registered himself or to have some Person nominated by him be registered as the holder of the Share (but the Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution or any other case than by transfer, as the case may be). If the notice is not complied with within ninety days of being received or deemed to be received (as determined pursuant to the Articles), the Directors may thereafter withhold payment of all Dividends, other distributions, bonuses or other monies payable in respect of the Share until the requirements of the notice have been complied with.

# 17 Rights and Restrictions Attaching to Ordinary Shares

Except as otherwise included in these Articles (including Articles 17.3 and 23.1), the rights attaching to the Class A Shares and Class B Shares shall have the same rights and powers, and rank *pari passu* in all respects (including as to dividends, distributions,

17.1 and upon the occurrence of any liquidation or winding up of the Company), share ratably and be identical in all respects as to all matters, unless different treatment of the shares of each such class is approved by the affirmative vote of the holders of the Class A Shares and the Class B Shares, each voting exclusively and as a separate class.

Holders of Ordinary Shares have the right to receive notice of, attend, speak and vote at general meetings of the Company. Except as set forth in these Articles (including Articles 17.3, 17.4 and 23.1), holders of Class A Shares and Class B Shares shall at all times vote together as one class on all matters submitted to a vote for Members' consent. On all matters subject to a vote of the Members, Ordinary Shares shall be entitled to the voting rights as set forth in Article 23.

In addition to any rights provided by applicable law or otherwise set forth in these Articles, the Company shall not, without the approval by vote or written consent of the holders of a majority of the voting power of the Class B Shares, voting exclusively and as a separate class, directly or indirectly, or whether by amendment or through merger, recapitalization, consolidation or otherwise:

(a) increase the number of authorized Class B Shares;

17.3

issue any Class B Shares or securities convertible into or exchangeable for Class B Shares, other than to Founder or his(b) Affiliates (provided, that his Affiliate shall have executed an original, a counterpart of, or a deed of adherence with respect to, a Proxy and Voting Deed in respect of such Class B Shares);

(c) create, authorize, issue, or reclassify into, any Preference Shares or any shares in the capital of the Company that carry more than one (1) vote per share;

(d) reclassify any Class B Shares into any other class of shares or consolidate or combine any Class B Shares without proportionately increasing the number of votes per Class B Share; or

amend, restate, waive, adopt any provision inconsistent with or otherwise vary or alter any provision of the Memorandum(e) or these Articles relating to the voting, conversion or other rights, powers, preferences, privileges or restrictions of the Class B Shares.

17.4 Optional and Automatic Conversion of Class B Shares:

each Class B Share is convertible into one (1) Class A Share (as adjusted for share splits, share combinations and similar(a) transactions occurring after the Merger Effective Date) at any time at the option of the holder thereof. In no event shall any Class A Share be convertible into any Class B Shares;

each Class B Share will automatically convert into one (1) Class A Share (as adjusted for share splits, share combinations(b) and similar transactions occurring after the Merger Effective Date) on the earliest to occur of 5:00 p.m., Cayman Islands time:

(i) on the first anniversary of Founder's death or Incapacity; or

on a date determined by the Board during the period commencing 90 days after, and ending 180 days after, the date on which Founder is terminated for Cause (and in the event of a dispute regarding whether there was Cause, Cause will

 (ii) which is terminated for Cause (and in the event of a dispute regarding whether there was cause, cause with be deemed not to exist unless and until an affirmative ruling regarding such Cause has been made by a court or arbitral panel of competent jurisdiction, and such ruling has become final and non-appealable); (c) no Class B Shares shall be issued by the Company after conversion of all Class B Shares into Class A Shares;

upon any sale, pledge, transfer, assignment or other disposition of Class B Shares by a holder thereof to any Person which is not a Permitted Transferee of such holder, each such Class B Share shall be automatically and immediately converted into one (1) Class A Share (as adjusted for share splits, share combinations and similar transactions occurring after the Acquisition Effective Time); provided that, notwithstanding anything to the contrary in these Articles, any pledge of Class B Shares by a holder thereof that creates a security interest in such Class B Shares pursuant to a bona fide loan or indebtedness

- (d) transaction shall be permitted (and not result in any such conversion) for so long as Founder or his Affiliates continue to control, directly or indirectly, the exercise of the voting rights of such pledged Class B Shares; provided, further, however, that a foreclosure on such Class B Shares or other similar action by the pledgee will result in automatic and immediate conversion of such Class B Shares into Class A Shares unless the transferee in such foreclosure or similar action qualifies as a Permitted Transferee at such time. For the avoidance of doubt, any sale, pledge, transfer, assignment or disposition of Class B Shares to a Permitted Transferee does not result in automatic conversion into Class A Shares.
- 17.5 The Company may, from time to time, establish such policies and procedures relating to the conversion of the Class B Shares to Class A Shares, including the issuance of share certificates with respect thereto, as it may deem necessary or advisable.

The Company shall at all times reserve and keep available out of its authorized but unissued Class A Shares, solely for the purpose of effecting the conversion of the Class B Shares, such number of its Class A Shares as shall from time to time be sufficient to effect the conversion of all outstanding Class B Shares; and if at any time the number of authorized but unissued Class A Shares shall not be sufficient to effect the conversion of all then outstanding Class B Shares.

17.6 Class A Shares shall not be sufficient to effect the conversion of all then-outstanding Class B Shares, the Company will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued Class A Shares to such numbers of shares as shall be sufficient for such purpose.

# 18 Amendments of Memorandum and Articles of Association and Alteration of Capital

- 18.1 The Company may by Ordinary Resolution, subject to the rights of the Class B Shares, including under Article 17.3:
  - (a) increase its share capital by such sum as the Ordinary Resolution shall prescribe and with such rights, priorities and privileges annexed thereto, as the Company in general meeting may determine;
  - (b) consolidate and divide all or any of its share capital into Shares of larger amount than its existing Shares;
  - by subdivision of its existing Shares or any of them divide the whole or any part of its share capital into Shares of smaller(c) amount; provided, that in the subdivision the amount paid and the amount, if any, unpaid on each reduced share shall be the same as it was in case of the share from which the reduced share is derived; or

(d) cancel any Shares that at the date of the passing of the Ordinary Resolution have not been taken or agreed to be taken by any Person and diminish the amount of its share capital by the amount of the Shares so cancelled.

All new Shares created in accordance with the provisions of the preceding Article shall be subject to the same provisions of the 18.2 Articles with reference to the payment of calls, liens, transfer, transmission, forfeiture and otherwise as the Shares in the original share capital.

- 18.3 Subject to the rights of the Class B Shares, including under Article 17.3(e), the provisions of the Statute, the provisions of the Articles as regards the matters to be dealt with by Ordinary Resolution, the Company may by Special Resolution:
  - (a) change its name;
  - (b) alter or add to the Articles;
  - (c) alter or add to the Memorandum with respect to any objects, powers or other matters specified therein; and

(d) reduce its share capital or any capital redemption reserve fund.

#### **19** Offices and Places of Business

Subject to the provisions of the Statute, the Company may by resolution of the Directors change the location of its Registered Office. The Company may, in addition to its Registered Office, maintain such other offices or places of business as the Directors determine.

# 20 General Meetings

20.1 All general meetings other than annual general meetings shall be called extraordinary general meetings.

The Company shall in each year hold a general meeting as its annual general meeting, and shall specify the meeting as such in the notices calling it. Any annual general meeting shall be held at such time and place as the Directors shall appoint. At these meetings the report of the Directors (if any) shall be presented.

20.3 The Directors, the chief executive officer or the Chairman may call general meetings, and, for the avoidance of doubt, Members shall not have the ability to call general meetings.

Members seeking to bring business before the annual general meeting or to nominate candidates for appointment as Directors at the annual general meeting must deliver notice to the principal executive offices of the Company not less than 120 calendar days before the date of the Company's proxy statement released to Members in connection with the previous year's annual general meeting or, if the Company did not hold an annual general meeting the previous year, or if the date of the current year's annual

20.4 meeting or, if the Company did not hold an annual general meeting the previous year, or if the date of the current year's annual general meeting has been changed by more than 30 days from the date of the previous year's annual general meeting, then the deadline shall be set by the board of Directors with such deadline being a reasonable time before the Company begins to print and send its related proxy materials.

# **21 Notice of General Meetings**

21.1

At least five clear days' notice shall be given of any general meeting. Every notice shall specify the place, the day and the hour of the meeting and the general nature of the business to be conducted at the general meeting and shall be given in the manner hereinafter mentioned or in such other manner if any as may be prescribed by the Company, provided that a general meeting of the Company shall, whether or not the notice specified in this Article has been given and whether or not the provisions of the Articles regarding general meetings have been complied with, be deemed to have been duly convened if it is so agreed:

(a) in the case of an annual general meeting, by all of the Members (or their proxies) entitled to attend and vote thereat; and

in the case of an extraordinary general meeting, by Members (or their proxies) having the right to attend and vote at the(b) meeting, together holding shares entitling the holders thereof to not less than two-thirds of the votes entitled to be cast at such extraordinary meeting.

21.2 The accidental omission to give notice of a general meeting to, or the non-receipt of notice of a general meeting by, any Person entitled to receive such notice shall not invalidate the proceedings of that general meeting.

#### 22 Proceedings at General Meetings

22.1 No business shall be transacted at any general meeting unless a quorum of Members is present. The holders of a majority of the Shares being individuals present in Person or by proxy or if a corporation or other non-natural Person by its duly authorised representative or proxy shall be a quorum for all purposes; provided, that the presence in person or by proxy of holders of a majority of the Class B Shares shall be required in any event.

- If provided by the Company, a Person may participate at a general meeting by conference telephone or other communications equipment by means of which all the Persons participating in the meeting can communicate with each other. Participation by a Person in a general meeting in this manner is treated as presence in Person at that meeting.
- 22.3 A resolution (including a Special Resolution) in writing (in one or more counterparts) signed by or on behalf of all of the Members for the time being entitled to receive notice of and to attend and vote at general meetings (or, being corporations or other non-natural Persons, signed by their duly authorised representatives) shall be as valid and effective as if the resolution had been passed at a general meeting of the Company duly convened and held.
- 22.4 If a quorum is not present within half an hour from the time appointed for the meeting to commence, the meeting shall stand adjourned to the same day in the next week at the same time and/or place or to such other day, time and/or place as the Directors may determine, and if at the adjourned meeting a quorum is not present within half an hour from the time appointed for the meeting to commence, the Members present shall be a quorum.
- 22.5 The Directors may, at the time prior to the time appointed for the meeting to commence, appoint any Person to act as a chairman of a general meeting or, if the Directors do not make any such appointment, the chairman, if any, of the board of Directors (as elected by a majority of the Directors in accordance with Article 31.7) shall preside as chairman at such general meeting. If there is no chairman present, or if the chairman is unwilling to act, the Directors present shall elect one of their number to be chairman of the meeting. The voting rights of the Chairman, if a Director, as to the matters to be decided by the Board of Directors shall be the same as other Directors.

22.6 The chairman may, with the consent of a meeting at which a quorum is present (and shall if so directed by the meeting) adjourn the meeting from time to time and from place to place (provided, that no special meeting called by the holders of Class B Shares may be adjourned unless a quorum does not exist), but no business shall be transacted at any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place.

- 22.7 When a general meeting is adjourned for thirty days or more, notice of the adjourned meeting shall be given as in the case of an original meeting. Otherwise, it shall not be necessary to give any such notice of an adjourned meeting.
- 22.8 A resolution put to the vote of the meeting shall be decided on a poll.
- 22.9 A poll shall be taken as the chairman directs, and the result of the poll shall be deemed to be the resolution of the general meeting at which the poll was demanded.

22.10 A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith (provided, that no special meeting called by a holder of Class B Shares may be adjourned unless a quorum does not exist). A poll demanded on any other question shall be taken at such date, time and place as the chairman of the general meeting directs, and any business other than that upon which a poll has been demanded or is contingent thereon may proceed pending the taking of the poll.

22.11 In the case of an equality of votes the chairman shall be entitled to a second or casting vote.

# 23 Votes of Members

Subject to any rights or restrictions attached to any Shares, including in Articles 17.3 and 17.4, each Class A Share shall be entitled to one (1) vote on all matters subject to a vote of Members, and each Class B Share shall be entitled to ten (10) votes on all matters subject to a vote of Members.

23.2 In the case of joint holders the vote of the senior holder who tenders a vote, whether in Person or by proxy (or, in the case of a corporation or other non-natural Person, by its duly authorised representative or proxy), shall be accepted to the exclusion of the votes of the other joint holders, and seniority shall be determined by the order in which the names of the holders stand in the Register of Members.

- A Member of unsound mind, or in respect of whom an order has been made by any court, having jurisdiction in lunacy, may
   vote by his committee, receiver, curator bonis, or other Person on such Member's behalf appointed by that court, and any such committee, receiver, curator bonis or other Person may vote by proxy.
- 23.4 No Person shall be entitled to vote at any general meeting unless he is registered as a Member on the record date for such meeting nor unless all calls or other monies then payable by him in respect of Shares have been paid.

No objection shall be raised as to the qualification of any voter except at the general meeting or adjourned general meeting at which the vote objected to is given or tendered and every vote not disallowed at the meeting shall be valid. Any objection made in due time in accordance with this Article shall be referred to the chairman whose decision shall be final and conclusive.

23.6 Votes may be cast either personally or by proxy (or in the case of a corporation or other non-natural Person by its duly authorised representative or proxy). A Member may appoint more than one proxy or the same proxy under one or more instruments to attend and vote at a meeting. Where a Member appoints more than one proxy the instrument of proxy shall specify the number of Shares in respect of which each proxy is entitled to exercise the related votes.

A Member holding more than one Share need not cast the votes in respect of his Shares in the same way on any resolution and therefore may vote a Share or some or all such Shares either for or against a resolution and/or abstain from voting a Share or some or all of the Shares and, subject to the terms of the instrument appointing him, a proxy appointed under one or more instruments may vote a Share or some or all of the Shares in respect of which he is appointed either for or against a resolution and/or abstain from voting a Share or some or all of the Shares in respect of which he is appointed.

# 24 Proxies

24.1 The instrument appointing a proxy shall be in writing and shall be executed under the hand of the appointor or of his attorney duly authorised in writing, or, if the appointor is a corporation or other non natural Person, under the hand of its duly authorised representative. A proxy need not be a Member.

The Directors may, in the notice convening any meeting or adjourned meeting, or in an instrument of proxy sent out by the Company, specify the manner by which the instrument appointing a proxy shall be deposited and the place and the time (being not later than the time appointed for the commencement of the meeting or adjourned meeting to which the proxy relates) at which the instrument appointing a proxy shall be deposited. In the absence of any such direction from the Directors in the notice convening any meeting or adjourned meeting or in an instrument of proxy sent out by the Company, the instrument appointing a proxy shall be deposited physically at the Registered Office not less than 48 hours before the time appointed for the meeting or adjourned meeting to commence at which the Person named in the instrument proposes to vote.

The chairman may in any event at his discretion declare that an instrument of proxy shall be deemed to have been duly deposited.
 An instrument of proxy that is not deposited in the manner permitted, or which has not been declared to have been duly deposited by the chairman, shall be invalid.

24.4 The instrument appointing a proxy may be in any usual or common form (or such other form as the Directors may approve) and a proxy shall be deemed to include the power to demand or join or concur in demanding a poll.

Votes given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous death or insanity of the principal or revocation of the proxy or of the authority under which the proxy was executed, or the transfer of the Share

24.5 in respect of which the proxy is given unless notice in writing of such death, insanity, revocation or transfer was received by the Company at the Registered Office before the commencement of the general meeting, or adjourned meeting at which it is sought to use the proxy.

#### 25 Corporate Members

Any corporation or other non-natural Person which is a Member may in accordance with its constitutional documents, or in the absence of such provision by resolution of its directors or other governing body, authorise such Person as it thinks fit to act as its representative at any meeting of the Company or of any class of Members, and the Person so authorised shall be entitled to exercise the same powers on behalf of the corporation which he represents as the corporation could exercise if it were an individual Member.

If a Clearing House (or its nominee(s)), being a corporation, is a Member, it may authorise such Persons as it sees fit to act as its representative at any meeting of the Company or at any meeting of any class of Members provided that the authorisation shall specify the number and class of Shares in respect of which each such representative is so authorised. Each Person so authorised under the provisions of this Article shall be deemed to have been duly authorised without further evidence of the facts and be entitled to exercise the same rights and powers on behalf of the Clearing House (or its nominee(s)) as if such Person was the registered holder of such Shares held by the Clearing House (or its nominee(s)).

#### 26 Shares that May Not be Voted

Treasury Shares and other shares in the Company that are beneficially owned by the Company shall not be voted, directly or indirectly, at any meeting and shall not be counted in determining the total number of outstanding Shares at any given time.

#### 27 Directors

27.2

27.1 There shall be a board of Directors consisting of not less than one Person provided however that the Company may by Ordinary Resolution increase or reduce the limits in the number of Directors.

The Directors shall be divided into three classes: Class I, Class II and Class III. The number of Directors in each class shall be as nearly equal as possible. Upon the adoption of the Articles, the existing Directors shall by resolution classify themselves as Class I, Class II or Class III Directors. The Class I Directors shall stand appointed for a term expiring at the Company's first annual general meeting, the Class II Directors shall stand appointed for a term expiring at the Company's second annual general meeting and the Class III Directors shall stand appointed for a term expiring at the Company's first annual general meeting. Commencing at the Company's first annual general meeting, and at each annual general meeting thereafter, Directors appointed to succeed those Directors whose terms expire shall be appointed for a term of office to expire at the third succeeding annual general meeting after their appointment. Except as the Statute or other Applicable Law may otherwise require, in the interim between annual general meetings or extraordinary general meetings called for the appointment of Directors and/or the removal of one or more Directors and the filling of any vacancy in that connection, additional Directors and any vacancies in the board of Directors, including unfilled vacancies resulting from the removal of Directors for cause, may be filled by the vote of a majority

of the remaining Directors then in office, although less than a quorum (as defined in the Articles), or by the sole remaining Director. All Directors shall hold office until the expiration of their respective terms of office and until their successors shall have been appointed and qualified. A Director appointed to fill a vacancy resulting from the death, resignation or removal of a Director shall serve for the remainder of the full term of the Director whose death, resignation or removal shall have created such vacancy and until his successor shall have been appointed and qualified.

24

#### 28 Powers of Directors

Subject to the provisions of the Statute, the Memorandum and the Articles and to any directions given by Special Resolution, the business of the Company shall be managed by the Directors who may exercise all the powers of the Company. No alteration of

28.1 the Memorandum or Articles and no such direction shall invalidate any prior act of the Directors which would have been valid if that alteration had not been made or that direction had not been given. A duly convened meeting of Directors at which a quorum is present may exercise all powers exercisable by the Directors.

All cheques, promissory notes, drafts, bills of exchange and other negotiable or transferable instruments and all receipts for monies paid to the Company shall be signed, drawn, accepted, endorsed or otherwise executed as the case may be in such manner as the Directors shall determine by resolution.

The Directors on behalf of the Company may pay a gratuity or pension or allowance on retirement to any Director who has held any other salaried office or place of profit with the Company or to his widow or dependants and may make contributions to any fund and pay premiums for the purchase or provision of any such gratuity, pension or allowance.

28.4 The Directors may exercise all the powers of the Company to borrow money and to mortgage or charge its undertaking, property and assets (present and future) and uncalled capital or any part thereof and to issue debentures, debenture stock, mortgages, bonds and other such securities whether outright or as security for any debt, liability or obligation of the Company or of any third party.

# 29 Appointment and Removal of Directors

29.1 The Directors may appoint any Person to be a Director, either to fill a vacancy or as an additional Director provided that the appointment does not cause the number of Directors to exceed any number fixed by or in accordance with the Articles as the maximum number of Directors.

29.2 The Company may by Ordinary Resolution appoint any Person to be a Director or may by Ordinary Resolution remove any Director.

# **30** Vacation of Office of Director

Notwithstanding anything in these Articles, the office of a Director shall be vacated if:

(a) the Director gives notice in writing to the Company that he resigns the office of Director; or

the Director absents himself (for the avoidance of doubt, without being represented by proxy) from three consecutive(b) meetings of the board of Directors without special leave of absence from the Directors, and the Directors pass a resolution that he has by reason of such absence vacated office; or

(c) the Director dies, becomes bankrupt or makes any arrangement or composition with his creditors generally; or

#### 25

(i) with respect to any Director (other than Founder), a licensed medical practitioner who has evaluated that Director

(d) gives a written opinion to the Company stating he or she has become physically or mentally incapable of acting as a Director (with reasonable accommodations) and may remain so for more than three (3) months, or (ii) with respect to Founder his Incapacity shall have been determined.

# 31 Proceedings of Directors

31.1 The quorum for the transaction of the business of the Directors may be fixed by the Directors, and unless so fixed shall be a majority of the Directors then in office.

Subject to the provisions of the Articles, the Directors may regulate their proceedings as they think fit. Questions arising at any meeting shall be decided by a majority of votes. In the case of an equality of votes, the Chairman shall have a second or casting vote.

A Person may participate in a meeting of the Directors or any committee of Directors by conference telephone or other communications equipment by means of which all the Persons participating in the meeting can communicate with each other 31.3 at the same time. Participation by a Person in a meeting in this manner is treated as presence in Person at that meeting. Unless otherwise determined by the Directors, the meeting shall be deemed to be held at the place where the Chairman is located at the start of the meeting. 31.4 A resolution in writing (in one or more counterparts) signed by all the Directors or all the members of a committee of the Directors or, in the case of a resolution in writing relating to the removal of any Director or the vacation of office by any Director, all of the Directors other than the Director who is the subject of such resolution shall be as valid and effectual as if it had been passed at a meeting of the Directors, or committee of Directors as the case may be, duly convened and held.

31.5 A Director may, or other Officer on the direction of a Director shall, call a meeting of the Directors by at least two days' notice in writing to every Director which notice shall set forth the general nature of the business to be considered unless notice is waived by all the Directors either at, before or after the meeting is held. To any such notice of a meeting of the Directors all the provisions of the Articles relating to the giving of notices by the Company to the Members shall apply *mutatis mutandis*.

31.6 The continuing Directors (or a sole continuing Director, as the case may be) may act notwithstanding any vacancy in their body, but if and so long as their number is reduced below the number fixed by or pursuant to the Articles as the necessary quorum of Directors the continuing Directors or Director may act for the purpose of increasing the number of Directors to be equal to such fixed number, or of summoning a general meeting of the Company, but for no other purpose.

The Directors may elect a chairman of their board and determine the period for which he is to hold office; but if no such Chairman is elected, or if at any meeting the Chairman is not present within five minutes after the time appointed for the meeting to commence, the Directors present may choose one of their number to be chairman of the meeting.

31.8 All acts done by any meeting of the Directors or of a committee of the Directors shall, notwithstanding that it is afterwards discovered that there was some defect in the appointment of any Director, and/or that they or any of them were disqualified, and/ or had vacated their office and/or were not entitled to vote, be as valid as if every such Person had been duly appointed and/or not disqualified to be a Director and/or had not vacated their office and/or had been entitled to vote, as the case may be.

26

31.9 A Director may be represented at any meetings of the board of Directors by a proxy appointed in writing by him. The proxy shall count towards the quorum and the vote of the proxy shall for all purposes be deemed to be that of the appointing Director.

# 32 Presumption of Assent

A Director who is present at a meeting of the board of Directors at which action on any Company matter is taken shall be presumed to have assented to the action taken unless his dissent shall be entered in the minutes of the meeting or unless he shall file his written dissent from such action with the Person acting as the chairman or secretary of the meeting before the adjournment thereof or shall forward such dissent by registered post to such Person immediately after the adjournment of the meeting. Such right to dissent shall not apply to a Director who voted in favour of such action.

# 33 Directors' Interests

- 33.1 A Director may hold any other office or place of profit under the Company (other than the office of Auditor) in conjunction with his office of Director for such period and on such terms as to remuneration and otherwise as the Directors may determine.
- 33.2 A Director may act by himself or by, through or on behalf of his firm in a professional capacity for the Company and he or his firm shall be entitled to remuneration for professional services as if he were not a Director.
- 33.3 A Director may be or become a director or other officer of or otherwise interested in any company promoted by the Company or in which the Company may be interested as a shareholder, a contracting party or otherwise, and no such Director shall be accountable to the Company for any remuneration or other benefits received by him as a director or officer of, or from his interest in, such other company.

33.4 No Person shall be disqualified from the office of Director or prevented by such office from contracting with the Company, either as vendor, purchaser or otherwise, nor shall any such contract or any contract or transaction entered into by or on behalf of the Company in which any Director shall be in any way interested be or be liable to be avoided, nor shall any Director so contracting or being so interested be liable to account to the Company for any profit realised by or arising in connection with

any such contract or transaction by reason of such Director holding office or of the fiduciary relationship thereby established. A Director shall be at liberty to vote in respect of any contract or transaction in which he is interested provided that the nature of the interest of any Director in any such contract or transaction shall be disclosed by him at or prior to its consideration and any vote thereon.

33.5

A general notice that a Director is a shareholder, director, officer or employee of any specified firm or company and is to be regarded as interested in any transaction with such firm or company shall be sufficient disclosure for the purposes of voting on a resolution in respect of a contract or transaction in which he has an interest, and after such general notice it shall not be necessary to give special notice relating to any particular transaction.

#### 34 Minutes

The Directors shall cause minutes to be made in books kept for the purpose of recording all appointments of Officers made by the Directors, all proceedings at meetings of the Company or the holders of any class of Shares and of the Directors, and of committees of the Directors, including the names of the Directors present at each meeting.

#### **35 Delegation of Directors' Powers**

35.1 The Directors may delegate any of their powers, authorities and discretions, including the power to sub-delegate, to any committee consisting of one or more Directors (including, without limitation, the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee). Any such delegation may be made subject to any conditions the Directors may impose and either collaterally with or to the exclusion of their own powers and any such delegation may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of a committee of Directors shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

35.2 The Directors may establish any committees, local boards or agencies or appoint any Person to be a manager or agent for managing the affairs of the Company and may appoint any Person to be a member of such committees, local boards or agencies. Any such appointment may be made subject to any conditions the Directors may impose, and either collaterally with or to the exclusion of their own powers and any such appointment may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of any such committee, local board or agency shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

The Directors may adopt formal written charters for committees. Each of these committees shall be empowered to do all things necessary to exercise the rights of such committee set forth in the Articles and shall have such powers as the Directors may delegate pursuant to the Articles and as required by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law. Each of the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee, if established, shall consist of such number of Directors as the Directors shall from time to time determine (or such minimum number as may be required from time to time by the rules and regulatory authority or otherwise under Applicable Law). For so long as any class of Shares is listed on the Designated Stock Exchange, the Audit Committee, the Nominating and Corporate Governance Committee shall be made up of such number of Independent Directors as is required from time to time by the rules shall be made up of such number of Independent Directors as is required from time to time by the rules and regulations of the Designated Stock Exchange Commission and/or any other competent Stock Exchange, the Securities and Exchange and Corporate Governance Committee shall be made up of such number of Independent Directors as is required from time to time by the rules and regulations of the Designated Stock Exchange Commission and/or any other competent Stock Exchange, the Securities and Exchange Commission and/or any other competent Stock Exchange, the Securities and Exchange Commission and/or any other competent stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise and Exchange Commission and/or any other competent regulatory authority or securities and Exchange Commission and/or any other competent regulatory authority or otherwise and Exchange Commission and/or any other co

The Directors may by power of attorney or otherwise appoint any Person to be the agent of the Company on such conditions as the Directors may determine, provided that the delegation is not to the exclusion of their own powers and may be revoked by the Directors at any time.

35.5 The Directors may by power of attorney or otherwise appoint any company, firm, Person or body of Persons, whether nominated directly or indirectly by the Directors, to be the attorney or authorised signatory of the Company for such purpose and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under the Articles) and for such period and subject to such conditions as they may think fit, and any such powers of attorney or other appointment may contain

such provisions for the protection and convenience of Persons dealing with any such attorneys or authorised signatories as the Directors may think fit and may also authorise any such attorney or authorised signatory to delegate all or any of the powers, authorities and discretions vested in him.

35.6 The Directors may appoint such Officers as they consider necessary on such terms, at such remuneration and to perform such duties, and subject to such provisions as to disqualification and removal as the Directors may think fit. Unless otherwise specified in the terms of his appointment an Officer may be removed by resolution of the Directors or Members. An Officer may vacate his office at any time if he gives notice in writing to the Company that he resigns his office.

#### 36 No Minimum Shareholding

The Company in general meeting may fix a minimum shareholding required to be held by a Director, but unless and until such a shareholding qualification is fixed a Director is not required to hold Shares.

#### **37** Remuneration of Directors

37.1 The remuneration to be paid to the Directors, if any, shall be such remuneration as the Directors shall determine. The Directors shall also be entitled to be paid all travelling, hotel and other expenses properly and reasonably incurred by them in connection with their attendance at meetings of Directors or committees of Directors, or general meetings of the Company, or separate meetings of the holders of any class of Shares or debentures of the Company, or otherwise in connection with the business of the Company or the discharge of their duties as a Director, or to receive a fixed allowance in respect thereof as may be determined by the Directors, or a combination partly of one such method and partly the other.

The Directors may by resolution approve additional remuneration to any Director for any services which in the opinion of the Directors go beyond his ordinary routine work as a Director. Any fees paid to a Director who is also counsel, attorney or solicitor to the Company, or otherwise serves it in a professional capacity shall be in addition to his remuneration as a Director.

# 38 Seal

The Company may, if the Directors so determine, have a Seal. The Seal shall only be used by the authority of the Directors or of a committee of the Directors authorised by the Directors. Every instrument to which the Seal has been affixed shall be signed by at least one Person who shall be either a Director or some Officer or other Person appointed by the Directors for the purpose.

The Company may have for use in any place or places outside the Cayman Islands a duplicate Seal or Seals each of which shall be a facsimile of the common Seal of the Company and, if the Directors so determine, with the addition on its face of the name of every place where it is to be used.

A Director or Officer, representative or attorney of the Company may without further authority of the Directors affix the Seal over his signature alone to any document of the Company required to be authenticated by him under seal or to be filed with the Registrar of Companies in the Cayman Islands or elsewhere wheresoever.

29

#### **39** Dividends, Distributions and Reserve

Subject to the Statute and this Article and except as otherwise provided by the rights attached to any Shares, the Directors may resolve to pay Dividends and other distributions on Shares in issue and authorise payment of the Dividends or other distributions
out of the funds of the Company lawfully available therefor. A Dividend shall be deemed to be an interim Dividend unless the terms of the resolution pursuant to which the Directors resolve to pay such Dividend specifically state that such Dividend shall be a final Dividend. No Dividend or other distribution shall be paid except out of the realised or unrealised profits of the Company,

out of the share premium account or as otherwise permitted by law. Notwithstanding anything contained in these Articles, no Dividend shall be made on any Class A Shares unless a Dividend in equal proportion is made on the Class B Shares.

Except as otherwise provided by the rights attached to any Shares, all Dividends and other distributions shall be paid according to the par value of the Shares that a Member holds. If any Share is issued on terms providing that it shall rank for Dividend as from a particular date, that Share shall rank for Dividend accordingly.

39.3 The Directors may deduct from any Dividend or other distribution payable to any Member all sums of money (if any) then payable by him to the Company on account of calls or otherwise.

The Directors may resolve that any Dividend or other distribution be paid wholly or partly by the distribution of specific assets and in particular (but without limitation) by the distribution of shares, debentures, or securities of any other company or in any one or more of such ways and where any difficulty arises in regard to such distribution, the Directors may settle the same as they think expedient and in particular may issue fractional Shares and may fix the value for distribution of such specific assets or any part thereof and may determine that cash payments shall be made to any Members upon the basis of the value so fixed in order to adjust the rights of all Members and may vest any such specific assets in trustees in such manner as may seem expedient to the Directors.

Except as otherwise provided by the rights attached to any Shares, Dividends and other distributions may be paid in any currency. 39.5 The Directors may determine the basis of conversion for any currency conversions that may be required and how any costs involved are to be met.

The Directors may, before resolving to pay any Dividend or other distribution, set aside such sums as they think proper as a reserve or reserves which shall, at the discretion of the Directors, be applicable for any purpose of the Company and pending such application may, at the discretion of the Directors, be employed in the business of the Company.

Any Dividend, other distribution, interest or other monies payable in cash in respect of Shares may be paid by wire transfer to the holder or by cheque or warrant sent through the post directed to the registered address of the holder or, in the case of joint holders, to the registered address of the holder who is first named on the Register of Members or to such Person and to such address as such holder or joint holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the Person to whom it is sent. Any one of two or more joint holders may give effectual receipts for any Dividends, other distributions, bonuses, or other monies payable in respect of the Share held by them as joint holders.

39.8 No Dividend or other distribution shall bear interest against the Company.

30

39.9 Any Dividend or other distribution which cannot be paid to a Member and/or which remains unclaimed after six months from the date on which such Dividend or other distribution becomes payable may, in the discretion of the Directors, be paid into a separate account in the Company's name, provided that the Company shall not be constituted as a trustee in respect of that account and the Dividend or other distribution shall remain as a debt due to the Member. Any Dividend or other distribution which remains unclaimed after a period of six years from the date on which such Dividend or other distribution becomes payable shall be forfeited and shall revert to the Company.

# 40 Capitalisation

The Directors may at any time capitalise any sum standing to the credit of any of the Company's reserve accounts or funds (including the share premium account and capital redemption reserve fund) or any sum standing to the credit of the profit and loss account or otherwise available for distribution; appropriate such sum to Members in the proportions in which such sum would have been divisible amongst such Members had the same been a distribution of profits by way of Dividend or other distribution; and apply such sum on their behalf in paying up in full unissued Shares for allotment and distribution credited as fully paid-up to and amongst them in the proportion aforesaid. In such event the Directors shall do all acts and things required to give effect to such capitalisation, with full power given to the Directors to make such provisions as they think fit in the case of Shares becoming distributable in fractions (including provisions whereby the benefit of fractional entitlements accrue to the Company rather than to the Members concerned). The Directors may authorise any Person to enter on behalf of all of

the Members interested into an agreement with the Company providing for such capitalisation and matters incidental or relating thereto and any agreement made under such authority shall be effective and binding on all such Members and the Company.

# 41 Books of Account

41.1 The Directors shall cause proper books of account (including, where applicable, material underlying documentation including contracts and invoices) to be kept with respect to all sums of money received and expended by the Company and the matters in respect of which the receipt or expenditure takes place, all sales and purchases of goods by the Company and the assets and liabilities of the Company. Such books of account must be retained for a minimum period of five years from the date on which they are prepared. Proper books shall not be deemed to be kept if there are not kept such books of account as are necessary to give a true and fair view of the state of the Company's affairs and to explain its transactions.

41.2 The Directors shall determine whether and to what extent and at what times and places and under what conditions or regulations the accounts and books of the Company or any of them shall be open to the inspection of Members not being Directors and no Member (not being a Director) shall have any right of inspecting any account or book or document of the Company except as conferred by Statute or authorised by the Directors or by the Company in general meeting.

41.3 The Directors may cause to be prepared and to be laid before the Company in general meeting profit and loss accounts, balance sheets, group accounts (if any) and such other reports and accounts as may be required by law.

31

#### 42 Audit

42.1 The Directors may appoint an Auditor of the Company who shall hold office on such terms as the Directors determine.

Without prejudice to the freedom of the Directors to establish any other committee, if the Shares (or depositary receipts therefor) are listed or quoted on the Designated Stock Exchange, and if required by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, the Directors shall establish and maintain an Audit Committee as a committee of the Directors and shall adopt

42.2 Applicable Law, the Directors shall establish and maintain an Addit Committee as a committee of the Directors and shall adopt a formal written Audit Committee charter and review and assess the adequacy of the formal written charter on an annual basis. The composition and responsibilities of the Audit Committee shall comply with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law.

If the Shares (or depositary receipts therefor) are listed or quoted on the Designated Stock Exchange, the Company shall conduct
 an appropriate review of all related party transactions on an ongoing basis and shall utilise the Audit Committee for the review and approval of potential conflicts of interest.

42.4 The remuneration of the Auditor shall be fixed by the Audit Committee (if one exists).

If the office of Auditor becomes vacant by resignation or death of the Auditor, or by his becoming incapable of acting by reason of illness or other disability at a time when his services are required, the Directors shall fill the vacancy and determine the remuneration of such Auditor.

Every Auditor of the Company shall have a right of access at all times to the books and accounts and vouchers of the Company 42.6 and shall be entitled to require from the Directors and Officers such information and explanation as may be necessary for the performance of the duties of the Auditor.

Auditors shall, if so required by the Directors, make a report on the accounts of the Company during their tenure of office at the next annual general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an ordinary company, and at the next extraordinary general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an exempted company, and at any other time during their term of office, upon request of the Directors or any general meeting of the Members. At least one member of the Audit Committee shall be an "audit committee financial expert" as determined by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law. The "audit committee financial expert" shall have such past employment experience in finance or accounting, requisite professional certification in accounting, or any other comparable experience or background which results in the individual's financial sophistication.

#### 43 Notices

43.3

Notices shall be in writing and may be given by the Company to any Member either personally or by sending it by courier, post, cable, telex, fax or e-mail to him or to his address as shown in the Register of Members (or where the notice is given by e-mail by sending it to the e-mail address provided by such Member). Notice may also be served by Electronic Communication

43.1 e-mail by sending it to the e-mail address provided by such Member). Notice may also be served by Electronic Communication in accordance with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/ or any other competent regulatory authority or by placing it on the Company's Website.

#### 43.2 Where a notice is sent by:

courier; service of the notice shall be deemed to be effected by delivery of the notice to a courier company, and shall be(a) deemed to have been received on the third day (not including Saturdays or Sundays or public holidays) following the day on which the notice was delivered to the courier;

post; service of the notice shall be deemed to be effected by properly addressing, pre paying and posting a letter containing(b) the notice, and shall be deemed to have been received on the fifth day (not including Saturdays or Sundays or public holidays in the Cayman Islands) following the day on which the notice was posted;

(c) cable, telex or fax; service of the notice shall be deemed to be effected by properly addressing and sending such notice and shall be deemed to have been received on the same day that it was transmitted;

e-mail or other Electronic Communication; service of the notice shall be deemed to be effected by transmitting the e-mail(d) to the e-mail address provided by the intended recipient and shall be deemed to have been received on the same day that it was sent, and it shall not be necessary for the receipt of the e-mail to be acknowledged by the recipient; and

(e) placing it on the Company's Website; service of the notice shall be deemed to have been effected one hour after the notice or document was placed on the Company's Website.

A notice may be given by the Company to the Person or Persons which the Company has been advised are entitled to a Share or Shares in consequence of the death or bankruptcy of a Member in the same manner as other notices which are required to be given under the Articles and shall be addressed to them by name, or by the title of representatives of the deceased, or trustee of the bankrupt, or by any like description at the address supplied for that purpose by the Persons claiming to be so entitled, or at the option of the Company by giving the notice in any manner in which the same might have been given if the death or bankruptcy had not occurred.

43.4 Notice of every general meeting shall be given in any manner authorised by the Articles to every holder of Shares carrying an entitlement to receive such notice on the record date for such meeting except that in the case of joint holders the notice shall be sufficient if given to the joint holder first named in the Register of Members and every Person upon whom the ownership of a Share devolves by reason of his being a legal personal representative or a trustee in bankruptcy of a Member where the Member but for his death or bankruptcy would be entitled to receive notice of the meeting, and no other Person shall be entitled to receive notices of general meetings.

# 44 Winding Up

- 44.1 If the Company shall be wound up, the liquidator shall apply the assets of the Company in satisfaction of creditors' claims in such manner and order as such liquidator thinks fit. Subject to the rights attaching to any Shares, in a winding up:
  - if the assets available for distribution amongst the Members shall be insufficient to repay the whole of the Company's issued
  - (a) share capital, such assets shall be distributed so that, as nearly as may be, the losses shall be borne by the Members in proportion to the par value of the Shares held by them; or

if the assets available for distribution amongst the Members shall be more than sufficient to repay the whole of the Company's issued share capital at the commencement of the winding up, the surplus shall be distributed amongst the(b) Members in proportion to the par value of the Shares held by them at the commencement of the winding up subject to a

(b) Members in proportion to the par value of the Shares held by them at the commencement of the winding up subject to a deduction from those Shares in respect of which there are monies due, of all monies payable to the Company for unpaid calls or otherwise.

If the Company shall be wound up the liquidator may, subject to the rights attaching to any Shares and with the approval of a Special Resolution of the Company and any other approval required by the Statute, divide amongst the Members in kind the whole or any part of the assets of the Company (whether such assets shall consist of property of the same kind or not) and may for that purpose value any assets and determine how the division shall be carried out as between the Members or different classes of Members. The liquidator may, with the like approval, vest the whole or any part of such assets in trustees upon such trusts for the benefit of the Members as the liquidator, with the like approval, shall think fit, but so that no Member shall be compelled to accept any asset upon which there is a liability.

# 45 Indemnity and Insurance

45.1

Every Director and Officer (which for the avoidance of doubt, shall not include auditors of the Company), together with every former Director and former Officer (each an "**Indemnified Person**") shall be indemnified out of the assets of the Company against any liability, action, proceeding, claim, demand, costs, damages or expenses, including legal expenses, whatsoever which they or any of them may incur as a result of any act or failure to act in carrying out their functions other than such liability (if any) that they may incur by reason of their own actual fraud, wilful neglect or wilful default. No Indemnified Person shall be liable to the Company for any loss or damage incurred by the Company as a result (whether direct or indirect) of the carrying out of their functions unless that liability arises through the actual fraud, wilful neglect or wilful default of such Indemnified Person. No Person shall be found to have committed actual fraud, wilful neglect or wilful default under this Article unless or until a court of competent jurisdiction shall have made a finding to that effect.

The Company shall advance to each Indemnified Person reasonable attorneys' fees and other costs and expenses incurred in connection with the defence of any action, suit, proceeding or investigation involving such Indemnified Person for which indemnity will or could be sought. In connection with any advance of any expenses hereunder, the Indemnified Person shall execute an undertaking to repay the advanced amount to the Company if it shall be determined by final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification pursuant to this Article. If it shall be determined by a final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification with respect to such judgment, costs or expenses, then such party shall not be indemnified with respect to such judgment, costs or expenses and any advancement shall be returned to the Company (without interest) by the Indemnified Person.

34

The Directors, on behalf of the Company, may purchase and maintain insurance for the benefit of any Director or Officer against
 any liability which, by virtue of any rule of law, would otherwise attach to such Person in respect of any negligence, default, breach of duty or breach of trust of which such Person may be guilty in relation to the Company.

# 46 Financial Year

Unless the Directors otherwise prescribe, the financial year of the Company shall end on 31st December in each year and, following the year of incorporation, shall begin on 1st January in each year.

#### 47 Transfer by Way of Continuation

If the Company is exempted as defined in the Statute, it shall, subject to the provisions of the Statute and with the approval of a Special Resolution, have the power to register by way of continuation as a body corporate under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.

#### 48 Mergers and Consolidations

The Company shall have the power to merge or consolidate with one or more other constituent companies (as defined in the Statute) upon such terms as the Directors may determine and (to the extent required by the Statute) with the approval of a Special Resolution.

#### 49 Business Opportunities

49.1

To the fullest extent permitted by Applicable Law, no individual serving as a Director or an Officer ("**Management**") shall have any duty, except and to the extent expressly assumed by contract, to refrain from engaging directly or indirectly in the same or similar business activities or lines of business as the Company. To the fullest extent permitted by Applicable Law, the Company renounces any interest or expectancy of the Company in, or in being offered an opportunity to participate in, any potential transaction or matter which may be a corporate opportunity for Management, on the one hand, and the Company, on the other. Except to the extent expressly assumed by contract, to the fullest extent permitted by Applicable Law, Management shall

have no duty to communicate or offer any such corporate opportunity to the Company and shall not be liable to the Company or its Members for breach of any fiduciary duty as a Member, Director and/or Officer solely by reason of the fact that such party pursues or acquires such corporate opportunity for itself, himself or herself, directs such corporate opportunity to another Person, or does not communicate information regarding such corporate opportunity to the Company.

49.2 Except as provided elsewhere in this Article, the Company hereby renounces any interest or expectancy of the Company in, or in being offered an opportunity to participate in, any potential transaction or matter which may be a corporate opportunity for both the Company and Management, about which a Director and/or Officer who is also a member of Management acquires knowledge.

To the extent a court might hold that the conduct of any activity related to a corporate opportunity that is renounced in this Article to be a breach of duty to the Company or its Members, the Company hereby waives, to the fullest extent permitted by

49.3 Applicable Law, any and all claims and causes of action that the Company may have for such activities. To the fullest extent permitted by Applicable Law, the provisions of this Article apply equally to activities conducted in the future and that have been conducted in the past.

35

# BUSINESS COMBINATION AGREEMENT by and among TRADEUP GLOBAL CORPORATION, as Purchaser,

# TGC MERGER SUB,

as Merger Sub,

and

SAITECH LIMITED,

# as the Company, Dated as of September 27, 2021

# **TABLE OF CONTENTS**

Page

ARTICLE	I THE MERGER	2
1.1	The Merger	2
1.2	Closing Statements	4
1.3	Withholding	4
1.4	Tax Consequences	4
1.5	Dissenter's Rights	4
1.6	Exchange Agent	5
ARTICLE	II REPRESENTATIONS AND WARRANTIES OF PURCHASER	6
2.1	Organization and Standing	6
2.2	Authorization: Binding Agreement	6
2.3	Governmental Approvals	6
2.4	Non-Contravention	7
2.5	Capitalization	7
2.6	SEC Filings: Purchaser Financials: Internal Controls	8
2.7	Absence of Certain Changes	9
2.8	Compliance with Laws	9
2.9	Actions; Orders; Permits	9
2.10	Taxes and Returns	9
2.11	Employees and Employee Benefit Plans	10
2.12	Properties	10
2.13	Material Contracts	10
2.14	Transactions with Affiliates	10
2.15	Investment Company Act; JOBS Act	11
2.16	Finders and Brokers	11
2.17	Certain Business Practices	11
2.18	Insurance	11
2.19	Information Supplied	11
2.20	Independent Investigation	12
2.21	Trust Account	12
2.22	EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES	12
	III REPRESENTATIONS AND WARRANTIES OF MERGER SUB	13
3.1	Organization and Standing	13
3.2	Authorization: Binding Agreement	13
3.3	Governmental Approvals	13
3.4	Non-Contravention	13
3.5	Capitalization	14
3.6	Merger Sub Activities	14
3.7	Compliance with Laws	14

3.8	Actions; Orders	14
3.9	Transactions with Related Persons	14
3.10	Finders and Brokers	14
3.11	Investment Company Act	14

i

	3.12	Intended Tax Treatment	14
	3.13	Information Supplied	14
	3.14	EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES	15
AR'		IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY	15
	4.1	Organization and Standing	15
	4.2	Authorization: Binding Agreement	15
	4.3	Capitalization	16
	4.4	Subsidiaries	16
	4.5	Governmental Approvals	17
	4.6	Non-Contravention	17
	4.7	Financial Statements	17
	4.8	Absence of Certain Changes	18
	4.9	Compliance with Laws	18
	4.10	Company Permits	18
	4.11	Litigation	18
	4.12	Material Contracts	19
	4.13	Intellectual Property	20
	4.14	Privacy and Data Security	22
	4.15	Taxes and Returns	22
	4.16	Title; Real Property	24
	4.17	Employee and Labor Matters	25
	4.18	Benefit Plans	25
	4.19	Transactions with Related Persons	27
	4.20	Certain Business Practices	27
	4.21	Investment Company Act	27
	4.22	Finders and Brokers	27
	4.23	Insurance	27
	4.24	Ownership of Digital Assets	28
	4.25	Information Supplied	28
	4.26	Independent Investigation	28
	4.27	EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES	28
			•••
AR		V COVENANTS	<b>29</b>
	5.1	Access and Information	29
	5.2	Conduct of Business of the Company	30
	5.3	Conduct of Business of Purchaser	32
	5.4	Purchaser Public Filings	34
	5.5	No Solicitation	34
	5.6	No Trading	35
	5.7	Efforts	35
	5.8	Further Assurances	36
	5.9	The Registration Statement	36
	5.10	Required Company Shareholder Approval	38
	5.12	Public Announcements	38
	5.13	Confidential Information	39
	5.14	Post-Closing Board of Directors and Executive Officers	40

	5.15	Purchaser Equity Plan	40	
	5.16	Indemnification of Directors and Officers	40	
	5.17	Section 16 Matters	41	
	5.18	Trust Account Proceeds	41	
	5.19	Tax Matters	41	
	5.20	280G	42	
	5.21	Financials	42	
	5.22	Purchaser	42	
	5.23	Amendment to Disclosure Schedules	42	
	0.20			
ARTICLE VI SURVIVAL 43				
	6.1	Non-Survival of Representations, Warranties and Covenants	43	
	011			
ARTICLE VII CLOSING CONDITIONS 43				
	7.1	Conditions to Each Party's Obligations	43	
	7.2	Conditions to Obligations of the Company	43	
	7.3	Conditions to Obligations of Purchaser	44	
	7.4	Frustration of Conditions	45	
	,.ı		15	
ART	ICLEV	VIII TERMINATION AND EXPENSES	46	
	8.1	Termination	46	
	8.2	Effect of Termination	46	
	8.3	Fees and Expenses	47	
	0.5	r ees und Expenses	- 17	
ART	ICLEI	X WAIVERS AND RELEASES	47	
	9.1	Waiver of Claims Against Trust	47	
	<i>.</i>		1,	
ARTICLE X MISCELLANEOUS 48				
	10.1	Notices	48	
	10.1	Binding Effect: Assignment	49	
	10.2	Third Parties	49	
	10.4	Governing Law; Jurisdiction	49	
	10.5	WAIVER OF JURY TRIAL	49	
	10.6	Specific Performance	49	
	10.7	Severability	49	
	10.8	Amendment	50	
	10.0	Waiver	50	
		Entire Agreement	50	
		Interpretation	50	
		Counterparts	50	
		No Recourse	51	
		Legal Representation	51	
	10.14		51	
ARTICLE XI DEFINITIONS 52				
ANI	11.1	Certain Definitions	52 52	
	11.1	Section References	61	
	11.4		01	

## **INDEX OF EXHIBITS**

<u>Exhibit</u>	Description
Exhibit A-1	Form of Lock-Up Agreement (Sponsor, Affiliates and Other Parties)
Exhibit A-2	Form of Lock-Up Agreement (Founder and Management)
Exhibit A-3	Form of Lock-Up Agreement (Other Company Shareholders)
Exhibit B	Form of Letter Agreement Amendment
Exhibit C	Form of Company Shareholder Support Agreement
Exhibit D	Form of Sponsor Support Agreement
Exhibit E	Forms of Employment Agreements
Exhibit F	Form of Registration Rights Agreement
Exhibit G	Form of Surviving Company Memorandum and Articles of Association
Exhibit H	Form of Amended Purchaser Charter
Exhibit I	Form of Waiver and Release
Exhibit J	Form of Purchaser Equity Plan

iv

# **BUSINESS COMBINATION AGREEMENT**

This Business Combination Agreement (this "Agreement") is made and entered into as of September 27, 2021 by and among TradeUp Global Corporation, a Cayman Islands exempted company incorporated with limited liability with company number 370735 (together with its successors, "Purchaser"), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and a direct wholly owned subsidiary of Purchaser ("Merger Sub") and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "Company"). Purchaser, Merger Sub and the Company are sometimes referred to herein individually as a "Party" and, collectively, as the "Parties".

## **RECITALS:**

WHEREAS, Purchaser is a blank check company and was formed for the purpose of effecting a merger, share exchange, asset acquisition, share purchase, reorganization or similar business combination with one or more businesses;

WHEREAS, Merger Sub is a newly incorporated Cayman Islands exempted company, wholly owned by Purchaser, and was formed for the purpose of effectuating the Merger;

WHEREAS, the Parties desire and intend to effect a business combination transaction whereby Merger Sub will merge with and into the Company (the "*Merger*") with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser, such Merger to occur upon the terms and subject to the conditions set forth in this Agreement and in accordance with the applicable provisions of the Cayman Act;

WHEREAS, simultaneously with the execution and delivery of this Agreement, TradeUp Global Sponsor LLC (the "Sponsor"), certain Affiliates of Purchaser and the Sponsor and other parties have each entered into a letter agreement with Purchaser, the form of which is attached as Exhibit B hereto (the "Letter Agreement Amendment");

**WHEREAS**, simultaneously with the execution and delivery of this Agreement, the Founder, Risheng Li, and certain other Affiliates and shareholders of the Company have entered into a Support Agreement with Purchaser and the Company, the form of which is attached as <u>Exhibit C</u> hereto (the *"Company Shareholder Support Agreement"*);

**WHEREAS**, simultaneously with the execution and delivery of this Agreement, the Sponsor and certain Affiliates of the Purchaser have entered into a Support Agreement with Purchaser and the Company, the form of which is attached as <u>Exhibit D</u> hereto (the "*Sponsor Support Agreement*");

**WHEREAS**, simultaneously with the execution and delivery of this Agreement, each of the Chief Executive Officer and the Chief Financial Officer of the Company have entered into an Employment and Non-Competition Agreement in the respective forms attached as <u>Exhibit E</u> hereto (the "*Employment Agreements*"), which agreements will become effective as of the Closing;

**WHEREAS**, in connection with the consummation of the Merger, certain equityholders of Purchaser and certain shareholders of the Company will, on or prior to the Closing, enter into an amended and restated registration rights agreement to, among other matters, have such rights apply to the Purchaser Securities, and provide those shareholders of the Company with registration rights, the form of which is attached as <u>Exhibit F</u> hereto (the "*Registration Rights Agreement*");

WHEREAS, the boards of directors of Purchaser, Merger Sub and the Company have each (a) determined that the Merger and the other transactions contemplated by this Agreement and the Ancillary Documents (the "*Transactions*") are fair, advisable and in the best interests of their respective companies and shareholders, and (b) approved this Agreement and the Transactions, upon the terms and subject to the conditions set forth herein;

WHEREAS, Purchaser, as the sole shareholder of Merger Sub, has approved this Agreement and the Transactions, upon the terms and subject to the conditions set forth herein;

**WHEREAS**, for U.S. federal income tax purposes, (a) it is intended that the Merger will qualify as a "reorganization" under Section 368(a) of the Code and (b) this Agreement is intended to constitute and hereby is adopted as a "plan of reorganization" with respect to the Merger within the meaning of Treasury Regulations Sections 1.368- 2(g) and 1.368-3(a) for purposes of Section 368 of the Code and the Treasury Regulations thereunder (collectively, the "*Intended Tax Treatment*"); and

WHEREAS, certain capitalized terms used herein are defined in Article XI hereof.

1

**NOW, THEREFORE**, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and the representations, warranties, covenants and agreements contained in this Agreement, and intending to be legally bound hereby, the Parties hereto agree as follows:

#### ARTICLE I THE MERGER

1.1 The Merger.

(a) <u>Closing</u>. As promptly as practicable following (and in any event no later than the third Business Day) following the satisfaction or (to the extent permitted by applicable Law) waiver of all the conditions set forth in <u>Sections 7.1 to 7.3</u> (other than any conditions that by their terms or nature are to be satisfied at the Closing) the consummation of the transactions contemplated by this Agreement with respect to the Merger (the *"Closing"*) shall take place electronically by the mutual exchange of electronic signatures (including portable document format (.PDF)) (the date on which the Closing is actually held being the *"Closing Date"*). On the Merger Effective Date, and subject to and upon the terms and conditions of this Agreement, and in accordance with the Cayman Act, the Company and Merger Sub shall consummate the Merger, pursuant to which Merger Sub shall be merged with and into the Company with the Company being the surviving entity, following which the separate corporate existence of Merger Sub shall cease and the Company shall continue as the surviving company. The Company, as the surviving company after the Merger, is hereinafter sometimes referred to as the "*Surviving Company*" (provided, that references to the Company for periods after the Merger Effective Date shall include the Surviving Company).

(b) <u>Merger Effective Date</u>. On the terms and subject to the conditions set forth herein and in accordance with the Cayman Act, on (or prior to, but effective on), the Closing Date, the Company and Merger Sub shall cause a plan of merger, in a form reasonably satisfactory to the Company and Purchaser (with such modifications, amendments or supplements thereto as may be required to comply with the Cayman Act), along with all other documentation and declarations required under the Cayman Act in connection with the Merger, to be duly executed and properly filed with the Cayman Registrar, in accordance with the relevant provisions of the Cayman Act (together, the "*Merger Documents*"). The Merger shall become effective on the date and time at which the Merger Documents have been duly filed with the Cayman Registrar or on a subsequent date and time as is agreed by Purchaser and the Company and specified

in the Merger Documents in accordance with the Cayman Act (the time the Merger becomes effective being referred to herein as the "Merger Effective Date").

(c) <u>Effect of the Merger</u>. On and after the Merger Effective Date, the effect of the Merger shall be as provided in this Agreement, the Merger Documents and the applicable provisions of the Cayman Act. Without limiting the generality of the foregoing, and subject thereto, under the Cayman Act, on the Merger Effective Date, all the property, rights, privileges, agreements, powers and franchises, debts, Liabilities, duties and obligations of Merger Sub and the Company shall become the property, rights, privileges, agreements, powers and franchises, debts, Liabilities, duties and obligations of the Surviving Company, which shall include the assumption by the Surviving Company of any and all agreements, covenants, duties and obligations of Merger Sub and the Company set forth in this Agreement to be performed after the Merger Effective Date, and the Surviving Company shall continue its existence as a wholly-owned Subsidiary of Purchaser.

(d) <u>Organizational Documents of the Surviving Company</u>. On the Merger Effective Date, in accordance with the Merger Documents, the form of amended and restated memorandum and articles of association of Surviving Company in the form attached as <u>Exhibit G</u> hereto, shall be adopted as the amended and restated memorandum and articles of association of the Surviving Company, until thereafter amended in accordance with the applicable provisions of the Cayman Act and such memorandum and articles of association.

(e) <u>Directors and Officers of the Surviving Company.</u> On the Merger Effective Date, the board of directors and officers of Merger Sub shall cease to hold office, and the board of directors and the officers of the Surviving Company shall be appointed as determined by the Company, each to hold office in accordance with the memorandum and articles of association of the Surviving Company until their respective successors are duly elected or appointed and qualified.

2

(f) Effect of the Merger on Issued Securities of the Company, At the Merger Effective Date, by virtue of the Merger and without any action on the part of any Party or the holders of securities of the Company, Purchaser or Merger Sub:

(i) *Company Preferred Shares*. Immediately prior to the Merger Effective Date, each outstanding Company Series A Preferred Share, Company Series Pre-A Preferred Share, Company Series Angel Preferred Share and Company Series Seed Preferred Share (other than those described in <u>Section 1.1(f)(iii)</u> below and the Dissenting Company Shares) shall automatically convert into one Company Class A Ordinary Share in accordance with the Organizational Documents of the Company. On the Merger Effective Date, every issued and outstanding Company Class A Ordinary Share resulting from the conversion described in the preceding sentence shall be converted automatically into such number of newly issued Purchaser Class A Ordinary Shares that is equal to the Exchange Ratio; <u>provided</u>, that notwithstanding anything to the contrary contained herein, no fraction of a Purchaser Class A Ordinary Share will be issued by virtue of the Merger or the other Transactions, and each Person who would otherwise be entitled to a fraction of a Purchaser Class A Ordinary Shares that otherwise would be received by such Person) shall instead have the number of Purchaser Class A Ordinary Shares issued to such Person rounded up in the aggregate to the nearest whole Purchaser Class A Ordinary Share, following which, all such Company Series A Preferred Share, Company Series Pre-A Preferred Share, Company Series Angel Preferred Share and Company Series Seed Preferred Share shall cease to be outstanding and shall automatically be canceled and shall cease to exist.

(ii) Company Class B Ordinary Shares. On the Merger Effective Date, every issued and outstanding Company Class B Ordinary Share shall be converted automatically into such number of newly issued Purchaser Class B Ordinary Shares that is equal to the Exchange Ratio; provided, that notwithstanding anything to the contrary contained herein, no fraction of a Purchaser Class B Ordinary Share will be issued by virtue of the Merger or the other Transactions, and each Person who would otherwise be entitled to a fraction of a Purchaser Class B Ordinary Shares that otherwise would be received by such Person) shall instead have the applicable class of Purchaser Class B Ordinary Shares issued to such Person rounded up in the aggregate to the nearest whole Purchaser Class B Ordinary Share, following which, all such Company Class B Ordinary Shares shall cease to be outstanding and shall automatically be canceled and shall cease to exist.

(iii) Company Class A Ordinary Shares. On the Merger Effective Date, every issued and outstanding Company Class A Ordinary Share shall be converted automatically into such number of newly issued Purchaser Class A Ordinary Shares that is equal to the Exchange Ratio; provided, that notwithstanding anything to the contrary contained herein, no fraction of

a Purchaser Class A Ordinary Share will be issued by virtue of the Merger or the other Transactions, and each Person who would otherwise be entitled to a fraction of a Purchaser Class A Ordinary Share (after aggregating all fractional shares of the applicable class of Purchaser Class A Ordinary Shares that otherwise would be received by such Person) shall instead have the number of Purchaser Class A Ordinary Shares issued to such Person rounded up in the aggregate to the nearest whole Purchaser Class A Ordinary Share, following which, all such Company Class A Ordinary Shares shall cease to be outstanding and shall automatically be canceled and shall cease to exist.

(iv) Cancellation of Share Capital Owned by the Company. On the Merger Effective Date, if there are any issued shares of the Company that are owned by the Company as treasury shares, such shares shall be canceled and extinguished without any conversion thereof or payment therefor.

(v) *No Liability.* Notwithstanding anything to the contrary in this <u>Section 1.1(f)</u>, none of the Surviving Company, Purchaser or any other Party hereto shall be liable to any Person for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar law.

(g) Effect of the Merger on Merger Sub Shares. On the Merger Effective Date, by virtue of the Merger and without any action on the part of any Party or any equityholder of Merger Sub, all of Merger Sub Ordinary Shares issued and outstanding immediately prior to the Merger Effective Date shall be converted into an equal number of ordinary shares of the Surviving Company, with the same rights, powers and privileges as the shares so converted and shall constitute the only outstanding shares of share capital of the Surviving Company.

3

(h) <u>Taking of Necessary Action</u>; <u>Further Action</u>. If, at any time after the Merger Effective Date, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Surviving Company with full right, title and possession to all assets, property, rights, privileges, powers and franchises of the Company and Merger Sub, the officers and directors of the Company and Merger Sub are fully authorized in the name of their respective entities to take, and will take, all such lawful and necessary action, so long as such action is not inconsistent with this Agreement.

## 1.2 Closing Statements.

(a) Three (3) Business Days prior to the Closing, Purchaser shall deliver to the Company a statement (the *"Purchaser Closing Statement*") setting forth, in each case as of the Reference Time: (i) the aggregate amount of cash in the Trust Account (prior to giving effect to the Redemption), (ii) the aggregate amount of all payments required to be made in connection with the Redemption, (iii) the net cash of Purchaser, after giving effect to the Redemption, (iv) the Purchaser Transaction Expenses, including the amount owed to each payee thereof and payment instructions therefor, (v) the Purchaser Indebtedness and (vi) a calculation of Excess Purchaser Indebtedness and Liability Amount, if any.

(b) Three (3) Business Days prior to the Closing, but in any case following receipt of the Purchaser Closing Statement, the Company shall deliver to Purchaser a statement (the "*Company Closing Statement*") setting forth, (i) a schedule setting forth Digital Assets owned by the Target Companies as of the date of the Company Closing Statement and the Digital Asset Market Value of such Digital Assets and (ii) in each case as of the Reference Time, (A) the Closing Company Cash, (B) the Company Indebtedness, (C) the Company Transaction Expenses, including the amount owed to each payee thereof and payment instructions therefor and (D) the resulting amount of the Merger Consideration and Exchange Ratio.

(c) From and after the delivery of the Purchaser Closing Statement or the Company Closing Statement, as the case may be, until the Closing Date, each of the Company and Purchaser shall (i) provide the other Parties and their Representatives with reasonable access to information reasonably requested by the Company or Purchaser or any of their respective Representatives in connection with the review of the Purchaser Closing Statement or the Company Closing Statement, as the case may be, (ii) consider in good faith any comments to the Purchaser Closing Statement or the Company Closing Statement, as the case may be, provided by any other Party prior to the Closing Date and (iii) revise the Purchaser Closing Statement or the Company Closing Statement, as the case may be, as needed to reflect any reasonable comments and any other comments that, based on its good faith assessment, are warranted or appropriate and deliver such revised Purchaser Closing Statement or Company Closing Statement, as the case may be, to any other Party prior to the Closing Date reflecting any such changes. It is understood and agreed that whether or not the Parties have fully resolved all comments to the Company Closing Statement or the Purchaser Closing Statement, such failure shall not affect, condition or delay the Closing, and the Closing shall occur based on the information set forth in the last agreed upon version of the Purchaser Closing Statement and the Company Closing Statement, as applicable.

1.3 Withholding. Each of Purchaser, the Surviving Company, Merger Sub and the Exchange Agent (and each of their respective Affiliates) (each, a *"Withholding Party"*) shall be entitled to deduct and withhold (or cause to be deducted and withheld) from any consideration payable pursuant to this Agreement such amounts as are required to be deducted and withheld under applicable Tax Law; provided, that prior to deducting and withholding any amounts with respect to any payment to be made pursuant to this Agreement, the applicable Withholding Party shall use commercially reasonable efforts to notify the Person in respect of which such deduction and withholding and to cooperate with such Person to reduce or eliminate any such deduction or withholding to the extent permitted by applicable Law. To the extent that amounts are so withheld and timely remitted to the applicable Governmental Authority, such withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made.

1.4 <u>Tax Consequences</u>. Each of the Parties intend that the Merger qualify for the Intended Tax Treatment and agree not to take any inconsistent position on any Tax Return or during the course of any audit, litigation or other proceeding with respect to Taxes except to the extent required by a "determination" within the meaning of Section 1313(a) of the Code. Each of the Parties acknowledge and agree that each has had the opportunity to obtain independent legal and Tax advice with respect to the transactions contemplated by this Agreement.

1.5 Dissenter's Rights. Notwithstanding any provision of this Agreement to the contrary and to the extent available under the Cayman Act, Company Shares that are outstanding immediately prior to the Merger Effective Date and that are held by shareholders of the Company who shall have demanded properly in writing dissenters' rights for such Company Shares in accordance with Section 238 of the Cayman Act and otherwise complied with all of the provisions of the Cayman Act relevant to the exercise and perfection of dissenters' rights (the "Dissenting Company Shares") shall not be converted into, and such shareholders shall have no right to receive, the applicable Merger Consideration unless and until such shareholder fails to perfect or withdraws or otherwise loses his, her or its right to dissenters' rights under the Cayman Act. The Company Shares owned by any shareholder of the Company who fails to perfect or who effectively withdraws or otherwise loses his, her or its dissenters' rights pursuant to the Cayman Act shall thereupon be deemed to have been converted into, and to have become exchangeable for, as of the Merger Effective Date, the right to receive the applicable Merger Consideration, without any interest thereon. Prior to the Closing, the Company shall give Purchaser prompt notice of any demands for dissenters' rights received by the Company and any withdrawals of such demands and the Company shall have complete control over all negotiations and proceedings with respect to such dissenters' rights (including the ability to make any payment with respect to any exercise by a shareholder of its rights to dissenter rights or demands).

4

#### 1.6 Exchange Agent.

(a) Prior to the Merger Effective Date, Purchaser shall appoint VStock Transfer, LLC as exchange agent, or another exchange agent reasonably acceptable to the Company and Purchaser (in such capacity, the "*Exchange Agent*"), for the purpose of exchanging Company Shares for a number of Purchaser Ordinary Shares in accordance with the provisions of this Agreement and the Merger Documents, as applicable. At or prior to the Merger Effective Date, Purchaser shall deposit, or cause to be deposited, with the Exchange Agent, for the benefit of the holders of Company Shares, for exchange in accordance with <u>Section 1.1(f)</u>, the Purchaser Ordinary Shares to be issued pursuant to <u>Section 1.1(f)</u>. At or as promptly as practicable following the Merger Effective Date, Purchaser shall send, or shall cause the Exchange Agent to send, to each holder of Company Shares a letter of transmittal for use in such exchange, in a form reasonably acceptable to the Company and Purchaser (a "*Letter of Transmittal*").

(b) Notwithstanding any other provision of this <u>Section 1.6</u>, any obligation on Purchaser under this Agreement to issue Purchaser Ordinary Shares to any holder of Company Shares entitled to receive Purchaser Ordinary Shares shall be satisfied by Purchaser issuing such Purchaser Ordinary Shares to the DTC or to such other clearing service or issuer of depositary receipts (or their nominees, in either case) as may be necessary or expedient, and each holder of Company Shares shall hold such Purchaser Ordinary Shares in bookentry form or through a holding of depositary receipts and the DTC or its nominee or the relevant clearing service or issuer of depositary receipts (or their nominees, as the case may be), will be the holder of record of such Purchaser Ordinary Shares. (c) Each holder of Company Shares shall be entitled to receive either Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares, as set forth in <u>Section 1.1(f)(i)</u>, (ii) and (iii) (excluding any Company Shares described in <u>Section 1.1(f)(iv)</u> or Dissenting Company Shares), as soon as reasonably practicable after the Merger Effective Date, but subject to the delivery to the Exchange Agent of the following items prior thereto: (i) the certificate(s), if any, representing such Company Shares (the "*Shareholder Certificates*" (or a Lost Certificate Affidavit) and (ii) a properly completed and duly executed Letter of Transmittal (the documents to be submitted to the Exchange Agent pursuant to this sentence may be referred to herein collectively as the "*Transmittal Documents*"). Until so surrendered, each such Shareholder Certificate shall represent after the Merger Effective Date for all purposes only the right to receive either Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares, as set forth in <u>Section 1.1(f)(i)</u>, (ii) and (iii).

(d) If any Purchaser Class A Ordinary Share or Purchaser Class B Ordinary Share is to be delivered or issued to a Person other than the Person in whose name the surrendered Shareholder Certificate is registered immediately prior to the Merger Effective Date, it shall be a condition to such delivery that (i) the transfer of such Company Shares shall have been permitted in accordance with the terms of the Organizational Documents of the Company, (ii) the recipient of such Purchaser Class A Ordinary Share or Purchaser Class B Ordinary Share, or the Person in whose name such Purchaser Class A Ordinary Share or Purchaser Class B Ordinary Share and delivered duly executed counterparts to the applicable Transmittal Documents as are reasonably deemed necessary by the Exchange Agent and (iii) the Person requesting such delivery shall have paid to the Exchange Agent any transfer or other Taxes required as a result of such delivery to a Person other than the registered holder of such Shareholder Certificate, or establish to the satisfaction of the Exchange Agent that such Tax has been paid or is not payable.

(e) Notwithstanding anything to the contrary contained herein, in the event that any Shareholder Certificate shall have been lost, stolen or destroyed, in lieu of delivery of a Shareholder Certificate to the Exchange Agent, the holder of Company Shares may instead deliver to the Exchange Agent an affidavit of lost certificate and indemnity of loss in form and substance reasonably acceptable to Purchaser (a *"Lost Certificate Affidavit"*), which at the reasonable discretion of Purchaser may include a requirement that the owner of such lost, stolen or destroyed Shareholder Certificate deliver a bond in such sum as it may reasonably direct as indemnity against any claim that may be made against Purchaser or the Surviving Company with respect to the Company Shares represented by the Shareholder Certificates alleged to have been lost, stolen or destroyed. Any Lost Certificate Affidavit properly executed and delivered in accordance with this <u>Section 1.6(e)</u> shall, unless the context otherwise requires, be treated as a Shareholder Certificate for all purposes of this Agreement.

(f) After the Merger Effective Date, the register of members of the Company shall be closed, and thereafter there shall be no further registration on the register of members of the Surviving Company of transfers of Company Shares that were issued and outstanding immediately prior to the Merger Effective Date. No dividends or other distributions declared or made after the date of this Agreement with respect to Purchaser Ordinary Shares with a record date after the Merger Effective Date will be paid to the holders of any Company Shares shall have provided the applicable Transmittal Documents in accordance with Section 1.6(c). Subject to applicable Law, following the delivery of the applicable Transmittal Documents, the Exchange Agent shall promptly deliver to the record holders thereof, without interest, the applicable Purchaser Ordinary Share and the amount of any such dividends or other distributions with a record date after the Merger Effective Date theretofore paid with respect to such Purchaser Ordinary Shares.

5

(g) All securities issued upon the surrender of Shareholder Certificates (or delivery of a Lost Certificate Affidavit) in accordance with the terms hereof shall be deemed to have been issued in full satisfaction of all rights pertaining to Company Shares represented by such Shareholder Certificates. Any Purchaser Ordinary Shares made available to the Exchange Agent pursuant to <u>Section 1.6(a)</u> that remains unclaimed by any holder of Company Shares one year after the Merger Effective Date shall be returned to Purchaser, upon demand, and any such holder of Company Shares who has not exchanged its Company Shares for the applicable number of Purchaser Ordinary Shares in accordance with this <u>Section 1.6</u> prior to that time shall thereafter look only to Purchaser for payment in respect of such Company Shares without any interest thereon (but with any dividends paid with respect thereto). Notwithstanding the foregoing, none of the Surviving Company, Purchaser or any party hereto or any Representative of any of the foregoing shall be liable to any Person for any amount properly paid to a public official pursuant to any applicable abandoned property, escheat or similar Law.

## ARTICLE II REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as set forth in the disclosure schedules delivered by Purchaser to the Company on the date of this Agreement (the "*Purchaser Disclosure Schedules*"), each section of which qualifies the correspondingly numbered representation or warranty specified therein and such other representation or warranty where its relevance as an exception to (or disclosure for purposes of) such other representation or warranty is reasonably apparent on the face of such disclosure, or other than with respect to <u>Sections 2.1</u> to <u>2.5</u> and <u>2.16</u>, the SEC Reports that are available on the SEC's website through EDGAR and at least two (2) Business Days prior to the date of this Agreement (excluding any risk factors, forward-looking statements or similar predictive statements), Purchaser represents and warrants to the Company, as of the date of this Agreement, as follows:

2.1 Organization and Standing. Purchaser is an exempted company duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. Purchaser has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary. Purchaser has made available to the Company accurate and complete copies of its Organizational Documents, each as currently in effect. Purchaser is not in violation of any provision of its Organizational Documents in any material respect.

2.2 Authorization: Binding Agreement. Purchaser has all requisite corporate power and authority to execute and deliver this Agreement and each Ancillary Document to which it is a party, to perform its obligations hereunder and thereunder and, to the Knowledge of the Purchaser, the Purchaser shareholders party to the Sponsor Support Agreement, and to consummate the transactions contemplated hereby and thereby, subject to obtaining the Required Purchaser Shareholder Approval. The execution and delivery of this Agreement and each Ancillary Document to which it is a party and the consummation of the transactions contemplated hereby and thereby (a) have been duly and validly authorized by the board of directors of Purchaser and (b) other than the Required Purchaser Shareholder Approval, no other corporate proceedings, other than as set forth elsewhere in the Agreement (including the filing of the Amended Purchaser Charter), on the part of Purchaser are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party and, to the Knowledge of the Purchaser, the Purchaser shareholders party to the Sponsor Support Agreement, or to consummate the transactions contemplated hereby and thereby. On or prior to the date of this Agreement, Purchaser's board of directors, at a duly called and held meeting, unanimously (i) determined that this Agreement and the transactions contemplated hereby, including the Merger, are advisable, fair to and in the best interests of Purchaser's shareholders in accordance with the Cayman Act, (ii) approved and adopted this Agreement, (iii) recommended that Purchaser's shareholders vote in favor of the approval of this Agreement, the Merger and the other Purchaser Shareholder Approval Matters in accordance with the Cayman Act and the Purchaser Charter (the "Purchaser Recommendation") and (iv) directed that this Agreement and the Purchaser Shareholder Approval Matters be submitted to Purchaser's shareholders for their approval. This Agreement has been, and each Ancillary Document to which Purchaser is a party shall be when delivered, duly and validly executed and delivered by Purchaser and, assuming the due authorization, execution and delivery of this Agreement and such Ancillary Documents by the other parties hereto and thereto, constitutes, or when delivered shall constitute, the valid and binding obligation of Purchaser, enforceable against Purchaser, and, to the Knowledge of the Purchaser, the Purchaser shareholders party to the Sponsor Support Agreement, enforceable against the Purchaser shareholders, in accordance with its terms, except to the extent that enforceability thereof may be limited by applicable bankruptcy, insolvency, reorganization and moratorium laws and other laws of general application affecting the enforcement of creditors' rights generally and subject to general principles of equity (collectively, the "Enforceability Exceptions").

2.3 Governmental Approvals. Except as otherwise described in <u>Schedule 2.3</u> of the Purchaser Disclosure Schedules, no Consent of or with any Governmental Authority, is required to be obtained or made in connection with the execution, delivery or performance by Purchaser of this Agreement and each Ancillary Document to which it is a party or the consummation by Purchaser of the transactions contemplated hereby and thereby, other than pursuant to Antitrust Laws, (b) such filings as expressly contemplated by this Agreement (including the filing of the Amended Purchaser Charter), (c) any filings required with NASDAQ or the SEC with respect to the Transactions, (d) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder, and (e) where the failure to obtain or make such Consents or to make such filings or notifications, would not reasonably be expected to have a Material Adverse Effect on Purchaser or materially impair or delay the ability of Purchaser to consummate the Transactions.

2.4 <u>Non-Contravention</u>. Except as otherwise described in <u>Schedule 2.4</u> of the Purchaser Disclosure Schedules, the execution and delivery by Purchaser of this Agreement and each Ancillary Document to which it is a party, the consummation by Purchaser of the transactions contemplated hereby and thereby, and compliance by Purchaser with any of the provisions hereof and thereof, will not (a)

conflict with or violate any provision of Purchaser's Organizational Documents, (b) subject to obtaining the Consents from Governmental Authorities referred to in <u>Section 2.3</u> hereof, and the waiting periods referred to therein having expired, and any condition precedent to such Consent or waiver having been satisfied, conflict with or violate any Law, Order or Consent applicable to Purchaser or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by Purchaser under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of Purchaser under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any Purchaser Material Contract, except for any deviations from any of the foregoing clauses (b) or (c) that would not reasonably be expected to have a Material Adverse Effect on Purchaser or materially impair or delay the ability of Purchaser to consummate the Transactions.

#### 2.5 Capitalization.

(a) Purchaser is authorized to issue 221,000,000 shares, of which 200,000,000 shares are Purchaser Class A Shares, 20,000,000 are Purchaser Class B Shares and 1,000,000 are Purchaser Preference Shares. The issued and outstanding Purchaser Securities as of the date of this Agreement are set forth on <u>Schedule 2.5(a)</u> of the Purchaser Disclosure Schedules. There are no issued or outstanding Purchaser Preference Shares. All outstanding Purchaser Shares are duly authorized, validly issued, fully paid and non-assessable and not subject to or issued in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the Cayman Act, Purchaser's Organizational Documents or any Contract to which Purchaser is a party. None of the outstanding Purchaser Securities has been issued in violation of any applicable securities Laws. Prior to giving effect to the transactions contemplated by this Agreement, Purchaser does not have, and has not had, any Subsidiaries or own any equity interests in any other Person other than Merger Sub.

(b) Except as set forth in <u>Schedule 2.5(a)</u> or <u>Schedule 2.5(b)</u> of the Purchaser Disclosure Schedules, there are no (i) outstanding options, warrants, puts, calls, convertible or exchangeable securities, "phantom" share rights, share appreciation rights, sharebased units, preemptive or similar rights, (ii) bonds, debentures, notes or other Indebtedness having general voting rights or that are convertible or exchangeable into securities having such rights or (iii) subscriptions or other rights, agreements, arrangements, Contracts or commitments of any character (other than this Agreement and the Ancillary Documents), (A) relating to the issued or unissued securities of Purchaser or (B) obligating Purchaser to issue, transfer, deliver or sell or cause to be issued, transferred, delivered, sold or repurchased any options or shares or securities convertible into or exchangeable for any capital shares, or (C) obligating Purchaser to grant, extend or enter into any such option, warrant, call, subscription or other right, agreement, arrangement or commitment for any capital shares. Other than the Redemption or as expressly set forth in this Agreement, there are no outstanding obligations of Purchaser to repurchase, redeem or otherwise acquire any shares of Purchaser or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any Person. Except as set forth in <u>Schedule 2.5(b)</u> of the Purchaser Disclosure Schedules, there are no shareholders agreements, voting trusts or other agreements or understandings to which Purchaser is a party with respect to the voting of any shares of Purchaser.

(c) All Indebtedness and unpaid Purchaser Transaction Expenses as of the date of this Agreement is disclosed on <u>Schedule 2.5(c)</u> of the Purchaser Disclosure Schedules. No Indebtedness of Purchaser contains any restriction upon: (i) the prepayment of any of such Indebtedness, (ii) the incurrence of Indebtedness by Purchaser or (iii) the ability of Purchaser to grant any Lien on its properties or assets.

(d) Since the date of formation of Purchaser, and except as contemplated by this Agreement, Purchaser has not declared or paid any distribution or dividend in respect of its shares and has not repurchased, redeemed or otherwise acquired any of its shares, and Purchaser's board of directors has not authorized any of the foregoing.

7

#### 2.6 SEC Filings: Purchaser Financials: Internal Controls.

(a) Purchaser, since the IPO, has filed all forms, reports, schedules, statements, registration statements, prospectuses and other documents required to be filed or furnished by Purchaser with the SEC under the Securities Act and/or the Exchange Act, together with any amendments, restatements or supplements thereto, and will file all such forms, reports, schedules, statements and other

documents required to be filed subsequent to the date of this Agreement. Except to the extent available on the SEC's web site through EDGAR, Purchaser has delivered to the Company copies in the form filed with the SEC of all of the following: (i) Purchaser's Form S-I, (ii) Purchaser's annual reports on Form 10-K for each fiscal year of Purchaser beginning with the first year Purchaser was required to file such a form, (iii) Purchaser's quarterly reports on Form 10-Q for each fiscal quarter that Purchaser filed such reports to disclose its quarterly financial results in each of the fiscal years of Purchaser referred to in clause (ii) above, (iv) all other forms, reports, registration statements, prospectuses and other documents (other than preliminary materials) filed by Purchaser with the SEC since the beginning of the first fiscal year referred to in clause (ii) above (the forms, reports, registration statements, prospectuses and other documents referred to in clauses (ii), (iii) and (iv) above, whether or not available through EDGAR, are, collectively, the "SEC Reports") and (v) all certifications and statements required by (A) Rules 13a-14 or 15d-14 under the Exchange Act, and (B) 18 U.S.C. §1350 (Section 906 of SOX) with respect to any report referred to in clause (ii) above (collectively, the "Public Certifications"). The SEC Reports (x) were prepared in all material respects in accordance with the requirements of the Securities Act and the Exchange Act, as the case may be, and the rules and regulations thereunder and (y) did not, as of their respective effective dates (in the case of SEC Reports that are registration statements filed pursuant to the requirements of the Securities Act) and at the time they were filed with the SEC (in the case of all other SEC Reports) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. The Public Certifications are each true as of their respective dates of filing. As used in this Section 2.6. the term "file" shall be broadly construed to include any manner permitted by SEC rules and regulations in which a document or information is furnished, supplied or otherwise made available to the SEC.

(b) As of the date of this Agreement, (A) the Purchaser Units, the Purchaser Class A Shares and the Purchaser Public Warrants are listed on NASDAQ, (B) Purchaser has not received any written deficiency notice from NASDAQ relating to the continued listing requirements of such Purchaser Securities, (C) there are no Actions pending or, to the Knowledge of Purchaser, threatened against Purchaser by the Financial Industry Regulatory Authority with respect to any intention by such entity to suspend, prohibit or terminate the quoting of such Purchaser Securities on NASDAQ and (D) such Purchaser Securities are in compliance with all of the applicable corporate governance rules of NASDAQ.

(c) The financial statements and notes of Purchaser contained or incorporated by reference in the SEC Reports (the *"Purchaser Financials"*), fairly present in all material respects the financial position and the results of operations, changes in shareholders' equity, and cash flows of Purchaser at the respective dates of and for the periods referred to in such financial statements, all in accordance with (i) GAAP methodologies applied on a consistent basis throughout the periods involved, (ii) Regulation S-X or Regulation S-K, as applicable (except as may be indicated in the notes thereto and for the omission of notes and audit adjustments in the case of unaudited quarterly financial statements to the extent permitted by Regulation S-X or Regulation S-K, as applicable), and (iii) audited in accordance with the standards of the Public Company Accounting Oversight Board.

(d) Except as and to the extent reflected or reserved against in the Purchaser Financials, Purchaser has not incurred any Liabilities or obligations of the type required to be reflected on a balance sheet in accordance with GAAP that is not adequately reflected or reserved on or provided for in the Purchaser Financials, other than Liabilities of the type required to be reflected on a balance sheet in accordance with GAAP that have been incurred since Purchaser's formation in the ordinary course of business consistent with past practice. Purchaser does not maintain any "off-balance sheet arrangement" within the meaning of Item 303 of Regulation S-K of the Securities Act. As of the date of this Agreement, no financial statements other than those of Purchaser are required by GAAP to be included in the financial statements of Purchaser.

(e) Since the IPO, Purchaser has not received from its independent auditors any written notification of any (i) "significant deficiency" in the internal controls over financial reporting of Purchaser, (ii) "material weakness" in the internal controls over financial reporting of Purchaser or other employees of Purchaser who have a significant role in the internal controls over financial reporting of Purchaser.

8

(f) Except as not required in reliance on exemptions from various reporting requirements by virtue of Purchaser's status as an "emerging growth company" within the meaning of the Securities Act, as modified by the JOBS Act, since the IPO, (i) Purchaser has established and maintained a system of internal controls over financial reporting (as defined in Rule 13a-15 and Rule 15d-15 under the Exchange Act) sufficient to provide reasonable assurance regarding the reliability of Purchaser's financial reporting and the preparation of Purchaser's financial statements for external purposes in accordance with GAAP and (ii) Purchaser has established and maintained disclosure controls and procedures (as defined in Rule 13a-15 and Rule 15d-15 under the Exchange Act) designed to ensure that material

information relating to Purchaser is made known to Purchaser's principal executive officer and principal financial officer by others within Purchaser, particularly during the periods in which the periodic reports required under the Exchange Act are being prepared.

(g) Purchaser has not taken any action prohibited by Section 402 of the Sarbanes-Oxley Act.

(h) Notwithstanding anything in this <u>Section 2.6</u> or otherwise in this Agreement, no representation or warranty is made as to the accounting treatment of Purchaser's issued and outstanding warrants, or as to any deficiencies in disclosure (including with respect to accounting and disclosure controls) arising from the treatment of such warrants as equity rather than liabilities in Purchaser's financial statements.

2.7 <u>Absence of Certain Changes</u>. As of the date of this Agreement, except as set forth in <u>Schedule 2.7</u> of the Purchaser Disclosure Schedules, Purchaser has (a) since its formation, conducted no business other than its formation, the public offering of its securities (and the related private offerings), public reporting and its search for an initial Business Combination as described in the IPO Prospectus (including the investigation of the Target Companies and the negotiation and execution of this Agreement) and related activities, (b) since its formation, not been subject to a Material Adverse Effect and (c) since December 31, 2020, not taken any action or committed or agreed to take any action that would be prohibited by <u>Section 5.3</u> (without giving effect to <u>Schedule 5.3</u>) if such action were taken on or after the date of this Agreement without the consent of Purchaser.

2.8 <u>Compliance with Laws</u>. Purchaser is, and has since its formation been, in material compliance with all Laws applicable to it and the conduct of its business, and Purchaser has not received written notice alleging any violation of applicable Law in any material respect by Purchaser. Purchaser has duly and promptly filed all information required under ES Legislation within the time periods prescribed under the ES Legislation.

2.9 <u>Actions; Orders; Permits</u>. There is no pending or, to the Knowledge of Purchaser, threatened Action to which Purchaser is subject. There is no Action that Purchaser has pending against any other Person. Purchaser is not subject to any Orders of any Governmental Authority, nor, to the Knowledge of the Purchaser, are any such Orders pending. Purchaser holds all material Permits necessary to lawfully conduct its business as presently conducted, and to own, lease and operate its assets and properties, all of which are in full force and effect.

## 2.10 Taxes and Returns.

(a) Purchaser has or will have timely filed, or caused to be timely filed, all material Tax Returns required to be filed by it (taking into account all available extensions), which Tax Returns are true, accurate, correct and complete in all material respects, and has paid, collected or withheld, or caused to be paid, collected or withheld, all material Taxes required to be paid, collected or withheld, other than such Taxes for which adequate reserves in the Purchaser financials have been established in accordance with GAAP. <u>Schedule 2.10(a)</u> of the Purchaser Disclosure Schedules sets forth each jurisdiction where Purchaser files or, to the Knowledge of Purchaser, is required to file a Tax Return. There are no claims, assessments, audits, examinations, investigations or other Actions pending against Purchaser in respect of any material Tax, and Purchaser has not been notified in writing of any material proposed Tax claims or assessments against Purchaser (other than, in each case, claims or assessments for which adequate reserves in the Purchaser financials have been established in accordance with GAAP). There are no Liens with respect to any Taxes upon any of Purchaser's assets, other than Permitted Liens. Purchaser has no outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes. There are no outstanding requests by Purchaser for any extension of time within which to file any Tax Return or within which to pay any Taxes shown to be due on any Tax Return outside of the ordinary course of business. Purchaser is not, and never has been, (A) a U.S. real property holding corporation within the meaning of Section 897(c)(2) of the Code, or (B) a member of any consolidated, combined, unitary or affiliated group of corporations for any Tax purposes.

(b) Since the date of its formation, Purchaser has not (i) changed any Tax accounting methods, policies or procedures except as required by a change in Law, (ii) made, revoked, or amended any material Tax election, (iii) filed any amended Tax Returns or claim for refund or (iv) entered into any closing agreement affecting or otherwise settled or compromised any material Tax Liability or refund.

(c) Purchaser is not nor has been a party to any "listed transaction" as defined in Section 6707A of the Code and Treasury Regulations Section 1.6011-4 (or any corresponding or similar provision of state, local or non-U. S. income Tax Law).

(d) Purchaser has neither taken nor agreed to take any action not contemplated by this Agreement and/or any Ancillary Documents that could reasonably be expected to prevent the Merger from qualifying for the Intended Tax Treatment. To the Knowledge of Purchaser, no facts or circumstances exist that could reasonably be expected to prevent the Merger from qualifying for the Intended Tax Treatment.

(e) Purchaser is treated as a corporation for U.S. federal (and applicable U.S. state and local) income Tax purposes.

(f) Notwithstanding anything to the contrary herein, the representations and warranties in this <u>Section 2.10</u> and <u>Section 2.11</u> are the sole and exclusive representations and warranties of the Purchaser concerning Tax matters.

2.11 Employees and Employee Benefit Plans. Except as set forth on Schedule 2.11 of the Purchaser Disclosure Schedules, Purchaser does not, and has not ever, (a) had any employees directors, officers, or other service provider, or (b) maintained, sponsored, contributed to or otherwise had any Liability (contingent or otherwise) under, any Benefit Plans, and Foreign Plan, any benefit plan sponsored by an ERISA Affiliate of Purchaser, or any other benefit or compensation plan, Contract, program, policy, or arrangement. Neither the execution and delivery of this Agreement or the Ancillary Documents nor the consummation of the transactions contemplated by this Agreement and the Ancillary Documents (either alone or in conjunction with any other event) will (i) result in any payment or benefit (including severance, unemployment compensation, golden parachute, bonus or otherwise) becoming due to any director, officer, employee or other service provider of Purchaser, (ii) result in the acceleration of the time of payment or vesting of any such payment or benefit, or (iii) result in any "parachute payment" under Section 280G of the Code.

2.12 <u>Properties</u>. Purchaser does not own, license or otherwise have any right, title or interest in any material Intellectual Property. Purchaser does not own or lease any material real property or Personal Property.

### 2.13 Material Contracts.

(a) Except as set forth on <u>Schedule 2.13(a)</u> of the Purchaser Disclosure Schedules or the Purchaser SEC Reports, other than this Agreement and the Ancillary Documents, there are no Contracts to which Purchaser is a party or by which any of its properties or assets may be bound, subject or affected, which (i) creates or imposes a Liability greater than \$5 million, or (ii) prohibits, prevents, restricts or impairs in any material respect any business practice of Purchaser or any acquisition of material property by Purchaser, or restricts in any material respect the ability of Purchaser from engaging in business as conducted as of the date of this Agreement by it (each, a *"Purchaser Material Contract"*). All Purchaser Material Contracts have been made available to the Company other than those that are exhibits to the SEC Reports.

(b) With respect to each Purchaser Material Contract: (i) the Purchaser Material Contract was entered into at arm's length and in the ordinary course of business consistent with past practice; (ii) the Purchaser Material Contract is legal, valid, binding and enforceable in all material respects against Purchaser and, to the Knowledge of Purchaser, the other parties thereto, and is in full force and effect (except, in each case, as such enforcement may be limited by the Enforceability Exceptions); (iii) Purchaser is not in breach or default in any material respect, and to the Knowledge of Purchaser, no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default in any material respect by Purchaser, no other party to any Purchaser Material Contract is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default in one event has occurred that with the passage of time or giving of notice is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default party, or permit termination or acceleration by Purchaser Material Contract is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default by such other party, or permit termination or acceleration by Purchaser under any Purchaser Material Contract.

2.14 <u>Transactions with Affiliates</u>. Purchaser has provided the Company with, and <u>Schedule 2.14</u> of the Purchaser Disclosure Schedules sets forth a true, correct and complete list of the Contracts and arrangements that are in existence as of the date of this Agreement under which there are any existing or future Liabilities or obligations between Purchaser, on the one hand, and (a) any present or former director, officer, employee, manager, direct or indirect equityholder (including Sponsor) or Affiliate of either Purchaser or Sponsor, or any immediate family member of any of the foregoing Persons, or (b) record or beneficial owner of more than five percent (5%) of Purchaser's outstanding shares as of the date of this Agreement, on the other hand.

2.15 <u>Investment Company Act</u>; JOBS Act. Purchaser is not an "investment company" or a Person directly or indirectly "controlled" by or acting on behalf of a person subject to registration and regulation as an "investment company", in each case within the meaning of the Investment Company Act. Purchaser constitutes an "emerging growth company" within the meaning of the JOBS Act.

2.16 <u>Finders and Brokers</u>. Except as set forth in <u>Schedule 2.16</u> of the Purchaser Disclosure Schedules, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission from Purchaser, the Target Companies or any of their respective Affiliates in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of Purchaser.

## 2.17 Certain Business Practices.

(a) Neither Purchaser, nor any of its Representatives acting on its behalf, has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 or any other local or foreign anticorruption or bribery Law, (iii) made any other unlawful payment or (iv) since the formation of Purchaser, directly or indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or other Person who is or may be in a position to help or hinder Purchaser or assist it in connection with any actual or proposed transaction.

(b) The operations of Purchaser are and have been conducted at all times in compliance with money laundering statutes in all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority, and no Action involving Purchaser with respect to any of the foregoing is pending or, to the Knowledge of Purchaser, threatened.

(c) None of Purchaser, any of its Subsidiaries, or any of their directors, officers or employees, or, to the Knowledge of Purchaser, any other Representative acting on behalf of Purchaser is currently, or has been in the last five (5) years, (i) identified on the list of specially designated nationals or other blocked persons or otherwise currently subject to any sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("*OFAC*"), the U.S. Department of State, or other applicable Governmental Authority; (ii) organized, resident, or located in, or a national of a comprehensively sanctioned country (currently, Cuba, Iran, North Korea, and the Crimea region of Ukraine); or (iii) in the aggregate, fifty (50) percent or greater owned, directly or indirectly, or otherwise controlled, by a person identified in (i) or (ii); and Purchaser has not, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any Subsidiary, joint venture partner or other Person, in connection with any sales or operations in any other country sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC or the U.S. Department of State in the last five (5) fiscal years.

2.18 <u>Insurance</u>. <u>Schedule 2.18</u> of the Purchaser Disclosure Schedules lists all insurance policies (by policy number, insurer, coverage period, coverage amount, annual premium and type of policy) held by Purchaser relating to Purchaser or its business, properties, assets, directors, officers and employees, copies of which have been provided to the Company. All premiums due and payable under all such insurance policies have been timely paid and Purchaser is otherwise in material compliance with the terms of such insurance policies. All such insurance policies are in full force and effect, and to the Knowledge of Purchaser, there is no threatened termination of, or material premium increase with respect to, any of such insurance policies. There have been no insurance claims made by Purchaser. Purchaser has reported to its insurers all material claims and pending circumstances that would reasonably be expected to result in a claim.

2.19 Information Supplied. None of the information supplied or to be supplied by Purchaser expressly for inclusion or incorporation by reference: (a) in any current report on Form 8-K, and any exhibits thereto or any other report, form, registration or other filing made with any Governmental Authority (including the SEC) with respect to the transactions contemplated by this Agreement or any Ancillary Documents; (b) in the Registration Statement; or (c) in the mailings or other distributions to Purchaser's or Purchaser's shareholders and/or prospective investors with respect to the consummation of the transactions contemplated by this Agreement or in any amendment to any of documents identified in (a) through (c), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading; provided, however, no representation or warranty is made as to the accounting treatment of Purchaser's issued and outstanding warrants, or as to any deficiencies in disclosure (including with respect to accounting and disclosure controls) arising from the treatment of such warrants as equity rather than liabilities in Purchaser's financial statements. None of the information supplied or to be supplied by Purchaser expressly for inclusion or incorporation by reference in any of the Signing Press Release, the Signing Filing, the Closing Filing and the Closing Press Release will, when filed or distributed, as applicable, contain any untrue statement of a material fact required to be state any material fact required to be stated therein or necessary in order to make the statements in any of the Signing Press Release, the Signing Filing, the Closing Filing and the Closing Press Release will, when filed or distributed, as applicable, contain any untrue statement of a material fact or omit to state

misleading. Notwithstanding the foregoing, Purchaser makes no representation, warranty or covenant with respect to any information supplied by or on behalf of Purchaser, the Target Companies or any of their respective Affiliates.

2.20 Independent Investigation. Purchaser has conducted its own independent investigation, review and analysis of the business, results of operations, condition (financial or otherwise) or assets of the Target Companies, Purchaser and Merger Sub and acknowledges that it has been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of the Target Companies, Purchaser and Merger Sub for such purpose. Purchaser acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, it has relied solely upon its own investigation and the express representations and warranties of the Company, Purchaser and Merger Sub set forth in this Agreement (including the related portions of the Company Disclosure Schedules) and in any certificate delivered to Purchaser pursuant hereto; and (b) none of the Company, Purchaser or Merger Sub or their respective Representatives have made any representation or warranty as to the Target Companies, Purchaser or Merger Sub or this Agreement, except as expressly set forth in this Agreement (including the related portions of the Company Disclosure Schedules) or in any certificate delivered to Purchaser pursuant hereto; of the Target Companies, Purchaser or Merger Sub or this Agreement, except as expressly set forth in this Agreement (including the related portions of the Company Disclosure Schedules) or in any certificate delivered to Purchaser pursuant hereto.

2.21 Trust Account. As of June 30, 2021, Purchaser has an amount of assets in the Trust Account equal to \$44,890,288. The funds held in the Trust Account are invested in U.S. government securities or money market funds meeting certain conditions under Rule 2a-7 promulgated under the Investment Company Act and held in trust pursuant to the Trust Agreement. The Trust Agreement is in full force and effect and is a legal, valid and binding obligation of Purchaser and the Trustee, enforceable in accordance with its terms. The Trust Agreement has not been terminated, repudiated, rescinded, amended, supplemented or modified, in any respect, and no such termination, repudiation, rescission, amendment, supplement or modification is contemplated. There are no separate Contracts, side letters or other arrangements or understandings (whether written or unwritten, express or implied) that would cause the description of the Trust Agreement in the SEC Reports to be inaccurate in any material respect or, to the Knowledge of Purchaser, that would entitle any Person (other than (i) in respect of deferred underwriting commissions or Taxes, (ii) the holders of Purchaser Securities prior to the Effective Time who shall have elected to redeem their Purchaser Class A Shares pursuant to Purchaser's Organizational Documents or (iii) if Purchaser fails to complete a Business Combination within the allotted time period and liquidates the Trust Account, subject to the terms of the Trust Agreement, Purchaser in limited amounts to permit Purchaser to pay the expenses of the Trust Account's liquidation and dissolution, and then Purchaser's public shareholders) to any portion of the funds in the Trust Account. Prior to the Closing, none of the funds held in the Trust Account have been released, except to pay Taxes from any interest income earned in the Trust Account, and to redeem Purchaser Class A Shares pursuant to Purchaser's Organizational Documents. As of the date of this Agreement, there are no Actions pending or, to the Knowledge of Purchaser, threatened with respect to the Trust Account.

2.22 EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES. NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE COMPANY, PURCHASER, MERGER SUB OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION (INCLUDING ANY FINANCIAL PROJECTIONS OR OTHER SUPPLEMENTAL DATA). EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS ARTICLE II, NEITHER PURCHASER NOR ANY OTHER PERSON MAKES, AND PURCHASER EXPRESSLY DISCLAIMS, ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE. EXPRESS OR IMPLIED, AS TO THE MATERIALS RELATING TO THE BUSINESS AND AFFAIRS OR HOLDINGS OF PURCHASER AND ITS SUBSIDIARIES THAT HAVE BEEN MADE AVAILABLE TO THE COMPANY, PURCHASER, MERGER SUB OR IN ANY PRESENTATION OF THE BUSINESS AND AFFAIRS OF PURCHASER AND ITS SUBSIDIARIES BY THE MANAGEMENT OF PURCHASER OR OTHERS IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY. NO STATEMENT CONTAINED IN ANY OF SUCH MATERIALS OR MADE IN ANY SUCH PRESENTATION SHALL BE DEEMED A REPRESENTATION OR WARRANTY HEREUNDER OR OTHERWISE OR DEEMED TO BE RELIED UPON BY THE COMPANY, PURCHASER AND MERGER SUB IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN THIS ARTICLE II, IT IS UNDERSTOOD THAT ANY COST ESTIMATES, PROJECTIONS OR OTHER PREDICTIONS, ANY DATA, ANY FINANCIAL INFORMATION OR ANY MEMORANDA OR OFFERING MATERIALS OR PRESENTATIONS, INCLUDING ANY OFFERING MEMORANDUM OR SIMILAR MATERIALS MADE AVAILABLE BY PURCHASER ARE NOT AND SHALL NOT BE DEEMED TO BE OR TO INCLUDE REPRESENTATIONS OR WARRANTIES OF PURCHASER, AND ARE NOT AND SHALL NOT BE DEEMED TO BE RELIED UPON BY THE COMPANY, PURCHASER AND MERGER SUB IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.

## ARTICLE III REPRESENTATIONS AND WARRANTIES OF MERGER SUB

Each of Purchaser and Merger Sub represents and warrants to the Company, as of the date of this Agreement and as of the Closing, as follows:

3.1 Organization and Standing. Merger Sub is an exempted company duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. Merger Sub has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Merger Sub is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary. Purchaser has heretofore made available to the Company accurate and complete copies of the Organizational Documents of Merger Sub, as currently in effect. Merger Sub is not in violation of any provision of its Organizational Documents in any material respect.

3.2 Authorization: Binding Agreement. Merger Sub has all requisite corporate power and authority to execute and deliver this Agreement and each Ancillary Document to which it is a party, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby, subject to the Required Merger Sub Shareholder Approval. Assuming that the Required Merger Sub Shareholder Approval has been obtained, the execution and delivery of this Agreement and each Ancillary Document to which it is a party and the consummation of the transactions contemplated hereby and thereby, (a) have been duly and validly authorized by the board of directors and, as applicable, shareholders of Merger Sub in accordance with the Merger Sub's Organizational Documents, the Cayman Act and any other applicable Law and (b) no other corporate proceedings, other than as expressly set forth elsewhere in the Agreement, on the part of Merger Sub are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party and, to the Knowledge of the Purchaser, the Purchaser shareholders party to the Sponsor Support Agreement, or to consummate the transactions contemplated hereby and thereby. This Agreement has been, and each Ancillary Document to which Merger Sub is a party has been or shall be when delivered, duly and validly executed and delivered and, assuming the due authorization, execution and delivery of this Agreement and such Ancillary Documents by the other parties hereto and thereto and the obtainment of the Required Merger Sub Shareholder Approval, constitutes, or when delivered shall constitute, the valid and binding obligation of the Merger Sub, enforceable against the Merger Sub in accordance with its terms, subject to the Enforceability Exceptions.

3.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority, on the part of Merger Sub is required to be obtained or made in connection with the execution, delivery or performance by such Party of this Agreement and each Ancillary Document to which it is a party or the consummation by such Party of the transactions contemplated hereby and thereby, other than (a) pursuant to Antitrust Laws, (b) such filings as expressly contemplated by this Agreement, (c) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder, and (e) where the failure to obtain or make such Consents or to make such filings or notifications, would not reasonably be expected to have a Material Adverse Effect on Purchaser.

3.4 <u>Non-Contravention</u>. The execution and delivery by Purchaser and Merger Sub of this Agreement and each Ancillary Document to which it is a party, the consummation by such Party of the transactions contemplated hereby and thereby, and compliance by such Party with any of the provisions hereof and thereof, will not (a) subject to the filing of the Amended Purchaser Charter, conflict with or violate any provision of such Party's Organizational Documents, (b) subject to obtaining the Consents from Governmental Authorities referred to in <u>Section 3.3</u> hereof, and the waiting periods referred to therein having expired, and any condition precedent to such Consent or waiver having been satisfied, conflict with or violate any Law, Order or Consent applicable to such Party or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by such Party under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of such Party under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of such Party, except for any deviations from any of the foregoing clauses (b) or (c) that would not reasonably be expected to have a Material Adverse Effect on Merger Sub.

#### 3.5 Capitalization.

(a) As of the date of this Agreement, Merger Sub is authorized to issue 1,000 Merger Sub Ordinary Shares, of which 1,000 shares are issued and outstanding, and all of which are owned by Purchaser. All such issued and outstanding Shares have been, or will be prior to such issuance, duly authorized, validly issued, fully paid and nonassessable and not subject to or issues issued in violation of any purchase option, right of first refusal, preemptive right, subscription right, call option or any similar right. No other shares or other equity interests of Merger Sub are issued, reserved for issuance or outstanding. Prior to giving effect to the transactions contemplated by this Agreement, Merger Sub does not have any Subsidiaries or own any equity interests in any other Person and prior to giving effect to the transactions contemplated by this Agreement, Merger Sub do not have any Subsidiaries or own any equity interests in any other Person.

(b) Except as set forth in its Organizational Documents, Merger Sub (i) has no obligation to issue, sell or transfer any equity securities of Merger Sub, (ii) is not party or subject to any contract that affects or relates to voting or giving of written consents with respect to, or the right to cause the redemption, or repurchase of, any equity interests of Merger Sub, (iii) has not granted any registration rights or information rights to any other Person, (iv) has not granted any phantom shares and there are no voting or similar agreements entered into by Merger Sub which relate to its capital or equity interests (v) has no outstanding bonds, debentures, notes or other obligations the holders of which have the right to vote (or convertible into or exercisable for voting interests of Merger Sub or equity interests of Merger Sub) with the owners or holders of Merger Sub on any matter or any agreements to issues such bonds, debentures, notes or other obligations and (vi) has no outstanding contractual obligations to provide funds to, or make any investment (other than the Transactions contemplated herein) in, any other Person.

3.6 <u>Merger Sub Activities</u>. Since its formation, Merger Sub has not engaged in any business activities other than as contemplated by this Agreement, does not own or control, directly or indirectly, any ownership, equity, profits or voting interest in any Person and has no assets or Liabilities except those incurred in connection with this Agreement and the Ancillary Documents to which it is a party and the Transactions, and, as of the date of this Agreement, other than this Agreement and the Ancillary Documents to which it is a party, Merger Sub is not party to or bound by any Contract.

3.7 <u>Compliance with Laws</u>. Merger Sub, is not, and since the date of its formation, has not been, in conflict or non-compliance with, or in default or violation of, any Laws applicable to it. Merger Sub, has not, since the date of its formation, received any written or, to the Knowledge of the Purchaser, oral notice of, is under investigation with respect to, any material conflict or non-compliance with, or material default or violation of, any applicable Laws by which it is or was bound.

3.8 <u>Actions</u>: Orders. There is no pending or, to the Knowledge of the Purchaser, threatened Action to which Merger Sub is subject and no such Action has been brought or, to the Knowledge of the Purchaser, threatened since the date of its respective formation. There is no Action that Merger Sub has pending against any other Person. Merger Sub, is not subject to any Orders of any Governmental Authority, nor, to the Knowledge of the Purchaser, are any such Orders pending and no such Order has been brought or, to the Knowledge of the Purchaser, has been threatened since the date of its respective formation.

3.9 <u>Transactions with Related Persons</u>. Except as set forth on <u>Schedule 3.9</u> of the Purchaser Disclosure Schedules, there are no transactions, Contracts or understandings between Merger Sub, on the one hand, and any Related Person of Merger Sub, on the other hand, either (a) currently in effect or (b) that would be required to be disclosed under Item 404 of Regulation S-K promulgated under the Securities Act.

3.10 <u>Finders and Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission from Purchaser, the Target Companies or any of their respective Affiliates in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of Merger Sub.

3.11 <u>Investment Company Act</u>. Merger Sub is not an "investment company" or, a Person directly or indirectly controlled by or acting on behalf of a person subject to registration and regulation as an "investment company", in each case within the meanings of the Investment Company Act.

3.12 <u>Intended Tax Treatment</u>. Merger Sub has not taken any action (nor permitted any action to be taken), and is not aware of any fact or circumstance, that would reasonably be expected to prevent, impair or impede the Intended Tax Treatment.

3.13 Information Supplied. None of the information supplied or to be supplied by Merger Sub expressly for inclusion or incorporation by reference: (a) in any Current Report on Form 8-K or 6-K, and any exhibits thereto or any other report, form, registration or other filing made with any Governmental Authority (including the SEC) with respect to the transactions contemplated by this Agreement or any Ancillary Documents; (b) in the Registration Statement; or (c) in the mailings or other distributions to Purchaser's shareholders and/or prospective investors with respect to the consummation of the transactions contemplated by this Agreement or in any amendment to any of documents identified in (a) through (c), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to be supplied by Merger Sub expressly for inclusion or incorporation by reference in any of the Signing Press Release, the Signing Filing, the Closing Filing and the Closing Press Release will, when filed or distributed, as applicable, contain any untrue statement of a material fact required to be stated therein or necessary in order to make the circumstances under which they are made, not misleading. None of the statements therein, in light of the circumstances will, when filed or distributed, as applicable, contain any untrue statement of a material fact required to be stated therein or necessary in order to make the statements under which they are made, not misleading. Norder to make the statements therein, in light of the closing Press Release will, when filed or distributed, as applicable, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. Notwithstanding the foregoing, Merger Sub makes no representation, warr

14

3.14 EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES. NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO PURCHASER, THE COMPANY, OR ANY OF THEIR RESPECTIVE REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION (INCLUDING ANY FINANCIAL PROJECTIONS OR OTHER SUPPLEMENTAL DATA), EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS ARTICLE III, MERGER SUB NOR ANY OTHER PERSON MAKES, AND MERGER SUB EXPRESSLY DISCLAIMS, ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AS TO THE MATERIALS RELATING TO THE BUSINESS AND AFFAIRS OR HOLDINGS MERGER SUB THAT HAVE BEEN MADE AVAILABLE TO PURCHASER OR THE COMPANY OR IN ANY PRESENTATION OF THE BUSINESS AND AFFAIRS OF MERGER SUB BY THE MANAGEMENT OF MERGER SUB OR OTHERS IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY. NO STATEMENT CONTAINED IN ANY OF SUCH MATERIALS OR MADE IN ANY SUCH PRESENTATION SHALL BE DEEMED A REPRESENTATION OR WARRANTY HEREUNDER OR OTHERWISE OR DEEMED TO BE RELIED UPON BY PURCHASER AND THE COMPANY IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN THIS ARTICLE III, IT IS UNDERSTOOD THAT ANY COST ESTIMATES, PROJECTIONS OR OTHER PREDICTIONS, ANY DATA, ANY FINANCIAL INFORMATION OR ANY MEMORANDA OR OFFERING MATERIALS OR PRESENTATIONS, INCLUDING ANY OFFERING MEMORANDUM OR SIMILAR MATERIALS MADE AVAILABLE BY MERGER SUB ARE NOT AND SHALL NOT BE DEEMED TO BE OR TO INCLUDE REPRESENTATIONS OR WARRANTIES OF MERGER SUB, AND ARE NOT AND SHALL NOT BE DEEMED TO BE RELIED UPON BY PURCHASER AND THE COMPANY IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.

## ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the disclosure schedules delivered by the Company to Purchaser on the date of this Agreement (the "*Company Disclosure Schedules*"), each section of which qualifies the correspondingly numbered representation or warranty specified therein and such other representation or warranty where its relevance as an exception to (or disclosure for purposes of) such other representation or warranty is reasonably apparent on the face of such disclosure, the Company hereby represents and warrants to Purchaser and Merger Sub, as of the date of this Agreement as follows:

4.1 Organization and Standing. The Company is an exempted company duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. The Company has all requisite corporate or other entity power and authority to own, lease and operate its properties and assets and to carry on its business as being conducted on the date of this Agreement. Each other Target Company is a corporation or other entity duly formed, validly existing and in good standing under the Laws of its jurisdiction of organization and has all requisite corporate or other entity power and authority to own, lease and operate its properties and assets and to carry on its business as being conducted as of the date of this Agreement. Each Target Company is duly qualified or licensed and in good standing (to the extent that such concept applies) in the jurisdiction in which it is incorporated or registered and in each other jurisdiction where it does business or operates to the extent that the character of the property owned, or leased or operated by it or the nature of the business

conducted by it makes such qualification or licensing necessary except if the failure to be so qualified or licensed or be in good standing would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. <u>Schedule 4.1</u> list all jurisdictions in which any Target Company is so licensed to conduct business. The Company has made available to Purchaser accurate and complete copies of the Organizational Documents of each Target Company, each as amended to date and as currently in effect as of the date of this Agreement. No Target Company is in violation of any provision of its Organizational Documents in any material respect.

4.2 Authorization: Binding Agreement. The Company has all requisite corporate power and authority to execute and deliver this Agreement and each Ancillary Document to which it is or is required to be a party, to perform the Company's obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby, subject to the Required Company Shareholder Approval. Assuming that the Required Company Shareholder Approval has been obtained, the execution and delivery of this Agreement and each Ancillary Document to which the Company is a party, and, to the Knowledge of the Company, the Company shareholders party to the Company Shareholder Support Agreement, and the consummation of the transactions contemplated hereby and thereby, (a) have been duly and validly authorized by the Company (as applicable) in accordance with the Company's Organizational Documents, the Cayman Act and any other applicable Law, and (b) no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement and each Ancillary Document to which it is a party, and, to the Knowledge of the Company, the Company shareholders party to the Company Shareholder Support Agreement, or to consummate the transactions contemplated hereby and thereby. This Agreement has been, and each Ancillary Document to which the Company is a party shall be, when delivered, duly and validly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement and any such Ancillary Document by the other parties hereto and thereto and the obtainment of the Required Company Shareholder Approval, constitutes, or when delivered shall constitute, the legal, valid and binding obligation of the Company, in each case, enforceable against the Company, and, to the Knowledge of the Company the Company, shareholders party to the Company Shareholder Support Agreement, enforceable against the Company shareholders, in accordance with its terms, subject to the Enforceability Exceptions. As of the date of this Agreement, the Company shareholders party to the Company Shareholder Support Agreement hold Company Shares representing the voting power sufficient to obtain the Required Company Shareholder Approval.

15

#### 4.3 Capitalization.

(a) The authorized share capital of the Company is: \$50,000 divided into (i) 387,448,504 Class A Ordinary Shares ("Company Class A Ordinary Shares"), (ii) 72,000,000 Class B Ordinary Shares ("Company Class B Ordinary Shares," and together with Company Class A Ordinary Shares, "Company Ordinary Shares"), 2,040,816 Series Seed Preferred Shares ("Company Series Seed Preferred Shares"), 7,288,630 Series Angel Preferred Shares ("Company Series Angel Preferred Shares"), 13,974,442 Series Pre-A Preferred Shares ("Company Series Pre-A Preferred Shares") and 17,247,608 Series A Preferred Shares ("Company Series A Preferred Shares," and together with Company Series Seed Preferred Shares, Company Series Angel Preferred Shares and Company Series Pre-A Preferred Shares, "Company Preferred Shares"), each such class of Company Ordinary Shares and Company Preferred Shares with a par value of \$0.0001 per share. The issued and outstanding capital shares of the Company as of the date of this Agreement consists of 28,000,000 Company Class A Ordinary Shares, 72,000,000 Company Class B Ordinary Shares, 2,040,816 Company Series Seed Preferred Shares, 7,288,630 Company Series Seed Preferred Shares, 13,974,442 Company Series Pre-A Preferred Shares and 17,247,608 Company Series A Preferred Shares, and there are no other authorized, issued or outstanding equity interests of the Company. All of the outstanding shares and other equity interests of the Company (i) have been duly authorized and validly issued, are fully paid and nonassessable, (ii) except as set out in the Company's Organizational Documents, are not subject to or issued in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the Cayman Act, any other applicable Law or any Contract to which the Company is a party or by which the Company is bound and (iii) as of the date of this Agreement are owned legally and of record by the Persons set forth on Schedule 4.3(a) of the Company Disclosure Schedules. None of the outstanding Company Shares has been issued in violation of any applicable securities Laws. The Company does not, directly or indirectly, hold any of its shares or other equity interests in treasury. The only Company Shares that will be issued and outstanding immediately after the Closing will be the Company Shares owned by Purchaser.

(b) As of the date of this Agreement, the Company has no shares reserved for issuance to, among others, officers, directors, employees and consultants of the Company pursuant to any Company incentive plan or pursuant to any stand-alone compensatory equity grant.

(c) Other than the Company Shares, or except as set forth in the Company's Organizational Documents or <u>Schedule</u> <u>4.3(c)</u> of the Company Disclosure Schedules, there are no (i) outstanding options, warrants, puts, calls, convertible or exchangeable securities, "phantom" share rights, share appreciation rights, share-based units, preemptive or similar rights, (ii) bonds, debentures, notes or other Indebtedness having general voting rights or that are convertible or exchangeable into securities having such rights or (iii) subscriptions or other rights, agreements, arrangements, Contracts or commitments of any character (other than this Agreement and the Ancillary Documents) (A) relating to the issued or unissued securities of the Company or (B) obligating the Company to issue, transfer, deliver or sell or cause to be issued, transferred, delivered, sold or repurchased any options or shares or securities convertible into or exchangeable for any capital shares, or (C) obligating the Company to grant, extend or enter into any such option, warrant, call, subscription or other right, agreement, arrangement or commitment for any capital shares. Other than as expressly set forth in this Agreement, there are no outstanding obligations of the Company to repurchase, redeem or otherwise acquire any equity interests or shares of the Company or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any Person. Except as set forth on <u>Schedule 4.3(c)</u> of the Company Disclosure Schedules, there are no shareholders agreements, voting trusts, proxies or other agreements or understandings with respect to the voting of the Company's equity interests. Except as contemplated by this Agreement, as a result of the consummation of the transactions contemplated by this Agreement, no equity interests of the Company are issuable and no rights in connection with any interests, warrants, rights, options or other securities of the Company accelerate or otherwise become triggered (whether as to vesting, exercisability, convertibility or otherwise).

(d) Except as disclosed in the Company Financials or as set forth on <u>Schedule 4.3(d)</u> of the Company Disclosure Schedules, since the date of its formation, no Target Company has declared or paid any distribution or dividend in respect of its equity interests and has not repurchased, redeemed or otherwise acquired any equity interests of such Target Company, and the board of directors of such Target Company has not authorized any of the foregoing.

4.4 Subsidiaries. Schedule 4.4 of the Company Disclosure Schedules contains a complete and accurate list of each Subsidiary of the Company as of the date of this Agreement and, with respect to each Subsidiary, (a) its name and jurisdiction of organization, (b) its authorized shares or other equity interests (if applicable), and (c) the number of issued and outstanding shares or other equity interests and the record holders and beneficial owners thereof. All of the outstanding equity securities of each Subsidiary of the Company are duly authorized and validly issued, fully paid and non-assessable (if applicable), and were offered, sold and delivered in compliance with all applicable securities Laws, and owned by one or more of the Target Companies free and clear of all Liens (other than those, if any, imposed by such Subsidiary's Organizational Documents or applicable securities Laws). Any capital raising efforts in connection with an issuance of equity securities of the Subsidiaries was conducted in compliance with all applicable Laws. The Company has provided Purchaser with all documentation relating to the Restructuring. There are no Contracts to which the Company or any of its Affiliates is a party or bound with respect to the voting (including voting trusts or proxies) of the equity interests of any Subsidiary of the Company other than the Organizational Documents of any such Subsidiary. There are no outstanding or authorized options, warrants, rights, agreements, subscriptions, convertible securities or commitments to which any Subsidiary of the Company is a party or which are binding upon any Subsidiary of the Company providing for the issuance or redemption of any equity interests of any Subsidiary of the Company. There are no outstanding equity appreciation, phantom equity, profit participation or similar rights granted by any Subsidiary of the Company. As of the date of this Agreement, except for the equity interests of the Subsidiaries listed on Schedule 4.4 of the Company Disclosure Schedules, the Company does not own or have any rights to acquire, directly or indirectly, any equity interests of, or otherwise Control, any Person.

16

4.5 <u>Governmental Approvals</u>. Except as otherwise described in <u>Schedule 4.5</u> of the Company Disclosure Schedules, no Consent of or with any Governmental Authority on the part of any Target Company is required to be obtained or made in connection with the execution, delivery or performance by the Company of this Agreement or any Ancillary Documents or the consummation by the Company of the transactions contemplated hereby or thereby other than (a) such filings as expressly contemplated by this Agreement, (b) pursuant to Antitrust Laws, (c) any filings required with NASDAQ or the SEC with respect to the Transactions, (d) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder, and (e) where the failure to obtain or make such Consents or to make such filings or notifications would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect on the Company.

4.6 <u>Non-Contravention</u>. Except as otherwise described in <u>Schedule 4.6</u> of the Company Disclosure Schedules, the execution and delivery by the Company (or any other Target Company, as applicable) of this Agreement and each Ancillary Document to which any Target Company is a party, and the consummation by any Target Company of the transactions contemplated hereby and thereby and compliance by any Target Company with any of the provisions hereof and thereof, will not (a) conflict with or violate, or constitute a default under, any provision of such Target Company's Organizational Documents, (b) subject to obtaining the Consents from Governmental Authorities referred to in <u>Section 4.5</u> hereof, the waiting periods referred to therein having expired, and any condition

precedent to such Consent or waiver having been satisfied, conflict with or violate any Law, Order or Consent applicable to such Target Company or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by such Target Company under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of such Target Company under, (viii) give rise to any obligation to provide notice to, or obtain any third party Consent from, any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of any Company Material Contract, except for any deviations from any of the foregoing clauses (b) or (c) that would not reasonably be expected to, individually or in the aggregate, have a Material Adverse Effect on the Company.

#### 4.7 Financial Statements.

(a) <u>Schedule 4.7(a)</u> of the Company Disclosure Schedules contains true and correct copies of the Audited Financial Statements and the Interim Financial Statements. The Audited Financial Statements and the Interim Financial Statements (A) were prepared from the books and records of the Target Companies as of the times and for the periods referred to therein, (B) were prepared in accordance with GAAP, consistently applied throughout and among the periods involved (except as may be indicated in the notes thereto), and (C) fairly present in all material respects the consolidated financial position of the Target Companies as of the respective dates thereof and the consolidated results of the operations of the Target Companies for the periods indicated, except that the Interim Financial Statements (x) are subject to normal year-end adjustments and (y) do not include footnotes required under GAAP. No Target Company has ever been subject to the reporting requirements of Sections 13(a) and 15(d) of the Exchange Act.

(b) The PCAOB Audited Financial Statements when delivered by the Company in accordance with <u>Section 5.20</u> will, when so delivered, (A) be prepared from the books and records of the Target Companies as of the times and for the periods referred to therein, (B) be prepared in accordance with GAAP, consistently applied throughout and among the periods involved (except as may be indicated in the notes thereto), (C) fairly present in all material respects the consolidated financial position of the Target Companies as of the respective dates thereof and the consolidated results of the operations and cash flows of the Target Companies for the periods indicated, and (D) will comply in all material respects with the applicable accounting requirements and with the rules and regulations of the SEC and the Securities Act in effect as of such date.

(c) No Target Company is subject to any Liabilities that would be required to be reflected on the face of a balance sheet prepared in accordance with GAAP (or in footnotes thereto) except (i) as set forth on the consolidated balance sheet of the Company and its Subsidiaries as of the Interim Balance Sheet Date contained in the Company Financials, (ii) as set forth on <u>Schedule 4.7(c)</u> of the Company Disclosure Schedules, (iii) for Liabilities incurred after the Interim Balance Sheet Date in the Ordinary Course of Business, which Liabilities are not, individually or in the aggregate, material to the Target Companies taken as a whole, and (iv) for Liabilities incurred in connection with the negotiation, preparation or execution of this Agreement or any Ancillary Document, the performance of their respective covenants or agreements in this Agreement or any Ancillary Document or the consummation of the transactions contemplated hereby or thereby.

17

(d) Since January 31, 2021, except as reflected on <u>Schedule 4.7(d)</u> of the Company Disclosure Schedule, the Company (including the Company's personnel and independent accountants who participated in the preparation or review of financial statements or the internal accounting controls employed by the Company) have not identified nor been made aware of (i) any significant deficiency or material weakness in the system of internal accounting controls utilized by the Company, (ii) any fraud, whether or not material, that involves management of the Company or any personnel involved in financial reporting or (iii) any written claim or allegation regarding any of the foregoing.

(e) The Target Entities make and keep books, records and accounts that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of assets and maintain systems of internal accounting controls sufficient to provide reasonable assurances that (i) transactions are executed in accordance with management's general or specific authorization, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain accountability for assets, (iii) access to assets is permitted only in accordance with management's general or specific authorization and (iv) the recorded

accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

## 4.8 Absence of Certain Changes.

(a) Except as set forth on <u>Schedule 4.8</u> of the Company Disclosure Schedules or for actions expressly contemplated by this Agreement or the Contribution Agreement, since the date of its formation to the date of this Agreement, each Target Company has (a) conducted its business in the Ordinary Course of Business, and (b) not been subject to a Material Adverse Effect.

(b) Since the Balance Sheet Date and except as expressly required or contemplated by this Agreement or as otherwise disclosed on Schedule 4.8(b) of the Company Disclosure Schedule, (a) the Target Companies have conducted their respective businesses, in all material respects, in the Ordinary Course of Business, and (b) no action has been taken by the Company that would require consent under Section 5.2 if such action were taken after signing of this Agreement and prior to Closing.

4.9 <u>Compliance with Laws</u>. Except as set forth on <u>Schedule 4.9</u> of the Company Disclosure Schedules, no Target Company is, or since the date of its formation, has been, in conflict or non-compliance with, or in default or violation of, any Laws applicable to it or the conduct of its business, except as would not, individually or in the aggregate, reasonably be expected to result in Liabilities that are material to the Target Company, oral notice that it is under investigation with respect to, any material conflict or non-compliance with, or material default or violation of, any applicable Laws by which it is or was bound, except as would not, individually or in the aggregate, reasonably be expected to result in Liabilities that are material default or violation of, any applicable Laws by which it is or was bound, except as would not, individually or in the aggregate, reasonably be expected to result in Liabilities that are material to the Target Companies taken as a whole.

4.10 <u>Company Permits</u>. Each Target Company (and, to the Knowledge of the Company, its employees who are legally required to be licensed by a Governmental Authority in order to perform his or her duties with respect to his or her employment with any Target Company), holds all Permits necessary to lawfully conduct its business as conducted as of the date of this Agreement and to own, lease and operate its assets and properties (collectively, the *"Company Permits"*), all of which are in full force and effect, except where the failure to hold such Company Permits would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Company. <u>Schedule 4.10</u> of the Company Disclosure Schedules correctly lists each material Company Permit except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Company. Since the date of its formation, no Target Company has received any written or, to the Knowledge of the Company, oral notice of any Actions relating to the revocation or modification of any Company Permit except as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on the Company. None of the Company Permits will, assuming any related third-party consents set forth on Schedule 4.10 have been obtained or waived prior to the Closing Date, if applicable, be terminated or become terminable as a result of the transactions contemplated hereby.

4.11 <u>Litigation</u>. Except as described on <u>Schedule 4.11</u> of the Company Disclosure Schedules, as of the date of this Agreement, there is no material (a) Action of any nature currently pending or, to the Knowledge of the Company, threatened (and no such Action has been brought or, to the Knowledge of the Company, threatened since the date of its formation), including any proceeding relating to any Target Company's information privacy or data security practices relating to personal information of consumers, including with respect to the access, disclosure or use of personal information maintained by or on behalf of any Target Company or (b) Order now pending or outstanding or that was rendered by a Governmental Authority since the date of its formation, in either case of (a) or (b), against any Target Company, or its directors or officers (in their capacity as such) or its business or assets or which in any manner challenges or seeks to prevent, enjoin, alter or delay the transaction contemplated hereby or by any Ancillary Document. Since its formation, none of the officers, senior management or directors of any Target Company have been the subject of any proceeding with a Governmental Authority involving the indictment for, or conviction of, (i) any felony or (ii) any crime involving fraud.

## 4.12 Material Contracts.

(a) <u>Schedule 4.12(a)</u> of the Company Disclosure Schedules sets forth a true, correct and complete list of, and the Company has made available to Purchaser true, correct and complete copies of, each Contract to which any Target Company is a party or by which any Target Company, or any of its properties or assets are bound (each Contract required to be set forth on <u>Schedule 4.12(a)</u> of the Company Disclosure Schedules, a *"Company Material Contract"*) that:

(i) contains covenants that limit in any material respect the ability of any Target Company (A) to compete in any line of business or with any Person or in any geographic area or to sell, or provide any service or product or solicit any Person, including any non-competition covenants, employee and customer non-solicit covenants, exclusivity restrictions, rights of first refusal or most-favored pricing clauses or (B) to purchase or acquire an interest in any other Person;

(ii) relates to the formation, creation, operation, management or control of any joint venture, profit-sharing, partnership, non-wholly-owned limited liability company or other similar agreement or arrangement, or involving the sharing of profits or losses;

(iii) evidences Indebtedness (whether incurred, assumed, guaranteed or secured by any asset) of any Target Company having an outstanding principal amount, individually or in the aggregate, in excess of \$1.0 million;

Schedules;

(iv) each Company IP License required to be set forth on <u>Schedule 4.13(a)(ii)</u> of the Company Disclosure

(v) involves the lease, license, sale, use acquisition or disposition, directly or indirectly (by merger or otherwise), of a business or assets with an aggregate value in excess of \$1.0 million (other than in the Ordinary Course of Business) or shares or other equity interests of any Target Company or another Person;

(vi) relates to any merger, consolidation or other business combination with any other Person or the acquisition or disposition of any other entity or its business or assets with a value above \$1.0 million or the sale of any Target Company or its business or assets with a value above \$1.0 million;

(vii) by its terms, individually or with all related Contracts, requires aggregate payments or receipts by the Target Companies under such Contract or Contracts of at least \$1.0 million per year or \$1.0 million in the aggregate;

(viii) obligates the Target Companies to (A) provide a guarantee of obligations of a third party after the date of this Agreement in excess of \$1.0 million or (B) indemnification arrangements and other hold harmless arrangements made or provided by any Target Company to a third party, in each case, other than those incurred in the Ordinary Course of Business;

(ix) obligates the Target Companies to make any capital commitment or expenditure in excess of \$1.0 million (including pursuant to any joint venture);

(x) relates to the waiver, compromise, conciliation, settlement or similar resolution of any Action under which any Target Company has material outstanding obligations (other than customary confidentiality or non-disparagement obligations);

(xi) is an employment or engagement Contract with any officer, director, employee or individual independent contractor of any Target Company under which any Target Company (A) has continuing obligations for payment of annual base compensation of at least \$200,000, (B) has severance or post termination obligations in excess of \$200,000 as measured as of the date of this Agreement or (C) has any other liability or financial obligation of the Company in excess of \$200,000;

(xii) relates to the voting or control of the equity interests of the Target Companies or the election of directors of the Target Company (other than the Organizational Documents of the Target Companies);

(xiii) can be terminated, or the provisions of which can be altered, as a result of the consummation of the transactions contemplated by this Agreement or any Ancillary Document to which any Target Company is a party;

(xiv) relates to benefits, compensation or payments (or the vesting thereof) with respect to a director, officer, employee or independent contractor of any Target Company that will be increased or accelerated by the consummation of the transactions contemplated hereby or the payment thereof will be calculated on the basis of any of the transactions contemplated by this Agreement;

(xv) is between any Target Company, on one hand, and any Related Person, on the other hand; and

(xvi) that will be required to be filed with the Registration Statement under applicable SEC requirements or would otherwise be required to be filed by the Company as an exhibit for a Form S-l pursuant to Items 601(b)(1), (2), (4), (9) or (10) of Regulation S-K under the Securities Act as if the Company was the registrant.

(b) Except as disclosed in <u>Schedule 4.12(b)</u> of the Company Disclosure Schedules, with respect to each Company Material Contract: (i) such Company Material Contract was entered into at arms' length and in the Ordinary Course of Business; (ii) such Company Material Contract is legal, valid and binding and enforceable in all respects against the Target Company party thereto and, to the Knowledge of the Company, each other party thereto, and is in full force and effect (except, in each case, as such enforcement may be limited by the Enforceability Exceptions); (iii) the applicable Target Company has duly performed all of its material obligations under each Company Material Contract to which it is a party to the extent that such obligations to perform have accrued and no Target Company is in material breach or default, and to the Knowledge of the Company, or permit termination or acceleration by the other party thereto, under such Company Material Contract; and (iv) to the Knowledge of the Company, no other party to such Company Material Contract is in material breach or default, and no event has occurred that with the passage of time or giving of notice or both would constitute a material breach or default, and no event has occurred that with the passage of to such Company Material breach or default by such other party, or permit termination or acceleration by any Target Company, under such Company Material breach or default by such other party, or permit termination or acceleration by any Target Company.

## 4.13 Intellectual Property.

(a) Schedule 4.13(a)(i) of the Company Disclosure Schedules sets forth, as of the date of this Agreement all Patents and Patent applications, trademarks and service mark registrations and applications, copyright registrations and applications and registered Internet Assets owned by, registered or issued to, or applied for in the name of a Target Company ("Company Registered IP"), specifying as to each item, as applicable: (A) the title of the item, if applicable, (B) the owner, registrant, or applicant (as applicable) of the item, (C) the jurisdictions in which the item is issued or registered or in which an application for issuance or registration has been filed and (D) the issuance, registration or application numbers and dates. Schedule 4.13(a)(ii) of the Company Disclosure Schedules sets forth all Intellectual Property licenses, sublicenses and other agreements or permissions that are material to the Target Companies' businesses as conducted as of the date of this Agreement ("Company IP Licenses") (other than "shrink wrap," "click wrap," "open source software" and "off the shelf' software agreements and other agreements for Software that is available on reasonable terms to the public generally with license, maintenance, support and other fees of less than \$200,000 per year for any such item of Software or group of related items of Software, which are not required to be listed, although such licenses are "Company IP Licenses" as that term is used herein), under which a Target Company is a licensee or otherwise is authorized to use or practice any material Intellectual Property. Each Target Company owns, free and clear of all Liens (other than Permitted Liens) each item of Company Registered IP that is claimed to be owned by, registered in the name of, issued to, or applied for in the name of, such Target Company as set forth in Schedule 4.13(a)(i) of the Company Disclosure Schedules. Except as set forth on Schedule 4.13(a)(iii) of the Company Disclosure Schedules, all material Company Registered IP is owned exclusively by the applicable Target Company without obligation to pay royalties, licensing fees or other fees, or otherwise account to any third party with respect to such Company Registered IP, except for fees and costs payable to file, apply for, register, patent or maintain Company Registered IP.

(b) Each Target Company has a valid and enforceable license to use all material Intellectual Property that is the subject of the Company IP Licenses applicable to such Target Company. Other than Company IP, the Company IP Licenses include all of the licenses, sublicenses and other agreements or permissions for material Intellectual Property necessary to operate the Target Companies as conducted as of the date of this Agreement. Each Target Company has performed all material obligations imposed on it in the applicable Company IP Licenses, has made all material payments required under the applicable Company IP Licenses to date, and such Target Company is not in material breach or material default thereunder. All Software developed by or for a Target Company specifically for the business as described in the Business Plan (the *"Material Software"*) is either owned by a Target Company or is otherwise used pursuant to a valid license or other enforceable right (including any applicable Open Source Software License). Except as would not, individually or in the aggregate, be reasonably expected to have a Material Adverse Effect: (x) none of the Software used in the business of any of the Target Companies is a "bootleg" version or unauthorized copy; and (y) to the Knowledge of the Company, no such unauthorized Software is used in the business of any Target Company. The Material Software and other information technology that is material to the operation of the businesses of the Target Companies as conducted as of the date of this Agreement: (i) are in satisfactory working order (apart from any that is under development); (ii) have reasonable security, backups, disaster recovery arrangements, and hardware and software support; and (iii) to the Knowledge of the Company, are free from bugs and other defects, in each case, except as would not be material and adverse to the Company. (c) None of the material Company Registered IP, or other Intellectual Property material to the businesses of the Target Companies as conducted as of the date of this Agreement and owned or claimed to be owned by the any Target Company (collectively, "*Company IP*") (i) are subject to any action challenging the validity, enforceability, or ownership of such Intellectual Property before any Government Authority or arbitration entity, and (ii) to the Knowledge of the Company are or have been subject to any written threats or claims of estoppel, invalidity, or other unenforceability. All rights in such Company IP are fully vested with the Company (either directly or by virtue of its ownership of a Target Company, and in the case of applied-for Intellectual Property such vesting is subject to successful completion of registration and issuance by the applicable Governmental Authority). Without limitation on the preceding representations and warranties:

(i) No proceeding or action before any Governmental Authority is pending or, to the Knowledge of the Company, threatened in writing against a Target Company that challenges the validity, enforceability, ownership, or right to use, sell, license or sublicense any Company IP.

(ii) During the past five (5) years, no Target Company has received any written notice or claim asserting that any infringement, misappropriation, violation, dilution or unauthorized use of the Intellectual Property of any other Person in material respects is occurring or has occurred, in each case, as a consequence of the material business activities of any Target Company.

(iii) There are no Orders to which any Target Company is a party or is otherwise bound that (i) restrict the rights of a Target Company to use, transfer, license or enforce any material Intellectual Property owned by a Target Company,
 (ii) restrict the conduct of the business of a Target Company in any material respects in order to accommodate a third Person's Intellectual Property, or (iii) grant any third Person any right with respect to any Company IP that would restrict the conduct of the business of a Target Company in any material respect.

(iv) To the Knowledge of the Company, no Target Company is currently infringing as of the date of this Agreement, or has, in the past five (5) years: (x) infringed, misappropriated or violated any Intellectual Property of any other Person in any material respect that is reasonably likely to result in a material liability to any of the Target Companies, nor (y) been the subject of any written and unresolved material claim of misappropriation or violation of any Intellectual Property of another Person in any material respect.

(v) Except with respect to Software that has been made available by a Target Company under an Open Source Software License, none of the Material Software with respect to which copyrights are included in the Company IP is subject to any term or condition of an Open Source Software License that as used by a Target Company requires or purports to require release of such source code to third parties.

(vi) To the Company's Knowledge, no third party is infringing upon, misappropriating or otherwise violating any Company IP in any material respect.

(vii) No Company IP used in the business of the Company and its Subsidiaries is owned, leased or licensed by an Affiliate or other Related Person of the Company or its shareholders (other than by the Company or any of its direct or indirect wholly owned subsidiaries).

(d) All employees and independent contractors of a Target Company who develop or developed Material Software have assigned to such Target Company material Intellectual Property arising from the services performed for a Target Company by such Persons that is included in the Company IP. To the Knowledge of the Company, no current or former officers, employees or independent contractors of a Target Company have claimed any ownership interest in any material Intellectual Property owned by a Target Company. The Company has made available to Purchaser true and complete copies of templates of written Contracts used by the Target Companies, and it is the policy of the Target Companies to require all employees and independent contractors of a Target Software to sign such agreements or agreements containing similar provisions for the assignment of Intellectual Property rights developed for such Target Company, and to the Knowledge of the Company, all such employee and independent contractors have signed such agreements. Excluding Software licensed under an Open Source Software License, each Target Company has taken commercially reasonable security measures for the purposes of protecting the secrecy and confidentiality of the material Trade Secrets included in Company IP.

(e) Except as would not result in Liabilities that are material to the Target Companies taken as a whole, the consummation of any of the transactions contemplated by this Agreement will not result in the breach, modification, cancellation,

termination, suspension of, or acceleration of any payments by a Target Company under, or release of source code for software included in Company IP because of: (i) any Contract providing for the license granted by a Target Company to a third party to use Intellectual Property owned by a Target Company, or (ii) any Company IP License. Except as would not result in Liabilities that are material to the Target Companies taken as a whole, following the Closing, the Company shall be permitted to exercise, directly or indirectly through its Subsidiaries, all of the Target Companies' rights under such Contracts or Company IP Licenses to the same or similar extent that the Target Companies would have been able to exercise had the transactions contemplated by this Agreement not occurred, without the payment of any additional amounts or consideration other than ongoing fees, royalties or payments which the Target Companies would otherwise be required to pay in the absence of such transactions.

21

(f) Notwithstanding anything to the contrary herein, the representations and warranties in this <u>Section 4.13</u> and <u>Schedule 4.13</u> are the sole and exclusive representations and warranties of the Company concerning Intellectual Property matters.

### 4.14 Privacy and Data Security.

(a) Except as would not result in Liabilities that are material to the Target Companies taken as a whole, to the extent that a Target Company collects any personally identifiable information ("**PII**") from third party individual persons as of the date of this Agreement, such Target Company has a privacy policy (which may be a group wide policy covering affiliated entities) regarding the collection, use and disclosure of such PII in connection with the operation of the its business as conducted as of the date of this Agreement, and each Target Company is and has been in compliance with any such privacy policy applicable to it.

(b) Except as would not result in Liabilities that are material to the Target Companies taken as a whole, all Target Companies have complied at all times in all respects with all applicable Laws regarding the collection, retention, use and protection of personal information. There is no claim pending or threatened in writing against any Target Company regarding any violation of or noncompliance with such applicable Laws.

(c) Except as would not result in Liabilities that are material to the Target Companies taken as a whole, the Target Companies are in compliance with the terms of all contracts to which such Target Company or Target Companies are a party related to data privacy, security or breach notification (including provisions that impose conditions or restrictions on the collection, use, disclosure, transmission, destruction, maintenance, storage or safeguarding of personal information).

(d) To the Knowledge of the Company, none of the Target Companies have experienced any loss, damage, or unauthorized access, disclosure, use or breach of security of any PII in their possession, custody or control, or otherwise held or processed on their behalf.

(e) To the Knowledge of the Company, in the last three (3) years, there has been no unauthorized material access, intrusion or breach of security, or material failure, breakdown, performance reduction or other adverse event affecting any of the Target Company's systems, that has caused or could reasonably be expected to cause any: (i) material disruption of or interruption in or to the use of such systems or the conduct of the business of any Target Company ; or (ii) material loss, destruction, damage or harm to any Target Company or any of their material operations, personnel, property or other material assets. Each Target Company has taken reasonable actions, consistent with industry practices, to protect the integrity and security of the Target Company's systems and the data and other information stored thereon.

## 4.15 Taxes and Returns.

(a) <u>Schedule 4.15(a)</u> of the Company Disclosure Schedules sets forth each jurisdiction where any Target Company files or, to the Knowledge of the Company, is required to file a Tax Return.

(b) No Target Company has constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock intended to qualify for tax-free treatment under Section 355 of the Code in the two (2) years prior to the date of this Agreement.

(c) No Target Company has been or will be required to include any amount in income after the Closing by reason of Section 965(a) of the Code or has made an election described in Section 965(h) of the Code.

(d) There are no Liens with respect to any Taxes upon any Target Company's assets, other than Permitted Liens.

(e) No Target Company has any outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes and there are no outstanding requests by a Target Company for any extension of time within which to file any Tax Return or within which to pay any Taxes shown to be due on any Tax Return outside of the ordinary course of business.

(f) No Target Company is, or ever has ever been, a U.S. real property holding corporation within the meaning of Section 897(c)(2) of the Code.

(g) No Target Company (i) has been a member of an affiliated group filing a consolidated U.S. federal income Tax Return (other than a group the common parent of which was a Target Company or any of its current Affiliates) or a member of any other consolidated, combined, unitary or affiliated group of corporations for any Tax purpose, or (ii) has any material Liability for the Taxes of any Person (other than a Target Company or any of its current Affiliates) under Section 1.1502-6 of the Treasury Regulations (or any similar provision of U.S. state, local or non-United States Law), as a transferee or successor or by Contract (other than any Contract the principal purpose of which does not relate to Taxes).

(h) No Target Company has any material Liability for the Taxes of another Person (other than another Target Company) as a transferee or successor or by contract (excluding commercial agreements entered into in the ordinary course of business the primary purpose of which was not the sharing of Taxes).

(i) No Target Company has taken or agreed to take any action not contemplated by this Agreement and/or any Ancillary Document that could reasonably be expected to prevent the Transactions from qualifying for the Intended Tax Treatment. To the Knowledge of the Company and any Target Company, no facts or circumstances exist that could reasonably be expected to prevent the Merger from qualifying for the Intended Tax Treatment.

(j) No Target Company is or has been a party to any "listed transaction" as defined in Section 6707A of the Code and Treasury Regulations Section 1.6011-4 (or any corresponding or similar provision of state, local or non-U.S. income Tax Law).

(k) Since the date of its formation, each Target Company has not (i) changed any Tax accounting methods, policies or procedures except as required by a change in Law, (ii) made, revoked, or amended any material Tax election, (iii) filed any amended Tax Returns or claim for refund or (iv) entered into any closing agreement affecting or otherwise settled or compromised any material Tax Liability or refund.

(l) No Target Company will be required to include any material item of income in, or exclude any material item or deduction from, taxable income for any taxable period beginning after the Closing Date or, in the case of any taxable period beginning on or before and ending after the Closing Date, the portion of such period beginning after the Closing Date, as a result of: (i) an installment sale or open transaction disposition that occurred on or prior to the Closing Date other than in the ordinary course of business; (ii) any change in method of accounting on or prior to the Closing Date, including by reason of the application of Section 481 of the Code (or any analogous provision of U.S. state, local or non-U.S. Law); (iii) any prepaid amount received or deferred revenue recognized on or prior to the Closing Date, other than in respect of such amounts reflected in the balance sheet included in the Interim Financial Statements, or received in the Ordinary Course of Business since the Interim Balance Sheet Date; (iv) to the Target Company's Knowledge, any intercompany transaction described in Treasury Regulations under Section 1502 (or any corresponding or similar provision of U.S. state, local or non-U.S. Law) or (v) an election under 108(i) of the Code.

(m) Except as would not reasonably be expected to delay or have a lasting impact on the ability of the Target Companies, taken as a whole, to implement the Business Plan in all material respects:

(i) Each Target Company has timely filed, or caused to be timely filed, all Tax Returns required to be filed by it (taking into account all available extensions), which Tax Returns are true, accurate, correct and complete in all respects, and has paid, collected or withheld, or caused to be paid, collected or withheld, all Taxes required to be paid, collected or withheld, other than such Taxes for which adequate reserves in the Company Financials have been established in accordance with GAAP.

(ii) The Target Companies have complied in all respects with all Laws relating to the withholding and remittance of all amounts of Taxes, and all amounts of Taxes required by any Law to be withheld by the Target Companies have been withheld and paid over to the appropriate Governmental Entity.

(iii) Within the last five (5) years, no claim has been made by any Governmental Entity in a jurisdiction in which any Target Company does not file Tax Returns that it is or may be subject to Tax by, or required to file Tax Returns in, that jurisdiction.

(iv) There are no claims, assessments, audits, examinations, investigations or other Actions pending against any Target Company in respect of any Tax, and no Target Company has been notified in writing of any proposed Tax claims or assessments against such Target Company (other than, in each case, claims or assessments for which adequate reserves in the Company Financials have been established in accordance with GAAP).

23

(v) No Target Company has received any written or, to the Knowledge of the Company, other communication from a taxing authority alleging that such Target Company (x) should be classified as having a permanent establishment (within the meaning of an applicable Tax treaty), or (y) otherwise has an office, fixed place of business, trade or business, or taxable presence in a country other than the country in which it is organized.

(vi) Each Target Company is registered for the purposes of sales, use, value added or similar Taxes in the jurisdiction where such Target Company is organized to the extent such registration is required by Law, in each case, in all respects, and each Target Company has complied in all material respects with all Laws relating to such Taxes. No Target Company has received a written communication from any taxing authority in a jurisdiction where such Target Company has not registered for sales, use, value added or similar Taxes alleging that such Target Company is required to register for such Taxes in such jurisdiction.

(vii) Each Target Company is in compliance in all respects with all applicable transfer pricing laws and regulations.

(viii) No Target Company is currently as of the date of this Agreement being audited by any Tax authority or has been notified in writing or, to the Knowledge of the Company, orally, by any Tax authority that any such audit is contemplated or pending.

(n) The Company is treated as a corporation for U.S. federal (and applicable U.S. state and local) income Tax purposes.

(o) Notwithstanding anything to the contrary herein, the representations and warranties in this <u>Section 4.15</u> and <u>Section 4.15</u> and <u>Section 4.15</u> are the sole and exclusive representations and warranties of the Company concerning Tax matters.

## 4.16 Title; Real Property.

(a) Except with respect to Intellectual Property (which is addressed by <u>Section 4.13</u>) and Digital Assets (which are addressed by <u>Section 4.24</u>), (i) the Target Companies have good, valid and indefeasible title in and to, or in the case of the Personal Property or assets which are leased or licensed pursuant to Contracts, a valid leasehold interest or license in or a right to use, all of their Personal Property and assets reflected on the Interim Financial Statements or acquired after the date of the Interim Financial Statements, (ii) no such Personal Property or asset is subject to any Liens other than Permitted Liens and (iii) no such Personal Property or asset used in the business of the Company and its Subsidiaries is owned, leased or licensed by an Affiliate or other Related Person of the Company or its members.

(b) Schedule 4.16 of the Company Disclosure Schedules contains a complete and accurate list as of the date of this Agreement of all premises currently leased or subleased by a Target Company for the operation of the business of a Target Company, and of all current leases as of the date of this Agreement related thereto, including all amendments, terminations and modifications thereof or waivers thereto (collectively, the "Company Real Property Leases"). The Company has provided to Purchaser a true and complete copy of each of the Company Real Property Leases. The Company Real Property Leases are valid, binding and enforceable against the Target

Company party thereto and, to the Knowledge of the Company, each other party thereto, in accordance with their terms, and are in full force and effect (except, in each case, as such enforcement may be limited by the Enforceability Exceptions). To the Knowledge of the Company, no event has occurred which (whether with or without notice, lapse of time or both or the happening or occurrence of any other event) would constitute a default on the part of a Target Company or any other party under any of the Company Real Property Leases. No Target Company has received any written notice of any such condition. No Target Company owns or has ever owned any real property or any interest in real property (other than the leasehold interests in the Company Real Property Leases).

## 4.17 Employee and Labor Matters.

(a) The Target Companies have for the last three (3) years materially complied, and are in material compliance with, all applicable Laws regarding employment and employment practices, terms and conditions of employment and wages and hours, including those relating to discrimination, harassment, retaliation, leaves of absence, payroll and withholding taxes, immigration, occupational safety and health in the workplace (including applicable workplace orders regarding COVID-19), hours of work, payment of wages and overtime, meal and rest periods, notice and severance obligations, and termination of employment (the *"Labor Laws"*). As of the date of this Agreement, no Target Company employs or has agreed to employ any Person who is not permitted to work in the jurisdiction in which such Person was employed. None of the Target Companies has any direct or indirect material liability with respect to any misclassification of any Person as an independent contractor rather than as an employee, or as an "exempt" employee rather than a "non-exempt" employee (within the meaning of the Fair Labor Standards Act of 1938 or comparable state law).

(b) The Company has provided to Purchaser, in each case, as of August 31, 2021, (i) a true and complete list of each individual whose employment (as well as all associated Liabilities, including, but not limited to, paid to off) will transfer to the Target Companies prior to, at or following the Closing in accordance with the Contribution Agreement, together with all of the Target Companies' employees; (ii) the current base compensation of each such employee; (iii) each such employee's eligibility for bonus, commissions, or other incentive pay; (iv) the job title of each such employee; (v) the work location of each such employee; and (vi) the classification of each such employee as exempt or non-exempt for purposes of applicable wage payment Laws (if any). The final list shall be provided to Purchaser no later than seven (7) days prior to the Closing.

(c) <u>Schedule 4.17(c)</u> of the Company Disclosure Schedule sets forth, as of the date of this Agreement, (i) a true and complete list of all of the Target Companies' contractors or consultants who are providing services on an individual basis; and (ii) a description of fees and services for each such consultant or contractor.

(d) There are no charges, audits, investigations, grievances, or complaint proceedings pending, or, to the Knowledge of the Company, threatened, against the Target Companies before any worker's compensation board or any international, federal, state, or local agency responsible for the prevention of unlawful employment, occupational safety and health, or wage and hour practices. To the Knowledge of the Company, no Target Company has received notice from any governmental agency evidencing its intent to conduct an audit or an investigation relating to any employees or employment law compliance, occupational safety, or wage payment practices, and no such investigations are as of the date of this Agreement currently in progress.

(e) Except as set forth on <u>Schedule 4.17(e)</u> of the Company Disclosure Schedule, no Target Company is a party to any collective bargaining agreement or other Contract with any labor union, works council, employee association, or other labor organization. There has not been any actual or, to the Knowledge of the Company, threatened strike, lockout, or other material labor dispute against any Target Company since the date of its formation. There is no unfair labor practice, charge or complaint pending, unresolved or, to the Knowledge of the Company, threatened before any Governmental Entities where the Target Companies conduct business. There are no grievance or arbitration proceedings against any Target Company pending or, to the Knowledge of the Company, threatened.

## 4.18 Benefit Plans.

(a) Set forth on <u>Schedule 4.18(a)</u> of the Company Disclosure Schedules is a true and complete list of each Foreign Plan and Benefit Plan maintained, sponsored or contributed to by a Target Company or with respect to which a Target Company has any Liability (each, a *"Company Benefit Plan"*).

(b) The Company has made available to Purchaser, if applicable, copies of (i) the current plan document for each Company Benefit Plan (or a written summary if a plan document is not available) and any amendments thereto, and any related insurance policies, trust agreements and other funding arrangements, (ii) the most recent summary plan description for each Company Benefit Plan for which such a summary plan description is required and any summary of material modifications thereto, (iii) the most recent favorable determination, opinion or advisory letter, as applicable, from the IRS with respect to each Company Benefit Plan intended to be qualified under Section 401(a) of the Code, (iv) the three (3) most recent annual reports (Form 5500 series) with respect to any Company Benefit Plan, (v) the most recent financial statements with respect to any Company Benefit Plan, (vi) all non-routine notices that were issued by, or non-routine correspondence received from, the IRS, U.S. Department of Labor, or any other Governmental Entity with respect to any Company Benefit Plan within the last three years, and (vii) any material associated administrative agreements with respect to any Company Benefit Plan, in each case, as of the date of this Agreement.

(c) Each Company Benefit Plan has been adopted, established, registered (where required) administered, maintained and operated in material compliance with its terms and the requirements of applicable Law. Each Company Benefit Plan that is intended to meet the requirements of a "qualified plan" under Section 401(a) of the Code has received a favorable determination letter from the Internal Revenue Service (*"IRS"*) or is in the form of a prototype or volume submitter plan with respect to which the IRS has issued a favorable opinion or advisory letter, as applicable, and to the Knowledge of the Company, no event has occurred and no condition exists that would reasonably be expected to result in the loss of any such qualified status of such Company Benefit Plan. All contributions or premiums required to be remitted by the applicable Target Company under the terms of each Company Benefit Plan, or by applicable Law, have in all material respects been remitted in a timely fashion in accordance with applicable Law and the terms of the applicable Company Benefit Plan, and all Liabilities of the Target Companies related to all Company Benefit Plans have been fully and accurately disclosed in accordance with applicable accounting principles. No Target Company has incurred or is reasonably expected to incur any Liability for any Tax imposed under Chapter 43 of the Code or civil Liability under Section 502(i) or (1) of ERISA.

(d) No Company Benefit Plan is a (i) "defined benefit plan" (as defined in Section 3(35) of ERISA) or "multiemployer plan" (as defined in Section 4001(a) (3) of ERISA), in each case, that is or was subject to Section 412 of the Code or Title IV of ERISA, (ii) multiple employer plan described in Section 413 of the Code or Section 210 of ERISA, (iii) nonqualified deferred compensation plan within the meaning of Section 409A(d)(l) of the Code, (iv) "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA (whether or not subject thereto), (v) provides medical or life insurance benefits beyond termination of employment, except as required by applicable Law or until the end of the month of termination, or (vi) provides for any reimbursement, indemnity or gross-up to compensate an employee for any Tax Liability. No Target Company has any Liability under Title IV of ERISA or on account of at any time being considered a single employer under Section 414 of the Code with any other Person.

(e) The consummation of the transactions contemplated by this Agreement and the Ancillary Documents will not: (i) entitle any director, employee or individual independent contractor of a Target Company to severance pay or other payment, benefits or compensation under any Company Benefit Plan; (ii) accelerate the time of payment or vesting, or increase the amount of any compensation due, or in respect of, any director, employee or individual independent contractor of a Target Company; (iii) give rise to any forgiveness of indebtedness of any director, employee or individual independent contractor of a Target Company for amounts due or owing to a Target Company; (iv) cause any Target Company to transfer or set aside any assets to fund any benefits under any Company Benefit Plan, or (v) result in an excess parachute payment (within the meaning of Section 280G of the Code) or limit the Tax deductibility of any amount payable under any Company Benefit Plan by operation of Section 280G.

(f) There are no pending audits or investigations by any Governmental Authority involving any Company Benefit Plan and, to the Knowledge of the Company, no threatened or pending material claims (except for individual claims for benefits payable in the normal operations of the Company Benefit Plans, as applicable), suits or proceedings involving any Company Benefit Plan, nor, to the Knowledge of the Company, are there any facts which would reasonably be expected to give rise to any material liability in the event of any such audit, investigation, claim, suit or proceeding.

4.19 <u>Transactions with Related Persons</u>. Except as set forth on <u>Schedule 4.19</u> of the Company Disclosure Schedules, there are no Contracts or understandings between or binding on any Target Company on the one hand, and any Related Person of any Target Company, on the other hand, either (a) currently in effect or binding on any Target Company or (b) that would be required to be disclosed under

Item 404 of Regulation S-K promulgated under the Securities Act (treating for purposes of this <u>Section 4.19</u>, each Target Company as an issuer of securities for which a Securities Act registration statement is required to be filed and/or an issuer of securities with respect to which an Exchange Act registration statement or report is required to be filed).

## 4.20 Certain Business Practices.

(a) Since the date of its formation, no Target Company, nor, to the Knowledge of the Company, any of their respective Representatives acting on their behalf, has (i) used any funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, (ii) made any unlawful payment to foreign or domestic government officials or employees, to foreign or domestic political parties or campaigns or violated any provision of the U.S. Foreign Corrupt Practices Act of 1977 any other local or foreign anti-corruption or bribery Law or (iii) made any other unlawful payment. Since the date of its formation, no Target Company, nor, to the Knowledge of the Company, any of their respective Representatives acting on their behalf has directly or, knowingly, indirectly, given or agreed to give any unlawful gift or similar benefit in any material amount to any customer, supplier, governmental employee or any other Person who is or may be in a position to help or hinder any Target Company or assist any Target Company in connection with any actual or proposed transaction.

(b) Since the date of its formation, the operations of each Target Company are and have been conducted at all times in compliance with money laundering statutes in all applicable jurisdictions that govern the operations of the Target Company, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Authority that have jurisdiction on the Target Companies, and no Action involving a Target Company with respect to any of the foregoing is pending or, to the Knowledge of the Company, threatened.

(c) No Target Company or any of their respective directors, officers or employees, or, to the Knowledge of the Company, any other Representative acting on behalf of a Target Company is currently, or has been in the last five (5) years, (i) identified on the specially designated nationals or other blocked person list or otherwise currently subject to any sanctions administered by OFAC, the U.S. Department of State, or other applicable Governmental Authority, (ii) organized, resident, or located in, or a national of a comprehensively sanctioned country (currently, Cuba, Iran, North Korea, and the Crimea region of Ukraine); or (iii) in the aggregate, fifty (50) percent or greater owned, directly or indirectly, or otherwise controlled, by a person identified in (i) or (ii); and no Target Company has, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any Subsidiary, joint venture partner or other Person, in connection with any sales or operations in any country sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC or the U.S. Department of State since the date of its formation.

4.21 <u>Investment Company Act</u>. No Target Company is an "investment company" or a Person directly or indirectly "controlled" by or acting on behalf of a person subject to registration and regulation as an "investment company", in each case within the meaning of the Investment Company Act.

4.22 <u>Finders and Brokers</u>. Except as set forth in <u>Schedule 4.22</u> of the Company Disclosure Schedules, no broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission from Purchaser, the Target Companies or any of their respective Affiliates in connection with the transactions contemplated hereby based upon arrangements made by or on behalf of any Target Company.

4.23 <u>Insurance</u>. <u>Schedule 4.23</u> of the Company Disclosure Schedules lists all insurance policies (by policy number, insurer, coverage period, coverage amount, annual premium and type of policy) held by any Target Company relating to such Target Company or its business, properties, assets, directors, officers and employees, copies of which have been provided to Purchaser. All premiums due and payable under all such insurance policies have been timely paid and the applicable Target Company is otherwise in material compliance with the terms of such insurance policies. All such insurance policies are in full force and effect, and to the Knowledge of the Company, there is no threatened termination of, or material premium increase with respect to, any of such insurance policies. Except as set forth on <u>Schedule 4.23</u> of the Company Disclosure Schedules, there have been no insurance claims made by the Target Companies. Each Target Company has reported to its insurers all material claims and pending circumstances that would reasonably be expected to result in a claim.

4.24 <u>Ownership of Digital Assets</u>. As of the date of this Agreement, the Target Companies own and have the exclusive ability to control, including by use of "private keys" or other equivalent means or through custody arrangements or other equivalent means, all of the crypto-currencies, blockchain-based tokens, and other blockchain asset equivalents (collectively, "*Digital Assets*") set forth on <u>Schedule 4.24(i)</u> of the Company Disclosure Schedules, free and clear of all Liens except for Permitted Liens; <u>provided</u>, <u>however</u>, that such ownership and exclusive ability to control Digital Assets is subject to the continued existence, validity, legality, governance and public availability of the relevant blockchains. Except as set forth on <u>Schedule 4.24(ii)</u> of the Company Disclosure Schedules, the Target Companies have taken no actions where any of them owns a substantial portion of all outstanding tokens in the then existing issued and circulating supply of such tokens on a blockchain to effectuate change through the governance process of that relevant blockchain that could reasonably foreseeably disrupt the continued existence, validity, legality, governance or public availability of the relevant blockchain to effectuate change through the governance or public availability of the relevant blockchain to effectuate change through the governance or public availability of the relevant blockchain that could reasonably foreseeably disrupt the continued existence, validity, legality, governance or public availability of the relevant blockchains.

4.25 Information Supplied. None of the information supplied or to be supplied by the Company expressly for inclusion or incorporation by reference: (a) in any current report on Form 8-K, and any exhibits thereto or any other report, form, registration or other filing made with any Governmental Authority (including the SEC) with respect to the transactions contemplated by this Agreement or any Ancillary Documents; (b) in the Registration Statement; or (c) in the mailings or other distributions to Purchaser's shareholders and/or prospective investors with respect to the consummation of the transactions contemplated by this Agreement or any amendment to any of documents identified in (a) through (c), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. None of the Signing Press Release, the Signing Filing, the Closing Press Release and the Closing Filing will, when filed or distributed, as applicable, contain any untrue statement of a material fact required to be stated therein or necessary in order to make the closing Filing. Notwithstanding the foregoing, the Company makes no representation, warranty or covenant with respect to any information supplied by or on behalf of Purchaser or its Affiliates.

4.26 Independent Investigation. The Company has conducted its own independent investigation, review and analysis of the business, results of operations, condition (financial or otherwise) or assets of Purchaser and Merger Sub and acknowledges that it has been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of Purchaser and Merger Sub for such purpose. The Company acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, it has relied solely upon its own investigation and the express representations and warranties of Purchaser and Merger Sub set forth in this Agreement (including the related portions of the Purchaser Disclosure Schedules) and in any certificate delivered to the Company pursuant hereto, and the information provided by or on behalf of Purchaser or Merger Sub for the Registration Statement; and (b) none of Purchaser, Merger Sub or their respective Representatives have made any representation or warranty as to Purchaser or Merger Sub or this Agreement, except as expressly set forth in this Agreement (including the related portions of the Purchaser Disclosure Schedules) or in any certificate delivered to Company pursuant hereto.

4.27 EXCLUSIVITY OF REPRESENTATIONS AND WARRANTIES. NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO PURCHASER, MERGER SUB OR ANY OF ITS REPRESENTATIVES OF ANY DOCUMENTATION OR OTHER INFORMATION (INCLUDING ANY FINANCIAL PROJECTIONS OR OTHER SUPPLEMENTAL DATA), EXCEPT AS OTHERWISE EXPRESSLY SET FORTH IN THIS ARTICLE IV, NEITHER THE COMPANY NOR ANY OTHER PERSON MAKES, AND THE COMPANY EXPRESSLY DISCLAIMS, ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND OR NATURE, EXPRESS OR IMPLIED, AS TO THE MATERIALS RELATING TO THE BUSINESS AND AFFAIRS OR HOLDINGS OF THE COMPANY AND ITS SUBSIDIARIES THAT HAVE BEEN MADE AVAILABLE TO PURCHASER, MERGER SUB OR IN ANY PRESENTATION OF THE BUSINESS AND AFFAIRS OF THE COMPANY AND ITS SUBSIDIARIES BY THE MANAGEMENT OF THE COMPANY OR OTHERS IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED HEREBY. NO STATEMENT CONTAINED IN ANY OF SUCH MATERIALS OR MADE IN ANY SUCH PRESENTATION SHALL BE DEEMED A REPRESENTATION OR WARRANTY HEREUNDER OR OTHERWISE OR DEEMED TO BE RELIED UPON BY PURCHASER OR MERGER SUB IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY SET FORTH IN THIS ARTICLE IV. IT IS UNDERSTOOD THAT ANY COST ESTIMATES, PROJECTIONS OR OTHER PREDICTIONS, ANY DATA. ANY FINANCIAL INFORMATION OR ANY MEMORANDA OR OFFERING MATERIALS OR PRESENTATIONS, INCLUDING ANY OFFERING MEMORANDUM OR SIMILAR MATERIALS MADE AVAILABLE BY THE COMPANY OR ANY OF ITS SUBSIDIARIES ARE NOT AND SHALL NOT BE DEEMED TO BE OR TO INCLUDE REPRESENTATIONS OR WARRANTIES OF THE COMPANY, AND ARE NOT AND SHALL NOT BE DEEMED TO BE RELIED UPON BY PURCHASER OR MERGER SUB IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT OR ANY ANCILLARY DOCUMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND THEREBY.

### ARTICLE V COVENANTS

5.1 Access and Information.

(a) During the period from the date of this Agreement and continuing until the earlier of the termination of this Agreement in accordance with Section 8.1 or the Closing Date (the "Interim Period"), subject to Section 5.9, each of the Target Companies shall give, and shall cause its Representatives to give, Purchaser and its Representatives, at reasonable times during normal business hours and at reasonable intervals and upon reasonable advance notice, reasonable access to all offices and other facilities and to all properties, Contracts, books and records, financial and operating data and other similar information (including Tax Returns, internal working papers, client files, client Contracts and director service agreements) of the Target Companies or their Affiliates as Purchaser or its Representatives may reasonably request regarding the Target Companies or their Affiliates and their respective businesses, assets, Liabilities, financial condition, operations, management, employees and other aspects (including unaudited balance sheets and income statements, a copy of each material report, schedule and other document filed with or received by or from a Governmental Authority, and independent public accountants' work papers (subject to the consent or any other conditions required by such accountants, if any) in each case, if the financial statements or other documents already exist) and cause each of the Representatives of the Target Companies to reasonably cooperate with Purchaser and its Representatives in their investigation, and, except as provided in Section 5.9, the Target Companies or their Affiliates are not required to produce new reports or information that otherwise are not already in existence; provided, however, that Purchaser and its Representatives shall conduct any such activities in such a manner as not to unreasonably interfere with the business or operations of the Target Companies, Purchaser, Merger Sub or their Affiliates; provided, further, that (i) such access may be limited to the extent any of the Target Companies, Purchaser, Merger Sub or their Affiliates reasonably determines, in light of COVID-19 or COVID-19 Measures, that such access would jeopardize the health and safety of any employee of any of the Target Companies, Purchaser, Merger Sub or their Affiliates and (ii) nothing in this Agreement shall be deemed to provide Purchaser and its Representatives with the right to have access to any of the offices or information of any of the equityholders of any of the Target Companies, that is not otherwise related to any Target Company or the transactions contemplated by this Agreement or any Ancillary Document. Purchaser hereby agrees that, during the Interim Period, (x) it shall not contact any employee (excluding executive officers), customer, supplier, distributor or other material business relation of any Target Company regarding any Target Company or (y) conduct or perform any invasive or subsurface investigations of the properties or facilities of any Target Company or its Affiliates, in each case, without the prior written consent of the Company. Notwithstanding the foregoing, the Company shall not be required to provide access to any information (i) that is prohibited from being disclosed pursuant to the terms of a confidentiality agreement with a third party, (ii) the disclosure of which would violate any applicable Law or (iii) the disclosure of which would constitute a waiver of attorney-client, attorney work product or other legal privilege.

(b) During the Interim Period, subject to Section 5.9, Purchaser shall give, and shall cause its Representatives to give, the Company and its Representatives, at reasonable times during normal business hours and at reasonable intervals and upon reasonable advance notice, reasonable access to all offices and other facilities and to all properties, Contracts, books and records, financial and operating data and other similar information (including Tax Returns, internal working papers, client files, client Contracts and director service agreements), of Purchaser or its Subsidiaries, as the Company or its Representatives may reasonably request regarding Purchaser, its Subsidiaries and their respective businesses, assets, Liabilities, financial condition, operations, management, employees and other aspects (including unaudited quarterly financial statements, including a consolidated quarterly balance sheet and income statement, a copy of each material report, schedule and other document filed with or received by or from a Governmental Authority, and independent public accountants' work papers (subject to the consent or any other conditions required by such accountants, if any) in each case, if such financial statements or other documents already exist) and cause each of Purchaser's Representatives to reasonably cooperate with the Company and its Representatives in their investigation; provided, however, that the Company and its Representatives shall conduct any such activities in such a manner as not to unreasonably interfere with the business or operations of Purchaser or any of its Subsidiaries; provided, further, that such access may be limited to the extent Purchaser or its Subsidiaries reasonably determines, in light of COVID-19 or COVID-19 Measures, that such access would jeopardize the health and safety of any employee of Purchaser or its Subsidiaries. Notwithstanding the foregoing, Purchaser shall not be required to provide access to any information (i) that is prohibited from being disclosed pursuant to the terms of a confidentiality agreement with a third party, (ii) the disclosure of which would violate any applicable Law or (iii) the disclosure of which would constitute a waiver of attorney-client, attorney work product or other legal privilege.

### 5.2 Conduct of Business of the Company.

(a) Unless Purchaser shall otherwise consent in writing (including e-mail or other forms of electronic communications) (such consent not to be unreasonably withheld, conditioned or delayed), during the Interim Period, except as expressly contemplated by this Agreement or any Ancillary Document or in furtherance of or consistent with the Business Plan, as set forth on <u>Schedule 5.2(a)</u> of the Company Disclosure Schedules, or as required by applicable Law (including for the avoidance of doubt, any COVID-19 Measures), the Company shall, and shall cause its Subsidiaries to (i) conduct their respective businesses, in all material respects, in the Ordinary Course of Business which, for the avoidance of doubt, shall include any reasonable actions or omission in response to any change, effect, event, occurrence, state of facts or development attributable to COVID-19 or COVID-19 Measures and (ii) use commercially reasonable efforts to preserve intact, in all material respects, their respective business organizations, to keep available the services of their officers and employees providing services to the Target Companies, and to preserve the possession, control and condition of their respective material assets.

(b) Notwithstanding anything contained herein to the contrary, except as contemplated by this Agreement, any Ancillary Document or in furtherance of or consistent with the Business Plan, as set forth on <u>Schedule 5.2(b)</u> of the Company Disclosure Schedules, as required by applicable Law (including for the avoidance of doubt, any COVID-19 Measures), during the Interim Period, without the prior written consent (including e-mail or other forms of electronic communications) of Purchaser (such consent not to be unreasonably withheld, conditioned or delayed) or with respect to clauses (ix), (xi) and (xix) below as would not in the aggregate result in the Company acquiring or disposing of any material amount of assets (other than, for the avoidance of doubt, Digital Assets, with respect to which there are no restrictions to the extent such Digital Assets are acquired or disposed of in the Ordinary Course of Business as consideration for services or in connection with the performance of mining operations for Digital Assets and the value of such Digital Assets does not exceed \$500,000 or involve the payment of cash by the Target Companies in excess of \$500,000), the Company shall not, and shall cause its Subsidiaries to not:

(i) amend, waive or otherwise change, in any respect, its Organizational Documents;

(ii) authorize for issuance, issue, grant, sell, pledge, dispose of or propose to issue, grant, sell, pledge or dispose of any of its equity securities or any options, warrants, commitments, subscriptions or rights of any kind to acquire or sell any of its equity securities, or other securities, including any securities convertible into or exchangeable for any of its equity securities or other security interests of any class and any other equity-based awards, or engage in any hedging transaction with a third Person with respect to such securities;

(iii) split, combine, recapitalize or reclassify any of its shares or other equity interests or issue any other securities in respect thereof or pay or set aside any non-cash dividend or other non-cash distribution in respect of its equity interests, or directly or indirectly redeem, purchase or otherwise acquire or offer to acquire any of its securities, pay any cash dividend or other cash distribution;

(iv) incur, create, assume or otherwise become liable for any Indebtedness (which for this purpose excludes trade payables or similar obligations) in excess of \$5,000,000 in the aggregate, make a loan or advance to or investment in any third party (other than advancement of expenses to employees in the Ordinary Course of Business), or guarantee or endorse any Indebtedness, Liability or obligation of any Person in excess of \$5,000,000 individually or in the aggregate, in each case, except for Indebtedness for which the Target Companies will not have any liability after the Closing;

(v) make or rescind any material election relating to Taxes, settle any claim or Action relating to Taxes, file any amended Tax Return or claim for refund, take any action or knowingly fail to take any action where such action or failure could reasonably be expected to prevent the Merger from qualifying for the Intended Tax Treatment, or make any material change in its accounting or Tax policies or procedures, in each case except as required by applicable Law or in compliance with GAAP, or change its jurisdiction for Tax residency;

(vi) take any action or agree to take any action which would have the effect of changing the classification of the Company, or a Subsidiary, under the ES Legislation, except that:

(1) if any requirement to report under the ES Legislation falls during the Interim Period, the Company shall report, by the applicable deadline, on the basis that during the relevant financial period the Company and/or its Subsidiaries, as is relevant, did not carry on any relevant activity as such term is defined in the ES Legislation, and shall consult with the Purchaser as to the prescribed information to be filed;

(2) the Company shall promptly inform the Purchaser of any communication received by the Company and/or its Subsidiaries, its directors, shareholder(s) or registered agent from the TIA and shall consult with the Purchaser and shall give the Purchaser the opportunity to comment as to any reply or any other actions to be taken in response to such communication;

(vii) transfer or license to any Person other than the license of IP in the Ordinary Course of Business or otherwise extend, materially amend or modify, permit to lapse or fail to preserve any material Company Registered IP or other Company IP;

(viii) terminate, or waive or assign any material right under, any Company Material Contract;

(ix) acquire, including by merger, consolidation, acquisition of equity interests or assets, or any other form of business combination, any corporation, partnership, limited liability company, other business organization or any division thereof, or any material amount of assets outside the Ordinary Course of Business (including Digital Assets in the Ordinary Course of Business);

(x) establish any non-wholly owned Subsidiary;

(xi) make any capital expenditures that are not contemplated by the Business Plan (excluding, for the avoidance of doubt, incurring any Company Transaction Expenses), unless such amount has been reserved in the Company Financials;

(xii) voluntarily incur any Liability or obligation (whether absolute, accrued, contingent or otherwise) in excess of \$5,000,000 individually or in the aggregate (excluding the incurrence of any Company Transaction Expenses or any Liability or obligation pursuant to any of the Company Material Contracts or in the Ordinary Course of Business or Indebtedness in compliance with clause (iv) above);

(xiii) amend, modify, supplement, waive or otherwise change, in any material respect, the Business Plan;

(xiv) except in the Ordinary Course of Business, enter into any partnership or joint venture with any Person;

(xv) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization;

(xvi) settle any Action (including any Action relating to this Agreement or the transactions contemplated hereby), other than settlements that involve only the payment of monetary damages (and not the imposition of equitable relief on, or the admission of wrongdoing by, the Company, Purchaser, Merger Sub or any of their respective Subsidiaries), or otherwise pay, discharge or satisfy any Action or any Liabilities related thereto, unless such amount has either been reserved in the Company Financials or involves an amount less than \$500,000;

31

(xvii) sell, lease, license, transfer, exchange or swap, mortgage or otherwise pledge or encumber (including securitizations), or otherwise dispose of any material portion of its properties, assets or rights (subject to the limitations set forth on <u>Schedule 5.2(b)</u> of the Company Disclosure Schedules, other than Digital Assets in the Ordinary Course of Business as noted above);

(xviii) enter into any agreement, understanding or arrangement with respect to the voting of its equity securities;

(xix) enter into, amend, renew, waive or terminate (other than terminations in accordance with their terms) any Contract with any officer, director, manager, employee, consultant or security holder of any Target Company or any of their Affiliates;

(xx) commence any Action where the amount claimed exceeds \$5,000,000;

(xxi) (A) establish, enter into, adopt, amend, terminate or increase or accelerate the funding, payment or vesting of the compensation or benefits provided under, any Company Benefit Plan or any other benefit or compensation plan, Contract, program, policy, or arrangement that would be a Company Benefit Plan if in effect on the date of this Agreement (except as may be required to comply with applicable Law or by the terms of any of the foregoing as in effect on the date of this Agreement), (B) materially increase the aggregate compensation or benefits of, or loan or advance any money or other property to any employee or current or former independent contractor who is a natural person (other than base salary or hourly wage increases in the Ordinary Course of Business), (C) change the key management structure of any Target Company, including the hiring of additional officers or the termination of existing officers (other than for cause), or (D) grant or promise to grant, any bonuses, change in control payments, deferred compensation, severance, retention, equity or equity-based rights, or other compensatory payments or benefits (other than base salary or hourly wages in the Ordinary Course of Business), to any present or former employee, director, officer, or other service provider who is a natural person; or

(xxii) authorize or agree to do any of the foregoing actions.

## 5.3 Conduct of Business of Purchaser.

(a) Unless the Company shall otherwise consent in writing (including e-mail or other forms of electronic communications) (such consent not to be unreasonably withheld, conditioned or delayed), during the Interim Period, except as expressly contemplated by this Agreement or any Ancillary Document, as set forth on <u>Schedule 5.3(a)</u> of the Purchaser Disclosure Schedules, or as required by applicable Law (including for the avoidance of doubt, any COVID-19 Measures), Purchaser shall, and shall cause its Subsidiaries to, (i) conduct their respective businesses, in all material respects, in the ordinary course of business consistent with past practice which, for the avoidance of doubt, shall include any actions or omission in response to any change, effect, event, occurrence, state of facts or development attributable to COVID-19 or COVID-19 Measures and (ii) use commercially reasonable efforts to preserve intact, in all material respects, their respective business organizations, to keep available the services of their respective managers, directors and officers, and to preserve the possession, control and condition of their respective material assets, all as consistent with past practice. Notwithstanding anything to the contrary in this <u>Section 5.3</u>, nothing in this Agreement shall prohibit or restrict Purchaser from extending one or more times, in accordance with the Purchaser Charter and IPO Prospectus, the deadline by which it must complete its Business Combination, and no consent of any other Party shall be required in connection therewith.

(b) Without limiting the generality of <u>Section 5.3(a)</u> and except as contemplated by this Agreement or any Ancillary Document, as set forth on <u>Schedule 5.3(b)</u> of the Purchaser Disclosure Schedules, or as required by applicable Law (including for the avoidance of doubt, any COVID-19 Measures), during the Interim Period, without the prior written consent of the Company (such consent not to be unreasonably withheld, conditioned or delayed), Purchaser shall not, and shall cause its Subsidiaries to not:

32

#### (i) amend, waive or otherwise change, in any respect, its Organizational Documents;

(ii) authorize for issuance, issue, grant, sell, pledge, dispose of or propose to issue, grant, sell, pledge or dispose of any of its equity securities or any options, warrants, commitments, subscriptions or rights of any kind to acquire or sell any of its equity securities, or other securities, including any securities convertible into or exchangeable for any of its equity securities or other security interests of any class and any other equity-based awards, or engage in any hedging transaction with a third Person with respect to such securities;

(iii) split, combine, recapitalize or reclassify any of its shares or other equity interests or issue any other securities in respect thereof or pay or set aside any dividend or other distribution (whether in cash, equity or property or any combination thereof) in respect of its equity interests, or directly or indirectly redeem, purchase or otherwise acquire or offer to acquire any of its securities;

(iv) incur, create, assume or otherwise become liable for any Indebtedness, make a loan or advance to or investment in any third party, or guarantee or endorse any Indebtedness, Liability or obligation of any Person (<u>provided</u>, that this <u>Section 5.3(b)(iv</u>) shall not prevent Purchaser from borrowing funds necessary to finance its ordinary course administrative costs and expenses and Purchaser Transaction Expenses incurred in connection with the consummation of the Transactions, up to aggregate additional Indebtedness during the Interim Period of \$4.5 million);

(v) make or rescind any material election relating to Taxes, settle any claim or Action relating to Taxes, file any amended Tax Return or claim for refund, take any action or knowingly fail to take any action where such action or failure could reasonably be expected to prevent the Merger from qualifying for the Intended Tax Treatment, or make any material change in its accounting or Tax policies or procedures, in each case except as required by applicable Law or in compliance with GAAP, or change its jurisdiction of Tax residency; <u>provided</u>, that Purchaser may, without the consent of the Company, change the accounting treatment of Purchaser's issued and outstanding warrants with respect to the treatment of such warrants as equity rather than liabilities in Purchaser's financial statements;

(vi) take any action or agree to take any action which would have the effect of changing the classification of the Purchaser, or a Subsidiary, under the ES Legislation, except that:

(1) if any requirement to report under the ES Legislation falls during the Interim Period, Purchaser shall report, by the applicable deadline, on the basis that during the relevant financial period Purchaser and/or its Subsidiaries, as is relevant, did not carry on any relevant activity as such term is defined in the ES Legislation, and shall consult with the Company as to the prescribed information to be filed;

(2) Purchaser shall promptly inform the Company of any communication received by Purchaser and/or its Subsidiaries, its directors, shareholder(s) or registered agent from the TIA and shall consult with the Company and shall give the Company the opportunity to comment as to any reply or any other actions to be taken in response to such communication;

(vii) amend, waive or otherwise change the Trust Agreement in any manner adverse to Purchaser's ability to consummate the transactions contemplated by this Agreement;

(viii) terminate, waive or assign any material right under any material agreement to which it is a party;

(ix) establish any Subsidiary or enter into any new line of business;

(x) revalue any of its material assets or make any change in accounting methods, principles or practices, except to the extent required to comply with GAAP, and after consulting Purchaser's outside auditors; <u>provided</u> that Purchaser may, without the consent of the Company, change the accounting treatment of Purchaser's issued and outstanding warrants with respect to the treatment of such warrants as equity rather than liabilities in Purchaser's financial statements;

33

(xi) waive, release, assign, settle or compromise any Action (including any Action relating to this Agreement or the transactions contemplated hereby), other than waivers, releases, assignments, settlements or compromises that involve the payment of monetary damages (and not the imposition of equitable relief on, or the admission of wrongdoing by, Purchaser or its Subsidiary), or otherwise pay, discharge or satisfy any Actions, Liabilities or obligations, unless such amount has been reserved in the Purchaser Financials;

(xii) acquire, including by merger, consolidation, acquisition of equity interests or assets, or any other form of business combination, any corporation, partnership, limited liability company, other business organization or any division thereof, or any material amount of assets;

(xiii) make any capital expenditures (excluding, for the avoidance of doubt, incurring any Purchaser Transaction Expenses);

(xiv) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization (other than with respect to the Merger);

(xv) subject to the restrictions contained in <u>Section 5.22</u>, voluntarily incur any Liability or obligation (whether absolute, accrued, contingent or otherwise) in excess of \$1 million individually or \$5 million in the aggregate (excluding the incurrence of any Purchaser Transaction Expenses);

(xvi) sell, lease, license, transfer, exchange or swap, mortgage or otherwise pledge or encumber (including securitizations), or otherwise dispose of any material portion of its properties, assets or rights;

(xvii) hire any employees or adopt, sponsor, contribute to or otherwise incur any Liability (contingent or otherwise) under any Benefit Plans, Foreign Plans, any benefit plan sponsored by an ERISA Affiliate of Purchaser, or any other benefit or compensation plan, Contract, program, policy, or arrangement;

securities;

(xviii) enter into any agreement, understanding or arrangement with respect to the voting of its equity

(xix) enter into, amend, renew, waive or terminate (other than terminations in accordance with their terms) any transaction or Contract with any officer, director, manager, employee, consultant or security holder of Purchaser or any of their Affiliates; or (xx) authorize or agree to do any of the foregoing actions.

5.4 <u>Purchaser Public Filings</u>. During the Interim Period, Purchaser will keep current and timely file all of its public filings with the SEC and otherwise comply in all material respects with applicable securities Laws and shall use its reasonable best efforts prior to the Merger to maintain the listing of the Purchaser Units, the Purchaser Class A Shares and the Purchaser Public Warrants on NASDAQ; provided, that the Parties acknowledge and agree that from and after the Closing, the Parties intend to list on NASDAQ only the Purchaser Ordinary Shares and the Purchaser Public Warrants. It is understood and agreed that any actions or inactions taken by Purchaser, based on advice by its legal counsel or auditors, in connection with the accounting treatment of Purchaser's issued and outstanding warrants, or any deficiencies in disclosure (including with respect to accounting and disclosure controls) arising from the treatment of such warrants as equity rather than liabilities in Purchaser's financial statements shall not be a breach of the requirements of this <u>Section 5.4</u>.

# 5.5 No Solicitation.

(a) For purposes of this Agreement, (i) an "Acquisition Proposal" means any inquiry, proposal or offer, or any indication of interest in making an offer or proposal, from any Person or group at any time relating to an Alternative Transaction, and (ii) an "Alternative Transaction" means (A) with respect to the Company, a transaction (other than the transactions contemplated by this Agreement or with the approval of Purchaser) concerning the sale of (x) all or any material part of the business or assets of such Person or (y) any material amount of the shares or other equity interests or profits of such Person, in any case, whether such transaction takes the form of a sale of shares or other equity interests, assets, merger, consolidation, issuance of debt securities, management Contract, joint venture or partnership, or otherwise and (B) with respect to Purchaser and its Affiliates, a transaction (other than the transactions contemplated by this Agreement) concerning a Business Combination for Purchaser.

34

(b) During the Interim Period, in order to induce the other Parties to continue to commit to expend management time and financial resources in furtherance of the transactions contemplated hereby, each Party shall not, and shall cause its Representatives not to, without the prior written consent of the Company and Purchaser, directly or indirectly, (i) solicit, initiate or knowingly facilitate or assist the making, submission or announcement of, or intentionally encourage, any Acquisition Proposal with respect to such Party, (ii) furnish any non-public information regarding such Party or its Affiliates or their respective businesses, operations, assets, Liabilities, financial condition, prospects or employees to any Person or group (other than a Party to this Agreement or their respective Representatives) in connection with or in response to an Acquisition Proposal, (iii) engage or participate in discussions or negotiations with any Person or group with respect to an Acquisition Proposal, other than to promptly inform such Person or group that it is subject to an exclusivity agreement that prohibits it from engaging or participating in any discussions with respect to any Acquisition Proposal, (iv) approve, endorse or recommend, or publicly propose to approve, endorse or recommend, any Acquisition Proposal, or (v) negotiate or enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement in furtherance of any Acquisition Proposal; provided, however, any Change in Recommendation by Purchaser's board of directors in accordance with <u>Section 5.9</u> shall not be a breach of this <u>Section 5.5(b)</u>.

5.6 No Trading. The Company acknowledges and agrees that it is aware, and that their respective Affiliates are aware (and each of their respective Representatives is aware or, upon receipt of any material nonpublic information of Purchaser, will be advised) of the restrictions imposed by U.S. federal securities laws and the rules and regulations of the SEC and NASDAQ promulgated thereunder or otherwise (the *"Federal Securities Laws"*) and other applicable foreign and domestic Laws on a Person possessing material nonpublic information about a publicly traded company. The Company hereby agrees that, while it is in possession of such material nonpublic information, it shall not purchase or sell any securities of Purchaser, communicate such information to any third party, take any other action with respect to Purchaser in violation of such Laws, or cause or encourage any third party to do any of the foregoing.

## 5.7 Efforts.

(a) Subject to the terms and conditions of this Agreement, each Party shall use its commercially reasonable efforts, and shall cooperate fully with the other Parties, to use its commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things reasonably necessary, proper or advisable under applicable Laws and regulations to receive all applicable Consents of Governmental Authorities needed to consummate the transactions contemplated by this Agreement and to comply as promptly as practicable with all requirements of Governmental Authorities applicable to the transactions contemplated by this Agreement; provided, that nothing in this Agreement shall be deemed to provide Purchaser and its Representatives with the right to obtain information regarding (or, other than as explicitly required by this Agreement or any Ancillary Document, require any action on behalf of) any of the equityholders of any of the Target Companies that is not otherwise related to any of the Target Companies or the transactions contemplated by this Agreement or any Ancillary Document.

(b) In furtherance and not in limitation of Section 5.7(a). to the extent required under any Laws that are designed to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade ("Antitrust Laws"), each Party hereto agrees to make any required filing or application under Antitrust Laws, as applicable, at Purchaser's sole cost and expense, with respect to the transactions contemplated hereby as promptly as practicable, to supply as promptly as reasonably practicable any additional information and documentary material that may be reasonably requested pursuant to Antitrust Laws and use its commercially reasonable efforts to take all other actions reasonably necessary, proper or advisable to cause the expiration or termination of the applicable waiting periods under Antitrust Laws as soon as practicable, including by requesting early termination of the waiting period provided for under the Antitrust Laws. Each Party shall, in connection with its efforts to obtain all requisite approvals and authorizations for the transactions contemplated by this Agreement under any Antitrust Law, use its commercially reasonable efforts to: (i) cooperate in all respects with each other Party or its Affiliates in connection with any filing or submission and in connection with any investigation or other inquiry; (ii) keep the other Parties reasonably informed of any material communication received by such Party or its Representatives from, or given by such Party or its Representatives to, any Governmental Authority, in each case regarding any of the transactions contemplated by this Agreement; (iii) permit a Representative of the other Parties and their respective outside counsel to review any substantive communication given by it to, and consult with each other in advance of any meeting or conference with, any Governmental Authority or with any other Person, and to the extent permitted by such Governmental Authority or other Person, give a Representative or Representatives of the other Parties the opportunity to attend and participate in such meetings and conferences; (iv) in the event a Party's Representative is prohibited from participating in or attending any meetings or conferences, the other Parties shall keep such Party promptly and reasonably apprised with respect thereto; and (v) use commercially reasonable efforts to cooperate in the filing of any memoranda, white papers, filings, correspondence or other written communications explaining or defending the transactions contemplated hereby, articulating any regulatory or competitive argument, and/or responding to requests or objections made by any Governmental Authority.

35

(c) As soon as reasonably practicable following the date of this Agreement, the Parties shall reasonably cooperate with each other and use (and shall cause their respective Affiliates to use) their respective commercially reasonable efforts to prepare and file with Governmental Authorities requests for approval of the transactions contemplated by this Agreement and shall use all commercially reasonable efforts to have such Governmental Authorities approve the transactions contemplated by this Agreement. Each Party shall give prompt written notice to the other Parties if such Party or any of its Representatives receives any notice from such Governmental Authorities in connection with the transactions contemplated by this Agreement and shall promptly furnish the other Parties with a copy of such Governmental Authority notice. If any Governmental Authority requires that a hearing or meeting be held in connection with its approval of the transactions contemplated hereby, each Party shall arrange for Representatives of such Party to be present for such hearing

or meeting. If any objections are asserted with respect to the transactions contemplated by this Agreement under any applicable Law or if any Action is instituted (or threatened to be instituted) by any applicable Governmental Authority challenging any of the transactions contemplated by this Agreement as violative of any applicable Law or which would otherwise prevent, materially impede or materially delay the consummation of the transactions contemplated hereby, the Parties shall use their commercially reasonable efforts to resolve any such objections or Actions so as to timely permit consummation of the transactions contemplated by this Agreement, including in order to resolve such objections or Actions which, in any case if not resolved, would reasonably be expected to prevent, materially impede or materially delay the consummation of the transactions contemplated hereby. In the event any Action is instituted (or threatened to be instituted) by a Governmental Authority challenging the transactions contemplated by this Agreement, the Parties shall, and shall cause their respective Representatives to, reasonably cooperate with each other and use their respective commercially reasonable efforts to contest and resist any such Action and to have vacated, lifted, reversed or overturned any Order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the transactions contemplated by this Agreement.

(d) Prior to the Closing, each Party shall use its commercially reasonable efforts to obtain any Consents of Governmental Authorities or other third Persons as may be necessary for the consummation by such Party or its Affiliates of the transactions contemplated by this Agreement or required as a result of the execution or performance of, or consummation of the transactions contemplated by, this Agreement by such Party or its Affiliates, and the other Parties shall provide reasonable cooperation in connection with such efforts.

5.8 Further Assurances. The Parties hereto shall further cooperate with each other and use their respective commercially reasonable efforts to take or cause to be taken all actions, and do or cause to be done all things, necessary, proper or advisable on their part under this Agreement and applicable Laws to consummate the transactions contemplated by this Agreement as soon as reasonably practicable, including preparing and filing as soon as practicable all documentation to effect all necessary notices, reports and other filings; provided, that nothing in this Agreement shall be deemed to provide Purchaser and its Representatives with the right to obtain information regarding (or, other than as explicitly required by this Agreement or any Ancillary Document, require any action on behalf of) any of the equityholders of the Company that is not otherwise related to any of the Target Companies or the transactions contemplated by this Agreement or any Ancillary Document.

#### 5.9 The Registration Statement.

(a) As promptly as practicable after the date of this Agreement, Purchaser and the Company shall jointly prepare and mutually agree upon (such agreement not to be unreasonably withheld, conditioned or delayed by any of the Parties), and Purchaser shall (at the sole cost and expense of Purchaser) file with the SEC a registration statement on Form F-4 (as amended or supplemented from time to time, and including the Proxy Statement contained therein, the "Registration Statement") in connection with the registration under the Securities Act of the Purchaser Securities to be issued under this Agreement to the holders of securities of the Company prior to the Merger Effective Date, which Registration Statement will also contain a proxy statement of Purchaser (as amended, the "Proxy Statement") for the purpose of soliciting proxies or votes from Purchaser shareholders for the matters to be acted upon at the Special Meeting and providing the Public Shareholders an opportunity in accordance with Purchaser's Organizational Documents and the IPO Prospectus to have their Purchaser Class A Shares redeemed (the "Redemption") in conjunction with the shareholder vote on the Purchaser Shareholder Approval Matters. The Proxy Statement shall include proxy materials for the purpose of soliciting proxies from Purchaser shareholders to vote, at an extraordinary general meeting of Purchaser shareholders to be called and held for such purpose (the "Special Meeting"), in favor of resolutions approving the following proposals (or such other proposals as may be agreed upon from time to time between the Company and Purchaser) (A) the adoption and approval of this Agreement and the Transactions by the holders of Purchaser Shares in accordance with Purchaser's Organizational Documents, the Cayman Act and the rules and regulations of the SEC and NASDAQ, (B) the adoption of the Amended Purchaser Charter, (C) the adoption and approval of a new equity incentive plan for Purchaser in substantially the form of Exhibit J or as the Company and Purchaser otherwise mutually agree on pursuant to Section 5.15 (the "Purchaser Equity Plan"), (D) the appointment of the members of the Post-Closing Purchaser Board, in each case in accordance with Section 5.14 hereof, (E) such other matters (if any) as the Company and Purchaser shall hereafter mutually determine to be necessary or appropriate in order to effect the Transactions (the approvals described in foregoing clauses (A) through (E), collectively, the "Purchaser Shareholder Approval Matters", and the approvals described in clauses (A) through (B), the "Required Purchaser Shareholder Approval Matters"), and (F) the adjournment of the Special Meeting, if necessary or desirable in the reasonable determination of Purchaser.

(b) Purchaser, acting through its board of directors (or a committee thereof), shall (i) make the Purchaser Recommendation and include such Purchaser Recommendation in the Proxy Statement and (ii) use its commercially reasonable efforts to solicit from its shareholders proxies or votes in favor of the approval of the Purchaser Shareholder Approval Matters, and (iii) use its commercially reasonable efforts to secure the approval of the Purchaser Shareholder Approval Matters; provided, however, that Purchaser's board of directors may change, withdraw, withhold, qualify or modify, or publicly propose to change, withdraw, withhold, qualify or modify, its recommendation (a "Change in Recommendation") if it determines in good faith, after consultation with its outside legal counsel and/or financial advisors, that a failure to make a Change in Recommendation would reasonably be expected to constitute a breach by Purchaser's board of directors of its fiduciary obligations to Purchaser's shareholders under applicable Law. If on the date for which the Special Meeting is scheduled, Purchaser has not received proxies and votes representing a sufficient number of shares to obtain the Required Purchaser Shareholder Approval Matters, whether or not a quorum is present, Purchaser may make one or more successive postponements or adjournments of the Special Meeting. In connection with the Registration Statement, Purchaser will, with the agreement of the Company prior to any such filing (such agreement not to be unreasonably withheld, conditioned or delayed), file with the SEC financial and other information about the transactions contemplated by this Agreement in accordance with applicable Law, Purchaser's Organizational Documents, the Cayman Act and the rules and regulations of the SEC and NASDAQ. Purchaser shall cooperate and provide the Company (and its counsel) with a reasonable opportunity to review and agree upon (such agreement not to be unreasonably withheld, conditioned or delayed) the Registration Statement and any amendment or supplement thereto prior to filing the same with the SEC. The Company shall provide Purchaser with such information concerning the Target Companies and their equity holders, officers, directors, employees, assets, Liabilities, condition (financial or otherwise), business and operations that may be required or appropriate for inclusion in the Registration Statement, or in any amendments or supplements thereto, which information provided by the Company shall be true and correct and not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made, in light of the circumstances under which they were made, not materially misleading.

(c) Purchaser shall take any and all reasonable and necessary actions required to satisfy the requirements of the Securities Act, the Exchange Act and other applicable Laws in connection with the Registration Statement, the Special Meeting and the Redemption. Each of Purchaser and the Company shall, and shall cause each of its Subsidiaries to, make their respective directors, officers and employees, upon reasonable advance notice, available to the Company, Purchaser and their respective Representatives in connection with the drafting of the public filings with respect to the transactions contemplated by this Agreement, including the Registration Statement, and responding in a timely manner to comments from the SEC. Each Party shall promptly correct any information provided by it for use in the Registration Statement (and other related materials) if and to the extent that such information has become false or misleading in any material respect or as otherwise required by applicable Laws. Purchaser shall, with the agreement of the Company prior to filing any such amendment or supplement (such agreement not to be unreasonably withheld, conditioned or delayed), amend or supplement the Registration Statement and Purchaser shall (at the sole cost and expense of Purchaser), with the agreement of the Company prior to any such filing (such agreement not to be unreasonably withheld, conditioned or delayed), file the Registration Statement, as so amended or supplemented, with the SEC and to be disseminated to Purchaser's shareholders, in each case as and to the extent required by applicable Laws and subject to the terms and conditions of this Agreement and Purchaser's Organizational Documents.

(d) Purchaser shall respond to any SEC comments on the Registration Statement and shall otherwise use their commercially reasonable efforts to cause the Registration Statement to "clear" comments from the SEC and become effective. Purchaser shall provide the Company with copies of any written comments, and shall inform the Company of any material oral comments, that Purchaser or its Representatives receive from the SEC or its staff with respect to the Registration Statement, the Special Meeting and the Redemption promptly after the receipt of such comments and shall give the Company a reasonable opportunity under the circumstances to review and agree upon (such agreement not to be unreasonably withheld, conditioned or delayed) any proposed written or material oral responses to such comments.

(e) As soon as practicable following the Registration Statement "clearing" comments from the SEC and becoming effective, Purchaser shall distribute the Registration Statement to Purchaser's shareholders and, Purchaser shall call the Special Meeting in accordance with the Cayman Act for a date no later than thirty (30) days following the effectiveness of the Registration Statement.

37

(f) Purchaser shall comply with all applicable Laws, any applicable rules and regulations of NASDAQ, Purchaser's Organizational Documents and this Agreement in the preparation, filing and distribution of the Registration Statement, any solicitation of proxies thereunder, the calling and holding of the Special Meeting and the Redemption.

(g) For the avoidance of doubt, the Company shall not be required by this Section 5.9 to take, or cause to be taken, any action in response to comments of the SEC staff made in connection with the Registration Statement, that would or could reasonably be expected to result in (x) any material and lasting change in the business of the Target Companies as contemplated under the Business Plan as of the date of this Agreement or (y) any Target Company being required to hold, obtain or apply for any material Permit from any Government Authority other than those set forth in Schedule 4.10 of the Company Disclosure Schedules.

5.10 Required Company Shareholder Approval; Series A SPA and Company Shareholders Agreement Waivers. As promptly as reasonably practicable (and in any event within one Business Day) following the Registration Statement "clearing" comments from the SEC and becoming effective, the Company shall duly seek unanimous written consent of its shareholders or duly call and convene a meeting of its shareholders, for the purposes of obtaining the requisite shareholders' consent for the adoption and approval of this Agreement and the Transactions (including approval of the Merger and the adoption of the amended and restated memorandum and articles of association of the Surviving Company in the form of Exhibit G), in accordance with the Company's Organizational Documents and the Cayman Act (such consent, either obtained at a meeting of the Company's shareholders or in writing, the "*Required Company Shareholder Approval*"). The Company, acting through its board of directors (or a committee thereof), shall not withhold, withdraw, qualify, amend or modify, or publicly propose or resolve to withhold, withdraw, qualify, amend or modify, its recommendation that the Company's shareholders vote in favor of the approval of this Agreement and the Transactions. As promptly as practicable after the date of this Agreement, the Company shall seek the Series A SPA and Company Shareholders Agreement Waivers and use its commercially reasonable effort to obtain all such waivers prior to the Registration Statement "clearing" comments from the SEC and becoming effective.

5.11 <u>Required Merger Sub Shareholder Approval</u>. Subject to Purchaser Shareholder adoption and approval of this Agreement and the Transactions at the Special Meeting, prior to the Merger Effective Date, the Purchaser shall execute and deliver a unanimous written consent as the sole shareholder of Merger Sub for the approving the Merger and the adoption of the amended and restated memorandum and articles of association of the Surviving Company substantially in the form of <u>Exhibit G</u>), in accordance with the Merger Sub's Organizational Documents and the Cayman Act (such consent in writing, the "*Required Merger Sub Shareholder Approval*").

## 5.12 Public Announcements.

(a) The Parties agree that, during the Interim Period, no public release, filing or announcement concerning this Agreement or the Ancillary Documents or the transactions contemplated hereby or thereby shall be issued by any Party or any of their Affiliates without the prior written consent (not to be unreasonably withheld, conditioned or delayed) of Purchaser and the Company, except as such release or announcement may be required by applicable Law or the rules or regulations of any securities exchange, in which case the applicable Party shall use commercially reasonable efforts to allow the other Parties reasonable time to comment on, and arrange for any required filing with respect to, such release or announcement in advance of such issuance.

(b) The Parties shall mutually agree upon and, as promptly as practicable after the execution of this Agreement (but in any event within four (4) Business Days thereafter), issue a press release announcing the execution of this Agreement (the "Signing Press Release"). Promptly after the issuance of the Signing Press Release (but in any event within four (4) Business Days after the execution of this Agreement), Purchaser shall file a current report on Form 8-K (the "Signing Filing") with the Signing Press Release and a description of this Agreement as required by Federal Securities Laws, which the Company shall review and comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) prior to filing (with a draft of which shall be provided to the Company for reviewing and comment promptly following the execution of this Agreement); provided, however, if the Company does not approve of the Form 8-K on or prior to the date such filing is required to be made pursuant to Federal Securities Laws, the failure to secure the approval of the Company shall not prevent Purchaser from making such filing in accordance with Federal Securities Laws. Prior to Closing, Purchaser and the Company shall prepare a current report on Form 8-K to be filed by Purchaser announcing the Closing, together with, or incorporating by reference, the financial statements prepared by the Company and its accountant, and such other information that may be required to be disclosed with respect to the Transactions in any report or form to be filed with the SEC ("Closing Filing"), provided, however, if the Company does not approve of the Form 8-K on or prior to the date such filing is required to be made pursuant to Federal Securities Laws, the failure to secure the approval of the Company shall not prevent Purchaser from making such filing in accordance with Federal Securities Laws. Purchaser shall review, comment upon and approve (which approval shall not be unreasonably withheld, conditioned or delayed) the Closing Filing prior to filing. The Parties shall mutually agree upon and, as promptly as practicable after the Closing Date (but in any event within four (4) Business Days thereafter), issue a press release announcing the consummation of the transactions contemplated by this Agreement (the "Closing Press Release"). In connection with the preparation of the Signing Press Release, the Signing Filing, the Closing Filing or the Closing Press Release, or any other report, statement, filing notice or application made by or on behalf of a Party to any Governmental Authority or other third party in the Interim Period in connection with the transactions contemplated hereby, each Party shall, upon request by any other Party, furnish the Parties with all information concerning themselves, their respective directors, officers and equity holders, and such other matters as may be reasonably necessary or advisable in connection with the transactions contemplated hereby, or any other report, statement, filing, notice or application made by or on behalf of a Party to any third party and/ or any Governmental Authority in connection with the transactions contemplated hereby.

#### 5.13 Confidential Information.

(a) The Company, agrees that during the Interim Period and, in the event this Agreement is terminated in accordance with Article VIII, for a period of two (2) years after such termination, they shall, and shall cause their respective Representatives to: (i) treat and hold in strict confidence any Purchaser Confidential Information that is provided to such Person, and will not use for any purpose (except in connection with the consummation of the transactions contemplated by this Agreement or the Ancillary Documents, performing their obligations hereunder or thereunder or enforcing their rights hereunder or thereunder), nor directly or indirectly disclose, distribute, publish, disseminate or otherwise make available to any third party any of the Purchaser Confidential Information without Purchaser's prior written consent; and (ii) in the event that the Company or any of its Representatives, during the Interim Period or, in the event that this Agreement is terminated in accordance with Article VIII, for a period of two (2) years after such termination, becomes legally compelled to disclose any Purchaser Confidential Information, (A) provide Purchaser to the extent legally permitted with prompt written notice of such requirement so that Purchaser or an Affiliate thereof may seek, at Purchaser's cost, a protective Order or other remedy or waive compliance with this Section 5.13(a) and (B) in the event that such protective Order or other remedy is not obtained, or Purchaser waives compliance with this Section 5.13(a), furnish only that portion of such Purchaser Confidential Information which is legally required to be provided as advised by outside counsel and to exercise its commercially reasonable efforts to obtain assurances that confidential treatment will be accorded such Purchaser Confidential Information. In the event that this Agreement is terminated and the transactions contemplated hereby are not consummated, the Company shall, and shall cause its Representatives to, promptly deliver to Purchaser or destroy (at Purchaser's election) any and all copies (in whatever form or medium) of Purchaser Confidential Information and destroy all notes, memoranda, summaries, analyses, compilations and other writings related thereto or based thereon. The Company shall, and shall cause its Representatives to, treat and hold in strict confidence any Trade Secret of Purchaser disclosed to such Person until such information ceases to be a Trade Secret and (z) the Company and its Representatives shall be permitted to retain copies of Purchaser Confidential Information to the extent required by internal compliance policies or applicable Laws or to satisfy requirements of a Governmental Authority.

(b) Purchaser and Merger Sub hereby agree that during the Interim Period and, in the event that this Agreement is terminated in accordance with Article VIII, for a period of two (2) years after such termination, they shall, and shall cause their respective Representatives to: (i) treat and hold in strict confidence any Company Confidential Information that is provided to such Person, and will not use for any purpose (except in connection with the consummation of the transactions contemplated by this Agreement or the Ancillary Documents, performing its obligations hereunder or thereunder or enforcing its rights hereunder or thereunder), nor directly or indirectly disclose, distribute, publish, disseminate or otherwise make available to any third party any of the Company Confidential Information without the Company's prior written consent; and (ii) in the event that Purchaser, Merger Sub or any of their respective Representatives, during the Interim Period or, in the event that this Agreement is terminated in accordance with Article VIII, for a period of two (2) years after such termination, becomes legally compelled to disclose any Company Confidential Information, (A) provide the Company to the extent legally permitted with prompt written notice of such requirement so that the Company may seek, at the Company's sole expense, a protective Order or other remedy or waive compliance with this Section 5.13(a) and (B) in the event that such protective Order or other remedy is not obtained, or the Company waives compliance with this Section 5.13(b), furnish only that portion of such Company Confidential Information which is legally required to be provided as advised by outside counsel and to exercise its commercially reasonable efforts to obtain assurances that confidential treatment will be accorded such Company Confidential Information. In the event that this Agreement is terminated and the transactions contemplated hereby are not consummated, Purchaser and Merger Sub shall, and shall cause their respective Representatives to, promptly deliver to the Company or destroy (at the Company's election) any and all copies (in whatever form or medium) of Company Confidential Information and destroy all notes, memoranda, summaries, analyses, compilations and other writings related thereto or based thereon. Notwithstanding the foregoing, (x) Purchaser and its Representatives shall be permitted to disclose any and all Company Confidential Information to the extent required by Federal Securities Laws, as advised by outside counsel and after giving the Company a reasonable opportunity under the circumstances to review such disclosure and considering in good faith any comments made by the Company regarding such disclosure, (y) Purchaser and Merger Sub shall, and shall cause their respective Representatives to, treat and hold in strict confidence any Trade Secret of any Target Company disclosed to such Person until such information ceases to be a Trade Secret and (z) Purchaser, Merger Sub and their respective Representatives shall be permitted to retain copies of Company Confidential Information to the extent required by internal compliance policies or applicable Laws or to satisfy requirements of a Governmental Authority.

(c) For the avoidance of doubt, the obligations set forth in this <u>Section 5.13</u> are in addition to and shall not supersede any continuing obligations under any confidentiality agreement between or among the Parties.

#### 5.14 Post-Closing Board of Directors and Executive Officers.

(a) The Parties shall take all necessary action, including causing the directors of the Purchaser to resign, so that effective as of the Closing, Purchaser's board of directors (the "*Post-Closing Purchaser Board*") will consist of the individuals listed on <u>Schedule</u> <u>5.14(a)</u> of the Company Disclosure Schedules. At or prior to the Closing, Purchaser will provide each director with a customary director indemnification agreement, in form and substance reasonably acceptable to such director.

(b) The Parties shall take all action necessary, including causing the executive officers of Purchaser to resign, so that, effective as of the Closing, the executive officers of Purchaser will consist of the individuals listed on <u>Schedule 5.14(b)</u> of the Company Disclosure Schedules.

5.15 <u>Purchaser Equity Plan</u>. Prior to the effectiveness of the Registration Statement / Proxy Statement, the board of directors of Purchaser shall approve and adopt the Purchaser Equity Plan, substantially in the form of <u>Exhibit J</u> or as the Company and Purchaser otherwise mutually agree (such agreement not to be unreasonably withheld, conditioned or delayed by either the Company or Purchaser, as applicable), and in the manner prescribed under applicable Laws, effective as of the Closing Date, reserving for grant thereunder the number of Purchaser Class A Ordinary Shares set forth on <u>Section 5.15</u> of the Company Disclosure Schedules.

#### 5.16 Indemnification of Directors and Officers.

(a) From and after the Closing Date, the Surviving Company and Purchaser shall jointly and severally indemnify and hold harmless (i) each present and former director and officer of the Target Company, Purchaser or Merger Sub, and (ii) in addition, solely with respect to the Target Company, named senior executives of the Target Company (in each case, solely to the extent acting in his or her capacity as such and to the extent such activities are related to the business of the relevant Target Company, Purchaser or Merger Sub, respectively) (the **"D&O Indemnified Parties"**) against any costs or expenses (including reasonable attorneys' fees), judgments, fines, losses, claims, damages or liabilities incurred in connection with any Action, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the Closing Date, whether asserted or claimed prior to, at or after the Closing Date (each, a "**Claim**"), to the fullest extent that the relevant Target Company, Purchaser or Merger Sub, respectively, would have been permitted under applicable Law and subject to the limitations of its respective Organizational Documents and indemnification agreements, if any, in effect from time to time at or prior to the Closing to indemnify such D&O Indemnified Parties (including the advancing of expenses as incurred to the fullest extent permitted under applicable Law). After the Closing Date, in the event that any D&O Indemnified Party becomes involved in any capacity in any Action based in whole or in part on, or arising in whole or in part out of, any matter, including the transactions contemplated hereby, existing or occurring at or prior to the Closing Date, the D&O Indemnified Party may retain counsel reasonably satisfactory to them after consultation with Purchaser; provided, however, that Purchaser shall have the right to assume the defense thereof with counsel reasonably satisfactory to the D&O Indemnified Parties.

(b) Prior to the Merger Effective Date, Purchaser shall use its commercially reasonable efforts to purchase and obtain, as of the Closing Date a "tail" insurance policy, to the extent available on commercially reasonable terms and at an aggregate cost of no higher than 330% of the annual premium of Purchaser's directors' and officers' liabilities insurance policy as of the date of this Agreement, extending coverage for an aggregate period of six (6) years (or such other coverage period as mutually agreed by Purchaser and the Company) providing directors' and officers' liability insurance with respect to claims arising from facts or events that occurred on or before the Closing covering (as direct beneficiaries) those persons who are as of the date of this Agreement currently covered by the Purchaser's directors' and officers' liability insurance policy, of the type and with the amount of coverage no less favorable than those of the directors' and officers' liability insurance maintained as of the date of this Agreement by, or for the benefit of, the Purchaser; <u>provided</u>, <u>however</u>, that to the extent a policy as permitted by this <u>Section 5.16(c)</u> is purchased by Purchaser, the aggregate cost of such policy shall be deemed a Purchaser Transaction Expense but shall in no case be included in the calculation of Excess Purchaser Indebtedness and Liability Amount.

(c) Notwithstanding the foregoing (i) none of the Surviving Company or Purchaser shall be obligated to indemnify a D&O Indemnified Party with respect to any amount in relation to a Claim of any type whatsoever to the extent such Claim (or part thereof) has been paid to the D&O Indemnified Party (or paid directly to a third party on a D&O Indemnified Party's behalf) by any directors and officers, or other type, of insurance maintained by any of the Surviving Company or Purchaser, and (ii) no D&O Indemnified Party shall settle any Claim without the prior written consent of the Surviving Company and Purchaser (which consents shall not be unreasonably withheld, conditioned or delayed), nor shall any of the Surviving Company or Purchaser: (A) settle any Claim without either (x) the written consent of all D&O Indemnified Parties against whom such Claim was made (which consents shall not be unreasonably withheld, conditioned or delayed), or (y) obtaining an unconditional general release from all liability arising out of the proceeding to which the Claim relates for all D&O Indemnified Parties without admission nor finding of wrongdoing as a condition of such settlement, or (B) be liable to a D&O Indemnified Party for any amounts paid in settlement of any threatened or pending Claim effected without its prior written consent (which consents shall not be unreasonably withheld, conditioned or delayed).

(d) On or prior to the Closing Date, Purchaser shall enter into customary indemnification agreements reasonably satisfactory to each of the Company and Purchaser with, or for the benefit of, the D&O Indemnified Parties, which indemnification agreements shall continue to be effective following the Closing Date. To the extent applicable, on or prior to the Closing Date, Purchaser shall countersign such indemnification agreements with respect to any D&O Indemnified Party that was a director or officer of Purchaser prior to the Merger for the purposes of acknowledging the termination of any applicable indemnification agreements between such D&O Indemnified Party and Purchaser.

(e) The provisions of this <u>Section 5.16</u> shall survive the Closing and are intended to be for the benefit of, and shall be enforceable by, each of the D&O Indemnified Parties and their respective heirs and representatives.

# 5.17 Section 16 Matters.

(a) Prior to the Closing Date, the Parties shall take all such steps (to the extent permitted under applicable Law) as are reasonably necessary to cause any acquisition or disposition of Purchaser Ordinary Shares or any derivative thereof that occurs or is deemed to occur by reason of or pursuant to the Transactions by each Person who is or will be or may become subject to Section 16 of the Exchange Act with respect to Purchaser, including by virtue of being deemed a director by deputization, to be exempt under Rule 16b-3 promulgated under the Exchange Act.

5.18 <u>Trust Account Proceeds</u>. The Parties agree that after the Closing, the funds in the Trust Account, after taking into account payments for the Redemption shall first be used to pay (i) the Purchaser Transaction Expenses, (ii) any loans owed by Purchaser to Sponsor for the payment of Purchaser Transaction Expenses and other administrative costs and expenses incurred by or on behalf of Purchaser prior to the Closing Date and (iii) the Company Transaction Expenses. Such amounts, as well as any expenses that are required or permitted to be paid by delivery of Purchaser Securities, will be paid at the Closing; <u>provided</u>, <u>however</u>, that in the case of clauses (i) and (ii) above, such funding shall be subject to the limitations set forth in <u>Section 5.22</u>. Any remaining cash will be transferred to a Target Company or Purchaser and used for working capital and general corporate purposes.

#### 5.19 Tax Matters.

(a) Each Party (i) shall use its respective commercially reasonable efforts to cause the Merger to qualify, and agree not to, and not to permit or cause any of their Affiliates or Subsidiaries to, take any action which to its Knowledge could reasonably be expected to prevent or impede the Transactions from qualifying, for the Intended Tax Treatment, (ii) shall report the Merger consistently with the Intended Tax Treatment unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code and (iii) acknowledges and agrees that it is not a condition to the Closing that the Merger qualifies as a "reorganization" within the meaning of Section 368(a).

(b) In the event the SEC requests or requires tax opinions, each Party shall use best efforts to execute and deliver customary tax representation letters (not to be inconsistent with this Agreement or the other Ancillary Documents) as the applicable tax advisor may reasonably request in form and substance reasonably satisfactory to such advisor.

(c) Each of the Parties shall (and shall cause their respective Affiliates to) cooperate fully, as and to the extent reasonably requested by another Party, in connection with the filing of relevant Tax Returns, any claim for a refund of any Tax, and any audit or tax proceeding. Such cooperation shall include the retention and (upon the other Party's request) the provision (with the right to make copies) of records and information reasonably relevant to any tax proceeding or audit, making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder.

5.20 280G. To the extent that any "disqualified individual" (within the meaning of Section 280G(c) of the Code and the regulations thereunder) has the right to receive any payments or benefits that are contingent (within the meaning of Section 280G of the Code) on the transactions contemplated by this Agreement and that could be deemed to constitute "parachute payments" (within the meaning of Section 280G(b)(2)(A) of the Code and the regulations thereunder) (the "Section 280G Payments"), then, prior to the Closing the Company shall, or shall cause the applicable Affiliate to, use commercially reasonable best efforts to (i) solicit and obtain from each individual who is, as of the Closing Date, a "disqualified individual" a waiver of such disqualified individual's rights, subject to the approval described in clause (ii) below, to some or all of such payments or benefits (the "Waived 280G Benefits") so that any remaining payments or benefits shall not be deemed to be "excess parachute payments" (within the meaning of Section 280G of the Code and the regulations thereunder), and (ii) with respect to each individual who executes the waiver described in clause (i), submit to a vote in the manner required under Section 280G(b)(5)(B) of the Code and the regulations promulgated thereunder, along with adequate disclosure intended to satisfy such requirements (including Q&A-7 of Treasury Regulations § 1.280G-1), the right of any such "disqualified individual" to receive the Waived 280G Benefits. In connection with the foregoing, Purchaser (a) will, no later than ten (10) Business Days prior to the Closing Date, provide the Company with any contract, agreement or plan entered into by the Purchaser (or at the direction of the Purchaser) and a disqualified individual (or reasonably be expected to be entered into at or prior to the Closing Date) the (the "Purchaser Arrangements") to determine whether any payments made or to be made, or benefits granted or to be granted, pursuant to such Purchaser Arrangements could reasonably be considered to be "parachute payments" within the meaning of Section 280G(b)(2) of the Code, and (b) shall cooperate with the Company in good faith to provide the Company with information necessary for the Company to make such determination and for the Company to prepare the required disclosure at least five (5) Business Days prior to the Company obtaining the waivers and soliciting the vote as set forth in this Section 5.20. At least one (1) Business Dav prior to the Closing Date, the Company shall deliver to Purchaser documents evidencing the results of such vote. If any of the Waived 280G Benefits fail to be approved as contemplated above, such Waived 280G Benefits shall not be made or provided. At least three (3) Business Days prior to soliciting such waivers and soliciting such shareholder approval as contemplated in this Section 5.20, the Company shall provide to Purchaser drafts of such waivers and such shareholder approval materials (including the calculations and analysis supporting such documentation)for Purchaser's review and reasonable opportunity to comment, and the Company shall in good faith consider any such comments. Nothing contained in this Section 5.20 shall be deemed to require (i) the Company to obtain a waiver from any "disqualified individual" or (ii) any specific outcome of the vote described in this Section 5.20.

5.21 <u>Financials</u>. As soon as reasonably practicable after the date of this Agreement, and in any case prior to the date of filing of the Registration Statement, except to the extent such failure is due to Purchaser's failure to comply with its obligations pursuant to <u>Section 5.9</u>, the Company shall deliver to Purchaser the PCAOB Audited Financial Statements and any other audited and unaudited consolidated balance sheets and the related audited or unaudited consolidated accounts of the Company that are required to be included in the Registration Statement. The Company and Purchaser shall each use its reasonable best efforts (a) to assist the other, upon advance written notice, during normal business hours and in a manner such as to not unreasonably interfere with the normal operation of any Target Company or Purchaser, in preparing in a timely manner any other financial information or statements (including customary pro forma financial statements) that are required to be included in the Registration Statement and any other filings to be made by Purchaser with the SEC in connection with the Transactions and (b) to obtain the consents of its auditors with respect thereto as may be required by applicable Law or requested by the SEC in connection therewith.

## 5.22 Purchaser Working Capital Funding Obligations; Purchaser Liability Limitations.

(a) During the Interim Period, Purchaser shall fund its ordinary working capital expenses from cash on hand, and if such amount is insufficient, shall cause Sponsor or its Affiliates (excluding the Purchaser or any of its Subsidiaries) to fund such working capital expenses in accordance with the terms of the Letter Agreement Amendment.

(b) As of the date of this Agreement through the expiration of the Interim Period, Purchaser and Merger Sub shall not incur an aggregate amount of Indebtedness, Purchaser Transaction Expenses (whether due before, on or after the Closing) (including deferred underwriting fees, but expressly excluding any D&O tail insurance policy costs pursuant to <u>Section 5.16(b)</u>) or other Liabilities in excess of \$4,500,000 (the "*Purchaser Debt and Liability Cap*"), without the prior written consent of the Company; <u>provided, however</u>, in the event the aggregate amount of such Indebtedness, Purchaser Transaction Expenses or other Liabilities incurred by Purchaser for purposes of funding working capital exceeds the Purchaser Debt and Liability Cap, Purchaser shall cause Sponsor or its Affiliates (other than the Purchaser or any of its Subsidiaries) to fund such excess amount in accordance with the terms of the Letter Agreement Amendment. If the Sponsor or its Affiliates (other than the Purchaser or any of its Subsidiaries) to elect by providing written notice to the Purchaser after the Purchaser Closing Statement is delivered to the Company and prior to Closing to treat such unfunded amount as the Excess Purchaser Indebtedness and Liability Amount.

#### 5.23 Amendment to Disclosure Schedules.

(a) Purchaser may from time to time prior to Closing, by written notice to the Company, supplement or amend the Purchaser Disclosure Schedules (including adding new Sections to the Purchaser Disclosure Schedule related to the provisions of this Agreement that do not contemplate qualification by the Purchaser Disclosure Schedule) with new or updated information with respect to matters that occur or arise after the Execution Date to correct any disclosed item or matter that would constitute a breach of a representation or warranty of Purchaser or Merger Sub in <u>Article II</u> or <u>Article III</u> as of the Closing Date (such new or updated information being "*New Purchaser Information*"); provided, however, New Purchaser Information shall not include information that is updated is in the Ordinary Course of Business of Purchaser during the Interim Period and does not violate the covenants in <u>Section 5.3(b)</u>. For purposes of determining whether the Company's conditions set forth in <u>Section 7.2</u> have been fulfilled, the Purchaser Disclosure Schedule shall be deemed to exclude the New Purchaser Information.

(b) The Company may from time to time prior to Closing, by written notice to the Purchaser, supplement or amend the Company Disclosure Schedules (including adding new Sections to the Company Disclosure Schedule related to the provisions of this Agreement that do not contemplate qualification by the Company Disclosure Schedule) with new or updated information with respect to matters that occur or arise after the Execution Date to correct any disclosed item or matter that would constitute a breach of a representation or warranty of the Company in <u>Article IV</u> as of the Closing Date (such new or updated information being "*New Company Information*"); <u>provided</u>, <u>however</u>, New Company Information shall not include information that is updated is in the Ordinary Course of Business of the Company during the Interim Period and does not violate the covenants in <u>Section 5.2(b)</u>. For purposes of determining whether the Company's conditions set forth in <u>Section 7.3</u> have been fulfilled, the Purchaser Disclosure Schedule shall be deemed to exclude the New Purchaser Information.

42

# ARTICLE VI <u>SURVIVAL</u>

6.1 Non-Survival of Representations, Warranties and Covenants. None of the representations, warranties, covenants, obligations or other agreements in this Agreement or in any certificate, statement or instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such representations, warranties, covenants, obligations or other agreements or other provisions, shall survive the Closing, and all of the representations, warranties, covenants, obligations or other agreements in this Agreement or in any certificate, statement or instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such representations, warranties, covenants, obligations, agreements or other provisions, shall terminate and expire upon the occurrence of the Closing (and there shall be no Liability after the Closing in respect thereof), in each case, except for (a) those covenants and agreements contained herein that by their terms expressly apply in whole or in part at or after the Closing and then only with respect to any breaches occurring at or after the Closing and (b) Article X.

#### ARTICLE VII CLOSING CONDITIONS

7.1 <u>Conditions to Each Party's Obligations</u>. The obligations of each Party to consummate the Closing shall be subject to the satisfaction or written waiver (where permissible) by the Company and Purchaser of the following conditions:

(a) *Required Purchaser Shareholder Approval*. The Required Purchaser Shareholder Approval Matters that are submitted to the vote of the shareholders of Purchaser at the Special Meeting in accordance with the Proxy Statement shall have been approved by the Required Purchaser Shareholder Approval.

(b) Required Company Shareholder Approval. The Required Company Shareholder Approval shall have been obtained.

(c) *Required Merger-Sub Shareholder Approval.* The Purchaser as sole shareholder of Merger Sub prior to the Merger Effective Date shall have approved the Merger and the adoption of the amended and restated memorandum and articles of the Surviving Company in the form of Exhibit G, which approval shall be required in accordance with <u>Section 5.11</u> if the Required Purchaser Shareholder Approval has been obtained as described in <u>Section 7.1(a)</u>.

(d) *Requisite Regulatory Approvals.* All Consents required to be obtained from or made with any Governmental Authority in order to consummate the Transactions set forth on <u>Schedule 7.1(c)</u> shall have been obtained or made.

(e) *No Law or Order*. No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any Law (whether temporary, preliminary or permanent) or Order that is then in effect and which has the effect of making the Closing illegal or which otherwise prevents or prohibits consummation of the transactions contemplated by this Agreement.

(f) *Net Tangible Assets*. Immediately prior to the Closing, after giving effect to the Redemption, Purchaser shall have net tangible assets of at least 5,000,001 (as determined in accordance with Rule 3a51-1(g)(1) of the Exchange Act).

(g) *Registration Statement*. The Registration Statement shall have been declared effective by the SEC and shall remain effective as of the Closing.

(h) *NASDAQ Listing*. The Purchaser's Ordinary Shares to be issued in connection with the Transactions shall have been approved for listing on the NASDAQ, subject to official notice of issuance.

(i) *Foreign Private Issuer*. The Company shall not have received evidence that Purchaser will not qualify as a "foreign private issuer" pursuant to Rule 3b-4 of the Exchange Act as of the Closing.

7.2 <u>Conditions to Obligations of the Company</u>. In addition to the conditions specified in <u>Section 7.1</u>, the obligation of the Company to consummate the Closing is subject to the satisfaction or written waiver (by the Company) of the following conditions:

(a) *Representations and Warranties*. All of the representations and warranties of Purchaser and Merger Sub set forth in this Agreement and in any certificate delivered by or on behalf of Purchaser pursuant hereto shall be true and correct on and as of the date of this Agreement and on and as of the Closing Date as if made on the Closing Date, except for (i) those representations and warranties that address matters only as of a particular date (which representations and warranties shall have been accurate as of such date), (ii) with respect to the Purchaser Fundamental Representations, any failures to be true and correct in all material respects, and (iii) for all other representations and warranties of Purchaser and Merger Sub, any failures to be true and correct that (without giving effect to any qualifications or limitations as to materiality or Material Adverse Effect), individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on, or with respect to, Purchaser or Merger Sub.

43

(b) Agreements and Covenants. Purchaser and Merger Sub shall have performed in all material respects all of their respective obligations and complied in all material respects with all of their respective agreements and covenants under this Agreement to be performed or complied with by such party on or prior to the Closing Date.

(c) *Minimum Cash*. The aggregate cash available to Purchaser at the Closing from the Trust Account after giving effect to the redemptions of any shares of Purchaser Class A Shares for holders that timely exercise and do not waive their redemptions rights in respect of the Transaction, but before giving effect to the consummation of the Closing and the payment of any outstanding Purchaser Transaction Expenses, Company Transaction Expenses and Indebtedness permitted under this Agreement, shall equal or exceed \$17,500,000.

(d) Certain Ancillary Documents. Each of the Purchaser Lock-Up Agreements shall be in full force and effect in accordance with the terms thereof as of the Closing Date.

(e) Closing Deliveries.

(i) *Officer Certificate.* Purchaser shall have delivered to the Company a certificate, dated the Closing Date, signed by an executive officer of Purchaser in such capacity, certifying as to the satisfaction of the conditions specified in <u>Sections 7.2(a)</u>, 7.2(b) and 7.2(c) with respect to Purchaser.

(ii) Secretary Certificate. Purchaser shall have delivered to the Company a certificate from its secretary or other executive officer certifying as to, and attaching, (A) copies of Purchaser's Organizational Documents as in effect as of the Closing Date (immediately prior to the Merger Effective Date), (B) the resolutions of Purchaser's board of directors authorizing

and approving the execution, delivery and performance of this Agreement and each of the Ancillary Documents to which it is a party or by which it is bound, and the consummation of the transactions contemplated hereby and thereby, (C) evidence that the Required Purchaser Shareholder Approval has been obtained and (D) the incumbency of officers authorized to execute this Agreement or any Ancillary Document to which Purchaser is a party or otherwise bound.

(iii) *Good Standing*. Purchaser shall have delivered to the Company a good standing certificate (or similar documents applicable for such jurisdictions) for Purchaser certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of Purchaser's jurisdiction of organization.

(iv) *Registration Rights Agreement*. The Company and Purchaser shall have received a duly executed copy of the Registration Rights Agreement, duly executed by Purchaser.

(v) *Purchaser Charter Amendment*. At or prior to Closing, Purchaser shall have amended and restated the memorandum and articles of association of Purchaser in substantially the form attached as <u>Exhibit H</u> hereto (the "*Amended Purchaser Charter*").

7.3 <u>Conditions to Obligations of Purchaser</u>. In addition to the conditions specified in Section 7.1. the obligations of Purchaser to consummate the Closing are subject to the satisfaction or written waiver (by Purchaser) of the following conditions:

(a) *Representations and Warranties*. All of the representations and warranties of the Company, Purchaser and Merger Sub set forth in this Agreement and in any certificate delivered by or on behalf of the Company, Purchaser or Merger Sub pursuant hereto shall be true and correct on and as of the date of this Agreement and on and as of the Closing Date as if made on the Closing Date shall be true and correct on and as of the Closing Date as if made on the Closing Date, in each case except for (i) those representations and warranties that address matters only as of a particular date (which representations and warranties shall have been accurate as of such date), (ii) with respect to the Company and Purchaser Fundamental Representations, any failures to be true and correct in all material respects and (iii) for all other representations or limitations as to materiality or Material Adverse Effect), individually or in the aggregate, have not had and would not reasonably be expected to have a Material Adverse Effect on, or with respect to, the Company, Purchaser, or Merger Sub, as applicable.

44

(b) Agreements and Covenants. The Company, Purchaser, and Merger Sub shall have performed in all material respects all of their respective obligations and complied in all material respects with all of their respective agreements and covenants under this Agreement to be performed or complied with by them on or prior to the Closing Date.

(c) *Waiver and Release under Series A SPA and Company Shareholders Agreement*. The Company shall have obtained and delivered to Purchaser a waiver and release as it relates to potential claims by any shareholder of the Company under (i) that certain Series A Preferred Share and Warrant Purchase Agreement dated April 22, 2021, by and among Saire Cayman, Zhencheng, Tyche, Fengyuan Capital, Fargo I, KeyMan, Tospring, EVERSUNNY, SUN Zeyu BVI and certain other parties and (ii) that certain Amended and Restated Shareholders Agreement of the Company dated August 31, 2021, including any predecessor agreement thereto, substantially in the form attached hereto as Exhibit I hereto (collectively, the "Series A SPA and Company Shareholders Agreement Waivers").

\$1,000,000.

(d) *Company Minimum Cash*. At Closing, the Company Closing Cash minus Company Indebtedness shall be at least 0.

(e) Certain Ancillary Documents. Each of the Company Lock-Up Agreements shall be in full force and effect in accordance with the terms thereof as of the Closing Date.

(f) Closing Deliveries.

(i) *Officer Certificate.* Purchaser shall have received a certificate from the Company, dated as the Closing Date, signed by an executive officer of the Company in such capacity, certifying as to the satisfaction of the conditions specified in <u>Sections 7.3(a).</u> 7.3(b) and 7.3(c) with respect to the Company. Purchaser shall have delivered to Purchaser a certificate, dated

the Closing Date, signed by an executive officer of Purchaser in such capacity, certifying as to the satisfaction of the conditions specified in Sections 7.3(a). 7.3(b) and 7.3(c) with respect to Purchaser and Merger Sub, as applicable.

(ii) Secretary Certificates. The Company and Purchaser shall each have delivered to Purchaser a certificate from its secretary or other executive officer certifying as to the validity and effectiveness of, and attaching, (A) copies of its Organizational Documents as in effect as of the Closing Date (immediately prior to the Effective Time), (B) the resolutions of its board of directors and shareholders, as applicable, authorizing and approving the execution, delivery and performance of this Agreement and each Ancillary Document to which it is a party or bound, and the consummation of the Transactions, and (C) the incumbency of its officers authorized to execute this Agreement or any Ancillary Document to which it is a party or otherwise bound.

(iii) *Good Standing*. The Company shall have delivered to Purchaser good standing certificates (or similar documents applicable for such jurisdictions) for each Target Company certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of the Target Company's jurisdiction of organization, to the extent that good standing certificates or similar documents are generally available in such jurisdictions) for each of Purchaser and Merger Sub certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of the Closing Date from the proper Governmental Authority of purchaser good standing certificates (or similar documents applicable for such jurisdictions) for each of Purchaser and Merger Sub certified as of a date no earlier than thirty (30) days prior to the Closing Date from the proper Governmental Authority of Purchaser's and Merger Sub' jurisdiction of organization, in each case to the extent that good standing certificates or similar documents are generally available in such jurisdiction.

(iv) *Registration Rights Agreement*. Purchaser shall have received a copy of the Registration Rights Agreement, duly executed by Purchaser and each Company shareholder party thereto.

(g) *Purchaser Charter Amendment*. At or prior to the Closing, the board of directors of the Purchaser and its requisite shareholders have adopted and approved, and the Purchaser has filed and made effective the Amended Purchaser Charter.

7.4 <u>Frustration of Conditions</u>. Notwithstanding anything contained herein to the contrary, no Party may rely on the failure of any condition set forth in this Article VII to be satisfied if such failure was caused by the failure of such Party or its Affiliates (or with respect to the Company, any Target Company, Purchaser or Merger Sub) to comply with or perform any of its covenants or obligations set forth in this Agreement in all material respects.

45

## ARTICLE VIII TERMINATION AND EXPENSES

8.1 <u>Termination</u>. This Agreement may be terminated and the transactions contemplated hereby may be abandoned at any time prior to the Closing as follows:

(a) by mutual written consent of Purchaser and the Company;

(b) by written notice by Purchaser or the Company if any of the conditions to the Closing set forth in <u>Article VII</u> have not been satisfied or waived by March 31, 2022 (the "*Outside Date*"), <u>provided</u>, that the right to terminate this Agreement under this <u>Section 8.1(b)</u> shall not be available to a Party if the breach or violation by such Party or its Affiliates (or with respect to the Purchaser or Merger Sub) of any representation, warranty, covenant or obligation under this Agreement was the proximate cause of, or proximately resulted in, the failure of the Closing to occur on or before the Outside Date;

(c) by written notice by either Purchaser or the Company if a Governmental Authority of competent jurisdiction shall have issued an Order or taken any other action permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement, and such Order or other action has become final and non-appealable; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this <u>Section 8.1(c)</u> shall not be available to a Party if the failure by such Party or its Affiliates (or with respect to the Purchaser or Merger Sub) to comply with any provision of this Agreement has been a substantial cause of, or substantially resulted in, such action by such Governmental Authority;

(d) by written notice by the Company to Purchaser, if (i) there has been a breach by Purchaser of any of its representations, warranties, covenants or agreements contained in this Agreement, or if any representation or warranty of Purchaser shall have become untrue or materially inaccurate, in any case, which would result in a failure of a condition set forth in Section 7.2(a) or Section 7.2(b) to be satisfied, and (ii) the breach or inaccuracy is incapable of being cured or is not cured before the earlier of (A) end of the twentieth day after written notice of such breach or inaccuracy is provided to Purchaser by the Company or (B) the Outside Date; provided, that the Company shall not have the right to terminate this Agreement pursuant to this Section 8.1(d) if at such time the Company is in uncured breach of this Agreement which would result in a failure of any condition set forth in Section 7.3(a) or Section 7.3(b) from being satisfied;

(e) by written notice by Purchaser to the Company, if (i) there has been a breach by the Company, Purchaser, or Merger Sub of any of their respective representations, warranties, covenants or agreements contained in this Agreement, or if any representation or warranty of such Parties shall have become untrue or inaccurate, in any case, which would result in a failure of a condition set forth in Section 7.3(a) or Section 7.3(b) to be satisfied, and (ii) the breach or inaccuracy is incapable of being cured or is not cured before the earlier of (A) end of the twentieth (20th) day after written notice of such breach or inaccuracy is provided to the Company by Purchaser or (B) the Outside Date; provided, that Purchaser shall not have the right to terminate this Agreement pursuant to this Section 8.1(e) if at such time Purchaser or Merger Sub is in uncured breach of this Agreement which would result in a failure of any condition set forth in Section 7.2(a) or Section 7.2(b) from being satisfied;

(f) by written notice by either Purchaser or the Company to the other if the Special Meeting is held (including any adjournment or postponement thereof) and has concluded, Purchaser's shareholders have duly voted, and the Required Purchaser Shareholder Approval was not obtained;

(g) by written notice by Purchaser to the Company if the PCAOB Audited Financial Statements have not been delivered to Purchaser on or prior to October 15, 2021; <u>provided</u>, that Purchaser shall not have the right to terminate this Agreement pursuant to this <u>Section 8.1(g)</u> if at such time Purchaser is in uncured breach of its obligations pursuant to <u>Section 5.9</u> of this Agreement and such uncured breach would result in a failure of the Company's ability to deliver the PCAOB Audited Financial Statements; or

(h) by written notice by the Company to Purchaser, if Purchaser's board of directors has made a Change in Recommendation.

8.2 Effect of Termination. This Agreement may only be terminated in the circumstances described in Section 8.1 and pursuant to a written notice delivered by the applicable Party to the other applicable Parties, which sets forth the basis for such termination, including the provision of Section 8.1 under which such termination is made. In the event of the valid termination of this Agreement pursuant to Section 8.1, this Agreement shall forthwith become void, and there shall be no Liability on the part of any Party, any of their respective Affiliates or any of their and their Affiliates' respective Representatives, and all rights and obligations of each Party shall cease, except: (i) Sections 5.11(a), 5.12, 8.3, 9.1, Article X and this Section 8.2 shall survive the termination of this Agreement, (ii) nothing herein shall relieve any Party from Liability for any Fraud against such Party prior to termination of this Agreement, and (iii) nothing herein shall relieve the Company, Purchaser or Merger Sub from Liability for willful breach (in each case of clauses (i), (ii) and (iii) above, subject to Section 9.1). Without limiting the foregoing, and except as provided in Sections 8.3 and this Section 8.2 (but subject to Section 9.1) and subject to the right to seek injunctions, specific performance or other equitable relief in accordance with Section 10.6), the Parties' sole right prior to the Closing with respect to any breach of any representation, warranty, covenant or other agreement contained in this Agreement by another Party or with respect to the transactions contemplated by this Agreement shall be the right, if applicable, to terminate this Agreement pursuant to Section 8.1.

46

8.3 Fees and Expenses. Subject to Section 9.1 unless otherwise provided for in this Agreement, all fees and expenses incurred in connection with this Agreement, the Ancillary Documents and the transactions contemplated hereby and thereby, including the fees and disbursements of counsel, financial advisors and accountants, shall be paid by the Party incurring such fees or expenses; provided that, for the avoidance of doubt, (a) if this Agreement is terminated in accordance with its terms, the Company shall pay, or cause to be paid, all unpaid Company Transaction Expenses and Purchaser shall pay, or cause to be paid, all unpaid Purchaser Transaction Expenses and all unpaid Purchaser Transaction Expenses, subject in the case of Purchaser Transaction Expenses, to the limitations set forth in Section 5.22.

# ARTICLE IX WAIVERS AND RELEASES

9.1 Waiver of Claims Against Trust. Reference is made to the IPO Prospectus. The Company and Merger Sub hereby represents and warrants that it understands that Purchaser has established the Trust Account containing the proceeds of the IPO and the overallotment shares acquired by Purchaser's underwriters and from certain private placements occurring simultaneously with the IPO (including interest accrued from time to time thereon) for the benefit of Purchaser's public shareholders (including overallotment shares acquired by Purchaser's underwriters) (the "Public Shareholders") and that, except as otherwise described in the IPO Prospectus, Purchaser may disburse monies from the Trust Account only: (a) to the Public Shareholders in the event they elect to redeem their shares of Purchaser Class A Shares in connection with the consummation of its initial business combination (as such term is used in the IPO Prospectus) ("Business Combination") or in connection with a shareholder vote to amend Purchaser's Organizational Documents to modify the substance or timing of Purchaser's obligation to provide holders of Purchaser Class A Shares the right to have their shares redeemed in connection with a Business Combination or to redeem 100% of Purchaser Class A Shares if Purchaser does not complete a Business Combination within 18 months from the closing of the IPO or with respect to any other provision relating to the rights of holders of Purchaser Class A Shares, (b) to the Public Shareholders if Purchaser fails to consummate a Business Combination within 18 months after the closing of the IPO (subject to further extension by amendment to Purchaser's Organizational Documents), (c) with respect to any interest earned on the amounts held in the Trust Account, up to \$50,000 of interest to pay dissolution expenses and any amounts necessary to pay for any income taxes, and (d) to Purchaser after the consummation of a Business Combination, in each case, subject to the Trust Agreement. For and in consideration of Purchaser entering into this Agreement and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Company hereby agrees on behalf of itself and its Affiliates that, notwithstanding anything to the contrary in this Agreement, neither of the Company nor any of its Affiliates do now or shall at any time hereafter have any right, title, interest or claim of any kind in or to any monies in the Trust Account, or make any claim against the Trust Account, regardless of whether such claim arises as a result of, in connection with or relating in any way to, this Agreement or any proposed or actual business relationship between Purchaser or any of its Representatives, on the one hand, and the Company or any of its Representatives, on the other hand, or any other matter, and regardless of whether such claim arises based on contract, tort, equity or any other theory of legal liability (collectively, the "Released Claims"). The Company, on behalf of itself and its Affiliates hereby irrevocably waives any Released Claims that any such Party or any of its Affiliates may have against the Trust Account now or in the future as a result of, or arising out of, any negotiations, contracts or agreements with Purchaser or its Representatives and will not seek recourse against the Trust Account for any reason whatsoever (including for an alleged breach of this Agreement or any other agreement with Purchaser or its Affiliates). The Company agrees and acknowledges that such irrevocable waiver is material to this Agreement and specifically relied upon by Purchaser and its Affiliates to induce Purchaser to enter in this Agreement, and the Company further intends and understands such waiver to be valid, binding and enforceable against such Party and each of its Affiliates under applicable Law. Notwithstanding anything herein to the contrary, (A) the Company and its Affiliates may commence any action or proceeding based upon, in connection with, relating to or arising out of any matter relating to Purchaser, Merger Sub or their respective Representatives, which proceeding seeks, in whole or in part, monetary relief against Purchaser, Merger Sub or their respective Representatives, against assets or funds held outside of the Trust Account (including any funds released from the Trust Account and assets that are acquired with such funds); provided that such claim shall not permit such Party or any of its Affiliates (or any Person claiming on any of their behaves or in lieu of them) to have any claim against the Trust Account or any amounts contained therein, and (B) nothing herein shall limit or prohibit the Company or any of its Affiliates from pursuing a claim against Purchaser or Merger Sub for specific performance or other equitable relief. This Section 9.1 shall survive termination of this Agreement for any reason.

# ARTICLE X MISCELLANEOUS

10.1 <u>Notices</u>. All notices, consents, waivers and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered (i) in person, (ii) by facsimile or other electronic means, with confirmation of receipt, (iii) one Business Day after being sent, if sent by reputable, nationally recognized overnight courier service or (iv) three (3) Business Days after being mailed, if sent by registered or certified mail, pre-paid and return receipt requested, in each case to the applicable Party at the following addresses (or at such other address for a Party as shall be specified by like notice):

If to Purchaser or Merger Sub at or prior to the Closing, to:

with a copy (which will not constitute notice) to:

Jianwei Li

David C. Buck

Chairman & Co-Chief Executive Officer 437 Madison Avenue, 27th Floor New York, New York 10022 (732) 910-9692 Jianwei@zhenchengcap.com

If to the Company at or prior to the Closing, to:

Ian Zou CFO, Vice President SAITECH Limited c/o Sustainable Available Innovative Pte. Ltd. 20 EMERALD HILL ROAD, SINGAPORE, 229302 +86 186 1174 3705 zoujian@sai.tech

If to Purchaser or the Company after the Closing, to:

Ian Zou CFO, Vice President SAITECH Limited c/o Sustainable Available Innovative Pte. Ltd. 20 EMERALD HILL ROAD, SINGAPORE, 229302 +86 186 1174 3705 zoujian@sai.tech Kayleigh McNelis Sidley Austin LLP 1000 Louisiana, Suite 5900 Houston, Texas 77002 Tel: (713) 494-4500 dbuck@sidley.com kmcnelis@sidley.com

#### and

Suzanne Correy Maples and Calder (Cayman) LLP PO Box 309 Ugland House, Grand Cayman, KY1-1104 Cayman Islands. Tel: +1 345 814 5434 suzanne.correy@maples.com

with a copy (which will not constitute notice) to:

Chris A. Ferazzi Michael J. Blankenship Winston & Strawn LLP 800 Capitol Street, Suite 2400 Houston, Texas 77002 Tel: (713) 651-2600 CFerazzi@winston.com MBlankenship@winston.com

and

George Weston Harney Westwood & Riegels LP, Craigmuir Chambers, PO Box 71, Road Town, Tortola VG1110, British Virgin Islands. Tel: +1 284 852 4333| george.weston@harneys.com

with a copy (which will not constitute notice) to:

Chris A. Ferazzi Michael J. Blankenship Winston & Strawn LLP 800 Capitol Street, Suite 2400 Houston, Texas 77002 Tel: (713) 651-2600 CFerazzi@winston.com MBlankenship@winston.com

and

George Weston Harney Westwood & Riegels LP, Craigmuir Chambers, PO Box 71, Road Town, Tortola VG1110, British Virgin Islands. Tel: +1 284 852 4333| george.weston@harneys.com 10.2 <u>Binding Effect: Assignment</u>. Subject to <u>Section 10.3</u>, this Agreement and all of the provisions hereof shall be binding upon and inure solely to the benefit of the Parties hereto and their respective successors and permitted assigns. This Agreement shall not be assigned by any Party, by operation of Law or otherwise, without the prior written consent of Purchaser and the Company, and any assignment without such consent shall be null and void; <u>provided</u> that no such assignment shall relieve the assigning Party of its obligations hereunder.

10.3 <u>Third Parties</u>. Except for the rights of the D&O Indemnified Parties set forth in <u>Section 5.15</u>, of Sidley Austin LLP ("*Sidley*") set forth in <u>Section 10.14(a)</u> and Winston & Strawn LLP ("*Winston*") set forth in <u>Section 10.14(b)</u>, which the Parties acknowledge and agree are express third party beneficiaries of this Agreement, nothing contained in this Agreement or in any instrument or document executed by any party in connection with the transactions contemplated hereby shall create any rights in, or be deemed to have been executed for the benefit of, any Person that is not a Party hereto or thereto or a successor or permitted assign of such a Party. For the avoidance of doubt, no shareholder of Purchaser after the Closing (in his, her or its capacity as such) is a third party beneficiary of this Agreement or shall have any rights hereunder.

10.4 Governing Law; Jurisdiction. This Agreement and all Actions (whether in contract, tort or otherwise) that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance hereof (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) shall be governed by, construed and enforced in accordance with the Laws (both substantive and procedural) of the State of Delaware, without regard to the conflict of laws principles thereof, except that the Merger, the internal affairs of Purchaser and any provisions of this Agreement that are expressly or otherwise required to be governed by the Cayman Act, shall be governed by the Laws of the Cayman Islands (without giving effect to choice of law principles thereof) in respect of which the Parties irrevocably submit to the non-exclusive jurisdiction of the Courts of the Cayman Islands. Subject to the immediately preceding sentence, all Actions arising out of or relating to this Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware, or to the extent such Court does not have subject matter jurisdiction, any federal court within the State of Delaware (and any courts having jurisdiction over appeals therefrom) (the "Specified Courts"). Each Party hereto hereby (a) submits to the exclusive personal and subject matter jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Agreement brought by any Party hereto and (b) irrevocably waives, and agrees not to assert by way of motion, defense or otherwise, in any such Action, any claim that it is not subject to the personal or subject matter jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any Specified Court. Each Party agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each Party irrevocably consents to the service of the summons and complaint and any other process in any other Action relating to the transactions contemplated by this Agreement, on behalf of itself, or its property, by personal delivery of copies of such process to such Party at the applicable address set forth in Section 10.1. Nothing in this Section 10.4 shall affect the right of any Party to serve legal process in any other manner permitted by Law.

10.5 <u>WAIVER OF JURY TRIAL</u>. EACH OF THE PARTIES HERETO HEREBY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY WITH RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY, IN EACH CASE, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF ANY ACTION, SEEK TO ENFORCE THAT FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 10.5</u>.

10.6 Specific Performance. Each Party acknowledges that the rights of each Party to consummate the transactions contemplated hereby are unique, recognizes and affirms that in the event of a breach of this Agreement by any Party, money damages may be inadequate and the non-breaching Parties may have not adequate remedy at law, and agree that irreparable damage may occur in the event that any of the provisions of this Agreement were not performed by an applicable Party in accordance with their specific terms or were otherwise breached. Accordingly, each Party shall be entitled to seek an injunction, restraining order or other equitable remedy to prevent or remedy any breach of this Agreement and to seek to enforce specifically the terms and provisions hereof, in each case, without the requirement to post any bond or other security or to prove that money damages would be inadequate, this being in addition to any other right or remedy to which such Party may be entitled under this Agreement, at law or in equity.

10.7 Severability. In case any provision in this Agreement shall be held invalid, illegal or unenforceable in a jurisdiction, such provision shall be modified or deleted, as to the jurisdiction involved, only to the extent necessary to render the same valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby nor shall the validity, legality or enforceability of such provision be affected thereby in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties will substitute for any invalid, illegal or unenforceable provision a suitable and equitable provision that carries out, so far as may be valid, legal and enforceable, the intent and purpose of such invalid, illegal or unenforceable provision.

49

10.8 <u>Amendment</u>. This Agreement may be amended, supplemented or modified only by execution of a written instrument signed by Purchaser and the Company.

10.9 Waiver. Each of Purchaser and the Company on behalf of itself and its Affiliates, may in its sole discretion (i) extend the time for the performance of any obligation or other act of any other non-Affiliated Party hereto, (ii) waive any inaccuracy in the representations and warranties by such other non-Affiliated Party contained herein or in any document delivered pursuant hereto and (iii) waive compliance by such other non-Affiliated Party with any covenant or condition contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the Party or Parties to be bound thereby. Notwithstanding the foregoing, no failure or delay by a Party in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise of any other right hereunder.

10.10 Entire Agreement. This Agreement and the documents or instruments referred to herein, including any exhibits, annexes and schedules attached hereto, which exhibits, annexes and schedules are incorporated herein by reference, together with the Ancillary Documents, embody the entire agreement and understanding of the Parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred to herein or the documents or instruments referred to herein, which collectively supersede all prior agreements and the understandings among the Parties with respect to the subject matter contained herein.

10.11 Interpretation. The table of contents and the Article and Section headings contained in this Agreement are solely for the purpose of reference, and shall not in any way affect the meaning or interpretation of this Agreement. In this Agreement, unless the context otherwise requires: (a) any pronoun used in this Agreement shall include the corresponding masculine, feminine or neuter forms, and words in the singular, including any defined terms, include the plural and vice versa; (b) reference to any Person includes such Person's successors and assigns and reference to a Person in a particular capacity excludes such Person in any other capacity; (c) any accounting term used and not otherwise defined in this Agreement or any Ancillary Document has the meaning assigned to such term in accordance with GAAP, based on the accounting principles used by the applicable Person; (d) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding or succeeding such term and shall be deemed in each case to be followed by the words "without limitation"; (e) the words "herein," "hereto," and "hereby" and other words of similar import in this Agreement shall be deemed in each case to refer to this Agreement as a whole and not to any particular Section or other subdivision of this Agreement; (f) the word "if' and other words of similar import when used herein shall be deemed in each case to be followed by the phrase "and only if"; (g) the term "or" means "and/or"; (h) the word "day" means calendar day unless Business Day is expressly specified; (i) any agreement, instrument, insurance policy, Law or Order defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument, insurance policy, Law or Order as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes, regulations, rules or orders) by succession of comparable successor statutes, regulations, rules or orders and references to all attachments thereto and instruments incorporated therein; (j) except as otherwise indicated, all references in this Agreement to the words "Section," "Article", "Schedule", "Annex" and "Exhibit" are intended to refer to Sections, Articles, Schedules, Annexes and Exhibits to this Agreement; and (k) the term "Dollars" or "\$" means United States dollars. Any reference in this Agreement to a Person's directors shall include any member of such Person's governing body and any reference in this Agreement to a Person's officers shall include any Person filling a substantially similar position for such Person. Any reference in this Agreement or any Ancillary Document to a Person's shareholders or stockholders shall include any applicable owners of the equity interests of such Person, in whatever form. The Parties have participated jointly in the negotiation and drafting of this Agreement. Consequently, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement. To the extent that any Contract, document, certificate or instrument is represented and warranted to by the Company to be given, delivered, provided or

made available by the Company, in order for such Contract, document, certificate or instrument to have been deemed to have been given, delivered, provided and made available to Purchaser or its Representatives, such Contract, document, certificate or instrument shall have been posted to the electronic data site maintained on behalf of the Company for the benefit of Purchaser and its Representatives and Purchaser and its Representatives have been given access to the electronic folders containing such information at least two (2) Business Days prior to the date of this Agreement.

10.12 <u>Counterparts</u>. This Agreement may be executed and delivered (including by facsimile, e-mail or other electronic transmission) in one or more counterparts, and by the different Parties hereto in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement.

50

10.13 <u>No Recourse</u>. Notwithstanding anything that may be expressed or implied in this Agreement, the Parties acknowledge and agree that no recourse under this Agreement or under any Ancillary Documents shall be had against any Person that is not a party to this Agreement or such Ancillary Document, including any past, present or future director, officer, agent, employee or other Representative of any past, present or future equity holder of any Party or of any Affiliate or successor or assignee thereof (collectively, the "*Non-Recourse Parties*"), as such, whether by the enforcement of any assessment or by any legal or equitable proceeding, or by virtue of any statute, regulation or other applicable Law; provided that, the Company can seek recourse against Sponsor if the Sponsor breaches its obligations under the Letter Agreement Amendment, solely to the extent of its payment obligations thereunder.

## 10.14 Legal Representation.

(a) The Parties agree that, notwithstanding the fact that Sidley may have, prior to Closing, jointly represented Purchaser and the Sponsor in connection with this Agreement, the Ancillary Documents and the Transactions, and has also represented Purchaser, Sponsor and/or their respective Affiliates in connection with matters other than the transaction that is the subject of this Agreement, Sidley will be permitted in the future, after the Closing, to represent the Sponsor or its Affiliates in connection with matters in which such Persons are adverse to Purchaser or any of its Affiliates, including any disputes arising out of, or related to, this Agreement. The Company and Purchaser hereby agree, in advance, to waive (and to cause their Affiliates to waive) any actual or potential conflict of interest that may hereafter arise in connection with Sidley's future representation of one or more of the Sponsor or its Affiliates, including any matters of such Person are adverse to the interests of Purchaser and/or the Company or any of their respective Affiliates, including any matters that arise out of this Agreement or that are substantially related to this Agreement or to any prior representation by Sidley of the Sponsor, Purchaser or any of their respective Affiliates. The Parties acknowledge and agree that, for the purposes of the attorney-client privilege, the Sponsor shall be deemed the client of Sidley with respect to the negotiation, execution and performance of this Agreement and the Ancillary Documents. All such communications shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to the Sponsor, shall be controlled by the Sponsor and shall not pass to or be claimed by Purchaser; provided, further, that nothing contained herein shall be deemed to be a waiver by Purchaser or any of its Affiliates of any applicable privileges or protections that can or may be asserted to prevent disclosure of any such communications to any third

(b) The Parties agree that, notwithstanding the fact that Winston may have, prior to the Closing, represented the Company in connection with this Agreement, the Ancillary Documents and the Transactions, and has also represented the Company and/ or its Affiliates in connection with matters other than the transaction that is the subject of this Agreement, Winston will be permitted in the future, after the Closing, to represent the shareholders or holders of other equity interests of the Company on or prior to the Closing or any of their respective directors, members, partners, officers, employees or Affiliates (other than Purchaser or the Surviving Company) (collectively, the "Company Shareholder Group") in connection with matters in which such Persons are adverse to Purchaser or the Surviving Company, including any disputes arising out of, or related to, this Agreement. The Company and Purchaser hereby agree, in advance, to waive (and to cause their Affiliates to waive) any actual or potential conflict of interest that may hereafter arise in connection with Winston's future representation of any member of the Company Shareholder Group in which the interests of such Person are adverse to the interests of Purchaser and/or the Company or any of their respective Affiliates, including any matters that arise out of this Agreement or that are substantially related to this Agreement or to any prior representation by Winston of any member of the Company Shareholder Group. The Parties acknowledge and agree that, for the purposes of the attorney-client privilege, the Company and the Company Shareholder Group shall be deemed the client of Winston with respect to the negotiation, execution and performance of this Agreement and the Ancillary Documents. All such communications shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to the Company Shareholder Group, shall be controlled by the Company Shareholder Group and shall not pass to or be claimed by Purchaser or the Surviving Company; provided, further, that nothing contained herein shall be deemed to be a waiver by Purchaser, the Surviving Company or any of their respective Affiliates of any applicable privileges or protections that can or may be asserted to prevent disclosure of any such communications to any third party.

#### ARTICLE XI DEFINITIONS

11.1 Certain Definitions. For purpose of this Agreement, the following capitalized terms have the following meanings:

*"Action"* means any charge, claim, demand, notice of noncompliance or violation, action, complaint, petition, investigation, appeal, suit, litigation, arbitration or other similar proceeding initiated or conducted by a mediator, arbitrator or Governmental Authority, whether administrative, civil, regulatory or criminal, and whether at law or in equity, or otherwise under any applicable Law.

"Affiliate" means, with respect to any Person, any other Person directly or indirectly Controlling, Controlled by, or under common Control with such Person. For the avoidance of doubt, Sponsor shall be deemed to be an Affiliate of Purchaser prior to the Closing.

*"Ancillary Documents"* means each agreement, instrument or document attached hereto as an Exhibit, including the Company Lock-Up Agreements, the Purchaser Lock-Up Agreements, the Company Shareholder Support Agreement, the Sponsor Support Agreement, the Registration Rights Agreement, the Employment Agreements and the Amended Purchaser Charter and the other agreements, certificates and instruments to be executed or delivered by any of the Parties hereto in connection with or pursuant to this Agreement.

*"Audited Financial Statements"* means the audited consolidated financial statements of the Target Companies (including, in each case, any related notes thereto), consisting of the audited consolidated balance sheets of the Target Companies as of the Balance Sheet Date and the related consolidated audited income statements, changes in shareholder equity and statements of cash flows for the year then ended.

"Balance Sheet Date" means December 31, 2020.

"Benefit Plans" of any Person means any deferred compensation, incentive compensation, equity purchase or other equitybased compensation plan, employment, severance or termination pay, holiday, vacation or other bonus plan or practice, hospitalization or other medical, life or other welfare benefit insurance, profit sharing, pension, or retirement plan and each other employee benefit plan, program, agreement or arrangement (whether written or unwritten), including each "employee benefit plan" as such term is defined under Section 3(3) of ERISA (whether or not subject to ERISA), maintained, sponsored or contributed to, or required to be contributed to, by a Person for the benefit of any current or former employee or individual service provider of such Person, or with respect to which such Person has any Liability.

*"Business Day"* means any day other than a Saturday, Sunday or a legal holiday on which commercial banking institutions in New York, New York, Hong Kong or the Cayman Islands are authorized to close for business.

*"Business Plan"* means the version of the business plan as posted as of the date of this Agreement to the electronic data site maintained on behalf of the Company for the benefit of the Purchaser and its Representatives.

"Cayman Act" means the Companies Act (as Revised) of the Cayman Islands.

"Cayman Registrar" means the Registrar of Companies of the Cayman Islands.

*"Closing Company Cash"* means, as of the Reference Time, the aggregate cash and cash equivalents of the Target Companies on hand or in bank accounts, including deposits in transit; provided, however, Closing Company Cash shall not include Digital Assets.

"Closing Date" means the date on which the Closing occurs.

"Code" means the U.S. Internal Revenue Code of 1986.

*"Company Confidential Information"* means all confidential or proprietary documents and information concerning the Target Companies, Purchaser, Merger Sub or any of their respective Representatives, furnished in connection with this Agreement or the transactions contemplated hereby; <u>provided</u>, <u>however</u>, that Company Confidential Information shall not include any information which, (i) at the time of disclosure by Purchaser or its Representatives, is generally available publicly and was not disclosed in breach of this Agreement or (ii) at the time of the disclosure by any Target Company, Purchaser, Merger Sub or their respective Representatives to Purchaser or its Representatives, without violation of Law or any confidentiality obligation by the Person receiving such Company Confidential Information.

"Company Financials" means the Audited Financial Statements and the Interim Financial Statements.

"Company Fundamental Representations" means the representations and warranties contained in <u>Sections 4.1</u> (Organization and Standing), <u>4.2</u> (Authorization; Binding Agreement), <u>4.3</u> (Capitalization), <u>4.4</u> (Subsidiaries) and <u>4.22</u> (Finders and Brokers).

*"Company Lock-Up Agreements"* means (a) the Lock-Up Agreement to be entered into at Closing by Energy Science Artist Holding Limited, LilOrange Holding Limited, Eternal Selead Technology Holding Limited, Haonan Technology Holding Limited, Shiningsunny Holding Limited and Make World Better Limited, the form of which shall be substantially in the form of <u>Exhibit A-2</u> hereto, and (b) the Lock-Up Agreements to be entered into at Closing by all other shareholders of the Company immediately prior to Closing, the form of which shall be substantially in the form of <u>Exhibit A-3</u> hereto.

"Company Ordinary Shares" means collectively the Class A Ordinary Shares and the Class B Ordinary Shares, each with a par value of \$0.0001 per share.

"Company Shares" means collectively, the Company Ordinary Shares and the Company Preferred Shares.

*"Company Transaction Expenses"* means all fees and expenses of any of the Target Companies, Purchaser or Merger Sub (and not otherwise expressly allocated to Purchaser pursuant to the terms of this Agreement or any Ancillary Document) incurred or payable as of the Closing Date in connection with the negotiation, preparation or execution of this Agreement or any Ancillary Document, the performance of their covenants or agreements in this Agreement or any Ancillary Document or the consummation of the transactions contemplated hereby or thereby, including (a) the fees and expenses of outside legal counsel, accountants, advisors, brokers, investment bankers, consultants, or other agents or service providers of the Target Companies, Purchaser or Merger Sub, (b) all bonuses, change in control payments, retention payments, severance payments or similar payments payable in connection with the execution of this Agreement or the consummation of the transactions contemplated hereby, and the amount of the employer portion of any employment Taxes payable with respect to such payments, and (c) any other fees, expenses, commissions or other amounts that are expressly allocated to any Target Company, Purchaser or Merger Sub pursuant to this Agreement or any Ancillary Document.

"Consent" means any consent, approval, waiver, authorization or Permit of, or notice to or declaration or filing with any Governmental Authority or any other Person.

"Contracts" means all binding contracts, agreements, arrangements, bonds, notes, indentures, mortgages, debt instruments, purchase order, licenses (and all other binding contracts, agreements or binding arrangements concerning Intellectual Property), franchises, leases and other instruments or obligations of any kind, written or oral (including any amendments and other modifications thereto).

"Control" of a Person means (a) the ownership of, or ability to direct the casting of, more than fifty percent (50%) of the total voting rights conferred by all the share then in issue and conferring the rights to vote at all general meetings of such Person or (b) the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract, or otherwise. "Controlled", "Controlling" and "under common Control with" have correlative meanings.

*"Copyrights"* means any works of authorship, mask works and all copyrights therein, including all renewals and extensions, copyright registrations and applications for registration and renewal, and non-registered copyrights.

*"COVID-19 Measures"* shall mean any quarantine, "shelter in place," "stay at home," workforce reduction, social distancing, shut down, closure, sequester, safety or similar Laws, guidelines or recommendations promulgated by any applicable Governmental Authority, including the Centers for Disease Control and Prevention and the World Health Organization, in each case, in connection with or in response to COVID-19 or any other epidemics, pandemics or disease outbreaks, including the CARES Act and Families First Act, for similarly situated companies.

"Digital Asset Market Value" means as of the date of the Company Closing Statement, the market value equivalent of Digital Assets set forth on the Company Closing Statement calculated based on the mechanism set forth on <u>Schedule 11.1-B</u> of the Company Disclosure Schedules.

"ERISA" means the U.S. Employee Retirement Income Security Act of 1974.

epidemics, pandemic or disease outbreaks.

*"ERISA Affiliate"* means any Person that, together with any other Person, would be, at any relevant time, deemed a "single employer" within the meaning of Section 4001(b)(1) of ERISA or Section 414 of the Code.

*"ES Legislation"* means The International Tax Co-operation (Economic Substance) Act (As Revised) of the Cayman Islands and all regulations and guidance notes issued thereunder.

*"Excess Purchaser Indebtedness and Liability Amount"* means the amount, if any, by which the aggregate outstanding Purchaser Indebtedness, unpaid Purchaser Transaction Expenses and other unpaid Liabilities immediately prior to the Closing exceeds \$4,500,000.

"Exchange Act" means the U.S. Securities Exchange Act of 1934.

*"Exchange Ratio"* means a number equal to the Merger Consideration divided by \$10 and divided by the number of Company Ordinary Shares issued and outstanding as of immediately prior to the Merger Effective Date.

*"Foreign Plan"* means any plan, fund (including any superannuation fund) or other similar program or arrangement established or maintained outside the United States by any Target Company primarily for the benefit of employees of any Target Company residing outside the United States, which plan, fund or other similar program or arrangement provides, or results in, retirement income, a deferral of income in contemplation of retirement or payments to be made upon termination of employment, other than (i) any of the foregoing maintained or to which contributions are required by a Governmental Authority and (ii) any Benefit Plan.

"Founder" means Energy Science Artist Holding Limited.

*"Fraud"* means, with respect to a Party, actual fraud by such Party (as determined pursuant to a final, non-appealable order of a court of competent jurisdiction) with regard to the representations and warranties (as modified by the Disclosure Schedules), agreements and covenants made by such Party in this Agreement or schedules and certificates delivered by such Party pursuant to this Agreement, which involves a knowing and intentional misrepresentation by such Party with respect to such representations, warranties, agreements or covenants or a knowing and intentional concealment of facts with respect to such representations, warranties, agreements, with the intent of inducing any other Party to act or refrain from acting in reliance upon it including entering into this Agreement.

"GAAP" means generally accepted accounting principles as in effect in the United States of America.

*"Governmental Authority"* means any federal, state, local, foreign or other governmental, quasi-governmental or administrative body, instrumentality, department, division, commission or agency or any court, tribunal, administrative hearing body, arbitration panel, commission, or other similar dispute resolving panel or body.

"Indebtedness" of any Person means, without duplication, (a) all indebtedness of such Person for borrowed money (including the outstanding principal and accrued but unpaid interest), (b) all obligations for the deferred purchase price of property or services (other than those incurred in the ordinary course of business), (c) any other indebtedness of such Person that is evidenced by a note, bond, debenture, credit agreement or similar instrument, (d) all obligations of such Person under leases that should be classified as capital leases in accordance with GAAP (as applicable to such Person), (e) all obligations of such Person for the reimbursement of any obligor on any line or letter of credit, banker's acceptance, guarantee or similar credit transaction, in each case, that has been drawn or claimed against and not settled, (f) all interest rate and currency swaps, caps, collars and similar agreements or hedging devices under which payments are obligated to be made by such Person, whether periodically or upon the happening of a contingency, (g) any premiums, prepayment fees or other penalties, fees, costs or expenses associated with payment of any Indebtedness of such Person, and (h) all obligations described in clauses (a) through (g) above of any other Person which is directly or indirectly guaranteed by such Person or which such Person has agreed (contingently or otherwise) to purchase or otherwise acquire or in respect of which it has otherwise assured a creditor against loss.

*"Intellectual Property"* means all of the following as they exist in any jurisdiction throughout the world: Patents, Trademarks, Copyrights, Trade Secrets, Internet Assets, intellectual property rights in Software, and other intellectual property.

"Interim Financial Statements" means the unaudited consolidated financial statements of the Target Companies consisting of the consolidated balance sheets of the Target Companies as of the Interim Balance Sheet Date and the related consolidated income statements for the six months then ended.

"Interim Balance Sheet Date" means June 30, 2021.

"Internet Assets" means any domain name registrations.

"Investment Company Act" means the U.S. Investment Company Act of 1940.

"IPO" means the initial public offering of Purchaser Units pursuant to the IPO Prospectus.

*"IPO Prospectus"* means the final prospectus of Purchaser, dated as of April 28, 2021, and filed with the SEC on April 30, 2021.

"JOBS Act" means the Jumpstart Our Business Startups Act of 2012.

*Knowledge*" means, with respect to (i) the Company, the actual knowledge of the individuals set forth on <u>Schedule 11.1-A</u> of the Company Disclosure Schedules or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the applicable subject matter, (ii) Purchaser, the actual knowledge of the individuals set forth on <u>Schedule 11.1-A</u> of the Purchaser Disclosure Schedules or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the applicable for the applicable subject matter or (iii) any other Party, (A) if an entity, the actual knowledge of its directors and executive officers or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the applicable subject matter or (B) if a natural person, the actual knowledge of such Party or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the applicable subject matter or (B) if a natural person, the actual knowledge of such Party or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the applicable subject matter or (B) if a natural person, the actual knowledge of such Party or the knowledge that any of them would have actually had following a reasonable inquiry with his or her direct reports directly responsible for the avoidance of doubt, other than such reasonable inquiry with direct reports, no such individual will be under any express or implied duty to investigate.

55

*"Law"* means any federal, state, local, municipal, foreign or other law, statute, legislation, principle of common law, ordinance, code, edict, decree, proclamation, treaty, convention, rule, regulation, directive, requirement, writ, injunction, settlement, Order or Consent that is or has been issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Governmental Authority.

"Liabilities" means any and all liabilities, Indebtedness, or obligations of any nature (whether absolute, accrued, contingent or otherwise, whether known or unknown, whether direct or indirect, whether matured or unmatured, whether due or to become due and whether or not required to be recorded or reflected on a balance sheet under GAAP or other applicable accounting standards), including Tax liabilities due or to become due.

*"Lien"* means any mortgage, pledge, security interest, attachment, right of first refusal, option, proxy, voting trust, encumbrance, lien or charge of any kind (including any conditional sale or other title retention agreement or lease in the nature thereof), restriction (on voting, sale, transfer or disposition), any subordination arrangement in favor of another Person, or any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code or any similar Law.

"Material Adverse Effect" means, with respect to any specified Person, any fact, event, occurrence, change or effect after the date of this Agreement that has had, or would reasonably be expected to have, individually or in the aggregate, a material adverse effect upon (a) the business, assets, Liabilities, results of operations or financial condition of such Person and its Subsidiaries, taken as a whole or (b) the ability of such Person and its Affiliates to consummate the transactions contemplated by this Agreement; provided, however, that any fact, event, occurrence, change or effect directly or indirectly attributable to, resulting from, relating to or arising out of the following (by themselves or when aggregated with any other facts, events, occurrences, changes or effects) shall not be taken into account when determining whether a Material Adverse Effect pursuant to clause (a) above has occurred: (i) general changes in the financial or securities markets or general economic or political conditions in the country or region in which such Person or any of its Subsidiaries do business; (ii) changes, conditions or effects that generally affect the industries in which such Person or any of its Subsidiaries principally operate; (iii) changes in GAAP or mandatory changes in the regulatory accounting requirements applicable to any industry in which such Person and its Subsidiaries principally operate; (iv) conditions caused by any acts of God, terrorism, war (whether or not declared) or natural disaster or any worsening thereof; (v) any epidemic, pandemic, plague or other outbreak of illness or disease or public health event (including COVID-19) or any COVID-19 Measures or any changes or prospective changes in such COVID-19 Measures or changes or prospective changes in the interpretation, implementation or enforcement thereof; (vi) any failure in and of itself by such Person and its Subsidiaries to meet any internal or published budgets, projections, forecasts or predictions of financial performance for any period (provided that the underlying cause of any such failure may be considered in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent not excluded by another exception herein); (vii) with respect to Purchaser, the consummation and effects of the Redemption; (viii) the announcement or the existence of, compliance with or performance under, this Agreement or the transactions contemplated hereby, including the impact thereof on the relationships, contractual or otherwise, of the Company or any of its Subsidiaries with employees, labor unions, works councils or other labor organizations, customers, suppliers or partners; (ix) any actions taken at the written request (including e-mail or other forms of electronic communications) or with the written consent of Purchaser (including e-mail or other forms of electronic communications); (x) any changes or prospective changes after the date of this Agreement in applicable Law (or interpretations, implementation or enforcement thereof), excluding GAAP or any other accounting principles (or authoritative interpretations thereof); (xi) any change in the price or relative value of any digital currency or cryptocurrency, or any other blockchain-based tokens or assets, including Bitcoin or EOS; (xii) any change in existence or legality of any digital currency or cryptocurrency, or any other blockchain-based token or asset, or any halt or suspension in trading of any such digital currency or cryptocurrency on any exchange, in each case including Bitcoin or EOS (except that this clause (xii) shall not exclude any changes in existence, public availability, legality, or trading volume of any digital currency or cryptocurrency, or any other blockchainbased token or asset, or any halt or suspension in trading of any such digital currency or cryptocurrency on any exchange, in each case including Bitcoin or EOS, which, reasonably foreseeably result from actions taken by the Target Companies); or (xiii) any change to block structure, methods and rules for adding transactions to the blockchain, methods for processing and adding new blocks, mining or staking rewards, or algorithms, in each case as embodied in the applicable blockchain (except that this clause (xiii) shall not exclude any changes to block structure, methods and rules for adding transactions to the blockchain, methods for processing and adding new blocks, mining or staking rewards, or algorithms, in each case as embodied in the applicable blockchain which, reasonably foreseeably result from actions taken by the Target Companies); provided further, however, that any event, occurrence, fact, condition, or change referred to in clauses (i), (ii), (iii), (iv), (v), (x), (xi) and (xii) immediately above shall be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur to the extent that such event, occurrence, fact, condition, or change has a disproportionate effect on such Person or any of its Subsidiaries compared to other participants in the industries and geographic location in which such Person or any of its Subsidiaries primarily conducts and operates its businesses. Notwithstanding the foregoing, with respect to Purchaser, the amount of the Redemption or failure to obtain the Required Purchaser Shareholder Approval shall not be deemed to be a Material Adverse Effect on or with respect to Purchaser, but for the avoidance of doubt, the conditions to closing under Section 7.1(a) (Required Purchaser Shareholder Approval and Section 7.2(c) (Minimum Cash) as they relate to the foregoing shall continue to apply.

*"Merger Consideration"* means an aggregate value equal to (i) Two Hundred Twenty-Eight Million U.S. Dollars (\$228,000,000), plus if the Company makes the election provided under <u>Section 5.22(b)</u>, the Excess Purchaser Indebtedness and Liability Amount.

"Merger Sub Ordinary Shares" means the ordinary shares, par value \$1.00 per share, of Merger Sub.

"NASDAQ" means the NASDAQ Capital Market.

*"Ordinary Course of Business"* means (i) any actions that are contemplated by or consistent with the Business Plan, (ii) any reasonable actions or omission in response to or otherwise related to any change, effect, event, occurrence, state of facts or development attributable to COVID-19 or COVID- 19 Measures, (iii) any actions taken with respect to setting up, preparing the operations of or reorganizing the Company or any of its Subsidiaries in accordance with, contemplated by or in furtherance of the Business Plan, this Agreement or any Ancillary Document and (iv) any other action reasonably deemed necessary in the good faith determination of the management or directors of the Company in connection with the operations of the Target Companies or the furtherance of the Transactions or the Business Plan.

*"Order"* means any order, decree, ruling, judgment, injunction, writ, determination, binding decision, verdict, judicial award or other Action that is or has been entered, rendered, or otherwise put into effect by or under the authority of any Governmental Authority.

"Organizational Documents" means, with respect to any Person, its memorandum and articles of association or similar organizational documents, in each case, as amended. With respect to Purchaser, Organizational Documents shall also include the Trust Agreement.

"Patents" means any patents and patent applications (including any divisional, provisionals, continuations, continuations-inpart, substitutions, or reissues thereof).

"PCAOB" means the U.S. Public Company Accounting Oversight Board (or any successor thereto).

*"PCAOB Audited Financial Statements"* means the audited consolidated financial statements of the Target Companies (including, in each case, any related notes thereto), consisting of the audited consolidated balance sheets of the Target Companies as of the Balance Sheet Date and the related consolidated audited income statements, changes in shareholder equity and statements of cash flows for the year then ended, each audited in accordance with PCAOB auditing standards by a PCAOB qualified auditor.

*"Permits"* means all federal, state, local or foreign or other third-party permits, grants, easements, consents, approvals, authorizations, exemptions, licenses, franchises, concessions, ratifications, permissions, clearances, confirmations, endorsements, waivers, certifications, designations, ratings, registrations, qualifications or orders of any Governmental Authority or any other Person.

"Permitted Liens" means (a) mechanic's, materialmen's, carriers', repairers' and other similar statutory Liens arising or incurred in the ordinary course of business, (b) Liens for Taxes or assessments and similar governmental charges or levies, which either are (i) not delinquent or (ii) being contested in good faith and by appropriate proceedings, and for which adequate reserves have been established with respect thereto, (c) encumbrances and restrictions on real property (including easements, covenants, conditions, rights of way and similar restrictions) that do not prohibit or materially interfere with any of the Target Companies' use or occupancy of such real property for the operation of their business, (d) other Liens imposed by operation of Law arising in the ordinary course of business for amounts which are not due and payable and as would not in the aggregate materially adversely affect the value of, or materially adversely interfere with the use of, the property subject thereto, (e) licenses of Intellectual Property in the ordinary course of business, (f) Liens arising under this Agreement or any Ancillary Document, or (g) with respect to digital currency or cryptocurrency, or any other blockchain-based tokens or assets (including Bitcoin and EOS), Liens that result from any existing or future encumbrance or restriction (on voting, sale, transfer or disposition) that is inherent in the existing (or any future) structure, governance, or Software underlying such digital currency, cryptocurrency, token or blockchain, such as changes made through governance processes associated therewith), in each case other than such encumbrances or restrictions that are the direct and intended result of the affirmative vote or action occurring after the date of this Agreement by a Target Company. *"Person"* means an individual, corporation, partnership (including a general partnership, limited partnership or limited liability partnership), limited liability company, association, trust or other entity or organization, including a government, domestic or foreign, or political subdivision thereof, or an agency or instrumentality thereof.

*"Personal Property"* means any machinery, equipment, tools, vehicles, furniture, leasehold improvements, office equipment, plant, parts and other tangible personal property.

"Purchaser Charter" means the Amended and Restated Memorandum and Articles of Association of Purchaser in effect under the Cayman Act; <u>provided</u>, that references herein to the Purchaser Charter for periods after the Merger Effective Date shall mean the Amended Purchaser Charter.

*"Purchaser Class A Shares"* means the Class A ordinary shares, par value \$0.0001 per share, of Purchaser, as provided for and fully described in the Purchaser Charter.

*"Purchaser Class B Shares"* means the Class B ordinary shares, par value \$0.0001 per share, of Purchaser, as provided for and fully described in the Purchaser Charter. For purposes clarification, the Purchaser Class B Shares prior to the Closing mean the Class B ordinary shares, par value \$0.0001 per share, of Purchaser being converted into Class A Shares at the Merger Effective Time in accordance with the existing Purchaser Charter, and the Purchaser Class B Shares after the Merger Effective Time mean the Class B ordinary shares, par value \$0.0001 per share, issued pursuant to the Amended Purchaser Charter.

"Purchaser Confidential Information" means all confidential or proprietary documents and information concerning Purchaser or any of its Representatives; provided, however, that Purchaser Confidential Information shall not include any information which, (i) at the time of disclosure by the Company, Purchaser, Merger Sub or any of their respective Representatives, is generally available publicly and was not disclosed in breach of this Agreement or (ii) at the time of the disclosure by Purchaser or its Representatives to by the Company, Purchaser, Merger Sub or any of their respective, was previously known by such receiving party, other than from Purchaser or its Representatives, without violation of Law or any confidentiality obligation by the Person receiving such Purchaser Confidential Information. For the avoidance of doubt, from and after the Closing, Purchaser Confidential Information will include the confidential or proprietary information of the Target Companies.

*"Purchaser Fundamental Representations"* means the representations and warranties contained in <u>Sections 2.1</u> (Organization and Standing), <u>2.2</u> (Authorization; Binding Agreement), <u>2.5</u> (Capitalization), <u>2.16</u> (Finders and Brokers), <u>3.1</u> (Organization and Standing), <u>3.2</u> (Authorization; Binding Agreement), <u>3.5</u> (Capitalization) and <u>3.10</u> (Finders and Brokers).

*"Purchaser Lock-up Agreements"* means the Lock-Up Agreements to be entered into at Closing by the Sponsor and the other Affiliates of Purchaser who are party to the Sponsor Support Agreement, the form of which shall be substantially in the form attached as <u>Exhibit A-1</u> hereto.

*"Purchaser Preference Shares"* means preference shares with a par value of \$0.0001 per share, with such designations, voting and other rights and preferences as may be determined from time to time by Purchasers' board of directors.

*"Purchaser Private Shares"* means an aggregate of 224,780 Purchaser Class A Shares issued to Sponsor in a private placement at the time of the consummation of the IPO (including the issuance of Purchaser Class A Shares in connection with the underwriters' partial exercise of an overallotment option), at a purchase price of \$10.00 per share.

*"Purchaser Public Warrants"* means the warrants that were included as part of each Purchaser Unit, entitling the holder thereof to purchase one (1) Purchaser Class A Share per whole warrant at a purchase price of \$11.50 per share.

"Purchaser Shares" means, collectively, the Purchaser Class A Shares, the Purchaser Class B Shares and the Purchaser Preference Shares.

"Purchaser Securities" means the Purchaser Units, the Purchaser Private Shares and the Purchaser Warrants, as appropriate.

"Purchaser Transaction Expenses" means all fees and expenses of Purchaser or Merger Sub (and not otherwise expressly allocated to any of the Target Companies pursuant to the terms of this Agreement or any Ancillary Document) incurred or payable as of the Closing Date in connection with the negotiation, preparation or execution of this Agreement or any Ancillary Document, the performance of their covenants or agreements in this Agreement or any Ancillary Document or the consummation of the transactions contemplated hereby or thereby, along with any and all deferred expenses (including fees or commissions payable to the underwriters and any legal fees) of the IPO payable upon consummation of a Business Combination, including (a) the fees and expenses of outside legal counsel, accountants, advisors, brokers, investment bankers, consultants, financial printer, proxy solicitor, or other agents or service providers of Purchaser, travel and entertainment incurred by Purchaser, and (b) any other fees, expenses, commissions or other amounts that are expressly allocated to Purchaser pursuant to this Agreement or any Ancillary Document.

*"Purchaser Units"* means the units issued in the IPO (including overallotment units acquired by Purchaser's underwriters) consisting of one (1) Purchaser Class A Share and one-half (1/2) of one Purchaser Public Warrant.

*"Reference Time"* means the close of business of the Company on the Business Day prior to the Closing Date (but without giving effect to the transactions contemplated by this Agreement, including any payments by Purchaser hereunder to occur at the Closing, but treating any obligations in respect of Indebtedness or other liabilities that are contingent upon the consummation of the Closing as currently due and owing without contingency as of the Reference Time).

"Related Person" has the meaning under Item 404 of Regulation S-K promulgated under the Securities Act.

*"Representatives"* means, as to any Person, such Person's Affiliates and the respective managers, directors, officers, employees, consultants, advisors (including financial advisors, counsel and accountants), agents and other legal representatives of such Person or its Affiliates.

*"Required Purchaser Shareholder Approval"* means the approval by the requisite majorities of the issued and outstanding shares of Purchaser in accordance with the Organizational Documents of Purchaser.

*"Restructuring"* means the restructuring of the Company's Subsidiaries, including the creation of new Subsidiaries and termination of control documents between the WFOE and the VIEs, all as more specifically described in the Restructuring Plan provided by the Company to Purchaser.

"SEC" means the U.S. Securities and Exchange Commission (or any successor Governmental Authority).

"Securities Act" means the U.S. Securities Act of 1933.

"Series A SPA and Company Shareholders Agreement Waivers" has the meaning set forth in Section 7.3(c).

"Software" means any computer software programs, including all source code, object code, and documentation related thereto.

"SOX" means the U.S. Sarbanes-Oxley Act of 2002.

"Subsidiary" means, with respect to any Person, any corporation, partnership, association or other business entity of which (i) if a corporation, a majority of the total voting power of capital shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a partnership, association or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons will be deemed to have a majority ownership interest in a partnership, association or other business entity gains or losses or will be or control the managing director, managing member, general partner or other managing Person of such partnership, association or other business entity also include any variable interest entity which is consolidated with such Person under applicable accounting rules.

"Target Company" means each of the Company and its direct and indirect Subsidiaries (excluding Purchaser and Merger Sub).

*"Tax"* means any U.S. federal, state, local or non-United States income, gross receipts, franchise, estimated, alternative minimum, sales, use, transfer, value added, excise, stamp, customs, duties, ad valorem, real property, personal property (tangible and intangible), capital stock, social security, unemployment, payroll, wage, employment, severance, occupation, registration, environmental, communication, mortgage, profits, license, lease, service, goods and services, withholding, premium, unclaimed property, escheat, turnover, windfall profits or other taxes of any kind whatever, whether computed on a separate or combined, unitary or consolidated basis or in any other manner, together with any interest, deficiencies, penalties, additions to tax, or additional amounts imposed by any Governmental Entity with respect thereto, whether disputed or not, and including any secondary liability for any of the aforementioned.

*"Tax Return"* means any return, declaration, report, claim for refund, information return or other documents (including any related or supporting schedules, statements or information) filed or required to be filed in connection with the determination, assessment or collection of any Taxes or the administration of any Laws or administrative requirements relating to any Taxes.

"TIA" means the Cayman Islands Tax Information Authority.

*"Trade Secrets"* means any trade secrets, confidential business information, concepts, ideas, designs, research or development information, processes, procedures, techniques, technical information, specifications, operating and maintenance manuals, engineering drawings, methods, know-how, data, mask works, discoveries, inventions, modifications, extensions, and improvements (whether or not patentable or subject to copyright, trademark, or trade secret protection), in each case, to the extent the foregoing are confidential, provide material competitive advantage, and protected by applicable Law.

"Trademarks" means any trademarks, service marks, trade dress, trade names, brand names, designs, logos, or corporate names (including, in each case, the goodwill associated therewith), whether registered or unregistered, and all registrations and applications for registration and renewal thereof.

*"Trust Account"* means the trust account established by Purchaser with the proceeds from the IPO pursuant to the Trust Agreement in accordance with the IPO Prospectus.

"Trust Agreement" means that certain Investment Management Trust Agreement, dated as of April 28, 2021, as it may be amended (including to accommodate the Merger), by and between Purchaser and the Trustee.

"Trustee" means Wilmington Trust, National Association, in its capacity as trustee under the Trust Agreement.

*"TUG Registration Rights Agreement"* means the Registration Rights Agreement, dated as of April 28, by and among Purchaser, Sponsor and the other "Holders" named therein.

*"VIEs"* means collectively, Beijing Saire Technology Co., Ltd., a company organized under the laws of the Republic of China (*"BST"*), and Xinjiang Silicon Substrate Heating Power Co., Ltd., a company organized under the laws of the Republic of China and wholly owned subsidiary of BST.

*"Warrant Agreement"* means that certain Warrant Agreement, dated as of April 28, 2021 by and between VStock Transfer, LLC and Purchaser.

*"WOFE"* means Hangzhou Dareruohan Technology Co., Ltd., a wholly foreign owned enterprise incorporated under the laws of the Republic of China and wholly owned subsidiary of Silicon Asset Investment, Limited, a company incorporated under the Laws of Hong Kong.

60

11.2 <u>Section References</u>. The following capitalized terms, as used in this Agreement, have the respective meanings given to them in the Section as set forth below adjacent to such terms:

Term Acquisition Proposal Agreement

Alternative Transaction	5.5(a)
Amended Purchaser Charter	7.2(e)
Antitrust Laws	5.7(b)
Business Combination	9.1
Claim	5.15(a)
Closing	1.1(a)
Closing Date	1.1(a)
Closing Filing	5.11(b)
Closing Press Release	5.11(b)
Company	Preamble

Term	Section
Company Benefit Plan	4.184.18(a)
Company Class A Ordinary Shares	4.3(a)
Company Class B Ordinary Shares	4.3(a)
Company Closing Statement	1.2(b)
Company Disclosure Schedules	Article IV
Company IP	4.13(c)
Company IP Licenses	4.13(a)
Company Letter of Transmittal	1.6(a)
Company Material Contract	4.12(a)
Company Permits	4.10
Company Ordinary Shares	4.3(a)
Company Preferred Shares	4.3(a)
Company Real Property Leases	4.16(b)
Company Registered IP	4.13(a)
Company Series A Preferred Shares	4.3(a)
Company Series Angel Preferred Shares	4.3(a)
Company Series Pre-A Preferred Shares	4.3(a)
Company Series Seed Preferred Shares	4.3(a)
Company Shareholder Support Agreement	Recitals
D&O Indemnified Parties	5.15(a)
Digital Assets	4.24
Dissenting Company Shares	1.5
Enforceability Exceptions	2.2
Exchange Agent	1.6(a)
Federal Securities Laws	5.6
Intended Tax Treatment	Recitals
Interim Period	5.1(a)
IRS	4.18(c)
Labor Laws	4.17(a)
Letter of Transmittal	1.6(a)
Lost Certificate Affidavit	1.6(e)
Material Software	4.13(b)
Merger	Recitals
Merger Documents	1.1(b)
Merger Effective Date	1.1(b)
Merger Sub	Preamble
New Company Information	5.23(b)
New Purchaser Information	5.23(a)
Non-Recourse Parties	10.13
OFAC	2.17(c)
Outside Date	8.1(b)
Party(ies)	Preamble
PII	4.14(a)
Post-Closing Purchaser Board	5.13(a)

Term	Section
Proxy Statement	5.9(a)
Public Certifications	2.6(a)
Public Shareholders	9.1
Purchaser	Preamble
Purchaser Arrangements	5.19
Purchaser Closing Statement	1.2(a)
Purchaser Disclosure Schedules	Article II
Purchaser Equity Plan	5.9(a)
Purchaser Financials	2.6(c)
Purchaser Material Contracts	2.13(a)
Purchaser Recommendation	2.2
Purchaser Shareholder Approval Matters	5.9(a)
Redemption	5.9(a)
Registration Statement	5.9(a)
Released Claims	9.1
Registration Rights Agreement	Recitals
Required Purchaser Shareholder Approval Matters	5.9(a)
SEC Reports	2.6(a)
Section 280G Payments	5.19
Shareholder Certificates	1.6(c)
Sidley	10.13
Signing Filing	5.11(b)
Signing Press Release	5.11(b)
Special Meeting	5.9(a)
Specified Courts	10.4
Sponsor	Recitals
Sponsor Support Agreement	Recitals
Surviving Company	1.1(a)
Target Support Agreements	Recitals
Transactions	Recitals
Transmittal Documents	1.6(c)
Waived 280G Benefits	5.19
Winston	10.3
Withholding Party	1.3

# [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK; SIGNATURE PAGE FOLLOWS]

62

IN WITNESS WHEREOF, each Party hereto has caused this Agreement to be signed and delivered by its respective duly authorized officer as of the date first written above.

Purchaser:

# TRADEUP GLOBAL CORPORATION

By: /s/ Huang Lei

Name: Huang Lei Title: Co-Chief Executive Officer

<u>Merger Sub</u>:

#### **TGC MERGER SUB**

By: /s/ Huang Lei

Name: Huang Lei Title: Co-Chief Executive Officer

#### [Signature Page to Business Combination Agreement]

63

The Company:

## SAITECH LIMITED

By: /s/Li Risheng

Name: Li Risheng Title: Director

[Signature Page to Business Combination Agreement]

64

#### Exhibit A-1

#### Form of Lock-Up Agreement (Sponsor, Affiliates and Other Parties)

**Final Form** 

[\_], 2021 SAI.TECH Global Corporation c/o ICS Corporate Services (Cayman) Limited 3-212 Governors Square, 23 Lime Tree Bay Avenue P.O. Box 30746, Seven Mile Beach Grand Cayman KY1-1203, Cayman Islands

#### **Re: Lock-Up Agreement for Company Shares**

Ladies and Gentlemen:

This letter (this "Letter Agreement") is being delivered to you in accordance with that certain Business Combination Agreement (as may be amended, restated or supplemented from time to time, the "Business Combination Agreement") entered into by and among SAI.TECH Global Corporation (f/k/a TradeUp Global Corporation), a Cayman Islands exempted company incorporated with limited liability with company number 370735 (together with its successors, "Purchaser"), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and a direct wholly owned subsidiary of Purchaser ("Merger Sub") and SAITECH Limited, a Cayman Islands exempted company incorporated with limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "Company"), pursuant to which, among other things, Merger Sub will merge with and into the Company (the "Merger"), with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser.

In order to induce Purchaser and the Company to proceed with the Merger and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned shareholder of Purchaser (the "*Shareholder*")<sup>1</sup> hereby agrees with Purchaser as follows:

Subject to the exceptions set forth herein, the Shareholder agrees not to, without the prior written consent of the board of directors of Purchaser, (i) sell, offer to sell, contract or agree to sell, assign, lend, offer, encumber, donate, hypothecate, pledge, grant any option, right or warrant to purchase or otherwise transfer, dispose of or agree to transfer or dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, (A) any Class A Ordinary Shares, par value \$0.0001 per share, of Purchaser (the "Purchaser Class A Ordinary Shares"), (B) any Class B Ordinary Shares, par value \$0.0001 per share, of Purchaser outstanding prior to the Merger (the "Prior Purchaser Class B Ordinary Shares") that will be converted into Purchaser Class A Ordinary Shares at the Merger Effective Date, or (C) any securities convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares, in each case, held by it immediately after the Merger Effective Date (as defined in the Business Combination Agreement) (the "Lock-up Shares"), (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any of the Lockup Shares, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii) (the actions specified in clauses (i)-(iii), collectively, "Transfer") until (a) for one-half of the Lock-up Shares, the earlier of (x) the date on which the volume weighted average trading price of the Purchaser Class A Ordinary Shares exceeds \$14.00 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after the Closing Date and (y) date of the first anniversary of the Closing Date and (b) for one-half of the Lock-up Shares, the earlier of (x) the date on which the volume weighted average trading price of the Purchaser Class A Ordinary Shares exceeds \$17.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after the Closing and (y) date of the first anniversary of the Closing Date (the "Lock-Up Period").

1 Parties that will sign the Form of Lock-up Agreement (Sponsor) will include: TradeUp Global Sponsor LLC, Michael Davidov, Tao Jiang and David Xianglin Li.

#### Ex A-1-1

2. The restrictions set forth in paragraph 1 shall not apply to:

1.

(i) (A) to another entity that is an affiliate of the undersigned, or to any investment fund or other entity controlling, controlled by, managing or managed by or under common control with the undersigned or affiliates of the undersigned or who shares a common investment advisor with the undersigned or (B) as part of a distribution to members, partners or shareholders of the undersigned via dividend or share repurchase;

- (ii) Transfers by virtue of the laws of the state of the entity's organization and the entity's organizational documents upon dissolution of the entity;
- transactions relating to Purchaser Class A Ordinary Shares or other securities convertible into or exercisable or
   (iii) exchangeable for Purchaser Class A Ordinary Shares acquired in open market transactions after the Merger Effective Date;

the exercise of stock options or warrants to purchase Purchaser Class A Ordinary Shares or the vesting of share awards of Purchaser Class A Ordinary Shares and any related transfer of Purchaser Class A Ordinary Shares to Purchaser in connection therewith (a) deemed to occur upon the "cashless" or "net" exercise of such options or warrants or (b) for

(iv) the purpose of paying the exercise price of such options or warrants or for paying taxes due as a result of the exercise of such options or warrants, the vesting of such options, warrants or share awards, or as a result of the vesting of such Purchaser Class A Ordinary Shares, it being understood that all Purchaser Class A Ordinary Shares received upon such exercise, vesting or transfer will remain subject to the restrictions of this Letter Agreement during the Lock-Up Period; surrender of Company Ordinary Shares (as defined in the Business Combination Agreement) or other securities
 convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares for cancellation pursuant to any contractual arrangement in effect at the Merger Effective Date;

the entry, by the Shareholder, at any time after the Merger Effective Date, of any trading plan providing for the sale of Purchaser Class A Ordinary Shares by the Shareholder, which trading plan meets the requirements of Rule 10b5-1(c)

- (vi) under the Exchange Act, <u>provided</u>, <u>however</u>, that such plan does not provide for, or permit, the sale of any Purchaser Class A Ordinary Shares during the Lock-Up Period and no public announcement or filing is voluntarily made or required regarding such plan during the Lock-Up Period;
- transactions in the event of completion of a liquidation, merger, stock exchange or other similar transaction which
   results in all of Purchaser's shareholders having the right to exchange their Purchaser Class A Ordinary Shares for cash, securities or other property; and

transactions to satisfy any U.S. federal, state, or local income tax obligations of the Shareholder (or its direct or indirect owners) arising from a change in the U.S. Internal Revenue Code of 1986, as amended (the "*Code*"), or the U.S. Treasury Regulations promulgated thereunder (the "*Regulations*") after the date on which the Business Combination

(viii) Agreement was executed by the parties, which change prevents the Merger from qualifying as either a "reorganization" pursuant to Section 368(a) of the Code or a transaction governed by Section 351 of the Code (and the Merger does not qualify for similar tax-free treatment pursuant to any successor or other provision of the Code or Regulations taking into account such changes), solely and to the extent necessary to cover any tax liability as a direct result of the transaction.

provided, however, that (A) in the case of clauses (i) through (iii), these permitted transferees must enter into a written agreement, in substantially the form of this Letter Agreement (it being understood that any references to "immediate family" in the agreement executed by such transferee shall expressly refer only to the immediate family of the Shareholder and not to the immediate family of the transferee), agreeing to be bound by these Transfer restrictions. For purposes of this paragraph, "immediate family" shall mean a spouse, domestic partner, child (including by adoption), father, mother, brother or sister of the undersigned, and lineal descendant (including by adoption) of the undersigned or of any of the foregoing persons; and "affiliate" shall have the meaning set forth in Rule 405 under the Securities Act of 1933, as amended.

#### Ex A-1-2

The Lock-Up Period shall terminate upon the earlier of (i) the date of the first anniversary following the Closing Date, or (ii) the closing of a merger liquidation stock exchange representation or other similar transaction after the Closing Date that results in all

 closing of a merger, liquidation, stock exchange, reorganization or other similar transaction after the Closing Date that results in all of the public shareholders of Purchaser having the right to exchange their Purchaser Class A Ordinary Shares for cash securities or other property.

In furtherance of the foregoing, Purchaser, and any duly appointed transfer agent for the registration or transfer of the securitiesdescribed therein, are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Letter Agreement.

This Letter Agreement embodies the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred herein or the documents or instrument referred to herein, which collectively supersedes all prior 5. agreements and the understandings between the parties hereto with respect to the subject matter contained herein. This Letter

5. agreements and the understandings between the parties hereto with respect to the subject matter contained herein. This Letter Agreement may be amended, supplemented or modified only by execution of a written instrument signed by the undersigned Shareholder and Purchaser (and with respect to Purchaser, only with the written consent of a majority of its directors, which shall include a majority of its independent directors).

This Letter Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their respective successors and permitted assigns. This Letter Agreement shall not be assigned by any party hereto, by operation of law or otherwise, without the prime prime shall be appreciated and any party hereto and the prime shall be applied with a prime shal

6. and permitted assigns. This better Agreement shall not be assigned by any party neteto, by operation of law of otherwise, without the prior written consent of the other party and any assignment without such consent shall be null and void; <u>provided</u>, that no such assignment shall relieve the assigning party of its obligations hereunder.

This Letter Agreement and any action, proceeding, claim or dispute (whether in contract, tort or otherwise) (each, an "Action") that may be based upon, arise out of or relate to this Letter Agreement or the negotiation, execution or performance hereof shall be governed by, construed and enforced in accordance with the laws (both substantive and procedural) of the State of Delaware, without regard to the conflicts of law principles thereof. All Actions arising out of or relating to this Letter Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware, or to the extent such Court does not have subject matter jurisdiction, any federal court within the State of Delaware (and any courts having jurisdiction over appeals therefrom) (the

- 7. "Specified Courts"). Each party hereto hereby (i) submits to the exclusive personal and subject matter jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Letter Agreement by any party hereto and (ii) irrevocably waives, and agrees not to assert by way of motion, defense, or otherwise, in any such Action, any claim that it is not subject to the personal or subject matter jurisdiction of the above named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any Specified Court. Each party hereto agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by laws.
- 8. This Letter Agreement shall become effective on the date hereof and terminate on the earlier of (i) the expiration of the Lock-up Period, (ii) termination of the Business Combination Agreement, and (iii) the liquidation of Purchaser.
- 9. This Letter Agreement hereby supersedes Section 5(a) of the Letter Agreement, dated April 28, 2021, among TradeUP Global Corporation, TradeUP Global Sponsor LLC and certain security holders named therein.

Capitalized terms used herein but not defined shall be defined as set forth in the Business Combination Agreement.

[Signature pages follow]

Ex A-1-3

IN WITNESS WHEREOF, each party has duly executed this Letter Agreement, as of the date first written above.

Very truly yours,

Signature:

Name:

Title:

[Signature Page to Lock-Up Agreement]

Ex A-1-4

Acknowledged and agreed by:

#### **TradeUp Global Corporation**

Signature:\_\_\_\_\_

Name:

Title:

[Signature Page to Lock-Up Agreement]

# Exhibit A-2

#### Form of Lock-Up Agreement (Founder and Management)

Final Form

[\_\_], 2021 SAI.TECH Global Corporation c/o ICS Corporate Services (Cayman) Limited 3-212 Governors Square, 23 Lime Tree Bay Avenue P.O. Box 30746, Seven Mile Beach Grand Cayman KY1-1203, Cayman Islands

#### **Re: Lock-Up Agreement for Company Shares**

Ladies and Gentlemen:

This letter (this "Letter Agreement") is being delivered to you in accordance with that certain Business Combination Agreement (as may be amended, restated or supplemented from time to time, the "Business Combination Agreement") entered into by and among SAI.TECH Global Corporation (f/k/a TradeUp Global Corporation), a Cayman Islands exempted company incorporated with limited liability with company number 370735 (together with its successors, "Purchaser"), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and a direct wholly owned subsidiary of Purchaser ("Merger Sub") and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "Company"), pursuant to which, among other things, Merger Sub will merge with and into the Company (the "Merger"), with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser.

In order to induce Purchaser to proceed with the Merger and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned shareholder of the Surviving Company (as defined in the Business Combination Agreement) (the "*Shareholder*")<sup>1</sup> hereby agrees with Purchaser as follows:

Subject to the exceptions set forth herein, the Shareholder agrees not to, without the prior written consent of the board of directors of Purchaser, (i) sell, offer to sell, contract or agree to sell, assign, lend, offer, encumber, donate, hypothecate, pledge, grant any option, right or warrant to purchase or otherwise transfer, dispose of or agree to transfer or dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, (A) any Class A Ordinary Shares, par value \$0.0001 per share, of Purchaser (the "Purchaser Class A Ordinary Shares"), (B) any Class B Ordinary Shares, par value \$0.0001 per share, of Purchaser (the "Purchaser Class B Ordinary Shares"), or (C) any securities convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares, in each case, held by it immediately after the Merger Effective Date (as defined in the Business Combination Agreement) (the "Lock-up Shares"), (ii) enter into any swap or other arrangement that transfers to another, in whole or 1 in part, any of the economic consequences of ownership of any of the Lock-up Shares, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii) (the actions specified in clauses (i)-(iii), collectively, "Transfer") until (a) for one-half of the Lock-up Shares, the earlier of (x) the date on which the volume weighted average trading price of the Purchaser Class A Ordinary Shares exceeds \$14.00 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after the Closing Date and (y) date of the first anniversary of the Closing Date and (b) for one-half of the Lock-up Shares, the earlier of (x) the date on which the volume weighted average trading price of the Purchaser Class A Ordinary Shares exceeds \$17.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days within any 30-trading day period commencing on the date that is 180 days after the Closing and (y) date of the first anniversary of the Closing Date (the "Lock-Up Period").

Parties that will sign the Form of Lock-up Agreement (Company Founders and Management) will include: Energy Science Artist Holding Limited, LilOrange Holding Limited, Eternal Selead Technology Holding Limited, Haonan Technology Holding Limited, Shiningsunny Holding Limited and Make World Better Limited.

## Ex A-2-1

2. The restrictions set forth in paragraph 1 shall not apply to:

1

- (A) to another entity that is an affiliate of the undersigned, or to any investment fund or other entity controlling, controlled by, managing or managed by or under common control with the undersigned or affiliates of the undersigned or who shares a common investment advisor with the undersigned or (B) as part of a distribution to members, partners or shareholders of the undersigned via dividend or share repurchase;
- (ii) Transfers by virtue of the laws of the state of the entity's organization and the entity's organizational documents upon dissolution of the entity;
- transactions relating to Purchaser Class A Ordinary Shares, Purchaser Class B Ordinary Shares or other securities
   (iii) convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares acquired in open market transactions after the Merger Effective Date;

the exercise of stock options or warrants to purchase Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares or the vesting of share awards of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares and any related transfer of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares to Purchaser in connection therewith (a) deemed to occur upon the "cashless" or "net" exercise of such options or warrants or (b) for

- (iv) the purpose of paying the exercise price of such options or warrants or for paying taxes due as a result of the exercise of such options or warrants, the vesting of such options, warrants or share awards, or as a result of the vesting of such Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares, it being understood that all Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares received upon such exercise, vesting or transfer will remain subject to the restrictions of this Letter Agreement during the Lock-Up Period;
- surrender of Company Ordinary Shares (as defined in the Business Combination Agreement) or other securities
   convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares for cancellation pursuant to any contractual arrangement in effect at the Merger Effective Date;

(vi) the entry, by the Shareholder, at any time after the Merger Effective Date, of any trading plan providing for the sale of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares by the Shareholder, which trading plan meets the requirements of Rule 10b5-1(c) under the Exchange Act, provided, however, that such plan does not provide for, or permit, the sale of any Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares during the Lock-Up Period and no public announcement or filing is voluntarily made or required regarding such plan during the Lock-Up Period;

transactions in the event of completion of a liquidation, merger, stock exchange or other similar transaction which
 results in all of Purchaser's shareholders having the right to exchange their Purchaser Class A Ordinary Shares or
 Purchaser Class B Ordinary Shares for cash, securities or other property; and

transactions to satisfy any U.S. federal, state, or local income tax obligations of the Shareholder (or its direct or indirect owners) arising from a change in the U.S. Internal Revenue Code of 1986, as amended (the "*Code*"), or the U.S. Treasury Regulations promulgated thereunder (the "*Regulations*") after the date on which the Business Combination

(viii) Agreement was executed by the parties, which change prevents the Merger from qualifying as either a "reorganization" pursuant to Section 368(a) of the Code or a transaction governed by Section 351 of the Code (and the Merger does not qualify for similar tax-free treatment pursuant to any successor or other provision of the Code or Regulations taking into account such changes), solely and to the extent necessary to cover any tax liability as a direct result of the transaction.

provided, however, that (A) in the case of clauses (i) through (iii), these permitted transferees must enter into a written agreement, in substantially the form of this Letter Agreement (it being understood that any references to "immediate family" in the agreement executed

by such transferee shall expressly refer only to the immediate family of the Shareholder and not to the immediate family of the transferee), agreeing to be bound by these Transfer restrictions. For purposes of this paragraph, "immediate family" shall mean a spouse, domestic partner, child (including by adoption), father, mother, brother or sister of the undersigned, and lineal descendant (including by adoption) of the undersigned or of any of the foregoing persons; and "affiliate" shall have the meaning set forth in Rule 405 under the Securities Act of 1933, as amended.

The Lock-Up Period shall terminate upon the earlier of (i) the date of the first anniversary of the Closing Date, or (ii) the closing of a merger, liquidation, stock exchange, reorganization or other similar transaction after the Closing Date that results in all of the public shareholders of Purchaser having the right to exchange their Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares for cash securities or other property.

In furtherance of the foregoing, Purchaser, and any duly appointed transfer agent for the registration or transfer of the securitiesdescribed therein, are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Letter Agreement.

This Letter Agreement embodies the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred herein or the documents or instrument referred to herein, which collectively supersedes all prior agreements and the understandings between the parties hereto with respect to the subject matter contained herein. This Letter Agreement may be amended, supplemented or modified only by execution of a written instrument signed by the undersigned Shareholder and Purchaser (and with respect to Purchaser, only with the written consent of a majority of its directors, which shall include a majority of its independent directors).

6. This Letter Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their respective successors and permitted assigns. This Letter Agreement shall not be assigned by any party hereto, by operation of law or otherwise, without the prior written consent of the other party and any assignment without such consent shall be null and void; provided, that no such assignment shall relieve the assigning party of its obligations hereunder.

This Letter Agreement and any action, proceeding, claim or dispute (whether in contract, tort or otherwise) (each, an "*Action*") that may be based upon, arise out of or relate to this Letter Agreement or the negotiation, execution or performance hereof shall be governed by, construed and enforced in accordance with the laws (both substantive and procedural) of the State of Delaware, without regard to the conflicts of law principles thereof. All Actions arising out of or relating to this Letter Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware, or to the extent such Court does not have subject matter jurisdiction, any federal court within the State of Delaware (and any courts having jurisdiction over appeals therefrom) (the

- 7. "Specified Courts"). Each party hereto hereby (i) submits to the exclusive personal and subject matter jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Letter Agreement by any party hereto and (ii) irrevocably waives, and agrees not to assert by way of motion, defense, or otherwise, in any such Action, any claim that it is not subject to the personal or subject matter jurisdiction of the above named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any Specified Court. Each party hereto agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by laws.
- 8. This Letter Agreement shall become effective on the date hereof and terminate on the earlier of (i) the expiration of the Lock-up Period, (ii) termination of the Business Combination Agreement, and (iii) the liquidation of Purchaser.

Capitalized terms used herein but not defined shall be defined as set forth in the Business Combination Agreement.

[Signature pages follow]

Ex A-2-3

IN WITNESS WHEREOF, each party has duly executed this Letter Agreement, as of the date first written above.

Very truly yours,

[Shareholder]

Signature:/s/

Name:

Title:

[Signature Page to Lock-Up Agreement]

Ex A-2-4

Acknowledged and agreed by:

**TradeUp Global Corporation** 

Signature: /s/

Name:

Title:

[Signature Page to Lock-Up Agreement]

Ex A-2-5

# Exhibit A-3

## Form of Lock-Up Agreement (Other Company Shareholders)

**Final Form** 

[\_\_], 2021 SAI.TECH Global Corporation c/o ICS Corporate Services (Cayman) Limited 3-212 Governors Square, 23 Lime Tree Bay Avenue P.O. Box 30746, Seven Mile Beach Grand Cayman KY1-1203, Cayman Islands

# **Re: Lock-Up Agreement for Company Shares**

Ladies and Gentlemen:

This letter (this "Letter Agreement") is being delivered to you in accordance with that certain Business Combination Agreement (as may be amended, restated or supplemented from time to time, the "Business Combination Agreement") entered into

by and among SAI.TECH Global Corporation (f/k/a TradeUp Global Corporation), a Cayman Islands exempted company incorporated with limited liability with company number 370735 (together with its successors, "*Purchaser*"), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and a direct wholly owned subsidiary of Purchaser ("*Merger Sub*") and SAITECH Limited, a Cayman Islands exempted company incorporated with limited, a Cayman Islands exempted company incorporated with limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "*Company*"), pursuant to which, among other things, Merger Sub will merge with and into the Company (the "*Merger*"), with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser.

In order to induce Purchaser to proceed with the Merger and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned shareholder of the Surviving Company (as defined in the Business Combination Agreement) (the "*Shareholder*")<sup>1</sup> hereby agrees with Purchaser as follows:

Subject to the exceptions set forth herein, the Shareholder agrees not to, without the prior written consent of the board of directors of Purchaser, (i) sell, offer to sell, contract or agree to sell, assign, lend, offer, encumber, donate, hypothecate, pledge, grant any option, right or warrant to purchase or otherwise transfer, dispose of or agree to transfer or dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended (the "*Exchange Act*"), and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, (A) any Class A Ordinary Shares, par value \$0.0001 per share, of Purchaser (the "*Purchaser Class A Ordinary Shares*"), (B) any Class B Ordinary Shares, par value \$0.0001 per share, of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares"), or (C) any securities convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares or purchaser S B Ordinary Shares, in each case, held by it immediately after the Merger Effective Date (as defined in the Business Combination Agreement) (the "*Lock-up Shares*"), (ii) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any of the Lock-up Shares, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise or (iii) publicly announce any intention to effect any transaction specified in clause (i) or (ii) (the actions specified in clauses (i)-(iii), collectively, "*Transfer*") for a period of one-hundred eighty (180) days commencing from the Closing Date (as defined in the Business Combination Agreement) (the "*Lock-Up Period*").

Parties that will sign the Form of Lock-up Agreement (Other Company Shareholders) will include: Lancer Venture Holding Limited, Tospring Technology Limited, Gongqingcheng Fargo I Venture Capital Partnership (Limited Partnership), EVERSUNNY ONE Limited, Suzhou Chaotou Island Venture Capital Center (Limited Partnership), Tyche Overseas Investment Company Limited, Nanjing Amino Equity Investment Fund LLP, Amino AI Academy (Nanjing) Co., Ltd, ZenGolden L.P., and PatternBase Inc.

Ex A-3-1

2. The restrictions set forth in paragraph 1 shall not apply to:

1.

1

(iv)

(A) to another entity that is an affiliate of the undersigned, or to any investment fund or other entity controlling, controlled by, managing or managed by or under common control with the undersigned or affiliates of the undersigned or who shares a common investment advisor with the undersigned or (B) as part of a distribution to members, partners or shareholders of the undersigned via dividend or share repurchase;

- (ii) Transfers by virtue of the laws of the state of the entity's organization and the entity's organizational documents upon dissolution of the entity;
- transactions relating to Purchaser Class A Ordinary Shares, Purchaser Class B Ordinary Shares or other securities
   (iii) convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares acquired in open market transactions after the Merger Effective Date;

the exercise of stock options or warrants to purchase Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares or the vesting of share awards of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares and any related transfer of Purchaser Class A Ordinary Shares to Purchaser in connection therewith (a) deemed to occur upon the "cashless" or "net" exercise of such options or warrants or (b) for the purpose of paying the exercise price of such options or warrants or for paying taxes due as a result of the exercise of such options or warrants, the vesting of such options, warrants or share awards, or as a result of the vesting of such Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares, it being understood that all Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares received upon such exercise, vesting or transfer will remain subject to the restrictions of this Letter Agreement during the Lock-Up Period;

surrender of Company Ordinary Shares (as defined in the Business Combination Agreement) or other securities
 convertible into or exercisable or exchangeable for Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares for cancellation pursuant to any contractual arrangement in effect at the Merger Effective Date;

the entry, by the Shareholder, at any time after the Merger Effective Date, of any trading plan providing for the sale of Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares by the Shareholder, which trading plan meets the requirements of Rule 10b5-1(c) under the Exchange Act, <u>provided</u>, <u>however</u>, that such plan does not provide for, or

- (vi) permit, the sale of any Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares during the Lock-Up Period and no public announcement or filing is voluntarily made or required regarding such plan during the Lock-Up Period;
- transactions in the event of completion of a liquidation, merger, stock exchange or other similar transaction which
   results in all of Purchaser's shareholders having the right to exchange their Purchaser Class A Ordinary Shares or
   Purchaser Class B Ordinary Shares for cash, securities or other property; and

transactions to satisfy any U.S. federal, state, or local income tax obligations of the Shareholder (or its direct or indirect owners) arising from a change in the U.S. Internal Revenue Code of 1986, as amended (the "*Code*"), or the U.S. Treasury Regulations promulgated thereunder (the "*Regulations*") after the date on which the Business Combination

(viii) Agreement was executed by the parties, which change prevents the Merger from qualifying as either a "reorganization" pursuant to Section 368(a) of the Code or a transaction governed by Section 351 of the Code (and the Merger does not qualify for similar tax-free treatment pursuant to any successor or other provision of the Code or Regulations taking into account such changes), solely and to the extent necessary to cover any tax liability as a direct result of the transaction.

provided, however, that (A) in the case of clauses (i) through (iii), these permitted transferees must enter into a written agreement, in substantially the form of this Letter Agreement (it being understood that any references to "immediate family" in the agreement executed by such transferee shall expressly refer only to the immediate family of the Shareholder and not to the immediate family of the transferee), agreeing to be bound by these Transfer restrictions. For purposes of this paragraph, "immediate family" shall mean a spouse, domestic partner, child (including by adoption), father, mother, brother or sister of the undersigned, and lineal descendant (including by adoption) of the undersigned or of any of the foregoing persons; and "affiliate" shall have the meaning set forth in Rule 405 under the Securities Act of 1933, as amended.

Ex A-3-2

- 3. The Lock-Up Period shall terminate upon the earlier of (i) the expiration of one-hundred eighty (180) days after the Closing Date, or (ii) the closing of a merger, liquidation, stock exchange, reorganization or other similar transaction after the Closing Date that results in all of the public shareholders of Purchaser having the right to exchange their Purchaser Class A Ordinary Shares or Purchaser Class B Ordinary Shares for cash securities or other property.
- In furtherance of the foregoing, Purchaser, and any duly appointed transfer agent for the registration or transfer of the securities
  described therein, are hereby authorized to decline to make any transfer of securities if such transfer would constitute a violation or breach of this Letter Agreement.

This Letter Agreement embodies the entire agreement and understanding of the parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred herein or the documents or instrument referred to herein, which collectively supersedes all prior agreements and the understandings between the parties hereto with respect to the subject matter contained herein. This Letter Agreement may be amended, supplemented or modified only by execution of a written instrument signed by the undersigned Shareholder and Purchaser (and with respect to Purchaser, only with the written consent of a majority of its directors, which shall include a majority of its independent directors).

5.

6. This Letter Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their respective successors and permitted assigns. This Letter Agreement shall not be assigned by any party hereto, by operation of law or otherwise, without

the prior written consent of the other party and any assignment without such consent shall be null and void; <u>provided</u>, that no such assignment shall relieve the assigning party of its obligations hereunder.

This Letter Agreement and any action, proceeding, claim or dispute (whether in contract, tort or otherwise) (each, an "Action") that may be based upon, arise out of or relate to this Letter Agreement or the negotiation, execution or performance hereof shall be governed by, construed and enforced in accordance with the laws (both substantive and procedural) of the State of Delaware, without regard to the conflicts of law principles thereof. All Actions arising out of or relating to this Letter Agreement shall be heard and determined exclusively in the Court of Chancery of the State of Delaware, or to the extent such Court does not have subject matter jurisdiction, any federal court within the State of Delaware (and any courts having jurisdiction over appeals therefrom) (the "Specified Courts"). Each party hereto hereby (i) submits to the exclusive personal and subject matter jurisdiction of any Specified Court for the purpose of any Action arising out of or relating to this Letter Agreement by any party hereto and (ii) irrevocably waives, and agrees not to assert by way of motion, defense, or otherwise, in any such Action, any claim that it is not subject to the personal or subject matter jurisdiction of the above named courts, that its property is exempt or immune from attachment or execution, that the Action is brought in an inconvenient forum, that the venue of the Action is improper, or that this Agreement or the transactions contemplated hereby may not be enforced in or by any Specified Court. Each party hereto agrees that a final judgment in any Action shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by laws.

8. This Letter Agreement shall become effective on the date hereof and terminate on the earlier of (i) the expiration of the Lock-up Period, (ii) termination of the Business Combination Agreement, and (iii) the liquidation of Purchaser.

Capitalized terms used herein but not defined shall be defined as set forth in the Business Combination Agreement.

7.

[Signature pages follow]

Ex A-3-3

IN WITNESS WHEREOF, each party has duly executed this Letter Agreement, as of the date first written above.

Very truly yours,

[Shareholder]

Signature: /s/

Name:

Title:

[Signature Page to Lock-Up Agreement]

Ex A-3-4

Acknowledged and agreed by:

## **TradeUp Global Corporation**

Signature:/s/

Name:

Title:

[Signature Page to Lock-Up Agreement]

# Exhibit B

#### Form of Letter Agreement Amendment

**EXECUTION COPY** 

September 27, 2021

TradeUp Global Corporation 437 Madison Ave., 27<sup>th</sup> Floor New York, New York 10022

RE: Letter Agreement Amendment

Ladies and Gentlemen:

This letter amends (this "Letter Agreement Amendment") that certain letter agreement, dated April 28, 2021, by and among Tradeup Global Corporation (the "Company"), TradeUp Global Sponsor LLC ("Sponsor"), Sponsor, and David X. Li, Tao Jiang and Michael Davidov (each an "Insider and collectively, the "Insiders") (the "Letter Agreement"). Capitalized terms used herein and not otherwise defined shall have the meanings ascribed to such terms in the Combination Agreement (as defined below).

This Letter Agreement Amendment is being entered into simultaneously with, among other documents, that certain Business Combination Agreement, dated the date hereof (the "<u>Combination Agreement</u>"), by and among the Company, TGC Merger Sub, a wholly owned subsidiary of the Company ("<u>Merger Sub</u>"), and SAITECH Limited ("<u>Target</u>"), pursuant to which, among other things, Merger Sub would merge with and into Target, with Target being the surviving company and becoming a wholly owned subsidiary of the Company (the "<u>Merger</u>"), subject to the terms and conditions set forth in the Combination Agreement.

In order to induce the Target to enter into the Combination Agreement and proceed with the Merger and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, each of the Company, Sponsor and Insiders hereby agree as follows:

1. By execution of this Letter Agreement Amendment, Sponsor hereby consents to the Company's entry into the Combination Agreement as required pursuant to Paragraph 3 of the Letter Agreement.

2. If a Closing of the Combination Agreement occurs, Paragraph 5 of the Letter Agreement will be deleted as of the Merger Effective Date, and instead Sponsor and the Insiders shall be subject to the terms of the Purchaser Lock-Up Agreements in the form attached to the Combination Agreement, which shall be executed and delivered by Sponsor and each of the Insiders in connection with the Closing in accordance with the terms of the Combination Agreement.

Ex B-1

3. The Letter Agreement is hereby amended to add a new Paragraph 19 relating to Sponsor funding obligations:

"19. Sponsor Funding Obligations.

(a) If as of the date hereof, the cash on hand at the Company as of such date is insufficient to fund the Company's ordinary working capital expenses of the Company and Merger Sub though the earlier of Closing or the termination of the Combination Agreement in accordance with its terms, then Sponsor shall fund (or cause its Affiliates (other than the Company or any Subsidiary of the Company) to fund) any such shortfall in working capital. Subject to the limitations set forth in Paragraph 19(b) below. funding may be made at the election of the Purchaser in equity, debt or convertible debt; *provided that*, (i) any such working capital funding for additional equity or convertible debt shall (A) not exceed \$1,200,000 and (B) shall be for the issuance of Class A Ordinary Shares of Purchaser at a price or conversion price of US\$10.00 per share (with the result that any debt convertible shall be converted immediately prior to the Closing into that number of shares of Class A Ordinary Shares of the Company equal to the quotient obtained by dividing the outstanding indebtedness balance immediately prior to Closing divided by 10), and (ii) in the case of any debt funding (whether convertible or non-convertible), such Indebtedness shall bear simple interest at a rate not to exceed 2% per annum.

(b) If from the date of this Agreement through the earlier of Closing or the termination of the Combination Agreement is Agreement, the Company and Merger Sub incur Indebtedness, Purchaser Transaction Expenses (whether due before, on or after the Closing) (including deferred underwriting fees, but expressly excluding any D&O tail insurance costs pursuant to Section 5.16(b) of the Combination Agreement) or other Liabilities incurred by Purchaser or Merger Sub for purposes of funding working capital that in the aggregate exceeds the Purchaser Debt and Liability Cap, without the prior express written consent of Target, then Sponsor shall fund (or cause its Affiliates (other than the Company or any Subsidiary of the Company) to fund) such excess amount in Sponsor's election either by (i) a cash contribution by Sponsor (for no additional equity interests) or (ii) advancing funds in the form of debt obligations that (A) will bear simple interest at a rate not to exceed 2% and (B) if a Closing occurs, such debt will remain an obligation of the Company, but Sponsor (or its Affiliates, as applicable) shall forfeit that number of Class A Ordinary Shares and or Class B Ordinary Shares of the Company immediately prior to Closing equal to the quotient obtained by dividing the outstanding indebtedness balance immediately prior to Closing divided by 10."

4. Target shall be an express intended third party beneficiary of the amendments set forth in this Letter Agreement Amendment, with full power to enforce the terms and provisions hereof against Sponsor and the Insiders, as applicable.

5. This Letter Agreement Amendment may be executed in any number of original or electronic counterparts, and each of such counterparts shall for all purposes be deemed to be an original, and all such counterparts shall together constitute one and the same instrument.

6. Except as amended herein, all terms and conditions of the Letter Agreement shall continue in full force and effect.

[Signature Page Follows]

Ex B-2

Sincerely,

# TRADEUP GLOBAL SPONSOR LLC

By: /s/ Jianwei Li

Name: Jianwei Li Title: Manager

/s/ David X. Li David X. Li

/s/ Tao Jiang Tao Jiang

/s/ Michael Davidov

Michael Davidov

ACKNOWLEDGED AND AGREED:

# TRADEUP GLOBAL CORPORATION

By: /s/ Hung Lei Name: Huang Lei Title: Co-Chief Executive Officer

#### Ex B-3

#### Exhibit C

#### Form of Company Shareholder Support Agreement

**Execution Version** 

## SUPPORT AGREEMENT

# (SAITECH LIMITED SHAREHOLDERS)

This SUPPORT AGREEMENT (this "Agreement") is made and entered into as of September 27, 2021, by and among SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability (the "Company"), TradeUP Global Corporation, a Cayman Islands exempted company incorporated with limited liability (the "Purchaser"), TradeUP Global Sponsor LLC, a Cayman Islands exempted limited liability company ("Sponsor"), and the undersigned parties who hold Subject Shares (the "Shareholders").

WHEREAS, the Company, Purchaser, and TGC Merger Sub, a Cayman Islands exempted company (the "Merger Sub"), are concurrently herewith entering into a Business Combination Agreement (as the same may be amended, restated or supplemented, the "Business Combination Agreement"; capitalized terms used but not defined herein shall have the meaning ascribed to such terms in the Business Combination Agreement) pursuant to which, among other things, Merger Sub will be merged with and into Company, with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser; and

WHEREAS, each Shareholder is, as of the date of this Agreement, the sole legal owner of the number of Company Class A Ordinary Shares, Company Class B Ordinary Shares and Company Preferred Shares set forth opposite such Shareholder's name on <u>Schedule A</u> hereto (such Company Class A Ordinary Shares, Company Class B Ordinary Shares and Company Preferred Shares, together with any other Company Shares acquired by such Shareholder after the date of this Agreement and during the term of this Agreement, including upon conversion of Company Shares, exercise of options or settlement of restricted share units, being collectively referred to herein as the "*Subject Shares*"); and

WHEREAS, as a condition to their willingness to enter into the Business Combination Agreement, Purchaser has requested that each Shareholder enter into this Agreement.

**NOW, THEREFORE**, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and the representations, warranties, covenants and agreements contained in this Agreement and the Business Combination Agreement, and intending to be legally bound hereby, the parties hereto agree as follows:

Ex C-1

## **ARTICLE I**

## **Representations and Warranties of Each Shareholder**

Each Shareholder hereby represents and warrants, severally and not jointly, to the Company and Purchaser as follows:

1.1 Organization and Standing; Authorization. Such Shareholder, (a) if a natural person, is of legal age to execute this Agreement and is legally competent to do so, and (b) if the Shareholder is not a natural person, (i) has been duly organized and is validly existing and in good standing under the Laws of the Cayman Islands, (ii) has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted, (iii) has all requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby and (iv) is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary. If the Shareholder is not a natural person, the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized and no other corporate proceedings on the part of such Shareholder are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby.

1.2 <u>Binding Agreement</u>. This Agreement has been or shall be when delivered, duly and validly executed and delivered by such Shareholder and, assuming the due authorization, execution and delivery of this Agreement by the other parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of such Shareholder subject to the Enforceability Exceptions.

1.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of such Shareholder is required to be obtained or made in connection with the execution, delivery or performance by such Shareholder of this Agreement or the consummation by such Shareholder of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of such Shareholder to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

1.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by such Shareholder will not (a) conflict with or violate any provision of the Organizational Documents of such Shareholder, if applicable, (b) conflict with or violate any Law, Order or Consent applicable to such Shareholder or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by such Shareholder under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) give rise to any obligation to make payments or sets of such Shareholder under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of such Shareholder, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of such Shareholder to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

1.5 <u>Subject Shares</u>. As of the date of this Agreement, such Shareholder has beneficial ownership of the Subject Shares set forth opposite such Shareholder's name on <u>Schedule A</u> hereto, and all such Subject Shares are owned by such Shareholder free and clear of all liens or encumbrances, other than liens or encumbrances pursuant to this Agreement, the Organizational Documents of the Company or applicable federal or state securities laws. Such Shareholder does not legally own any shares of the Company other than the Subject Shares. Such Shareholder has the sole right to vote the Subject Shares, and none of the Subject Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Subject Shares, except as contemplated by this Agreement, the Organizational Documents, or the Company Shareholder Support Agreement of the Company.

1.6 <u>Business Combination Agreement</u>. Such Shareholder understands and acknowledges that Purchaser and the Company are entering into the Business Combination Agreement in reliance upon such Shareholder's execution and delivery of this Agreement. Such Shareholder has received a copy of the Business Combination Agreement and is familiar with the provisions of the Business Combination Agreement.

# ARTICLE II

# **Representations and Warranties of Purchaser**

Purchaser hereby represents and warrants to the Shareholders and the Company as follows:

2.1 Organization and Standing. Purchaser is an exempted company with limited liability duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. Purchaser has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary.

2.2 <u>Authorization</u>; <u>Binding Agreement</u>. Purchaser has all requisite corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the board of directors of Purchaser and no other corporate proceedings on the part of Purchaser are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby. This Agreement has been or shall be when delivered, duly and validly executed and delivered by Purchaser and, assuming the due authorization, execution and delivery of this Agreement by the other parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of Purchaser, subject to the Enforceability Exceptions.

2.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of Purchaser is required to be obtained or made in connection with the execution, delivery or performance of this Agreement or the consummation by Purchaser of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Purchaser to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

2.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by Purchaser will not (a) conflict with or violate any provision of Organizational Documents of Purchaser, (b) conflict with or violate any Law, Order or Consent applicable to Purchaser or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by Purchaser under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of Purchaser under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of Purchaser, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Purchaser to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

# **ARTICLE III**

#### **Representations and Warranties of the Company**

The Company hereby represents and warrants to the Shareholders and Purchaser as follows:

3.1 Organization and Standing. The Company is an exempted company with limited liability duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. The Company has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. The Company is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary.

3.2 <u>Authorization; Binding Agreement</u>. The Company has all requisite corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the board of directors and shareholders of the Company and no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby. This Agreement has been or shall be when delivered, duly and validly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement by the other parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the Enforceability Exceptions.

Ex C-4

3.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of the Company is required to be obtained or made in connection with the execution, delivery or performance by the Company of this Agreement or the consummation by the Company of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of the Company to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

3.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by the Company will not (a) conflict with or violate any provision of Organizational Documents of the Company, (b) conflict with or violate any Law, Order or Consent applicable to the Company or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by the Company under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of the Company under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of the Company, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of the Company to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

Ex C-5

## ARTICLE IV

# Agreement to Vote; Certain Other Covenants of the Shareholders

Each Shareholder covenants and agrees during the term of this Agreement as follows:

4.1 Agreement to Vote.

(a) In Favor of Merger. At any meeting of the shareholders of the Company called to seek the Required Company Shareholder Approval, or at any adjournment thereof, or in connection with any written consent of the shareholders of the Company or in any other circumstances upon which a vote, consent or other approval with respect to the Business Combination Agreement, any other Ancillary Documents, the Merger, or any other Transaction is sought (including in accordance with Section 5.10 of the Business Combination Agreement), such Shareholder shall (i), if a meeting is held, appear at such meeting or otherwise cause the Subject Shares to be counted as present at such meeting for purposes of establishing a quorum, and (ii) vote or cause to be voted (including by class vote and/or written consent, if applicable) the Subject Shares in favor of granting the Required Company Shareholder Approval or, if there are insufficient votes in favor of granting the Required Company

Shareholder Approval, in favor of the adjournment such meeting of the shareholders of the Company to a later date but not past the Outside Date.

(b) Against Other Transactions. At any meeting of shareholders of the Company or at any adjournment thereof, or in connection with any written consent of the shareholders of the Company or in any other circumstances upon which such Shareholder's vote, consent or other approval is sought, such Shareholder shall vote (or cause to be voted) the Subject Shares (including by withholding class vote and/or written consent, if applicable) against (i) any business combination agreement, merger agreement or merger (other than the Business Combination Agreement and the Merger), scheme of arrangement, business combination, consolidation, combination, sale of substantial assets, reorganization, recapitalization, dissolution, liquidation or winding up of or by the Company or any public offering of any shares of the Company, any of its material Subsidiaries, or, in case of a public offering only, a newly-formed holding company of the Company or such material Subsidiaries, other than in connection with the Transactions, (ii) any Acquisition Proposal relating to an Alternative Transaction with respect to the Company or any of its material Subsidiaries, and (iii) other than any amendment to Organizational Documents of the Company permitted or required under Sections 1.1(d) and 5.10 of the Business Combination Agreement, any amendment of Organizational Documents of the Company or other proposal or transaction involving the Company or any of its Subsidiaries, which, in each of cases (i) and (iii) of this sentence, would be reasonably likely to in any material respect impede, interfere with, delay or attempt to discourage, frustrate the purposes of, result in a breach by the Company of, prevent or nullify any provision of the Business Combination Agreement or any other Ancillary Document, the Merger, or any other Transaction or change in any manner the voting rights of any class of the Company's share capital.

(c) <u>Revoke Other Proxies</u>. Such Shareholder represents and warrants that any proxies heretofore given in respect of the Subject Shares that may still be in effect are not irrevocable, and such proxies have been or are hereby revoked, other than the voting and other arrangements under the Organizational Documents of the Company.

Ex C-6

4.2 No Transfer. Other than (x) pursuant to this Agreement, (y) upon the consent of Purchaser or (z) to an Affiliate of such Shareholder (provided that such Affiliate shall enter into a written agreement, in form and substance reasonably satisfactory to Purchaser, agreeing to be bound by this Agreement to the same extent as such transferring Shareholder was with respect to such transferred Subject Shares), from the date of this Agreement until the date of termination of this Agreement, such Shareholder shall not, directly or indirectly, (i) (a) sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option, right or warrant to purchase or otherwise transfer, dispose of or agree to transfer or dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Exchange Act, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, any Subject Share, (b) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Subject Shares, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) publicly announce any intention to effect any transaction specified in clause (a) or (b) (the actions specified in clauses (a)-(c), collectively, "Transfer"), other than pursuant to the Merger, (ii) grant any proxies or enter into any voting arrangement, whether by proxy, voting agreement, voting trust, voting deed or otherwise (including pursuant to any loan of Subject Shares), or enter into any other agreement, with respect to any Subject Shares, in each case, other than as set forth in this Agreement or the voting and other arrangements under the Organizational Documents of the Company, (iii) take any action that would make any representation or warranty of such Shareholder herein untrue or incorrect, or have the effect of preventing or disabling such Shareholder from performing its obligations hereunder, or (iv) commit or agree to take any of the foregoing actions or take any other action or enter into any Contract that would reasonably be expected to make any of its representations or warranties contained herein untrue or incorrect or would have the effect of preventing or delaying such Shareholder from performing any of its obligations hereunder. Any action attempted to be taken in violation of the preceding sentence will be null and void. Such Shareholder agrees with, and covenants to, Purchaser and the Company that such Shareholder shall not request that the Company register the Transfer (by bookentry or otherwise) of any certificated or uncertificated interest representing any of the Subject Shares.

4.3 <u>Waiver of Dissenters' Rights</u>. Such Shareholder hereby irrevocably waives, and agrees not to exercise or assert, any dissenters' rights under Section 238 of the Cayman Act and any other similar statute in connection with the Merger and the Business Combination Agreement.

4.4 <u>New Shares</u>. In the event that prior to the Closing (i) any Company Shares or other securities are issued or otherwise distributed to a Shareholder pursuant to any stock dividend or distribution, or any change in any of the Company Shares or other share capital of the Company by reason of any stock split-up, recapitalization, combination, exchange of shares or the like, (ii) a Shareholder

acquires legal or beneficial ownership of any Company Shares after the date of this Agreement, including upon exercise of options or settlement of restricted share units or (iii) a Shareholder acquires the right to vote or share in the voting of any Company Share after the date of this Agreement (collectively, the "*New Securities*"), the terms "Subject Shares" shall be deemed to refer to and include such New Securities (including all such stock dividends and distributions and any securities into which or for which any or all of the Subject Shares may be changed or exchanged into).

4.5 <u>Termination</u>. This Agreement shall terminate upon the earliest of (i) the Merger Effective Date (<u>provided</u>, <u>however</u>, that upon such termination, Section 4.3, this Section 4.5, Section 4.6 and Article V, shall survive indefinitely) and (ii) the termination of the Business Combination Agreement in accordance with its terms, and upon such termination, no party shall have any liability hereunder other than for its willful and material breach of this Agreement prior to such termination.

4.6 <u>Additional Matters</u>. Such Shareholder shall, from time to time, (i) execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as Purchaser or the Company may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement, the Business Combination Agreement and the other Ancillary Documents and (ii) refrain from exercising any veto right, consent right or similar right (whether under the Organizational Documents of the Company or the Cayman Act) which would impede, disrupt, prevent or otherwise adversely affect the consummation of the Merger or any other Transaction.

## Ex C-7

## ARTICLE V

#### **General Provisions**

5.1 <u>Entire Agreement</u>. This Agreement, the Business Combination Agreement and documents referred to herein and therein constitute the entire agreement of the parties with respect to the subject matter of this Agreement, and supersede all prior agreements and undertakings, both written and oral, among the parties with respect to the subject matter of this Agreement, except as otherwise expressly provided in this Agreement.

5.2 <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) to Purchaser in accordance with Section 10.1 of the Business Combination Agreement and to such Shareholder at its address set forth set forth on <u>Schedule A</u> hereto (or at such other address for a party as shall be specified by like notice).

5.3 <u>Governing Law</u>. This Agreement and all Actions (whether in contract, tort or otherwise) that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance hereof (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) shall be governed by the Laws of the Delaware (without giving effect to choice of law principles thereof).

5.4 <u>Amendments and Waivers; Assignment</u>. Any provision of this Agreement may be amended or waived if, and only if, such amendment or waiver is in writing and signed by all parties. Notwithstanding the foregoing, no failure or delay by any party in exercising any right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise thereof preclude any other or further exercise of any other right hereunder. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assignable without the prior written consent of the Company and Purchaser (to be withheld or given in their sole discretion).

5.5 <u>No Third-Party Beneficiaries</u>. This Agreement shall be for the sole benefit of the parties and their respective successors and permitted assigns and is not intended, nor shall be construed, to give any Person, other than the parties and their respective successors and permitted assigns, any legal or equitable right, benefit or remedy of any nature whatsoever.

5.6 <u>Miscellaneous</u>. The provisions of Article X (other than the first sentence of Section 10.4) of the Business Combination Agreement are incorporated herein by reference, *mutatis mutandis*, as if set forth in full herein.

[Signature pages follow]

IN WITNESS WHEREOF, each party has duly executed this Agreement, all as of the date first written above.

# ENERGY SCIENCE ARTIST HOLDING LIMITED

Signature: Name: Li Risheng Title:

[Signature Page to Target Support Agreement]

Ex C-9

# LILORANGE HOLDING LIMITED

Signature: \_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

Ex C-10

# ETERNAL SELEAD TECHNOLOGY HOLDING LIMITED

Signature: \_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

Ex C-11

# HAONAN TECHNOLOGY HOLDING LIMITED

Signature: \_\_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

Copyright © 2022 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

# SHININGSUNNY HOLDING LIMITED

Signature: \_\_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

Ex C-13

## ZENGOLDEN L.P.

Signature: \_\_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

Ex C-14

# PATTERNBASE INC.

Signature: \_\_\_\_\_\_Name: \_\_\_\_\_\_Title:

[Signature Page to Target Support Agreement]

Ex C-15

# MAKE WORLD BETTER LIMITED

Signature: \_\_\_\_\_\_ Name:

Title:

[Signature Page to Target Support Agreement]

Ex C-16

# TRADEUP GLOBAL CORPORATION

Signature:	
Name:	Jianwei Li
Title:	Chairman and Co-Chief Executive Officer

[Signature Page to Target Support Agreement]

# Ex C-17

# SAITECH LIMITED

Signature: \_\_\_\_\_\_ Name: Title:

[Signature Page to Target Support Agreement]

# Ex C-18

## Schedule A

Name of Holder	Number of Company Class B Ordinary Shares	Number of Company Class A Ordinary Shares	Number of Company Preferred Shares
Energy Science Artist Holding Limited	72,000,000	0	0
LilOrange Holding Limited	0	7,999,600	0
Eternal Selead Technology Holding Limited	0	5,000,800	0
Haonan Technology Holding Limited	0	5,000,800	0
Shiningsunny Holding Limited	0	1,019,200	0
ZenGolden L.P.	0	0	12,012,948
PatternBase Inc.	0	0	4,138,915
Make World Better Limited	0	8,979,600	0

Address for Notice:

[\_\_\_]

Sch A-1

# <u>Exhibit D</u>

# Form of Sponsor Support Agreement

## SUPPORT AGREEMENT

This SUPPORT AGREEMENT (this "*Agreement*") is made and entered into as of September 27, 2021, by and among SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability (the "*Company*"), TradeUP Global Corporation, a Cayman Islands exempted company incorporated with limited liability (the "*Purchaser*"), TradeUP Global Sponsor LLC, a Cayman Islands exempted limited liability company ("*Sponsor*"), and the undersigned parties who hold Subject Shares subject to the Letter Agreement (as defined herein)(such parties, the "*Insiders*" and together with the Sponsor, the "*Founder Holders*").

WHEREAS, the Company, Purchaser, and TGC Merger Sub, a Cayman Islands exempted company (the "Merger Sub"), are concurrently herewith entering into a Business Combination Agreement (as the same may be amended, restated or supplemented, the "Business Combination Agreement"; capitalized terms used but not defined herein shall have the meaning ascribed to such terms in the Business Combination Agreement) pursuant to which, among other things, Merger Sub will be merged with and into Company, with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser; and

WHEREAS, each Founder Holder is, as of the date of this Agreement, the sole legal owner of the number of (i) outstanding Class B ordinary shares of the Purchaser ("*Purchaser Class B Shares*") and (ii) outstanding Class A ordinary shares of the Purchaser ("*Purchaser Class A Shares*") set forth opposite such Founder Holder's name on <u>Schedule A</u> hereto, and such Founder Holders do not own any outstanding Units of Purchaser (the "*Units*"), each consisting of one Purchaser Class A Share and one-half of one redeemable warrant exercisable for one Purchaser Class A Share (the "*Purchaser Public Warrants*") (such Purchaser Class B Shares and Purchaser Class A Shares owned by the Founder Holders, together with any other Purchaser Class B Shares or Purchaser Class A Shares acquired by a Founder Holder after the date of this Agreement and during the term of this Agreement, being collectively referred to herein as the "*Subject Shares*"); and

WHEREAS, as a condition to their willingness to enter into the Business Combination Agreement, Purchaser and the Company have requested that each Founder Holder enter into this Agreement.

**NOW, THEREFORE**, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and the representations, warranties, covenants and agreements contained in this Agreement and the Business Combination Agreement, and intending to be legally bound hereby, the parties hereto agree as follows:

Ex D-1

#### ARTICLE I

#### **Representations and Warranties of Each Founder Holder**

Each Founder Holder hereby represents and warrants, severally and not jointly, to the Company and the Purchaser as follows:

1.1 Organization and Standing; Authorization. Such Founder Holder, (a) if a natural person, is of legal age to execute this Agreement and is legally competent to do so, and (b) if the Founder Holder is not a natural person, (i) has been duly organized and is validly existing and in good standing under the Laws of the Cayman Islands, (ii) has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted, (iii) has all requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby and (iv) is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary. If the Founder Holder is not a natural person, the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized and no other corporate proceedings on the part of such Founder Holder are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby.

1.2 <u>Binding Agreement</u>. This Agreement has been or shall be when delivered, duly and validly executed and delivered by such Founder Holder and, assuming the due authorization, execution and delivery of this Agreement by the other parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of Sponsor, enforceable against such Founder Holder in accordance with its terms, subject to the Enforceability Exceptions.

1.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of such Founder Holder is required to be obtained or made in connection with the execution, delivery or performance by such Founder Holder of this Agreement or the consummation by such Founder Holder of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of such Founder Holder to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

1.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by such Founder Holder will not (a) conflict with or violate any provision of the Organizational Documents of such Founder Holder, if applicable, (b) conflict with or violate any Law, Order or Consent applicable to such Founder Holder or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by such Founder Holder under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of such Founder Holder under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of such Founder Holder, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of such Founder Holder to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

1.5 <u>Subject Shares</u>. As of the date of this Agreement, such Founder Holder has beneficial ownership of the Subject Shares set forth opposite such Founder Holder's name on <u>Schedule A</u> hereto, and all such Subject Shares are owned by such Founder Holder free and clear of all liens or encumbrances, other than liens or encumbrances pursuant to this Agreement, the Organizational Documents of Purchaser or applicable federal or state securities laws. Such Founder Holder does not legally own any Purchaser Shares other than the Subject Shares. Such Founder Holder has the sole right to vote the Subject Shares, and none of the Subject Shares is subject to any voting trust or other agreement, arrangement or restriction with respect to the voting of the Subject Shares, except as contemplated by this Agreement or the Organizational Documents of the Purchaser.

1.6 <u>Business Combination Agreement</u>. Such Founder Holder understands and acknowledges that Purchaser and the Company are entering into the Business Combination Agreement in reliance upon Such Founder Holder's execution and delivery of this Agreement. Such Founder Holder has received a copy of the Business Combination Agreement and is familiar with the provisions of the Business Combination Agreement.

Ex D-2

# ARTICLE II

# **Representations and Warranties of Purchaser**

Purchaser hereby represents and warrants to the Founder Holders and the Company as follows:

2.1 <u>Organization and Standing</u>. Purchaser is an exempted company duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. Purchaser has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Purchaser is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary.

2.2 <u>Authorization</u>; <u>Binding Agreement</u>. Purchaser has all requisite corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the board of directors of Purchaser and no other corporate proceedings on the part of Purchaser are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby. This Agreement has been or shall be when delivered, duly and validly executed and delivered by Purchaser and, assuming the due authorization, execution and delivery of this Agreement by the other

parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of Purchaser, subject to the Enforceability Exceptions.

2.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of Purchaser is required to be obtained or made in connection with the execution, delivery or performance of this Agreement or the consummation by Purchaser of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Purchaser to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

2.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by Purchaser will not (a) conflict with or violate any provision of Organizational Documents of Purchaser, (b) conflict with or violate any Law, Order or Consent applicable to Purchaser or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by Purchaser under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of Purchaser under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of Purchaser, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of Purchaser to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

Ex D-3

## **ARTICLE III**

#### **Representations and Warranties of the Company**

The Company hereby represents and warrants to the Founder Holders and Purchaser as follows:

3.1 Organization and Standing. The Company is an exempted company duly incorporated, validly existing and in good standing under the Laws of the Cayman Islands. The Company has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now being conducted. The Company is duly qualified or licensed and in good standing to do business in each jurisdiction in which the character of the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary.

3.2 <u>Authorization; Binding Agreement</u>. The Company has all requisite corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly and validly authorized by the board of directors and shareholders of the Company and no other corporate proceedings on the part of the Company are necessary to authorize the execution and delivery of this Agreement or to consummate the transactions contemplated hereby. This Agreement has been or shall be when delivered, duly and validly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement by the other parties hereto, constitutes, or when delivered shall constitute, the valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, subject to the Enforceability Exceptions.

3.3 <u>Governmental Approvals</u>. No Consent of or with any Governmental Authority on the part of the Company is required to be obtained or made in connection with the execution, delivery or performance by the Company of this Agreement or the consummation by the Company of the transactions contemplated hereby, other than (a) applicable requirements, if any, of the Securities Act, the Exchange Act, and/ or any state "blue sky" securities Laws, and the rules and regulations thereunder and (b) where the failure to obtain or make such Consents or to make such filings or notifications has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of the Company to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

3.4 <u>Non-Contravention</u>. The execution and delivery of this Agreement, the consummation of the transactions contemplated hereby and compliance with any of the provisions hereof by the Company will not (a) conflict with or violate any provision of Organizational Documents of the Company, (b) conflict with or violate any Law, Order or Consent applicable to the Company or any of its properties or assets, or (c) (i) violate, conflict with or result in a breach of, (ii) constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, (iii) result in the termination, withdrawal, suspension, cancellation or modification of, (iv) accelerate the performance required by the Company under, (v) result in a right of termination or acceleration under, (vi) give rise to any obligation to make payments or provide compensation under, (vii) result in the creation of any Lien (other than Permitted Lien) upon any of the properties or assets of the Company under, (viii) give rise to any obligation to obtain any third party Consent from any Person or (ix) give any Person the right to declare a default, exercise any remedy, accelerate the maturity or performance, cancel, terminate or modify any right, benefit, obligation or other term under, any of the terms, conditions or provisions of, any material Contract of the Company, except for any deviations from any of the foregoing clauses (b) or (c) that has not had, and would not reasonably be expected to have, individually or in the aggregate, a material adverse effect on the ability of the Company to enter into and perform this Agreement and to consummate the transactions contemplated hereby.

Ex D-4

#### ARTICLE IV

#### Agreement to Vote; Certain Other Covenants of the Founder Holders

Each Founder Holder covenants and agrees with the Company during the term of this Agreement as follows:

#### 4.1 Agreement to Vote.

(a) In Favor of Merger. At any meeting of the shareholders of Purchaser called to seek the Required Purchaser Shareholder Approval with respect to the Purchaser Shareholder Approval Matters, or at any adjournment thereof, or in connection with any written consent of the shareholders of Purchaser or in any other circumstances upon which a vote, consent or other approval with respect to the Business Combination Agreement, any other Ancillary Documents, the Merger, or any other Transaction is sought, each Founder Holder shall (i), if a meeting is held, appear at such meeting or otherwise cause the Subject Shares to be counted as present at such meeting for purposes of establishing a quorum, and (ii) vote or cause to be voted (including by class vote and/or written consent, if applicable) the Subject Shares in favor of granting the Required Purchaser Shareholder Approval or, if there are insufficient votes in favor of granting the Required Purchaser Shareholder Approval, in favor of the adjournment such meeting of the shareholders of Purchaser to a later date but not past the Outside Date.

(b) Against Other Transactions. At any meeting of shareholders of Purchaser or at any adjournment thereof, or in connection with any written consent of the shareholders of Purchaser or in any other circumstances upon which such Founder Holder's vote, consent or other approval is sought, such Founder Holder shall vote (or cause to be voted) the Subject Shares (including by withholding class vote and/or written consent, if applicable) against (i) any business combination agreement, merger agreement or merger (other than the Business Combination Agreement and the Merger), scheme of arrangement, business combination, consolidation, combination, sale of substantial assets, reorganization, recapitalization, dissolution, liquidation or winding up of or by Purchaser or any public offering of any shares of Purchaser, any of its material Subsidiaries, or, in case of a public offering only, a newly-formed holding company of Purchaser or such material Subsidiaries, other than in connection with the Transactions, (ii) any Acquisition Proposal relating to an Alternative Transaction with respect to Purchaser, and (iii) other than any amendment to Organizational Documents of Purchaser permitted under Section 2.2 and Section 2.3 of the Business Combination Agreement (and required as the Amended Purchaser Charter as a condition to closing under Section 7.2(e)(v) and Section 7.3(f) of the Business Combination Agreement), any amendment of Organizational Documents of Purchaser or other proposal or transaction involving Purchaser or any of its Subsidiaries, which, in each of cases (i) and (iii) of this sentence, would be reasonably likely to in any material respect impede, interfere with, delay or attempt to discourage, frustrate the purposes of, result in a breach by Purchaser of, prevent or nullify any provision of the Business Combination Agreement or any other Ancillary Document, the Merger, or any other Transaction or change in any manner the voting rights of any class of Purchaser's share capital.

(c) <u>Revoke Other Proxies</u>. Such Founder Holder represents and warrants that any proxies heretofore given in respect of the Subject Shares that may still be in effect are not irrevocable, and such proxies have been or are hereby revoked, other than the voting and other arrangements under the Organizational Documents of Purchaser.

4.2 No Transfer. Other than (x) pursuant to this Agreement, (y) upon the consent of the Company or (z) to an Affiliate of such Founder Holder (provided that such Affiliate shall enter into a written agreement, in form and substance reasonably satisfactory to Purchaser, agreeing to be bound by this Agreement to the same extent as such Founder Holder was with respect to such transferred Subject Shares), from the date of this Agreement until the date of termination of this Agreement, such Founder Holder shall not, directly or indirectly, (i) (a) sell, offer to sell, contract or agree to sell, hypothecate, pledge, grant any option, right or warrant to purchase or otherwise transfer, dispose of or agree to transfer or dispose of, directly or indirectly, or establish or increase a put equivalent position or liquidate or decrease a call equivalent position within the meaning of Section 16 of the Exchange Act, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder, any Subject Share, (b) enter into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any Subject Shares, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) publicly announce any intention to effect any transaction specified in clause (a) or (b) (the actions specified in clauses (a)-(c), collectively, "Transfer"), other than pursuant to the Merger, (ii) grant any proxies or enter into any voting arrangement, whether by proxy, voting agreement, voting trust, voting deed or otherwise (including pursuant to any loan of Subject Shares), or enter into any other agreement, with respect to any Subject Shares, in each case, other than as set forth in this Agreement or the voting and other arrangements under the Organizational Documents of Purchaser, (iii) take any action that would make any representation or warranty of such Founder Holder herein untrue or incorrect, or have the effect of preventing or disabling such Founder Holder from performing its obligations hereunder, or (iv) commit or agree to take any of the foregoing actions or take any other action or enter into any Contract that would reasonably be expected to make any of its representations or warranties contained herein untrue or incorrect or would have the effect of preventing or delaying such Founder Holder from performing any of its obligations hereunder. Any action attempted to be taken in violation of the preceding sentence will be null and void. Such Founder Holder agrees with, and covenants to, Purchaser and the Company that such Founder Holder shall not request that Purchaser register the Transfer (by book-entry or otherwise) of any certificated or uncertificated interest representing any of the Subject Shares.

4.3 <u>Waiver of Rights of Class B Ordinary Shares of the Company</u>. Such Founder Holder hereby irrevocably waives the anti-dilution adjustments set forth in Section 17.3 of Purchaser's Amended and Restated Memorandum and Articles of Association in connection with the issuance of Class A ordinary shares and Class B ordinary shares of the Purchaser pursuant to the Business Combination Agreement or any other issuance of equity interests or securities exchangeable for, or convertible into, such equity interests by the Purchaser during the Interim Period.

4.4 <u>Waiver of Dissenters' Rights</u>. Such Founder Holder hereby irrevocably waives, and agrees not to exercise or assert, any dissenters' rights under Section 238 of the Cayman Act and any other similar statute in connection with the Merger and the Business Combination Agreement.

4.5 <u>No Redemption</u>. Such Founder Holder irrevocably and unconditionally agrees that, from the date hereof and until the termination of this Agreement, such Founder Holder shall not elect to cause Purchaser to redeem any Subject Shares now or at any time legally or beneficially owned by such Founder Holder, or submit or surrender any of its Subject Shares for redemption, in connection with the transactions contemplated by the Business Combination Agreement or otherwise.

Ex D-6

4.6 <u>New Shares</u>. In the event that prior to the Closing (i) any Purchaser Shares or other securities are issued or otherwise distributed to such Founder Holder pursuant to any stock dividend or distribution, or any change in any of the Purchaser Shares or other share capital of Purchaser by reason of any stock split-up, recapitalization, combination, exchange of shares or the like, (ii) such Founder Holder acquires legal or beneficial ownership of any Purchaser Shares after the date of this Agreement, including upon exercise of options or settlement of restricted share units or (iii) such Founder Holder acquires the right to vote or share in the voting of any Purchaser Share after the date of this Agreement (collectively, the "*New Securities*"), the terms "Subject Shares" shall be deemed to refer to and include

such New Securities (including all such stock dividends and distributions and any securities into which or for which any or all of the Subject Shares may be changed or exchanged into).

# ARTICLE V

#### **Additional Agreements of the Parties**

5.1 Letter Agreement. Each Founder Holder and Purchaser hereby agree that from the date hereof until the termination of this Agreement, none of them shall, or shall agree to, amend, modify or vary that certain letter agreement dated April 28, 2021 by and among the Founder Holders and Purchaser (the "Letter Agreement"), except as otherwise provided for under this Agreement, the Business Combination Agreement or any Ancillary Document.

## 5.2 Mutual Release.

(a) Founder Holder Release. Sponsor, on its own behalf and on behalf of each of its Affiliates (other than Purchaser or any of Purchaser's Subsidiaries), and each other Founder Holder on its own behalf, and each of its and their successors, assigns and executors (each, a "Sponsor Releasor"), effective as at the Merger Effective Date, shall be deemed to have, and hereby does, irrevocably, unconditionally, knowingly and voluntarily release, waive, relinquish and forever discharge the Company, Purchaser, their respective Subsidiaries and each of their respective successors, assigns, heirs, executors, officers, directors, partners, managers and employees (in each case in their capacity as such) (each, a "Sponsor Releasee"), from (i) any and all obligations or duties the Company, Purchaser or any of their respective Subsidiaries has prior to or as of the Merger Effective Date to such Sponsor Releasor or (ii) all claims, demands, Liabilities, defenses, affirmative defenses, setoffs, counterclaims, actions and causes of action of whatever kind or nature, whether known or unknown, which any Sponsor Releasor has prior to or as of the Merger Effective Date, against any Sponsor Release arising out of, based upon or resulting from any Contract, transaction, event, circumstance, action, failure to act or occurrence of any sort or type, whether known or unknown, and which occurred, existed, was taken, permitted or begun prior to the Merger Effective Date (except in the event of fraud on the part of a Sponsor Releasee); provided, however, that nothing contained in this Section 5.2(a) shall release, waive, relinquish, discharge or otherwise affect the rights or obligations of any party (i) arising under this Agreement, the Business Combination Agreement, the Ancillary Documents, or Purchaser's Organizational Documents, (ii) for indemnification or contribution, in any Sponsor Releasor's capacity as an officer or director of Purchaser, (iii) arising under any then-existing insurance policy of Purchaser, (iv) pursuant to a contract and/or Purchaser policy, to reimbursements for reasonable and necessary business expenses incurred and documented prior to the Merger Effective Date, or (v) for any claim for fraud.

#### Ex D-7

(b) <u>Company Release</u>. Each of the Company, Purchaser and their respective Subsidiaries and each of its and their successors, assigns and executors (each, a "*Company Releasor*"), effective as at the Merger Effective Date, shall be deemed to have, and hereby does, irrevocably, unconditionally, knowingly and voluntarily release, waive, relinquish and forever discharge each Founder Holder and its respective successors, assigns, heirs, executors, officers, directors, partners, members, managers and employees (in each case in their capacity as such) (each, a "*Company Releasee*"), from (i) any and all obligations or duties such Company Releasee has prior to or as of the Merger Effective Date to such Company Releasor, (ii) all claims, demands, Liabilities, defenses, affirmative defenses, setoffs, counterclaims, actions and causes of action of whatever kind or nature, whether known or unknown, which any Company Releasor has, may have or might have or may assert now or in the future, against any Company Release arising out of, based upon or resulting from any Contract, transaction, event, circumstance, action, failure to act or occurrence of any sort or type, whether known or unknown, and which occurred, existed, was taken, permitted or begun prior to the Merger Effective Date (except in the event of fraud on the part of a Company Releasee); provided, however, that nothing contained in this Section 5.2(b) shall release, waive, relinquish, discharge or otherwise affect the rights or obligations of any party (i) arising under this Agreement, the Business Combination Agreement or the Ancillary Documents, or (ii) for any claim for fraud.

5.3 <u>Termination</u>. This Agreement shall terminate upon the earliest of (i) the Merger Effective Date (<u>provided</u>, <u>however</u>, that upon such termination, <u>Section 5.3</u>, <u>Section 6.2</u>, this <u>Section 6.3</u>, <u>Section 6.4</u>, <u>Section 7.1</u> and <u>Section 7.2</u> shall survive indefinitely) and (ii) the termination of the Business Combination Agreement in accordance with its terms, and upon such termination, no party shall have any liability hereunder other than for its willful and material breach of this Agreement prior to such termination.

5.4 <u>Additional Matters</u>. Each Founder Holder shall, from time to time, (i) execute and deliver, or cause to be executed and delivered, such additional or further consents, documents and other instruments as Purchaser or the Company may reasonably request for the purpose of effectively carrying out the transactions contemplated by this Agreement, the Business Combination Agreement and the other Ancillary Documents and (ii) refrain from exercising any veto right, consent right or similar right (whether under the Organizational Documents of Purchaser or the Cayman Act) which would impede, disrupt, prevent or otherwise adversely affect the consummation of the Merger or any other Transaction.

# ARTICLE VI

# **General Provisions**

6.1 <u>Notice</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or sent by overnight courier (providing proof of delivery) to the Company and Purchaser in accordance with Section 10.1 of the Business Combination Agreement and to such Founder Holder at its address set forth set forth on <u>Schedule A</u> hereto (or at such other address for a party as shall be specified by like notice).

6.2 <u>Governing Law</u>. This Agreement and all Actions (whether in contract, tort or otherwise) that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance hereof (including any claim or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) shall be governed by the Laws of the Cayman Islands (without giving effect to choice of law principles thereof).

6.3 <u>Miscellaneous</u>. The provisions of Article X (other than the first sentence of Section 10.4) of the Business Combination Agreement are incorporated herein by reference, *mutatis mutandis*, as if set forth in full herein.

[Signature pages follow]

Ex D-8

IN WITNESS WHEREOF, each party has duly executed this Agreement, all as of the date first written above.

#### SAITECH LIMITED

Signature: /s/ Li RishengName:Li RishengTitle:Director

[Signature Page to Sponsor Support Agreement]

Ex D-9

IN WITNESS WHEREOF, each party has duly executed this Agreement, all as of the date first written above.

# **TRADEUP GLOBAL CORPORATION**

Signature: /s/ Huang Lei

Name:Huang LeiTitle:Co-Chief Executive Officer

[Signature Page to Sponsor Support Agreement]

IN WITNESS WHEREOF, each party has duly executed this Agreement, all as of the date first written above.

# TRADEUP GLOBAL SPONSOR LLC

Signature: <u>/s/ Jianwei Li</u> Name: Jianwei Li Title: Manager

[Signature Page to Sponsor Support Agreement]

Ex D-11

IN WITNESS WHEREOF, each party has duly executed this Agreement, all as of the date first written above.

# MICHAEL DAVIDOV

/s/ Michael Davidov

TAO JIANG

/s/ Tao Jiang

DAVID XIANGLIN LI

/s/ David Xianglin Li

[Signature Page to Sponsor Support Agreement]

#### Ex D-12

## Schedule A

Name of Founder Holder	Number of Purchaser Class B Shares	Number of Purchaser Class A Shares	Number of Purchaser Public Warrants
TradeUP Global Sponsor LLC	212,247	1,074,780	0
Michael Davidov	20,000	0	0
Tao Jiang	20,000	0	0
David Xianglin Li	20,000	0	0
Addresses for Notice:			

**TradeUP Global Sponsor LLC** 

c/o TradeUP Global Corporation 437 Madison Avenue, 27th Floor New York, New York 10022 Attn: Jianwei Li Email: Jianwei@zhenchengcap.com

# MICHAEL DAVIDOV

c/o TradeUP Global Corporation 437 Madison Avenue, 27th Floor New York, New York 10022 Attn: Michael Davidov Email: michaeldavidov@gmail.com

# TAO JIANG

c/o TradeUP Global Corporation 437 Madison Avenue, 27th Floor New York, New York 10022 Attn: Tao Jiang Email: jiangtao@csdn.net

# DAVID XIANGLIN LI

c/o TradeUP Global Corporation 437 Madison Avenue, 27th Floor New York, New York 10022 Attn: David Xianglin Li Email: Davidx.li@yahoo.com

Sch A-1

# <u>Exhibit E</u>

## **Forms of Employment Agreements**

Execution Version

## **EMPLOYMENT AGREEMENT**

THIS EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of September 27, 2021, by and between TradeUp Global Corporation, a Cayman Islands exempted company incorporated with limited liability with company number 370735 (the "<u>Company</u>") and Risheng Li (the "<u>Executive</u>"), to be effective as of the Merger Effective Date (the "<u>Effective Date</u>"), as such term is defined in that certain Business Company Agreement, dated the date hereof, by and among the Company, TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and wholly owned subsidiary of the Company, and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "<u>Target</u>") (the "<u>Merger Agreement</u>"). In the event the Closing (as defined in the Merger Agreement) does not occur for any reason, this Agreement will be void *ab initio* and of no force or effect.

WHEREAS, the Executive possesses certain experience and expertise that qualifies the Executive to provide the direction and leadership required by the Company, including with respect to the Target and its Subsidiaries; and

WHEREAS, the Company desires to continue to employ the Executive as Chief Executive Officer of the Company following the Closing and the Executive wishes to accept such employment;

NOW, THEREFORE, in consideration of the mutual covenants contained herein and intending to be legally bound hereby, the Company and the Executive agree as follows:

# **1. EMPLOYMENT**

**1.1 Position**. Effective as of the Effective Date, the Executive will be employed by the Company, on a full-time basis, as its Chief Executive Officer. In addition, for so long as the Executive is employed by the Company, the Executive will serve as a director or officer of one or more of the Company's Subsidiaries as may be required from time to time, in each case, without further compensation. Effective immediately upon termination of the Executive's employment for any reason, the Executive will be deemed to have resigned from all such positions and offices the Executive may then hold at the Company or any of its Subsidiaries.

**1.2 Duties.** The Executive agrees to perform the duties of the Executive's position and such other duties as may reasonably be assigned to the Executive by the Board from time to time. The Executive also agrees that, while employed by the Company, the Executive will devote the Executive's full business time and best efforts, business judgment, skill and knowledge to the advancement of the business interests of the Company and its Subsidiaries and to the discharge of the Executive's duties and responsibilities for them; provided, that the Executive's service as a director, trustee or committee member of civic or charitable organizations or of any for-profit organization which is not in the same competitive space as the Company or its Subsidiaries or for those entities with whom Executive has a pre-existing relationship, which are listed on Exhibit A hereto will not be in violation of the foregoing, in each case, to the extent such service does not interfere in any material respect with the effective discharge of the Executive's duties and responsibilities hereunder, create a conflict of interest or violate the Executive's obligations under Section 3 hereof.

**1.3 Compliance with Policy**. The Executive agrees that, while employed by the Company, the Executive will comply with all applicable policies, practices and procedures, and codes of ethics or business conduct of the Company or any of its Subsidiaries, as in effect from time to time.

# 2. COMPENSATION AND BENEFITS

During the Executive's employment hereunder, as compensation for all services performed by the Executive for the Company and its Subsidiaries, the Company will provide the Executive the following compensation and benefits:

**2.1 Base Salary**. From and after the Effective Date, the Company will pay the Executive a base salary at the rate of US\$200,000 per year, payable in accordance with the ordinary payroll practices of the Company and subject to adjustment from time to time by the Board in its discretion (as adjusted from time to time, the "Base Salary").

**2.2 Annual Bonus**. The Executive will be eligible to earn an annual bonus (the "<u>Annual Bonus</u>") for each fiscal year completed during the Executive's employment hereunder, in a target amount of fifty percent (50%) of the Base Salary (for the first year, only with respect to the Base Salary payable from and after the Effective Date). The actual amount of any Annual Bonus will be determined by the Board in its discretion, based on financial, operational, individual and/or other targets established by the Board. In order to be eligible to receive any Annual Bonus hereunder, the Executive must be employed through the date such Annual Bonus is paid, except as otherwise provided in <u>Section 5.2</u>.

**2.3 Employee Benefits**. The Executive will be entitled to participate in all employee benefit plans as in effect from time to time for employees of the Company generally, except to the extent such plans are duplicative of benefits otherwise provided to the Executive under this Agreement. The Executive's participation will be subject to the terms of the applicable plan documents and generally applicable Company policies, as the same may be in effect from time to time, and any other restrictions or limitations imposed by law.

Ex E-1

**2.4 Vacations**. The Executive will be eligible to accrue vacation time in accordance with the policies of the Company as in effect from time to time, which vacation time will be in addition to any holidays observed by the Company. Vacation may be taken at such times and intervals as the Executive determines, subject to the business needs of the Company. Vacation otherwise will be subject to the policies of the Company as in effect from time to time.

**2.5 Business Expenses.** The Company will pay or reimburse the Executive for all reasonable business expenses incurred or paid by the Executive in the performance of the Executive's duties and responsibilities hereunder, subject to any expense reimbursement policies of the Company as in effect from time to time and to such reasonable substantiation and documentation as may be specified by the Company from time to time. The Executive's right to payment or reimbursement hereunder will be subject to the following additional rules: (i) the amount of expenses eligible for payment or reimbursement during any calendar year will not affect the expenses eligible for

payment or reimbursement in any other calendar year; (ii) payment or reimbursement will be made not later than December 31 of the calendar year following the calendar year in which the expense was incurred; and (iii) the right to payment or reimbursement will not be subject to liquidation or exchange for any other benefit.

# 3. CONFIDENTIAL INFORMATION AND RESTRICTED ACTIVITIES

**3.1 Confidential Information**. During the course of the Executive's employment with the Company, the Executive has learned and will continue to learn of Confidential Information and has developed and will continue to develop Confidential Information on behalf of the Company and its Subsidiaries. The Executive agrees not to use or disclose to any Person (except as required or permitted by applicable law or for the proper performance of the Executive's regular duties and responsibilities for the Company) any Confidential Information obtained by the Executive incident to the Executive's employment or any other association with the Company or any of its Subsidiaries (including, for the avoidance of doubt, any such information acquired with respect to the Executive's employment with SAITECH Limited or its Subsidiaries prior to the Effective Date). The Executive agrees that this restriction will continue to apply after the Executive's employment terminates, regardless of the reason for such termination. For the avoidance of doubt, (i) nothing contained in this Agreement limits, restricts or in any other way affects the Executive's communicating with any governmental agency or entity, or communicating with any official or staff person of a governmental agency or entity, concerning matters relevant to such governmental agency or entity; and (ii) the Executive will not be held criminally or civilly liable under any federal, national, provincial, state or local trade secret law for disclosing a trade secret (A) in confidence to a national, federal, provincial, state, or local government official, either directly or indirectly, or to an attorney, solely for the purpose of reporting or investigating a suspected violation of law, or (B) in a complaint or other document filed under seal in a lawsuit or other proceeding, provided, however, that notwithstanding this immunity from liability, the Executive may be held liable if the Executive unlawfully accesses or discloses trade secrets by unauthorized means.

**3.2 Protection of Documents.** All documents, records and files, in any media of whatever kind and description, relating to the business, present or otherwise, of the Company or any of its Subsidiaries, and any copies, in whole or in part, thereof (the "<u>Documents</u>"), whether or not prepared by the Executive, will be the sole and exclusive property of the Company. The Executive agrees to safeguard all Documents and to surrender to the Company, at the time the Executive's employment terminates or at such earlier time or times as the Board or its designee may specify, all Documents then in the Executive's possession or control. The Executive also agrees to disclose to the Company, at the time the Executive's engloyment terminates or at such earlier time or its designee may specify, all passwords necessary to obtain access to any Confidential Information or Company information that the Executive has password-protected on any computer equipment, network or system of the Company or any of its Subsidiaries.

Ex E-2

**3.3** Assignment of Rights to Intellectual Property. The Executive will promptly and fully disclose all Intellectual Property to the Company. The Executive hereby assigns and agrees to assign to the Company (or as otherwise directed by the Company) the Executive's full right, title and interest in and to all Intellectual Property. The Executive agrees to execute any and all applications for domestic and foreign patents, copyrights or other proprietary rights and to do such other acts (including, without limitation, the execution and delivery of instruments of further assurance or confirmation and the provision of good faith testimony in person or by declaration or affidavit) requested by the Company to assign the Intellectual Property to the Company (or as otherwise directed by the Company) and to permit the Company to secure, prosecute and enforce any patents, copyrights or other proprietary rights to the Intellectual Property. The Executive will not charge the Company or any of its Subsidiaries for time spent in complying with these obligations. All copyrightable works that the Executive creates during the Executive's employment will be considered "work made for hire" and will, upon creation, be owned exclusively by the Company.

**3.4 Restricted Activities**. The Executive agrees to the following restrictions on the Executive's activities during and, to the extent applicable, after the Executive's employment, and further agrees that such restrictions are necessary to protect the goodwill, Confidential Information, trade secrets and other legitimate interests of the Company and its Subsidiaries:

(a) *Non-Competition*. While the Executive is employed by the Company and for a period of twenty-four (24) months following the termination of the Executive's employment for any reason (collectively, the "<u>Restricted Period</u>"), the Executive will not, directly or indirectly, whether as an owner, partner, investor, consultant, agent, employee, co-venturer or otherwise, compete with, undertake any planning to compete with, or assist or encourage any other Person in competing with or undertaking any planning to compete with, the Company or any of its Subsidiaries. Specifically, but without limiting the foregoing, the Executive agrees not to work for or provide services to, in any capacity, whether with or without compensation, any Person that is engaged in any business anywhere that is competitive with the business of the Company or any of its Subsidiaries,

as conducted or in active planning at any time during the Executive's employment with the Company. Notwithstanding the foregoing, in no event shall ownership of five percent (5%) or less of the outstanding securities of any class of any issuer whose securities are registered or listed on any securities exchange or recognized securities market anywhere in the world so long as the Executive does not have, or exercise, any rights to manage or operate the business of such issuer other than rights as an equity holder thereof.

(b) *Non-Solicitation of Business Partners*. During the Restricted Period, the Executive will not, directly or indirectly, and will not assist or encourage any other Person to, (i) solicit or encourage any customer, vendor, supplier or other business partner of the Company or any of its Subsidiaries to terminate, diminish or otherwise change in any manner adverse to the Company or any of its Subsidiaries his, her or its relationship with any of them; or (ii) seek to persuade any such customer, vendor, supplier or business partner, or any prospective customer, vendor, supplier or business partner of the Company or any of its Subsidiaries any business or activity that such Person conducts or could conduct with the Company or any of its Subsidiaries.

(c) Non-Solicitation of Employees and Other Service Providers. During the Restricted Period, except as required for the proper performance of the Executive's regular duties and responsibilities hereunder, the Executive will not, directly or indirectly, and will not assist or encourage any other Person to, hire or engage any employee of the Company or any of its Subsidiaries. While the Executive is employed by the Company, except as required for the proper performance of the Executive's regular duties and responsibilities hereunder, the Executive will not, directly or indirectly, and will not assist or encourage any other Person to, directly or indirectly, and will not assist or encourage any other Person to, (i) solicit for hiring or engagement any employee of the Company or any of its Subsidiaries or seek to persuade any such employee to discontinue employment; or (ii) solicit or encourage any independent contractor providing services to the Company or any of its Subsidiaries to terminate, diminish or otherwise change in any manner adverse to the Company or any of its Subsidiaries his, her or its relationship with any of them.

Ex E-3

3.5 Enforcement. In signing this Agreement, the Executive gives the Company assurance that the Executive has carefully read and considered all the terms and conditions of this Agreement, including the restraints imposed on the Executive under this Section 3, that the Executive has not relied on any agreements or representations, express or implied, that are not set forth expressly in this Agreement, and that the Executive has entered into this Agreement knowingly and voluntarily. The Executive agrees that these restraints are necessary for the reasonable and proper protection of the Company and its Subsidiaries, and that each and every one of the restraints is reasonable in respect to subject matter, length of time and geographic area. The Executive further agrees that, were the Executive to breach any of the covenants contained in this Section 3, the damage to the Company and its Subsidiaries would be irreparable. The Executive therefore agrees that the Company, in addition and not in the alternative to any other remedies available to it, will be entitled to preliminary and permanent injunctive relief against any breach or threatened breach by the Executive of any such covenants, without having to post bond. In the event that any provision of this Section 3 is determined by any court of competent jurisdiction to be unenforceable by reason of its being extended over too great a time, too large a geographic area or too great a range of activities, that provision will be deemed to be modified to permit its enforcement to the maximum extent permitted by law. It is also agreed that each of the Company's Subsidiaries will have the right to enforce all of the Executive's obligations to that Subsidiary under this Agreement, including, without limitation, pursuant to this Section 3. No claimed breach of this Agreement or other violation of law attributed to the Company or any of its Subsidiaries, or change in the nature or scope of the Executive's employment or other relationship with the Company or any of its Subsidiaries, will operate to excuse the Executive from the performance of the Executive's obligations under this Section 3.

# 4. TERMINATION OF EMPLOYMENT

The Executive's employment under this Agreement will continue until terminated pursuant to this Section 4.

**4.1 In General**. The Executive's employment may be terminated by the Company or by the Executive at any time upon sixty (60) days' prior written notice to the other party; <u>provided</u>, <u>however</u>, that the Board may elect to waive such notice period or any portion thereof, in which case the Company will continue to pay the Base Salary for that portion of the notice period so waived.

**4.2 Termination by the Company for Cause.** Notwithstanding the provisions of <u>Section 4.1</u>, the Company may terminate the Executive's employment immediately for Cause upon written notice to the Executive setting forth in reasonable detail the nature of the Cause. For purposes of this Agreement, "<u>Cause</u>" means the occurrence of any of the following, as determined by the Board in its reasonable judgment: (i) the Executive's material failure to perform (other than by reason of disability), or substantial misconduct in the performance of, the Executive's duties and responsibilities for the Company or any of its Subsidiaries which causes material harm to the Company; (ii) the Executive's material and demonstrable breach of any provision of <u>Section 3</u> or of any other confidentiality, invention assignment, or other restrictive covenant obligation set forth in any written agreement by and between the Executive and the Company or any of its Subsidiaries; (iii) the Executive's material and demonstrable breach of any other provision of this Agreement or any other written agreement by and between the Executive and the Company or any of its Subsidiaries; (iv) the Executive's material violation of any applicable policy or code of conduct of the Company or any of its Subsidiaries which violation causes material reputational or financial harm to the Company; or (v) the Executive's indictment for, or plea of nolo contendere to, any felony or any crime involving moral turpitude. Notwithstanding anything to the contrary in the foregoing, a circumstance otherwise giving rise to Cause pursuant to the foregoing clause (i), (ii), (iii) or (iv), if capable of cure, will not constitute Cause if cured by the Executive within ten (10) days following the Company's notice to the Executive thereof; <u>provided</u>, <u>however</u>, that the Company will not be required to provide any such notice or opportunity to cure with respect to any subsequent substantially similar or related conduct.

**4.3 Resignation by the Executive for Good Reason**. Notwithstanding the provisions of <u>Section 4.1</u>, the Executive may terminate the Executive's employment for Good Reason upon written notice to the Company setting forth in reasonable detail the nature of the circumstances constituting Good Reason. For purposes of this Agreement, "<u>Good Reason</u>" means the occurrence of any of the following without the Executive's consent: (i) a material reduction in Base Salary, other than an across-the-board reduction applicable to similarly situated executives of the Company; (ii) a permanent relocation of the Executive's principal place of business that increases the Executives commute by more than sixty (60) miles in a single direction; or (iii) a material diminution of Executive's duties, authorities or responsibilities; <u>provided</u>, in each case, that (x) the Executive's first knowledge thereof, (y) the Company fails to cure such circumstance within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the expiration of such cure period.

**4.4 Death and Disability**. The Executive's employment will terminate automatically in the event of the Executive's death during employment. The Company may terminate the Executive's employment, upon notice to the Executive, in the event that the Executive becomes disabled during the Executive's employment hereunder through any illness, injury, accident or condition of either a physical or psychological nature and, as a result, is unable to perform substantially all of the Executive's duties and responsibilities for the Company and its Subsidiaries (notwithstanding the provision of any reasonable accommodation) for a period of ninety (90) days during any period of three hundred sixty-five (365) consecutive days. If any question arises as to whether the Executive is disabled to the extent that the Executive is unable to perform substantially all of the Executive's duties hereunder, the Executive will, at the Company's request, submit to a medical examination by a physician selected by the Company, and to whom the Executive or the Executive's guardian, if any, has no reasonable objection, to determine whether the Executive is so disabled, and such determination will be conclusive of the issue for purposes of this Agreement. If such a question arises and the Executive fails to submit to the requested medical examination, the Company's good faith, reasonable determination of the issue will be binding on the Executive.

Ex E-5

# 5. OTHER MATTERS RELATED TO TERMINATION

**5.1 Final Compensation**. In the event of termination of the Executive's employment with the Company, howsoever occurring, the Company will pay the Executive (i) the Base Salary for the final payroll period of the Executive's employment, through the date the Executive's employment terminates (the "Termination Date"); (ii) compensation at the rate of the Base Salary for any vacation time accrued in accordance with the policies of the Company but not used as of the Termination Date; and (iii) reimbursement, in accordance with Section 2.5 hereof, for business expenses incurred by the Executive but not yet paid to the Executive as of the Termination Date, provided that the Executive submits all expenses and supporting documentation required within sixty (60) days of the Termination Date and that such expenses are reimbursable under Company policies then in effect (all of the foregoing, the "Final Compensation"). Except as otherwise provided in the foregoing clause (iii), the Final Compensation will be paid to the Executive immediately upon termination of employment.

**5.2 Severance Payments.** In the event the Executive's employment is terminated by the Company without Cause or by the Executive for Good Reason, subject to the provisions of <u>Section 5.3</u>, the Company will pay the Executive, in addition to the Final Compensation, severance payments as provided in this <u>Section 5.2</u> (collectively, the "<u>Severance Payments</u>").

(a) Termination Other than in Connection with a Change of Control. Except as otherwise provided in <u>Section 5.2(b)</u>, the Severance Payments will include (i) payment in an amount equal to twelve months Base Salary, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company over the twelve (12) months following the Termination Date; (ii) payment of any earned but unpaid Annual Bonus for the fiscal year preceding the year in which the termination occurs, payable at the same time bonuses otherwise are paid to active employees of the Company; and (iii) payment of premiums for continued health benefits under the Company's health plans for Executive and his dependants, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company until the earlier of (A) twelve (12) months following the Termination Date and (B) the first date on which the Executive becomes eligible to receive health insurance coverage from another employer.

(b) Termination in Connection with a Change of Control. Notwithstanding the provisions of Section 5.2(a), in the event such termination occurs within the three (3) months prior to or twelve (12) months following the date of a Change of Control (and in the case of a termination by the Company without Cause that occurs prior to such Change of Control, solely to the extent such termination results from the request of another party to the Change of Control transaction), the Severance Payments will include (i) payment in an amount equal to (A) fifteen (15) months' Base Salary plus (B) the target amount of the Annual Bonus for the fiscal year in which the termination occurs, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company over the fifteen (15) months following the Termination Date; (ii) payment of any earned but unpaid Annual Bonus for the fiscal year preceding the year in which the termination occurs, payable at the same time bonuses otherwise are paid to active employees of the Company; (iii) payment of premiums for continued health benefits under the Company's health plans for Executive and his dependants, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company until the earlier of (A) fifteen (15) months following the Termination Date and (B) the first date on which the Executive becomes eligible to receive health insurance coverage from another employer; and (iv) notwithstanding anything to the contrary in the Company's incentive equity plans or any applicable award agreement, the Executive's equity awards under such agreement, the Executive's equity awards under such plans that are outstanding and unvested as of the Termination Date will become fully vested effective as of the Release Date (as defined below).

**5.3 Conditions to and Timing of the Severance Payments**. Any obligation of the Company to provide the Severance Payments is conditioned on the Executive signing and returning to the Company, without revocation, a timely and effective separation agreement containing a general release of claims and other customary terms in a form reasonably satisfactory to the Company (the "<u>Release</u>"). The Release must become effective, if at all, by the sixtieth (60th) calendar day following the Termination Date (the date the Release so becomes effective, the "<u>Release Date</u>"). Provided the foregoing conditions are satisfied, the first installment of the Severance Payments will be paid on the Company's next regular payroll date at least five (5) business days following the Release Date, the amount of which will be retroactive to the Termination Date.

**5.4 Benefits Termination**. Except for any right the Executive may have under COBRA or other applicable law to continue participation in the Company's group health and dental plans at the Executive's own cost, the Executive's participation in all employee benefit plans will terminate in accordance with the terms of the applicable benefit plans based on the Termination Date, without regard to any continuation of the Base Salary or other payment to the Executive following such termination, and the Executive will not be eligible to earn vacation or other paid time off following such termination.

**5.5 Survival.** The provisions of this Agreement will survive any termination of employment if so provided in this Agreement or if necessary or desirable to accomplish the purposes of other surviving provisions, including, without limitation, the Executive's obligations under <u>Section 3</u>. Upon termination by either the Executive or the Company, all rights, duties and obligations of the Executive and the Company to each other will cease, except as otherwise expressly provided in this Agreement.

# 6. CERTAIN DEFINITIONS

For purposes of this Agreement, the following definitions apply:

"Board" means the board of directors of the Company or any committee thereof, as applicable.

"Change of Control" has the meaning ascribed to such term in the Company's 2021 Equity Incentive Plan.

"Code" means the Internal Revenue Code of 1986, as amended, together with the regulations and guidance promulgated thereunder.

"<u>Confidential Information</u>" means any and all information of the Company and its Subsidiaries that is not generally available to the public. Confidential Information also includes any information received by the Company or any of its Subsidiaries from any Person with any understanding, express or implied, that it will not be disclosed. Confidential Information does not include information that enters the public domain other than through the Executive's breach of the Executive's obligations under this Agreement or any other agreement between the Executive and the Company or any of its Subsidiaries.

"Intellectual Property" means all Inventions conceived, made, created, developed or reduced to practice by the Executive (whether alone or with others, whether or not during normal business hours or on or off Company premises) during the Executive's employment that relate either to the business of the Company or any of its Subsidiaries or to any prospective activity of the Company or any of its Subsidiaries or that result from any work performed by the Executive for the Company or any of its Subsidiaries or that make use of Confidential Information or any of the equipment or facilities of the Company or any of its Subsidiaries.

"Invention" means any invention, discovery, design, development, improvement, method, process, procedure, plan, project, system, technique, strategy, information, composition, know-how, work, concept or idea, or any modification or derivative of any of the foregoing (whether or not patentable or copyrightable or constituting a trade secret).

"<u>Person</u>" means an individual, a corporation, a limited liability company, an association, a partnership, an estate, a trust or any other entity or organization, other than the Company or any of its Subsidiaries.

"Subsidiary" means, with respect to any Person, any corporation, partnership, association or other business entity of which (i) if a corporation, a majority of the total voting power of capital shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a partnership, association or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons will be deemed to have a majority ownership interest in a partnership, association or other business entity if such Person or Persons will be allocated a majority of partnership, association or other business entity gains or losses or will be or control the managing director, managing member, general partner or other managing Person of such partnership, association or other business entity. A Subsidiary of a Person will also include any variable interest entity which is consolidated with such Person under applicable accounting rules.

Ex E-8

### 7. NO CONFLICTING AGREEMENTS

The Executive hereby represents and warrants that the signing of this Agreement and the performance of the Executive's obligations hereunder will not breach or be in conflict with any other agreement to which the Executive is a party or by which the Executive is bound, and that the Executive is not now subject to any covenants against competition or similar covenants or any court order that could affect the performance of the Executive's obligations hereunder. The Executive agrees that the Executive will not disclose to or use on behalf of the Company any confidential or proprietary information of a third party without that party's consent.

# 8. MISCELLANEOUS

**8.1 Notices**. Any notices provided for in this Agreement will be in writing and will be effective when delivered in person or sent by the United States mail, postage prepaid, return receipt requested, or by a reputable overnight delivery service, and addressed to the Executive at the Executive's last known address on the books of the Company or, in the case of the Company, to it at its principal place of business, attention of the Chair of the Board, or to such other address as either party may specify by notice to the other actually received.

**8.2 Withholding**. All payments made by the Company under this Agreement will be reduced by any tax or other amounts required to be withheld by the Company to the extent required by applicable law.

**8.3** Assignment. Neither the Executive nor the Company may make any assignment of this Agreement or any interest in it, by operation of law or otherwise, without the prior written consent of the other; <u>provided</u>, <u>however</u>, that the Company may assign its rights and obligations under this Agreement without the Executive's consent to one of its Subsidiaries (for the avoidance of doubt, following the Closing, including Target) or to any Person with whom the Company hereafter effects a reorganization, consolidation or merger, or to whom the Company hereafter transfers all or substantially all of its properties or assets. This Agreement will inure to the benefit of and be binding upon the Executive and the Company and each of their respective successors, executors, administrators, heirs and permitted assigns.

**8.4 Severability.** If any portion or provision of this Agreement is declared illegal or unenforceable to any extent by a court of competent jurisdiction, to any extent, then the remainder of this Agreement, or the application of such portion or provision in circumstances other than those as to which it is so declared illegal or unenforceable, will not be affected thereby, and each portion and provision of this Agreement will be valid and enforceable to the fullest extent permitted by law.

**8.5 Other Matters**. This Agreement sets forth the entire agreement and understanding between the parties hereto relating to the subject matter hereof, and replaces all prior and contemporaneous communications, agreements and understandings, written or oral, relating to the same. This Agreement may not be modified or amended, and no breach will be deemed to be waived, unless agreed to in writing by the Executive and an expressly authorized representative of the Board. The headings and captions in this Agreement are for convenience only and in no way define or describe the scope or content of any provision herein. This Agreement may be executed in counterparts (and may be delivered by email or other electronic means), each of which will be an original and all of which together will constitute one and the same instrument. This Agreement will be governed by and construed in accordance with the laws of the State of Delaware, without regard to any conflict of laws principles that would result in the application of the laws of any other jurisdiction. The Executive and the Company agree to submit to the exclusive jurisdiction of the courts of or in the State of Delaware in connection with any dispute arising out of or otherwise related to this Agreement or the Executive's employment with the Company, in whole or in part, and agree that any such dispute will be brought and maintained solely in such courts.

[Remainder of Page Intentionally Left Blank]

Ex E-9

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

# TRADEUP GLOBAL CORPORATION

By: /s/ Huang Lei

Name: Huang Lei Title: Co-Chief Executive Officer

Accepted and agreed:

/s/ Risheng Li

Risheng Li

[Signature Page to Employment Agreement]

## <u>Exhibit A</u>

#### Certain Permitted Activities

## NONE

[Exhibit A]

Execution Version

#### **EMPLOYMENT AGREEMENT**

THIS EMPLOYMENT AGREEMENT (this "<u>Agreement</u>") is made and entered into as of September 27, 2021, by and between TradeUp Global Corporation, a Cayman Islands exempted company incorporated with limited liability with company number 370735 (the "<u>Company</u>") and Jian Zou (the "<u>Executive</u>"), to be effective as of the Merger Effective Date (the "<u>Effective Date</u>"), as such term is defined in that certain Business Company Agreement, dated the date hereof, by and among the Company, TGC Merger Sub, a Cayman Islands exempted company number 380624 and wholly owned subsidiary of the Company, and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability with company number 371049 (the "<u>Target</u>") (the "<u>Merger Agreement</u>"). In the event the Closing (as defined in the Merger Agreement) does not occur for any reason, this Agreement will be void *ab initio* and of no force or effect.

WHEREAS, the Executive possesses certain experience and expertise that qualifies the Executive to provide the direction and leadership required by the Company, including with respect to the Target and its Subsidiaries; and

WHEREAS, the Company desires to continue to employ the Executive as Chief Financial Officer of the Company following the Closing and the Executive wishes to accept such employment;

NOW, THEREFORE, in consideration of the mutual covenants contained herein and intending to be legally bound hereby, the Company and the Executive agree as follows:

# **1. EMPLOYMENT**

**1.1 Position**. Effective as of the Effective Date, the Executive will be employed by the Company, on a full-time basis, as its Chief Financial Officer. In addition, for so long as the Executive is employed by the Company, the Executive will serve as a director or officer of one or more of the Company's Subsidiaries as may be required from time to time, in each case, without further compensation. Effective immediately upon termination of the Executive's employment for any reason, the Executive will be deemed to have resigned from all such positions and offices the Executive may then hold at the Company or any of its Subsidiaries.

**1.2 Duties.** The Executive agrees to perform the duties of the Executive's position and such other duties as may reasonably be assigned to the Executive by the Board, Chief Executive Officer or other authorized officer of the Company from time to time. The Executive also agrees that, while employed by the Company, the Executive will devote the Executive's full business time and best efforts, business judgment, skill and knowledge to the advancement of the business interests of the Company and its Subsidiaries and to the discharge of the Executive's duties and responsibilities for them; provided, that the Executive's service as a director, trustee or committee member of civic or charitable organizations or of any for-profit organization which is not in the same competitive space as the Company or its Subsidiaries or for those entities with whom Executive has a pre-existing relationship, which are listed on Exhibit A hereto will not be in violation of the foregoing, in each case, to the extent such service does not interfere in any material respect with the effective discharge of the Executive's duties and responsibilities hereunder, create a conflict of interest or violate the Executive's obligations under <u>Section 3</u> hereof.

**1.3 Compliance with Policy**. The Executive agrees that, while employed by the Company, the Executive will comply with all applicable policies, practices and procedures, and codes of ethics or business conduct of the Company or any of its Subsidiaries, as in effect from time to time.

## 2. COMPENSATION AND BENEFITS

During the Executive's employment hereunder, as compensation for all services performed by the Executive for the Company and its Subsidiaries, the Company will provide the Executive the following compensation and benefits:

**2.1 Base Salary**. From and after the Effective Date, the Company will pay the Executive a base salary at the rate of US\$200,000 per year, payable in accordance with the ordinary payroll practices of the Company and subject to adjustment from time to time by the Board in its discretion (as adjusted from time to time, the "Base Salary").

**2.2 Annual Bonus**. The Executive will be eligible to earn an annual bonus (the "<u>Annual Bonus</u>") for each fiscal year completed during the Executive's employment hereunder, in a target amount of twenty-five percent (25%) of the Base Salary (for the first year, only with respect to the Base Salary payable from and after the Effective Date). The actual amount of any Annual Bonus will be determined by the Board in its discretion, based on financial, operational, individual and/or other targets established by the Board. In order to be eligible to receive any Annual Bonus hereunder, the Executive must be employed through the date such Annual Bonus is paid, except as otherwise provided in <u>Section 5.2</u>.

**2.3 Employee Benefits**. The Executive will be entitled to participate in all employee benefit plans as in effect from time to time for employees of the Company generally, except to the extent such plans are duplicative of benefits otherwise provided to the Executive under this Agreement. The Executive's participation will be subject to the terms of the applicable plan documents and generally applicable Company policies, as the same may be in effect from time to time, and any other restrictions or limitations imposed by law.

**2.4 Vacations**. The Executive will be eligible to accrue vacation time in accordance with the policies of the Company as in effect from time to time, which vacation time will be in addition to any holidays observed by the Company. Vacation may be taken at such times and intervals as the Executive determines, subject to the business needs of the Company. Vacation otherwise will be subject to the policies of the Company as in effect from time to time.

**2.5 Business Expenses**. The Company will pay or reimburse the Executive for all reasonable business expenses incurred or paid by the Executive in the performance of the Executive's duties and responsibilities hereunder, subject to any expense reimbursement policies of the Company as in effect from time to time and to such reasonable substantiation and documentation as may be specified by the Company from time to time. The Executive's right to payment or reimbursement hereunder will be subject to the following additional rules: (i) the amount of expenses eligible for payment or reimbursement during any calendar year will not affect the expenses eligible for payment or reimbursement will be made not later than December 31 of the calendar year following the calendar year in which the expense was incurred; and (iii) the right to payment or reimbursement will not be subject to liquidation or exchange for any other benefit.

# 3. CONFIDENTIAL INFORMATION AND RESTRICTED ACTIVITIES

**3.1 Confidential Information**. During the course of the Executive's employment with the Company, the Executive has learned and will continue to learn of Confidential Information and has developed and will continue to develop Confidential Information on behalf of the Company and its Subsidiaries. The Executive agrees not to use or disclose to any Person (except as required or permitted by applicable law or for the proper performance of the Executive's regular duties and responsibilities for the Company) any Confidential Information obtained by the Executive incident to the Executive's employment or any other association with the Company or any of its Subsidiaries (including, for the avoidance of doubt, any such information acquired with respect to the Executive's employment with SAITECH Limited or its Subsidiaries prior to the Effective Date). The Executive agrees that this restriction will continue to apply after the Executive's employment terminates, regardless of the reason for such termination. For the avoidance of doubt, (i) nothing contained in this Agreement limits, restricts or in any other way affects the Executive's communicating with any governmental agency or entity, and (ii) the Executive will not be held criminally or civilly liable under any federal, national, provincial, state or local trade secret law for disclosing a trade secret (A) in confidence to a federal, national, provincial state, or local government official, either directly or indirectly, or to an attorney, solely for the purpose of reporting or investigating a suspected violation of law, or (B) in a complaint or other document filed under seal in a lawsuit or other proceeding, provided, however, that notwithstanding this immunity from liability, the Executive may be held liable if the Executive unlawfully accesses or discloses trade secrets by unauthorized means.

**3.2 Protection of Documents**. All documents, records and files, in any media of whatever kind and description, relating to the business, present or otherwise, of the Company or any of its Subsidiaries, and any copies, in whole or in part, thereof (the "Documents"),

whether or not prepared by the Executive, will be the sole and exclusive property of the Company. The Executive agrees to safeguard all Documents and to surrender to the Company, at the time the Executive's employment terminates or at such earlier time or times as the Board or its designee may specify, all Documents then in the Executive's possession or control. The Executive also agrees to disclose to the Company, at the time the Executive's employment terminates or at such earlier time or times as the Board or its designee may specify, all passwords necessary to obtain access to any Confidential Information or Company information that the Executive has passwordprotected on any computer equipment, network or system of the Company or any of its Subsidiaries.

**3.3** Assignment of Rights to Intellectual Property. The Executive will promptly and fully disclose all Intellectual Property to the Company. The Executive hereby assigns and agrees to assign to the Company (or as otherwise directed by the Company) the Executive's full right, title and interest in and to all Intellectual Property. The Executive agrees to execute any and all applications for domestic and foreign patents, copyrights or other proprietary rights and to do such other acts (including, without limitation, the execution and delivery of instruments of further assurance or confirmation and the provision of good faith testimony in person or by declaration or affidavit) requested by the Company to assign the Intellectual Property to the Company (or as otherwise directed by the Company) and to permit the Company to secure, prosecute and enforce any patents, copyrights or other proprietary rights to the Intellectual Property. The Executive will not charge the Company or any of its Subsidiaries for time spent in complying with these obligations. All copyrightable works that the Executive creates during the Executive's employment will be considered "work made for hire" and will, upon creation, be owned exclusively by the Company.

**3.4 Restricted Activities**. The Executive agrees to the following restrictions on the Executive's activities during and, to the extent applicable, after the Executive's employment, and further agrees that such restrictions are necessary to protect the goodwill, Confidential Information, trade secrets and other legitimate interests of the Company and its Subsidiaries:

(a) *Non-Competition*. While the Executive is employed by the Company and for a period of twenty-four (24) months following the termination of the Executive's employment for any reason (collectively, the "<u>Restricted Period</u>"), the Executive will not, directly or indirectly, whether as an owner, partner, investor, consultant, agent, employee, co-venturer or otherwise, compete with, undertake any planning to compete with, or assist or encourage any other Person in competing with or undertaking any planning to compete with, the Company or any of its Subsidiaries. Specifically, but without limiting the foregoing, the Executive agrees not to work for or provide services to, in any capacity, whether with or without compensation, any Person that is engaged in any business anywhere that is competitive with the business of the Company or any of its Subsidiaries, as conducted or in active planning at any time during the Executive's employment with the Company. Notwithstanding the foregoing, in no event shall ownership of five percent (5%) or less of the outstanding securities of any class of any issuer whose securities are registered or listed on any securities exchange or recognized securities market anywhere in the world so long as the Executive does not have, or exercise, any rights to manage or operate the business of such issuer other than rights as an equity holder thereof.

(b) *Non-Solicitation of Business Partners*. During the Restricted Period, the Executive will not, directly or indirectly, and will not assist or encourage any other Person to, (i) solicit or encourage any customer, vendor, supplier or other business partner of the Company or any of its Subsidiaries to terminate, diminish or otherwise change in any manner adverse to the Company or any of its Subsidiaries his, her or its relationship with any of them; or (ii) seek to persuade any such customer, vendor, supplier or business partner, or any prospective customer, vendor, supplier or business partner of the Company or any of its Subsidiaries any business or activity that such Person conducts or could conduct with the Company or any of its Subsidiaries.

(c) Non-Solicitation of Employees and Other Service Providers. During the Restricted Period, except as required for the proper performance of the Executive's regular duties and responsibilities hereunder, the Executive will not, directly or indirectly, and will not assist or encourage any other Person to, hire or engage any employee of the Company or any of its Subsidiaries. While the Executive is employed by the Company, except as required for the proper performance of the Executive's regular duties and responsibilities hereunder, the Executive will not, directly or indirectly, and will not assist or encourage any employee of the Company or any of its Subsidiaries or encourage any employee of the Company or any of its Subsidiaries or seek to persuade any such employee to discontinue employment; or (ii) solicit or encourage any independent contractor providing services to the Company or any of its Subsidiaries to terminate, diminish or otherwise change in any manner adverse to the Company or any of its Subsidiaries his, her or its relationship with any of them.

3.5 Enforcement. In signing this Agreement, the Executive gives the Company assurance that the Executive has carefully read and considered all the terms and conditions of this Agreement, including the restraints imposed on the Executive under this Section 3, that the Executive has not relied on any agreements or representations, express or implied, that are not set forth expressly in this Agreement, and that the Executive has entered into this Agreement knowingly and voluntarily. The Executive agrees that these restraints are necessary for the reasonable and proper protection of the Company and its Subsidiaries, and that each and every one of the restraints is reasonable in respect to subject matter, length of time and geographic area. The Executive further agrees that, were the Executive to breach any of the covenants contained in this Section 3, the damage to the Company and its Subsidiaries would be irreparable. The Executive therefore agrees that the Company, in addition and not in the alternative to any other remedies available to it, will be entitled to preliminary and permanent injunctive relief against any breach or threatened breach by the Executive of any such covenants, without having to post bond. In the event that any provision of this Section 3 is determined by any court of competent jurisdiction to be unenforceable by reason of its being extended over too great a time, too large a geographic area or too great a range of activities, that provision will be deemed to be modified to permit its enforcement to the maximum extent permitted by law. It is also agreed that each of the Company's Subsidiaries will have the right to enforce all of the Executive's obligations to that Subsidiary under this Agreement, including, without limitation, pursuant to this Section 3. No claimed breach of this Agreement or other violation of law attributed to the Company or any of its Subsidiaries, or change in the nature or scope of the Executive's employment or other relationship with the Company or any of its Subsidiaries, will operate to excuse the Executive from the performance of the Executive's obligations under this Section 3.

## 4. TERMINATION OF EMPLOYMENT

The Executive's employment under this Agreement will continue until terminated pursuant to this Section 4.

**4.1 In General**. The Executive's employment may be terminated by the Company or by the Executive at any time upon sixty (60) days' prior written notice to the other party; <u>provided</u>, <u>however</u>, that the Board may elect to waive such notice period or any portion thereof, in which case the Company will continue to pay the Base Salary for that portion of the notice period so waived.

**4.2 Termination by the Company for Cause**. Notwithstanding the provisions of <u>Section 4.1</u>, the Company may terminate the Executive's employment immediately for Cause upon written notice to the Executive setting forth in reasonable detail the nature of the Cause. For purposes of this Agreement, "<u>Cause</u>" means the occurrence of any of the following, as determined by the Board in its reasonable judgment: (i) the Executive's material failure to perform (other than by reason of disability), or substantial misconduct in the performance of, the Executive's duties and responsibilities for the Company or any of its Subsidiaries which causes material harm to the Company; (ii) the Executive's material and demonstrable breach of any provision of <u>Section 3</u> or of any other confidentiality, invention assignment, or other restrictive covenant obligation set forth in any written agreement by and between the Executive and the Company or any of its Subsidiaries; (iii) the Executive's material and demonstrable breach of any other provision of this Agreement or any other written agreement by and between the Executive's material and demonstrable breach of any other provision of this Agreement or any other written agreement by and between the Executive's material wield the Company or any of its Subsidiaries; (iv) the Executive's material violation of any applicable policy or code of conduct of the Company or any of its Subsidiaries which violation causes material reputational or financial harm to the Company; or (v) the Executive's indictment for, or plea of nolo contendere to, any felony or any crime involving moral turpitude. Notwithstanding anything to the contrary in the foregoing, a circumstance otherwise giving rise to Cause pursuant to the foregoing clause (i), (ii), (iii) or (iv), if capable of cure, will not constitute Cause if cured by the Executive within ten (10) days following the Company's notice to the Executive thereof; <u>provided</u>, <u>however</u>, that the Company will not be required to provide any such notice or opp

**4.3 Resignation by the Executive for Good Reason**. Notwithstanding the provisions of <u>Section 4.1</u>, the Executive may terminate the Executive's employment for Good Reason upon written notice to the Company setting forth in reasonable detail the nature of the circumstances constituting Good Reason. For purposes of this Agreement, "<u>Good Reason</u>" means the occurrence of any of the following without the Executive's consent: (i) a material reduction in Base Salary, other than an across-the-board reduction applicable to similarly situated executives of the Company; (ii) a permanent relocation of the Executive's principal place of business that increases the Executives commute by more than sixty (60) miles in a single direction; or (iii) a material diminution of Executive's duties, authorities or responsibilities; <u>provided</u>, in each case, that (x) the Executive provides the Company with written notice of the circumstance constituting Good Reason within twenty (20) days following the Executive's first knowledge thereof, (y) the Company fails to cure such circumstance

within twenty (20) days following the receipt of such notice and (z) the Executive actually terminates employment within twenty (20) days following the expiration of such cure period.

**4.4 Death and Disability**. The Executive's employment will terminate automatically in the event of the Executive's death during employment. The Company may terminate the Executive's employment, upon notice to the Executive, in the event that the Executive becomes disabled during the Executive's employment hereunder through any illness, injury, accident or condition of either a physical or psychological nature and, as a result, is unable to perform substantially all of the Executive's duties and responsibilities for the Company and its Subsidiaries (notwithstanding the provision of any reasonable accommodation) for a period of ninety (90) days during any period of three hundred sixty-five (365) consecutive days. If any question arises as to whether the Executive is disabled to the extent that the Executive is unable to perform substantially all of the Executive's duties hereunder, the Executive will, at the Company's request, submit to a medical examination by a physician selected by the Company, and to whom the Executive or the Executive's guardian, if any, has no reasonable objection, to determine whether the Executive is so disabled, and such determination will be conclusive of the issue for purposes of this Agreement. If such a question arises and the Executive fails to submit to the requested medical examination, the Company's good faith, reasonable determination of the issue will be binding on the Executive.

Ex A-5

#### 5. OTHER MATTERS RELATED TO TERMINATION

**5.1 Final Compensation**. In the event of termination of the Executive's employment with the Company, howsoever occurring, the Company will pay the Executive (i) the Base Salary for the final payroll period of the Executive's employment, through the date the Executive's employment terminates (the "Termination Date"); (ii) compensation at the rate of the Base Salary for any vacation time accrued in accordance with the policies of the Company but not used as of the Termination Date; and (iii) reimbursement, in accordance with Section 2.5 hereof, for business expenses incurred by the Executive but not yet paid to the Executive as of the Termination Date, provided that the Executive submits all expenses and supporting documentation required within sixty (60) days of the Termination Date and that such expenses are reimbursable under Company policies then in effect (all of the foregoing, the "Final Compensation"). Except as otherwise provided in the foregoing clause (iii), the Final Compensation will be paid to the Executive immediately upon termination of employment.

**5.2 Severance Payments.** In the event the Executive's employment is terminated by the Company without Cause or by the Executive for Good Reason, subject to the provisions of <u>Section 5.3</u>, the Company will pay the Executive, in addition to the Final Compensation, severance payments as provided in this <u>Section 5.2</u> (collectively, the "<u>Severance Payments</u>").

(a) *Termination Other than in Connection with a Change of Control*. Except as otherwise provided in <u>Section 5.2(b)</u>, the Severance Payments will include (i) payment in an amount equal to twelve (12) months' Base Salary, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company over the twelve (12) months following the Termination Date; (ii) payment of any earned but unpaid Annual Bonus for the fiscal year preceding the year in which the termination occurs, payable at the same time bonuses otherwise are paid to active employees of the Company; and (iii) payment of premiums for continued health benefits under the Company's health plans for Executive and his dependants, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company until the earlier of (A) twelve (12) months following the Termination Date and (B) the first date on which the Executive becomes eligible to receive health insurance coverage from another employer.

(b) Termination in Connection with a Change of Control. Notwithstanding the provisions of Section 5.2(a), in the event such termination occurs within the three (3) months prior to or twelve (12) months following the date of a Change of Control (and in the case of a termination by the Company without Cause that occurs prior to such Change of Control, solely to the extent such termination results from the request of another party to the Change of Control transaction), the Severance Payments will include (i) payment in an amount equal to (A) fifteen (15) months' Base Salary plus (B) the target amount of the Annual Bonus for the fiscal year in which the termination occurs, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company over the fifteen (15) months following the Termination Date; (ii) payment of any earned but unpaid Annual Bonus for the fiscal year preceding the year in which the termination occurs, payable at the same time bonuses otherwise are paid to active employees of the Company; (iii) payment of premiums for continued health benefits under the Company's health plans for Executive and his dependants, payable in the form of salary continuation in accordance with the ordinary payroll practices of the Company until the earlier of (A) fifteen (15) months following the Termination Date; with the ordinary payroll practices of the Company until the earlier of (A) fifteen (15) months following the Termination Date and (B) the first date on which the Executive becomes eligible to receive health insurance coverage from another employee;

and (iv) notwithstanding anything to the contrary in the Company's incentive equity plans or any applicable award agreement, the Executive's equity awards under such plans that are outstanding and unvested as of the Termination Date will become fully vested effective as of the Release Date (as defined below).

**5.3 Conditions to and Timing of the Severance Payments**. Any obligation of the Company to provide the Severance Payments is conditioned on the Executive signing and returning to the Company, without revocation, a timely and effective separation agreement containing a general release of claims and other customary terms in a form reasonably satisfactory to the Company (the "<u>Release</u>"). The Release must become effective, if at all, by the sixtieth (60th) calendar day following the Termination Date (the date the Release so becomes effective, the "<u>Release Date</u>"). Provided the foregoing conditions are satisfied, the first installment of the Severance Payments will be paid on the Company's next regular payroll date at least five (5) business days following the Release Date, the amount of which will be retroactive to the Termination Date.

**5.4 Benefits Termination**. Except for any right the Executive may have under COBRA or other applicable law to continue participation in the Company's group health and dental plans at the Executive's own cost, the Executive's participation in all employee benefit plans will terminate in accordance with the terms of the applicable benefit plans based on the Termination Date, without regard to any continuation of the Base Salary or other payment to the Executive following such termination, and the Executive will not be eligible to earn vacation or other paid time off following such termination.

**5.5 Survival.** The provisions of this Agreement will survive any termination of employment if so provided in this Agreement or if necessary or desirable to accomplish the purposes of other surviving provisions, including, without limitation, the Executive's obligations under <u>Section 3</u>. Upon termination by either the Executive or the Company, all rights, duties and obligations of the Executive and the Company to each other will cease, except as otherwise expressly provided in this Agreement.

# 6. CERTAIN DEFINITIONS

For purposes of this Agreement, the following definitions apply:

"Board" means the board of directors of the Company or any committee thereof, as applicable.

"Change of Control" has the meaning ascribed to such term in the Company's 2021 Equity Incentive Plan.

Ex A-7

"Code" means the Internal Revenue Code of 1986, as amended, together with the regulations and guidance promulgated thereunder.

"<u>Confidential Information</u>" means any and all information of the Company and its Subsidiaries that is not generally available to the public. Confidential Information also includes any information received by the Company or any of its Subsidiaries from any Person with any understanding, express or implied, that it will not be disclosed. Confidential Information does not include information that enters the public domain other than through the Executive's breach of the Executive's obligations under this Agreement or any other agreement between the Executive and the Company or any of its Subsidiaries.

"Intellectual Property" means all Inventions conceived, made, created, developed or reduced to practice by the Executive (whether alone or with others, whether or not during normal business hours or on or off Company premises) during the Executive's employment that relate either to the business of the Company or any of its Subsidiaries or to any prospective activity of the Company or any of its Subsidiaries or that result from any work performed by the Executive for the Company or any of its Subsidiaries or that make use of Confidential Information or any of the equipment or facilities of the Company or any of its Subsidiaries. "Invention" means any invention, discovery, design, development, improvement, method, process, procedure, plan, project, system, technique, strategy, information, composition, know-how, work, concept or idea, or any modification or derivative of any of the foregoing (whether or not patentable or copyrightable or constituting a trade secret).

"<u>Person</u>" means an individual, a corporation, a limited liability company, an association, a partnership, an estate, a trust or any other entity or organization, other than the Company or any of its Subsidiaries.

"Subsidiary" means, with respect to any Person, any corporation, partnership, association or other business entity of which (i) if a corporation, a majority of the total voting power of capital shares entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof, or (ii) if a partnership, association or other business entity, a majority of the partnership or other similar ownership interests thereof is at the time owned or controlled, directly or indirectly, by any Person or one or more Subsidiaries of that Person or a combination thereof. For purposes hereof, a Person or Persons will be deemed to have a majority ownership interest in a partnership, association or other business entity gains or losses or will be or control the managing director, managing member, general partner or other managing Person of such partnership, association or other business entity also include any variable interest entity which is consolidated with such Person under applicable accounting rules.

Ex A-8

## 7. NO CONFLICTING AGREEMENTS

The Executive hereby represents and warrants that the signing of this Agreement and the performance of the Executive's obligations hereunder will not breach or be in conflict with any other agreement to which the Executive is a party or by which the Executive is bound, and that the Executive is not now subject to any covenants against competition or similar covenants or any court order that could affect the performance of the Executive's obligations hereunder. The Executive agrees that the Executive will not disclose to or use on behalf of the Company any confidential or proprietary information of a third party without that party's consent.

# 8. MISCELLANEOUS

**8.1 Notices**. Any notices provided for in this Agreement will be in writing and will be effective when delivered in person or sent by the United States mail, postage prepaid, return receipt requested, or by a reputable overnight delivery service, and addressed to the Executive at the Executive's last known address on the books of the Company or, in the case of the Company, to it at its principal place of business, attention of the Chair of the Board, or to such other address as either party may specify by notice to the other actually received.

**8.2 Withholding**. All payments made by the Company under this Agreement will be reduced by any tax or other amounts required to be withheld by the Company to the extent required by applicable law.

**8.3** Assignment. Neither the Executive nor the Company may make any assignment of this Agreement or any interest in it, by operation of law or otherwise, without the prior written consent of the other; <u>provided</u>, <u>however</u>, that the Company may assign its rights and obligations under this Agreement without the Executive's consent to one of its Subsidiaries (for the avoidance of doubt, following the Closing, including Target) or to any Person with whom the Company hereafter effects a reorganization, consolidation or merger, or to whom the Company hereafter transfers all or substantially all of its properties or assets. This Agreement will inure to the benefit of and be binding upon the Executive and the Company and each of their respective successors, executors, administrators, heirs and permitted assigns.

**8.4 Severability.** If any portion or provision of this Agreement is declared illegal or unenforceable to any extent by a court of competent jurisdiction, to any extent, then the remainder of this Agreement, or the application of such portion or provision in circumstances other than those as to which it is so declared illegal or unenforceable, will not be affected thereby, and each portion and provision of this Agreement will be valid and enforceable to the fullest extent permitted by law.

**8.5 Other Matters**. This Agreement sets forth the entire agreement and understanding between the parties hereto relating to the subject matter hereof, and replaces all prior and contemporaneous communications, agreements and understandings, written or oral, relating to the same. This Agreement may not be modified or amended, and no breach will be deemed to be waived, unless agreed to in writing by the Executive and an expressly authorized representative of the Board. The headings and captions in this Agreement are for

convenience only and in no way define or describe the scope or content of any provision herein. This Agreement may be executed in counterparts (and may be delivered by email or other electronic means), each of which will be an original and all of which together will constitute one and the same instrument. This Agreement will be governed by and construed in accordance with the laws of the State of Delaware, without regard to any conflict of laws principles that would result in the application of the laws of any other jurisdiction. The Executive and the Company agree to submit to the exclusive jurisdiction of the courts of or in the State of Delaware in connection with any dispute arising out of or otherwise related to this Agreement or the Executive's employment with the Company, in whole or in part, and agree that any such dispute will be brought and maintained solely in such courts.

[Remainder of Page Intentionally Left Blank]

Ex A-9

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

TRADEUP GLOBAL CORPORATION

By: /s/ Huang Lei

Name: Hung Lei Title: Co-Chief Executive Officer

Accepted and agreed:

/s/ Jian Zou

Jian Zou

[Signature Page to Employment Agreement]

Ex A-10

Exhibit A

Certain Permitted Activities

NONE

[Exhibit A]

Ex A-11

# <u>Exhibit F</u>

#### Form of Registration Rights Agreement

**Final Form** 

FORM OF REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "*Agreement*") is made and entered into as of [\_], 2021, by and among: (i) SAI.TECH Global Corporation (f/k/a TradeUP Global Corporation), a Cayman Islands exempted company (together with its successors, "*Purchaser*"), and (ii) the undersigned parties listed as "Investors" on the signature page hereto (each, an "*Investor*" and collectively, the "*Investors*").

WHEREAS, on September 27, 2021, (i) Purchaser, (ii) TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability and a direct wholly owned subsidiary of Purchaser ("Merger Sub"), and (iii) SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability (the "Company"), entered into that certain Business Combination Agreement (as may be amended, restated or supplemented from time to time, the "Business Combination Agreement"), pursuant to which, among other things, Merger Sub will merge with and into the Company (the "Merger" and the closing of the Merger, the "Closing"), with the Company being the surviving entity and becoming a wholly owned subsidiary of Purchaser;

WHEREAS, Purchaser and TradeUP Global Sponsor LLC, a Cayman Islands exempted limited liability company (the "Sponsor"), and certain other shareholders (collectively with Sponsor, the "Original Holders") are parties to that certain Registration Rights Agreement, dated as of April 28, 2021 (the "Original Registration Rights Agreement"), which shall be superseded by this Agreement and shall terminate upon the effectiveness hereof at the Closing;

WHEREAS, in connection with the execution of the Business Combination Agreement, certain of the Investors have entered into, or will prior to the Closing enter into, certain other agreements with Purchaser, pursuant to which, among other things, such Investor has agreed not to transfer the merger consideration held by such Investor for a certain period of time after the Closing (each such agreement, as amended from time to time in accordance with the terms thereof, a "Lock-Up Agreement"), in each case pursuant to the terms of such Lock-Up Agreement;

WHEREAS, in connection with the execution of the Business Combination Agreement, Purchaser and the Investors desire to enter into this Agreement, which shall replace the Original Registration Rights Agreement, in order to provide the Investors with registration rights on the terms set forth herein; and

WHEREAS, the parties desire for this Agreement to be effective only upon the consummation of the Closing.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. <u>Definitions</u>. Any capitalized term used but not defined in this Agreement will have the meaning ascribed to such term in the Business Combination Agreement. The following capitalized terms used herein have the following meanings:

"Agreement" is defined in the preamble of this Agreement.

"Business Combination Agreement" is defined in the recitals to this Agreement.

"Business Day" means any day other than a Saturday, Sunday or a legal holiday on which commercial banking institutions in New York, New York, Hong Kong or the Cayman Islands are authorized to close for business.

"Closing" is defined in the recitals to this Agreement.

"Company" is defined in the recitals to this Agreement.

"Concurrent Offering" shall have the meaning given in Section 2.1.2.

"Concurrent Purchaser Offering" shall have the meaning given in Section 2.1.2.

Ex F-1

"Concurrent Secondary Offering" shall have the meaning given in Section 2.1.2.

"Demanding Holders" shall have the meaning given in Section 2.1.1(b).

"Effectiveness Date" shall have the meaning given in Section 2.1.1(a).

*"Exchange Act"* means the Securities Exchange Act of 1934, as amended, and unless the context requires otherwise, the rules and regulations of the SEC promulgated thereunder, all as the same shall be in effect at the time.

*"Filing Date"* shall have the meaning given in <u>Section 2.1.1(a)</u>.

*"Founder Shares*" shall mean the 1,122,247 Purchaser Class A Shares issued upon conversion of prior Class B ordinary shares of the Purchaser (as designated as such prior to the Merger (the "*Prior Class B Shares*") prior to or in connection with the consummation of the Merger (including (i) 850,000 Purchaser Class A Shares issued to the Sponsor prior to the date hereof upon conversion of Prior Class B Shares, (ii) 212,247 issued to the Sponsor in connection with the Merger upon conversion of Prior Class B Shares, and (iii) 60,000 Purchaser Class A Shares in connection with the Merger upon conversion of Prior Class B Shares, and (iii) 60,000 Purchaser Class A Shares in connection with the Merger upon conversion of Prior Class B Shares, and (iii) 60,000 Purchaser Class A Shares in connection with the Merger upon conversion of Prior Class B Shares A Shares issuable upon conversion of Prior Class B Shares. For purposes of clarification, Founder Shares do not include any Purchaser Class A Shares issuable upon conversion of Purchaser Class B Shares (as defined in the Business Combination Agreement).

"Indemnified Party" is defined in Section 4.3.

"Indemnifying Party" is defined in Section 4.3.

"Investor Indemnified Party" is defined in Section 4.1.

*"Investor(s)"* is defined in the preamble to this Agreement, and includes any transferee of the Registrable Securities (so long as they remain Registrable Securities) of an Investor permitted under this Agreement and with respect to an Investor, its applicable Lock-Up Agreement.

"Lock-Up Agreement" is defined in the recitals to this Agreement.

"Maximum Number of Securities" is defined in Section 2.1.2.

"Merger Sub" is defined in the recitals to this Agreement.

"Merger" is defined in the recitals to this Agreement.

"Minimum Demand Threshold" shall mean \$10,000,000.

"Original Holders" is defined in the recitals to this Agreement.

"Original Registration Rights Agreement" is defined in the recitals to this Agreement.

"Other Selling Investors" shall have the meaning given in Section 2.1.1(b).

"Piggyback Registration" is defined in Section 2.2.1.

"Pro Rata" is defined in Section 2.1.2.

"Proceeding" is defined in Section 6.9.

"Purchaser" is defined in the recitals to this Agreement.

*"Register," "Registered"* and *"Registration"* mean a registration or offering effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

Ex F-2

"Registrable Securities" means (a) the Founder Shares (including any Purchaser Class A Shares or other equivalent equity security issued or issuable upon the conversion of any Founder Shares or exercisable for Purchaser Class A Shares), (b) the Purchaser Private Shares, (c) the Working Capital Shares and (d) the Ordinary Shares, in each case, set forth on the Schedule to this Agreement, including Ordinary Shares (i) issued or issuable to any Investor in exchange for shares of Purchaser pursuant to the Business Combination Agreement, (ii) issued or issuable pursuant to the exercise of the Purchaser Warrants, (iii) issued or issuable upon conversion of Purchaser Class B Ordinary Shares and (iv) all Purchaser Class A Shares issued or issuable to any holder with respect to such securities by way of a share capitalization or share split, or in connection with a combination of shares, recapitalization, share exchange, share reconstruction, amalgamation, contractual control arrangement or similar event. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when: (a) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (b) such securities shall have been otherwise transferred, new certificates (or evidence of book entry position) for them not bearing a legend restricting further transfer shall have been delivered by Purchaser and subsequent public distribution of them shall not require registration under the Securities Act; (c) such securities shall have ceased to be outstanding; or (d) such securities are freely saleable under Rule 144 without volume limitations or the requirement for Purchaser to be current in its Exchange Act reporting. Notwithstanding anything to the contrary contained herein, a Person shall be deemed to be an "Investor holding Registrable Securities" (or words to that effect) under this Agreement only if they are an Investor or a transferee of the applicable Registrable Securities (so long as they remain Registrable Securities) of any Investor permitted under this Agreement and any applicable Lock-Up Agreement.

*"Registration Statement"* means a registration statement filed by Purchaser with the SEC in compliance with the Securities Act for a public offering and sale of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities (other than a registration statement on Form S-4, F-4 or Form S-8, or their successors, or any registration statement covering only securities proposed to be issued in exchange for securities or assets of another entity).

"Rule 144" means Rule 144 promulgated under the Securities Act.

"SEC" means the United States Securities and Exchange Commission or any successor thereto.

"Securities Act" means the Securities Act of 1933, as amended, and unless the context requires otherwise, the rules and regulations of the SEC promulgated thereunder, all as the same shall be in effect at the time.

"Shelf Registrable Securities" shall have the meaning given in Section 2.1.1(b).

*"Shelf Registration Statement"* shall have the meaning given in <u>Section 2.1.1(a)</u>.

"Shelf Underwriting" shall have the meaning given in Section 2.1.1(b).

"Shelf Underwriting Notice" shall have the meaning given in Section 2.1.1(b).

"Shelf Underwriting Request" shall have the meaning given in Section 2.1.1(b).

"Specified Courts" is defined in Section 6.9.

"Sponsor" is defined in the recitals to this Agreement.

"Underwritten Block Trade" shall have the meaning give in Section 2.1.1(b).

"Underwritten Offering" shall mean a Registration in which securities of the Purchaser are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

"Underwriter" means a securities dealer who purchases any Registrable Securities as principal in an underwritten offering and not as part of such dealer's market-making activities.

*"Working Capital Shares"* means any loan by the Sponsor or certain of Purchaser's officers and directors to Purchaser of funds as Purchaser may require, of which up to \$1,200,000 of such loans may be convertible into Purchaser Class A Shares at a price of \$10.00 per share.

# 2. Registration Rights.

## 2.1 Demand Registration.

#### 2.1.1 Shelf Registration.

(a) As soon as practicable but no later than 30 calendar days after the Closing Date (the "Filing Date"), Purchaser shall prepare and file with (or confidentially submit to) the SEC a shelf registration statement under Rule 415 of the Securities Act (such registration statement, a "Shelf Registration Statement") covering the resale of all the Registrable Securities (determined as of two Business Days prior to such filing) on a delayed or continuous basis and shall use its commercially reasonable efforts to have such Shelf Registration Statement declared effective as soon as practicable after the filing thereof and no later than the earlier of (x) the 90th Business Day following the Filing Date if the SEC notifies Purchaser that it will "review" such Shelf Registration Statement) and (y) the tenth Business Day after the date Purchaser is notified (orally or in writing, whichever is earlier) by the SEC that such Shelf Registration Statement will not be "reviewed" or will not be subject to further review (such earlier date, the "Effectiveness Date"). Such Shelf Registration Statement shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available to, and requested by, any Investor named therein. Purchaser shall maintain the Shelf Registration Statement in accordance with the terms hereof, and shall prepare and file with the SEC such amendments, including posteffective amendments, and supplements as may be necessary to keep a Shelf Registration Statement continuously effective, available for use to permit all Investors named therein to sell their Registrable Securities included therein and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. If at any time Purchaser shall have qualified for the use of a Registration Statement on Form S-3, Form F-3 or any other form that permits incorporation of substantial information by reference to other documents filed by Purchaser with the SEC and at such time Purchaser has an outstanding Shelf Registration Statement on Form S-1 or Form F-1, then Purchaser shall use its commercially reasonably efforts to convert such outstanding Shelf Registration Statement on Form S-1 or F-1 into a Shelf Registration Statement on Form S-3 or Form F-3, as applicable.

(b) Subject to any applicable Lock-up Agreement, an Investor or Investors (the "Demanding Holders"), may make a written demand (a "Shelf Underwriting Request") from time to time to elect to offer for sale all or any part of their Registrable Securities, with a total offering price reasonably expected to exceed, in the aggregate, the Minimum Demand Threshold, pursuant to an Underwritten Offering pursuant to the Shelf Registration Statement, which written demand shall describe the amount and type of securities to be included in such Underwritten Offering and the intended method(s) of distribution thereof (the "Shelf Underwriting"). As promptly as practicable, but no later than five Business Days after receipt of a Shelf Underwriting Request, Purchaser shall give written notice (the "Shelf Underwriting Notice") of such Shelf Underwriting Request to the Investors holding other Registrable Securities registered on such Shelf Registration Statement ("Shelf Registrable Securities"). Purchaser and the Demanding Holders, subject to Section 2.1.2, shall include in such Shelf Underwriting (x) the Registrable Securities of the Demanding Holders and (y) the Shelf Registrable Securities of any other Investor of Shelf Registrable Securities which shall have made a written request to Purchaser for inclusion in such Shelf Underwriting (which request shall specify the maximum number of Shelf Registrable Securities intended to be disposed of by such Investor) within ten days after the receipt of the Shelf Underwriting Notice ("Other Selling Investors"). Purchaser shall, as expeditiously as possible (and in any event within 20 Business Days after the receipt of a Shelf Underwriting Request), file any prospectus supplement or, if the applicable Shelf Registration Statement is an automatic shelf registration statement, any post-effective amendments and otherwise take any action necessary to include therein all disclosure and language deemed necessary or advisable by the Demanding Holders or the Other Selling Investors to effect such Shelf Underwriting. Once a Shelf Registration Statement has been declared effective, Investors may request, and Purchaser shall be required to facilitate, an aggregate of four Shelf Underwritings pursuant to this Section 2.1.1(b) with respect to any or all Registrable Securities in any 12 month period; provided, however, that a Shelf Underwriting shall not be counted for such purposes unless all of the Registrable Securities requested by the Demanding Holders and the Other Selling Investors to be included in such Shelf Underwriting have been sold. Notwithstanding the foregoing, if a Demanding Holder wishes to engage in an underwritten block trade or similar transaction or other transaction with a 2-day or less marketing period (collectively, "Underwritten Block Trade") pursuant to a Shelf Registration Statement, then notwithstanding the foregoing time periods, such Demanding Holder only needs to notify Purchaser of the Underwritten Block Trade two Business Days prior to the day such offering is to commence and the other Investors shall not be entitled to notice of such Underwritten Block Trade and shall not be entitled to participate in such Underwritten Block Trade.

(c) Notwithstanding anything to the contrary in Sections 2.1.1(a) and 2.1.1(b), the rights granted in this Section 2.1.1 are subject to the following limitations: (i) Purchaser shall not be required to effect more than two Shelf Underwritings (including one Shelf Underwriting Request which shall be reserved for the Sponsor so long as the Sponsor owns Registrable Securities representing at least the Minimum Demand Threshold and has not exercised any Shelf Underwriting Request) for so long as the Shelf Registration Statement is on Form S-1, Form F-1 or any similar long-form registration statement at the request of the Investors in the aggregate; and (ii) if the Purchaser's board of directors, in its good faith judgment, determines that any registration of Registrable Securities or Shelf Underwriting should not be made or continued because it would materially and adversely interfere with any existing or potential financing, acquisition, corporate reorganization, merger, share exchange or other transaction or event involving Purchaser or any of its subsidiaries or would otherwise result in the public disclosure of information that the Purchaser's board of directors in good faith has a bona fide business purpose for keeping confidential (a "Valid Business Reason"), then (x) Purchaser may postpone filing or confidentially submitting the Shelf Registration Statement or a prospectus supplement relating to a Shelf Underwriting Request until five Business Days after such Valid Business Reason no longer exists, but in no event for more than 45 days after the date Purchaser's board of directors determines a Valid Business Reason exists or (y) if the Shelf Registration Statement has been filed or confidentially submitted or a prospectus supplement has been filed relating to a Shelf Underwriting Request, if the Valid Business Reason has not resulted in whole or in part from actions taken or omitted to be taken by Purchaser (other than actions taken or omitted with the consent of the Demanding Holder (not to be unreasonably withheld or delayed)), Purchaser may, to the extent determined in the good faith judgment of the Purchaser's board of directors to be reasonably necessary to avoid interference with any of the transactions described above, suspend use of or, if required by the SEC, cause such Shelf Registration Statement to be withdrawn and its effectiveness terminated or may postpone amending or supplementing such registration statement until five Business Days after such Valid Business Reason no longer exists, but in no event for more than 45 days after the date Purchaser's board of directors determines a Valid Business Reason exists (such period of postponement or withdrawal under this clause (ii), the "Postponement Period"). The Company shall give written notice to the Investors of its determination to postpone or suspend use of or withdraw the Shelf Registration Statement and of the fact that the Valid Business Reason for such postponement or suspension or withdrawal no longer exists, in each case, promptly after the occurrence thereof; provided, however, that Purchaser shall not be entitled to more than two Postponement Periods during any 12 month period.

#### Ex F-5

Each Investor holding Registrable Securities agrees that, upon receipt of any notice from Purchaser that Purchaser has determined to suspend use of, withdraw, terminate or postpone amending or supplementing the Shelf Registration Statement pursuant to clause (ii) above, such Investor will discontinue its disposition of Registrable Securities pursuant to the Shelf Registration Statement (including pursuant to a Shelf Underwriting). If Purchaser shall give any notice of suspension, withdrawal or postponement of the Shelf Registration Statement, Purchaser shall, not later than five Business Days after the Valid Business Reason that caused such suspension, withdrawal or postponement no longer exists (but, with respect to a suspension, withdrawal or postponement pursuant to clause (c)(iii) above, in no event later than 45 days after the date of the suspension, postponement or withdrawal), as applicable, permit use of the Shelf Registration Statement or use its reasonable best efforts to effect the registration under the Securities Act of the Registrable Securities covered by the withdrawn or postponed registration statement in accordance with this Section 2.1 (and any new registration statement shall thereupon be the "Shelf Registration Statement" for all purposes hereof), and following such permission or such effectiveness such registration shall no longer be deemed to be suspended, withdrawn or postponed pursuant to clause (ii) above. An Investor may deliver written notice (an "Opt-Out Notice") to Purchaser requesting that such Investor not receive notices from Purchaser otherwise required by Section 2.1.1(c); provided, however, that such Investor may later revoke any such Opt-Out Notice in writing. Following receipt of an Opt-Out Notice from an Investor (unless subsequently revoked), (i) Purchaser shall not deliver any such notices to such Investor and such Investor shall no longer be entitled to the rights associated with any such notice and (ii) each time prior to such Investor's intended use of an effective Shelf Registration Statement, such Investor will notify Purchaser in writing at least two Business Days in advance of such intended use, and if a notice of a Postponement Period was previously delivered and the related suspension period remains in effect, Purchaser will so notify such Investor, within one Business Day of such Investor's notification to Purchaser, by delivering to such Investor a copy of such previous notice of the Postponement Period Event, and thereafter will provide such Investor with the related notice of the conclusion of such Postponement Period as required hereby.

2.1.2 <u>Reduction of Offering</u>. If the managing Underwriter or Underwriters for a Shelf Underwriting advises Purchaser and the Demanding Holders in writing that the dollar amount or number of Registrable Securities that the Demanding Holders and the Other Selling Investors (if any) desire to sell, taken together with all other Purchaser Class A Shares or other securities which Purchaser desires to sell concurrently (a "*Concurrent Purchaser Offering*") and Purchaser Class A Shares or other securities, if any, as to which an Underwritten Offering pursuant to a Registration by Purchaser has been requested concurrently pursuant to written contractual registration rights held by other security holders of Purchaser (a "*Concurrent Secondary Offering*" and together with Concurrent Purchaser Offering, a "Concurrent Offering"), exceeds the maximum dollar amount or maximum number of shares that can be sold in such offerings without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of securities, as applicable, the "Maximum Number of Securities"), then the following shall be included in such Shelf Underwriting and any Concurrent Offering: (i) first, the Registrable Securities as to which the Demanding Holders have requested the Shelf Underwriting (pro rata in accordance with the number of securities that each applicable Person has requested be included in such registration, regardless of the number of securities held by each such Person (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities that Other Selling Investors have requested be included in such Shelf Underwriting (Pro Rata in accordance with the number of securities that each applicable Person has requested be included in such registration) that can be sold without exceeding the Maximum Number of Securities; (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), Purchaser Class A Shares or other securities Purchaser desires to sell in the Concurrent Purchaser Offering that can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), Purchaser Class A Shares or other securities, if any, to be included in the Concurrent Secondary Offering (Pro Rata in accordance with the number of securities that each applicable Person has requested be included in such registration) that can be sold without exceeding the Maximum Number of Securities. In the event that Purchaser securities that are convertible into Purchaser Class A Shares are included in the offering, the calculations under this Section 2.1.2 shall include such Purchaser securities on an as-converted to Purchaser Class A Share basis.

#### Ex F-6

2.1.3 <u>Withdrawal</u>. Any Investor shall have the right in its sole discretion to withdraw from a Shelf Underwriting upon written notification to Purchaser and the Underwriter or Underwriters (if any) of their intention to withdraw prior to the filing of a preliminary prospectus supplement setting forth the terms of the Shelf Underwriting with the SEC; *provided*, *however*, if after giving effect to all such withdrawals, the total offering price is not reasonably expected to exceed, in the aggregate, the Minimum Demand Threshold, Purchaser shall not be required to proceed with such Shelf Underwriting.

#### 2.2 Piggyback Registration.

2.2.1 Piggyback Rights. If at any time after the Closing, Purchaser proposes to file a Registration Statement under the Securities Act with respect to the Registration of or an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, by Purchaser for its own account or for security holders of Purchaser for their account (or by Purchaser and by security holders of Purchaser), other than the Shelf Registration Statement to be filed pursuant to Section 2.1.1, and other than a Registration Statement (i) filed in connection with any employee share option or other benefit plan, (ii) for an exchange offer or offering of securities solely to Purchaser's existing security holders, (iii) for an offering of debt that is convertible into equity securities of Purchaser, or (iv) for a dividend reinvestment plan, then Purchaser shall (x) give written notice of such proposed filing to Investors holding Registrable Securities as soon as practicable but in no event less than ten days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering or registration, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, of the offering, and (y) offer to Investors holding Registrable Securities in such notice the opportunity to register the sale of such number of Registrable Securities as such Investors may request in writing within five days following receipt of such notice (a "Piggyback Registration"). To the extent permitted by applicable securities laws with respect to such registration by Purchaser or another demanding security holder, Purchaser shall use its reasonable best efforts to cause (i) such Registrable Securities to be included in such registration and (ii) the managing Underwriter or Underwriters of a proposed underwritten offering to permit the Registrable Securities requested to be included in a Piggyback Registration on the same terms and conditions as any similar securities of Purchaser and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All Investors holding Registrable Securities proposing to distribute their securities through a Piggyback Registration that involves an Underwriter or Underwriters shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such Piggyback Registration.

2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters for a Piggyback Registration that is to be an underwritten offering advises Purchaser and Investors holding Registrable Securities proposing to distribute their Registrable Securities through such Piggyback Registration in writing that the dollar amount or number of Purchaser Class A Shares or other Purchaser securities that Purchaser desires to sell, taken together with Purchaser Class A Shares or other Purchaser securities hereunder, the Registrable Securities as to which registration has been requested under this Section 2.2, and Purchaser Class A Shares or other Purchaser securities, if any, as to which registration rights of other security holders of Purchaser, exceeds the Maximum Number of Securities, then Purchaser shall include in any such registration:

(a) If the registration is undertaken for Purchaser's account: (i) first, Purchaser Class A Shares or other securities that Purchaser desires to sell that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clause (i)</u>, the Registrable Securities of Investors as to which registration has been requested pursuant to this <u>Section 2.2</u>, Pro Rata among the holders thereof based on the number of Securities; and (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), Purchaser Class A Shares or other equity securities for the account of other Persons that Purchaser is obligated to register pursuant to separate written contractual arrangements with such Persons that can be sold without exceeding the Maximum Number of Securities;

(b) If the registration is a "demand" registration undertaken at the demand of Persons other than either Demanding Holders under <u>Section 2.1</u>: (i) first, Purchaser Class A Shares or other securities for the account of the demanding Persons that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), Purchaser Class A Shares or other securities that Purchaser desires to sell that can be sold without exceeding the Maximum Number of Securities; (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the Registrable Securities of Investors as to which registration has been requested pursuant to this <u>Section 2.2</u>, Pro Rata among the holders thereof based on the number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses thereof based on the number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), the Registration the foregoing clauses of the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), the Registration has been requested by such holders to be included in such registration, that can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), Purchaser Class A Shares or other equity securities for the account of other Persons that Purchaser is obligated to register pursuant to separate written contractual arrangements with such Persons that can be sold without exceeding the Maximum Number of Securities.

Ex F-8

In the event that Purchaser securities that are convertible into Purchaser Class A Shares are included in the offering, the calculations under this <u>Section 2.2.2</u> shall include such Purchaser securities on an as-converted to Purchaser Class A Share basis. Notwithstanding anything to the contrary above, to the extent that the registration of an Investor's Registrable Securities would prevent Purchaser or the demanding shareholders from effecting such registration and offering, such Investor shall not be permitted to exercise Piggyback Registration rights with respect to such registration and offering.

2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Investor holding Registrable Securities may elect to withdraw such Investor's request for inclusion of Registrable Securities in any Piggyback Registration by giving written notice to Purchaser of such request to withdraw at least two business days prior to the effectiveness of the Registration Statement filed with the SEC with respect to such Piggyback Registration. Purchaser (whether on its own determination or as the result of a withdrawal by Persons making a demand pursuant to written contractual obligations) may withdraw a Registration Statement at any time prior to the effectiveness of such Registration Statement without any liability to the applicable Investor, subject to the next sentence and the provisions of <u>Section 4</u>. Notwithstanding any such withdrawal, Purchaser shall be responsible for and pay all registration expenses incurred in connection with such Piggyback Registration prior to its withdrawal under this <u>subsection 2.2.3</u>.

2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a Shelf Underwriting effected under <u>Section 2.1</u> hereof.

3. <u>Registration Procedures.</u>

3.1 Whenever Purchaser is required to effect the registration of any Registrable Securities pursuant to <u>Section 2</u>, Purchaser shall use its reasonable best efforts to effect the registration and sale of such Registrable Securities in accordance with the intended method(s) of distribution thereof as expeditiously as practicable, and in connection with any such request:

3.1.1 Filing Registration Statement. Purchaser shall use its reasonable best efforts to, as expeditiously as possible, prepare and file with the SEC the Shelf Registration Statement, and all required amendments thereto on any form for which Purchaser then qualifies or which counsel for Purchaser shall deem appropriate and which form shall be available for the sale of all Registrable Securities to be registered thereunder in accordance with the intended method(s) of distribution thereof, and shall use its reasonable efforts to cause such Registration Statement and required amendments thereto to become effective and use its reasonable efforts to keep it effective for the period required hereby; *provided, however*, that Purchaser shall have the right to defer the Shelf Registration Statement and Shelf Underwriting for up to 90 days, and any Piggyback Registration for such period as may be applicable to deferment of any demand registration to which such Piggyback Registration relates, in each case if Purchaser shall furnish to Investors requesting to include their Registrable Securities in such registration a certificate signed by the Chief Executive Officer, Chief Financial Officer or Chairman of Purchaser stating that, in the good faith judgment of the Board of Directors of Purchaser, it would be materially detrimental to Purchaser and its shareholders for such Registration Statement to be effected at such time or the filing would require premature disclosure of material information which is not in the interests of Purchaser to disclose at such time; *provided further* that Purchaser shall not have the right to exercise the right set forth in the immediately preceding proviso more than twice in any 365-day period in respect of a demand registration hereunder.

Ex F-9

3.1.2 <u>Copies</u>. Purchaser shall, prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish without charge to Investors holding Registrable Securities included in such registration, and such Investors' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein, except for such exhibits and documents available on the SEC's Electronic Data Gathering, Analysis and Retrieval System), the prospectus included in such Registrable Securities included in such registration or legal counsel for any such Investors may request in order to facilitate the disposition of the Registrable Securities owned by such Investors.

3.1.3 <u>Amendments and Supplements</u>. Subject to <u>Section 2.1.1(c)</u>, Purchaser shall prepare and file with the SEC such amendments, including post-effective amendments, and supplements to such Registration Statement and the prospectus used in connection therewith as may be necessary to keep such Registration Statement effective and in compliance with the provisions of the Securities Act until all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or such securities have been withdrawn or until such time as the Registrable Securities cover to be Registrable Securities as defined by this Agreement.

3.1.4 Notification. After the filing of a Registration Statement, Purchaser shall promptly, and in no event more than five Business Days after such filing, notify Investors holding Registrable Securities included in such Registration Statement of such filing, and shall further notify such Investors promptly and confirm such advice in writing in all events within five Business Days after the occurrence of any of the following: (i) when such Registration Statement becomes effective; (ii) when any post-effective amendment to such Registration Statement becomes effective; (iii) the issuance or threatened issuance by the SEC of any stop order (and Purchaser shall take all actions required to prevent the entry of such stop order or to remove it if entered); and (iv) any request by the SEC for any amendment or supplement to such Registration Statement or any prospectus relating thereto or for additional information or of the occurrence of an event requiring the preparation of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of the securities covered by such Registration Statement, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and promptly make available to Investors holding Registrable Securities included in such Registration Statement any such supplement or amendment; except that before filing with the SEC a Registration Statement or prospectus or any amendment or supplement thereto, including documents incorporated by reference, Purchaser shall furnish to Investors holding Registrable Securities included in such Registration Statement and to the legal counsel for any such Investors, copies of all such documents proposed to be filed sufficiently in advance of filing to provide such Investors and legal counsel with a reasonable opportunity to review such documents and comment thereon; provided, that such Investors and their legal counsel must provide any comments promptly (and in any event within five Business Days) after receipt of such documents.

3.1.5 <u>State Securities Laws Compliance</u>. Purchaser shall use its reasonable efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as Investors holding Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may reasonably request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of Purchaser and do any and all other acts and things that may be necessary or advisable to enable Investors holding Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; *provided*, *however*, that Purchaser shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this paragraph or take any action to which it would be subject to general service of process or to taxation in any such jurisdiction where it is not then otherwise subject.

3.1.6 Agreements for Disposition. To the extent required by the underwriting agreement or similar agreements, Purchaser shall enter into reasonable customary agreements (including, if applicable, an underwriting agreement in customary form) and take such other actions as are reasonably required in order to expedite or facilitate the disposition of such Registrable Securities, including, if necessary, the engagement of a "qualified independent underwriter" in connection with the qualification of the underwriting arrangements with the Financial Industry Regulatory Authority, Inc. (*"FINRA"*) and arranging for the delivery of customary legal opinions and auditor "comfort letters." The representations, warranties and covenants of Purchaser in any underwriting agreement which are made to or for the benefit of any Underwriters, to the extent applicable, shall also be made to and for the benefit of Investors holding Registrable Securities included in such Registration Statement. No Investor holding Registrable Securities included in such Registration or warranties in the underwriting agreement except, if applicable, with respect to such Investor's organization, good standing, authority, title to Registrable Securities, lack of conflict of such sale with such Investor's material agreements and organizational documents, and with respect to written information relating to such Investor that such Investor has furnished in writing expressly for inclusion in such Registration Statement.

Ex F-10

3.1.7 <u>Cooperation</u>. The principal executive officer of Purchaser, the principal financial officer of Purchaser, the principal accounting officer of Purchaser and all other officers and members of the management of Purchaser shall reasonably cooperate in any offering of Registrable Securities hereunder, which cooperation shall include the preparation of the Registration Statement with respect to such offering and all other offering materials and related documents, and participation in meetings with Underwriters, attorneys, accountants and potential investors.

3.1.8 <u>Records</u>. Purchaser shall make available for inspection by Investors holding Registrable Securities included in such Registration Statement, any Underwriter participating in any disposition pursuant to such Registration Statement and any attorney, accountant or other professional retained by any Investor holding Registrable Securities included in such Registration Statement or any Underwriter, all financial and other records, pertinent corporate documents and properties of Purchaser, as shall be reasonably necessary to enable them to exercise their due diligence responsibility, and cause Purchaser's officers, directors and employees to supply all information reasonably requested by any of them in connection with such Registration Statement; *provided, however*, that Purchaser may require execution of a reasonable confidentiality agreement prior to sharing any such information.

3.1.9 Listing. Purchaser shall use its reasonable best efforts to cause all Registrable Securities that are Purchaser Class A Shares included in any registration to be listed on such exchanges or otherwise designated for trading in the same manner as similar securities issued by Purchaser are then listed or designated or, if no such similar securities are then listed or designated, in a manner satisfactory to Investors holding a majority-in-interest of the Registrable Securities included in such registration.

3.1.10 <u>Selection of Underwriters</u>. Purchaser shall have the right to select an Underwriter or Underwriters in connection with any Underwritten Offering, which Underwriter or Underwriters shall be subject to consultation with the Demanding Holder.

3.2 <u>Obligation to Suspend Distribution</u>. Upon receipt of any notice from Purchaser of the happening of any event of the kind described in <u>Section 3.1.4(iv)</u>, or in the event that the financial statements contained in the Registration Statement become stale, or in the event that the Registration Statement or prospectus included therein contains a misstatement of material fact or omits to state a material fact due to a bona fide business purpose, each Investor holding Registrable Securities included in any registration shall immediately discontinue disposition of such Registrable Securities pursuant to the Registration Statement covering such Registrable Securities until such Investor receives the supplemented or amended prospectus contemplated by <u>Section 3.1.4(iv)</u> or the Registration

Statement is updated so that the financial statements are no longer stale, as applicable, and, if so directed by Purchaser, each such Investor will deliver to Purchaser all copies, other than permanent file copies then in such Investor's possession, of the most recent prospectus covering such Registrable Securities at the time of receipt of such notice.

3.3 Registration Expenses. Subject to Section 4, Purchaser shall bear all reasonable costs and expenses incurred in connection with any demand registration pursuant to Section 2.1, any Piggyback Registration pursuant to Section 2.2, and all reasonable expenses incurred in performing or complying with its other obligations under this Agreement, whether or not the Registration Statement becomes effective, including: (i) all registration and filing fees; (ii) fees and expenses of compliance with securities or "blue sky" laws (including fees and disbursements of counsel in connection with blue sky qualifications of the Registrable Securities); (iii) printing expenses; (iv) Purchaser's internal expenses (including all salaries and expenses of its officers and employees); (v) the fees and expenses incurred in connection with the listing of the Registrable Securities as required by Section 3.1.11; (vi) FINRA fees; (vii) fees and disbursements of counsel for Purchaser and fees and expenses for independent certified public accountants retained by Purchaser (including the expenses or costs associated with the delivery of any opinions or comfort letters requested pursuant to Section 3.1.9); (viii) the reasonable fees and expenses of any special experts retained by Purchaser in connection with such registration; and (ix) the reasonable and documented fees and expenses (up to a maximum of \$30,000 in the aggregate in connection with such registration) of one legal counsel selected by Investors holding a majority-in-interest of the Registrable Securities included in such registration for such legal counsel's review, comment and finalization of the proposed Registration Statement and other relevant documents. Purchaser shall have no obligation to pay any underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions shall be borne by such holders. Additionally, in an underwritten offering, all selling security holders and Purchaser shall bear the expenses of the Underwriter pro rata in proportion to the respective amount of securities each is selling in such offering.

3.4 Information. Investors holding Registrable Securities included in any Registration Statement shall provide such information as may reasonably be requested by Purchaser, or the managing Underwriter, if any, in connection with the preparation of such Registration Statement, including amendments and supplements thereto, in order to effect the registration of any Registrable Securities under the Securities Act pursuant to Section 2 and in connection with the obligation to comply with federal and applicable state securities laws. Investors selling Registrable Securities in any offering must provide all questionnaires, powers of attorney, custody agreements, stock powers, and other documentation reasonably requested by Purchaser or the managing Underwriter. Notwithstanding anything else in this Agreement, Purchaser shall not be obligated to include such Investor's Registrable Securities to the extent Purchaser has not received such information, and received any other reasonably requested agreements or certificates, on or prior to the second business day prior to the first anticipated filing date of a Registration Statement (or in the case of an Underwritten Offering pursuant to Rule 415, at least two Business Days prior to the time of pricing of the applicable offering).

Ex F-12

# 4. Indemnification and Contribution.

4.1 Indemnification by Purchaser. Subject to the provisions of this Section 4.1 below, Purchaser agrees to indemnify and hold harmless each Investor, and each Investor's officers and directors and each Person, if any, who controls an Investor (within the meaning of the Securities Act) (each, an "Investor Indemnified Party"), from and against claims, suits, actions, or litigation brought by a third party, whether joint or several, that arise out of or are based upon any untrue statement of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to such Registration Statement, or that arise out of or are based upon any omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading ("Claim"), and any losses, damages, liabilities, costs (including, without limitation, reasonable and documented attorneys' fees) and expenses as incurred as a result of such Claim (provided, however, that the indemnity agreement contained in this Section 4.1 shall not apply to amounts paid in settlement of any such claim, loss, damage, liability or action if such settlement is effected without the consent of Purchaser, such consent not to be unreasonably withheld, delayed or conditioned); provided, however, that Purchaser will not be liable in any such case to the extent that any such expense, loss, claim, damage or liability arises out of or is based upon any untrue statement or omission made in such Registration Statement, preliminary prospectus, final prospectus, or summary prospectus, or any such amendment or supplement, in reliance upon and in conformity with information furnished to Purchaser, in writing, by such selling holder or Investor Indemnified Party expressly for use therein. Purchaser also shall indemnify any Underwriter of the Registrable Securities, their officers, affiliates, directors, partners, members and agents and each Person who controls such Underwriter (within the meaning of the Securities Act) on substantially the same basis as that of the indemnification provided above in this <u>Section 4.1</u>.

4.2 Indemnification by Investors Holding Registrable Securities. Subject to the provisions of this Section 4.2 below, each Investor selling Registrable Securities will, in the event that any Registration is being effected under the Securities Act pursuant to this Agreement of any Registrable Securities held by such selling Investor, indemnify and hold harmless Purchaser, each of its directors and officers and each other Person, if any, who controls Purchaser (within the meaning of the Securities Act), against any losses, claims, judgments, damages or liabilities, whether joint or several, insofar as such losses, claims, judgments, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to the Registration Statement, or arise out of or are based upon any omission to state a material fact required to be stated therein or necessary to make the statement therein not misleading, if the statement or omission was made in reliance upon and in conformity with information furnished in writing to Purchaser by such selling Investor expressly for use therein (provided, however, that the indemnity agreement contained in this Section 4.2 shall not apply to amounts paid in settlement of any such claim, loss, damage, liability or action if such settlement is effected without the consent of the indemnifying Investor, such consent not to be unreasonably withheld, delayed or conditioned), and shall reimburse Purchaser, its directors and officers and each other Person, if any, who controls Purchaser (within the meaning of the Securities Act) for any legal or other expenses reasonably incurred by any of them in connection with investigation or defending any such loss, claim, damage, liability or action. Each selling Investor's indemnification obligations hereunder shall be several and not joint and shall be limited to the amount of any net proceeds actually received by such selling Investor. The selling Investors also shall indemnify any Underwriter, their officers, directors, partners, members and agents and each person who controls such Underwriter (within the meaning of the Securities Act) on substantially the same basis as that of the indemnification provided above in this Section 4.2.

4.3 Conduct of Indemnification Proceedings. Promptly after receipt by any Person of any notice of any loss, claim, damage or liability or any action in respect of which indemnity may be sought pursuant to Sections 4.1 or 4.2, such Person (the "Indemnified Party") shall, if a claim in respect thereof is to be made against any other Person for indemnification hereunder, notify such other Person (the "Indemnifying Party") in writing of the loss, claim, judgment, damage, liability or action; provided, however, that the failure by the Indemnified Party to notify the Indemnifying Party shall not relieve the Indemnifying Party from any liability which the Indemnifying Party may have to such Indemnified Party hereunder, except and solely to the extent the Indemnifying Party is actually prejudiced by such failure. If the Indemnified Party is seeking indemnification with respect to any claim or action brought against the Indemnified Party, then the Indemnifying Party shall be entitled to participate in such claim or action, and, to the extent that it wishes, jointly with all other Indemnifying Parties, to assume control of the defense thereof with counsel satisfactory to the Indemnified Party. After notice from the Indemnifying Party to the Indemnified Party of its election to assume control of the defense of such claim or action, the Indemnifying Party shall not be liable to the Indemnified Party for any legal or other expenses subsequently incurred by the Indemnified Party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that in any action in which both the Indemnified Party and the Indemnifying Party are named as defendants, the Indemnified Party shall have the right to employ separate counsel (but no more than one such separate counsel) to represent the Indemnified Party and its controlling Persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought by the Indemnified Party against the Indemnifying Party, with the fees and expenses of such counsel to be paid by such Indemnifying Party if, based upon the written opinion of counsel of such Indemnified Party, representation of both parties by the same counsel would be inappropriate due to actual or potential differing interests between them. If such defense is assumed, the Indemnifying Party shall not be subject to any liability for any settlement made by the Indemnified Party without its consent. No Indemnifying Party shall, without the prior written consent of the Indemnified Party (acting reasonably), consent to entry of judgment or effect any settlement of any claim or pending or threatened proceeding in respect of which the Indemnified Party is or could have been a party and indemnity could have been sought hereunder by such Indemnified Party, unless such judgment or settlement includes an unconditional release of such Indemnified Party from all liability arising out of such claim or proceeding.

Ex F-13

4.4 Contribution.

4.4.1 If the indemnification provided for in the foregoing <u>Sections 4.1, 4.2</u> and <u>4.3</u> is unavailable to any Indemnified Party in respect of any loss, claim, damage, liability or action referred to herein, then each such Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such loss, claim, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the Indemnified Parties and the Indemnifying Parties in connection with the actions or omissions which resulted in such loss, claim, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of any Indemnified Party and any Indemnifying Party shall be determined by reference to, among other things, whether the untrue statement of a material fact or the omission to state a material fact relates to information supplied by such Indemnified Party or such Indemnifying Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.

4.4.2 The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>Section 4.4</u> were determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the immediately preceding <u>Section 4.4.1</u>.

4.4.3 The amount paid or payable by an Indemnified Party as a result of any loss, claim, damage, liability or action referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses incurred by such Indemnified Party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this <u>Section 4.4</u>, no Investor holding Registrable Securities shall be required to contribute any amount in excess of the dollar amount of the net proceeds (after payment of any underwriting fees, discounts, commissions or taxes) actually received by such Investor from the sale of Registrable Securities which gave rise to such contribution obligation. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

5. <u>Rule 144</u>. As long as any Investor shall own Registrable Securities, Purchaser, at all times while it shall be a reporting company under the Exchange Act, covenants to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by Purchaser after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act. Purchaser further covenants that it shall take such further action as any Investor may reasonably request, all to the extent required from time to time to enable such Investor to sell Ordinary Shares held by such Investor without registration under the Securities Act within the limitation of the exemptions provided by Rule 144, including providing any legal opinions. Upon the request of any Investor, Purchaser shall deliver to such Investor a written certification of a duly authorized officer as to whether it has complied with such requirements.

# 6. Miscellaneous.

6.1 <u>Other Registration Rights</u>. Purchaser represents and warrants that, as of the date of this Agreement, no Person, other than the holders of Registrable Securities, has any right to require Purchaser to register any of Purchaser's share capital for sale or to include Purchaser's share capital in any Registration filed by Purchaser for the sale of share capital for its own account or for the account of any other Person. Purchaser agrees with the Investors not to enter into any registration rights agreement that would conflict with the rights of the Investors hereunder, including without limitation <u>Sections 2.1.2</u> and <u>2.2.2</u>.

#### 6.2 Assignment; No Third Party Beneficiaries.

(a) This Agreement and the rights, duties and obligations of Purchaser hereunder may not be assigned or delegated by Purchaser in whole or in part, unless Purchaser first provides Investors holding Registrable Securities at least ten Business Days prior written notice; *provided*, *however*, that no assignment or delegation by Purchaser will relieve Purchaser of its obligations under this Agreement unless Investors holding a majority-in-interest of the Registrable Securities provide their prior written consent, which consent must not be unreasonably withheld, delayed or conditioned.

(b) This Agreement and the rights, duties and obligations of Investors holding Registrable Securities hereunder may be freely assigned or delegated by such Investor in conjunction with and to the extent of any transfer of Registrable Securities by such Investor which is permitted by such Investor's applicable Lock-Up Agreement; *provided*, *however*, that no assignment by any Investor of its rights, duties and obligations hereunder shall be binding upon or obligate Purchaser unless and until Purchaser shall have received (i) written notice of such assignment as provided in <u>Section 6.3</u> and (ii) the written agreement of the assignee, in a form reasonably satisfactory to Purchaser, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 6.2</u> shall be null and void. (c) This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties, to the permitted assigns of the Investors or of any assignee of the Investors.

(d) This Agreement is not intended to confer any rights or benefits on any Persons that are not party hereto other than as expressly set forth in <u>Section 4</u> and this <u>Section 6.2</u>.

6.3 <u>Notices</u>. All notices, consents, waivers and other communications hereunder shall be in writing and shall be deemed to have been duly given when delivered (i) in person, (ii) by facsimile or other electronic means, with confirmation of receipt, (iii) one Business Day after being sent, if sent by reputable, nationally recognized overnight courier service or (iv) three (3) Business Days after being mailed, if sent by registered or certified mail, pre-paid and return receipt requested, in each case to the applicable Party at the following addresses (or at such other address for a Party as shall be specified by like notice):

If to Purchaser, to: Jianwei Li Chairman & Co-Chief Executive Officer 437 Madison Avenue, 27th Floor New York, New York 10022 (732) 910-9692 Jianwei@zhenchengcap.com

With copies to (which shall not constitute notice): David C. Buck Kayleigh McNelis Sidley Austin LLP 1000 Louisiana, Suite 5900 Houston, Texas 77002 Tel: (713) 494-4500 dbuck@sidley.com kmcnelis@sidley.com

If to an Investor, to: the address set forth underneath such Investor's name on the signature page or to such Investor's address as found in Purchaser's books and records.

6.4 Severability.

In case any provision in this Agreement shall be held invalid, illegal or unenforceable in a jurisdiction, such provision shall be modified or deleted, as to the jurisdiction involved, only to the extent necessary to render the same valid, legal and enforceable, and the validity, legality and enforceability of the remaining provisions hereof shall not in any way be affected or impaired thereby nor shall the validity, legality or enforceability of such provision be affected thereby in any other jurisdiction. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties will substitute for any invalid, illegal or unenforceable provision that carries out, so far as may be valid, legal and enforceable, the intent and purpose of such invalid, illegal or unenforceable provision.

#### Ex F-15

6.5 Entire Agreement. This Agreement (together with the Business Combination Agreement and the Lock-Up Agreements to the extent incorporated herein, and including all agreements entered into pursuant hereto or thereto or referenced herein or therein and all certificates and instruments delivered pursuant hereto and thereto) embody the entire agreement and understanding of the Parties hereto in respect of the subject matter contained herein. There are no restrictions, promises, representations, warranties, covenants or undertakings, other than those expressly set forth or referred to herein or the documents or instruments referred to herein, which collectively supersede all prior agreements and the understandings among the Parties with respect to the subject matter contained herein.

6.6 Interpretation. Titles and headings of sections of this Agreement are for convenience only and shall not affect the construction of any provision of this Agreement. In this Agreement, unless the context otherwise requires: (i) any pronoun used in this Agreement shall include the corresponding masculine, feminine or neuter forms, and the singular form of nouns, pronouns and verbs shall include the plural and vice versa; (ii) "including" (and with correlative meaning "include") means including without limiting the generality of any description preceding or succeeding such term and shall be deemed in each case to be followed by the words "without limitation"; (iii) the words "herein," "hereto," and "hereby" and other words of similar import in this Agreement shall be deemed in each case to refer to this Agreement as a whole and not to any particular section or other subdivision of this Agreement; and (iv) the term "or" means "and/or". The parties have participated jointly in the negotiation and drafting of this Agreement. Consequently, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

6.7 <u>Amendments; Waivers</u>. Any term of this Agreement may be amended and the observance of any term of this Agreement may be waived (either generally or in a particular instance, and either retroactively or prospectively) only with the written agreement or consent of Purchaser and Investors holding a majority-in-interest of the Registrable Securities; provided, that any amendment or waiver of this Agreement which affects an Investor in a manner materially and adversely disproportionate to other Investors will also require the consent of such Investor. No failure or delay by a party in exercising any right hereunder shall operate as a waiver thereof. No waivers of or exceptions to any term, condition, or provision of this Agreement, in any one or more instances, shall be deemed to be or construed as a further or continuing waiver of any such term, condition, or provision.

6.8 <u>Remedies Cumulative</u>. In the event a party fails to observe or perform any covenant or agreement to be observed or performed under this Agreement, the other parties may proceed to protect and enforce its rights by suit in equity or action at law, whether for specific performance of any term contained in this Agreement or for an injunction against the breach of any such term or in aid of the exercise of any power granted in this Agreement or to enforce any other legal or equitable right, or to take any one or more of such actions, without being required to post a bond. None of the rights, powers or remedies conferred under this Agreement shall be mutually exclusive, and each such right, power or remedy shall be cumulative and in addition to any other right, power or remedy, whether conferred by this Agreement or now or hereafter available at law, in equity, by statute or otherwise.

6.9 Governing Law; Jurisdiction. This Agreement shall be governed by, construed and enforced in accordance with the Laws of the State of New York without regard to the conflict of laws principles thereof. Each party hereto hereby (i) submits to the exclusive jurisdiction of any state or federal court located in the County of New York in the State of New York (or in any appellate court thereof) (the "Specified Courts") for the purpose of any claim, action, litigation or other legal proceeding arising out of or relating to this Agreement or the transactions contemplated hereby (a "Proceeding"), and (b) irrevocably waives, and agrees not to assert by way of motion, defense or otherwise, in any such Proceeding, any claim that it is not subject personally to the jurisdiction of the above-named courts, that its property is exempt or immune from attachment or execution, that the Proceeding is brought in an inconvenient forum, that the venue of the Proceeding is improper, or that this Agreement or the transactions contemplated hereby agrees that a final judgment in any Proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law. Each party irrevocably consents to the service of the summons and complaint and any other process in any Proceeding, on behalf of itself, or its property, by personal delivery of copies of such process to such party at the applicable address set forth in <u>Section 6.3</u>. Nothing in this <u>Section 6.9</u> shall affect the right of any party to serve legal process in any other manner permitted by applicable Law.

6.10 <u>WAIVER OF TRIAL BY JURY</u>. EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES THE RIGHT TO A TRIAL BY JURY IN ANY ACTION, SUIT, COUNTERCLAIM OR OTHER PROCEEDING (WHETHER BASED ON CONTRACT, TORT OR OTHERWISE) ARISING OUT OF, CONNECTED WITH OR RELATING TO THIS AGREEMENT, THE TRANSACTIONS CONTEMPLATED HEREBY, OR THE ACTIONS OF THE INVESTORS IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF.

6.11 <u>Termination of Business Combination Agreement</u>. This Agreement shall be binding upon each party upon such party's execution and delivery of this Agreement, but this Agreement shall only become effective upon the Closing. In the event that the Business Combination Agreement is validly terminated in accordance with its terms prior to the Closing, this Agreement shall automatically terminate and become null and void and be of no further force or effect, and the parties shall have no obligations hereunder.

6.12 <u>Termination of Original Registration Rights Agreement</u>. Upon the effectiveness hereof on the Closing Date, the Original Registration Rights Agreement shall terminate without further action by Purchaser or the Investors party thereto.

6.13 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including by facsimile or pdf or other electronic document transmission), each of which shall be deemed an original, and all of which taken together shall constitute one and the same instrument.

(Remainder of page intentionally left blank)

Ex F-17

IN WITNESS WHEREOF, the parties have caused this Registration Rights Agreement to be executed and delivered as of the date first written above.

# **TradeUP Global Corporation**

By:

Name: Title:

**Investors:** 

## **TradeUp Global Sponsor LLC**

By:

Name: Title:

Michael Davidov

Tao Jiang

David Xianglin Li

#### **Energy Science Artist Holding Limited**

By:

Name: Li Risheng Title:

# LilOrange Holding Limited

By:

Name: Title:

Signature Page to Registration Rights Agreement

Ex F-18

## **Eternal Selead Technology Holding Limited**

By:

Name:

Title:

## Haonan Technology Holding Limited

By:

Name: Title:

## **Shiningsunny Holding Limited**

By:

Name: Title:

#### ZenGolden L.P.

By:

Name: Title:

Patternbase Inc.

#### By:

Name: Title:

#### **Make World Better Limited**

By:

Name: Title:

Signature Page to Registration Rights Agreement

Ex F-19

### <u>Exhibit G</u>

#### Form of Surviving Company Memorandum and Articles of Association

**Final Form** 

# THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES AMENDED AND RESTATED MEMORANDUM AND ARTICLES OF ASSOCIATION

Copyright © 2022 www.secdatabase.com. All Rights Reserved. Please Consider the Environment Before Printing This Document

## OF SAITECH LIMITED (ADOPTED BY SPECIAL RESOLUTION DATED [*DATE*] AND EFFECTIVE ON [*DATE*])

Ex G-1

**Final Form** 

# THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES AMENDED AND RESTATED MEMORANDUM OF ASSOCIATION OF SAITECH LIMITED (ADOPTED BY SPECIAL RESOLUTION DATED [*DATE*] AND EFFECTIVE ON [*DATE*])

1 The name of the Company is SAITECH Limited.

- The Registered Office of the Company shall be at the offices of ICS Corporate Services (Cayman) Limited, 3-212 Governors Square, 23 Lime Tree Bay Avenue, P.O. Box 30746, Seven Mile Beach, Grand Cayman KY1-1203, Cayman Islands, or at such other place within the Cayman Islands as the Directors may decide.
- 3 The objects for which the Company is established are unrestricted and the Company shall have full power and authority to carry out any object not prohibited by the laws of the Cayman Islands.
- 4 The liability of each Member is limited to the amount unpaid on such Member's shares.
- 5 The share capital of the Company is US\$[ ] divided into [ ] shares of a par value of US\$0.0001 each.
- 6 The Company has power to register by way of continuation as a body corporate limited by shares under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.
- 7 Capitalised terms that are not defined in this Amended and Restated Memorandum of Association bear the respective meanings given to them in the Amended and Restated Articles of Association of the Company.

Ex G-2

# THE COMPANIES ACT (AS REVISED) OF THE CAYMAN ISLANDS COMPANY LIMITED BY SHARES AMENDED AND RESTATED ARTICLES OF ASSOCIATION OF SAITECH LIMITED

# (ADOPTED BY SPECIAL RESOLUTION DATED [DATE] AND EFFECTIVE ON [DATE])

#### **1** Interpretation

1.1 In the Articles Table A in the First Schedule to the Statute does not apply and, unless there is something in the subject or context inconsistent therewith:

"Articles"	means these amended and restated articles of association of the Company.
"Auditor"	means the person for the time being performing the duties of auditor of the Company (if any).
"Company"	means the above named company.
"Directors"	means the directors for the time being of the Company.
"Dividend"	means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the Articles.
"Electronic Record"	has the same meaning as in the Electronic Transactions Act.
"Electronic Transactions Act"	means the Electronic Transactions Act (As Revised) of the Cayman Islands.
"Member"	has the same meaning as in the Statute.
"Memorandum"	means the amended and restated memorandum of association of the Company.
"Ordinary Resolution"	means a resolution passed by a simple majority of the Members as, being entitled to do so, vote in person or, where proxies are allowed, by proxy at a general meeting, and includes a unanimous written resolution. In computing the majority when a poll is demanded regard shall be had to the number of votes to which each Member is entitled by the Articles.
"Register of Members"	means the register of Members maintained in accordance with the Statute and includes (except where otherwise stated) any branch or duplicate register of Members.
"Registered Office"	means the registered office for the time being of the Company.
"Seal"	means the common seal of the Company and includes every duplicate seal.
"Share"	means a share in the Company and includes a fraction of a share in the Company.
"Special Resolution"	has the same meaning as in the Statute, and includes a unanimous written resolution.
"Statute"	means the Companies Act (As Revised) of the Cayman Islands.
"Subscriber"	means the subscriber to the Memorandum.
"Treasury Share"	means a Share held in the name of the Company as a treasury share in accordance with the Statute.

Ex G-3

# 1.2 In the Articles:

- (a) words importing the singular number include the plural number and vice versa;
- (b) words importing the masculine gender include the feminine gender;
- (c) words importing persons include corporations as well as any other legal or natural person;
- (d) "written" and "in writing" include all modes of representing or reproducing words in visible form, including in the form of an Electronic Record;

- (e) "shall" shall be construed as imperative and "may" shall be construed as permissive;
- (f) references to provisions of any law or regulation shall be construed as references to those provisions as amended, modified, re-enacted or replaced;
- (g) any phrase introduced by the terms "including", "include", "in particular" or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms;

the term "and/or" is used herein to mean both "and" as well as "or." The use of "and/or" in certain contexts in no respects qualifies or modifies the use of the terms "and" or "or" in others. The term "or" shall not be interpreted to

- (h) respects quarties or modifies the use of the terms and or or in others. The term of shall not be interpreted to be exclusive and the term "and" shall not be interpreted to require the conjunctive (in each case, unless the context otherwise requires);
- (i) headings are inserted for reference only and shall be ignored in construing the Articles;
- (j) any requirements as to delivery under the Articles include delivery in the form of an Electronic Record;
- (k) any requirements as to execution or signature under the Articles including the execution of the Articles themselves can be satisfied in the form of an electronic signature as defined in the Electronic Transactions Act;
- (l) sections 8 and 19(3) of the Electronic Transactions Act shall not apply;
- (m) the term "clear days" in relation to the period of a notice means that period excluding the day when the notice is received or deemed to be received and the day for which it is given or on which it is to take effect; and
- (n) the term "holder" in relation to a Share means a person whose name is entered in the Register of Members as the holder of such Share.

## 2 Commencement of Business

- 2.1 The business of the Company may be commenced as soon after incorporation of the Company as the Directors shall see fit.
- 2.2 The Directors may pay, out of the capital or any other monies of the Company, all expenses incurred in or about the formation and establishment of the Company, including the expenses of registration.

#### 3 Issue of Shares

3.1

Subject to the provisions, if any, in the Memorandum (and to any direction that may be given by the Company in general meeting) and without prejudice to any rights attached to any existing Shares, the Directors may allot, issue, grant options over or otherwise dispose of Shares (including fractions of a Share) with or without preferred, deferred or other rights or restrictions, whether in regard to Dividend or other distribution, voting, return of capital or otherwise and to such persons, at such times and on such other terms as they think proper, and may also (subject to the Statute and the Articles) vary such rights. Notwithstanding the foregoing, the Subscriber shall have the power to:

- (a) issue one Share to itself;
- (b) transfer that Share by an instrument of transfer to any person; and
- (c) update the Register of Members in respect of the issue and transfer of that Share.
- 3.2 The Company shall not issue Shares to bearer.

4.1 The Company shall maintain or cause to be maintained the Register of Members in accordance with the Statute.

The Directors may determine that the Company shall maintain one or more branch registers of Members in accordance with
 the Statute. The Directors may also determine which register of Members shall constitute the principal register and which shall constitute the branch register or registers, and to vary such determination from time to time.

# 5 Closing Register of Members or Fixing Record Date

For the purpose of determining Members entitled to notice of, or to vote at any meeting of Members or any adjournment thereof,
 or Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose, the Directors may provide that the Register of Members shall be closed for transfers for a stated period which shall not in any case exceed forty days.

5.2 In lieu of, or apart from, closing the Register of Members, the Directors may fix in advance or arrears a date as the record date for any such determination of Members entitled to notice of, or to vote at any meeting of the Members or any adjournment thereof, or for the purpose of determining the Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose.

If the Register of Members is not so closed and no record date is fixed for the determination of Members entitled to notice of, or to vote at, a meeting of Members or Members entitled to receive payment of a Dividend or other distribution, the date on which notice of the meeting is sent or the date on which the resolution of the Directors resolving to pay such Dividend or other distribution is passed, as the case may be, shall be the record date for such determination of Members. When a determination of Members entitled to vote at any meeting of Members has been made as provided in this Article, such determination shall apply to any adjournment thereof.

## 6 Certificates for Shares

5.3

A Member shall only be entitled to a share certificate if the Directors resolve that share certificates shall be issued. Share certificates representing Shares, if any, shall be in such form as the Directors may determine. Share certificates shall be signed by one or more Directors or other person authorised by the Directors. The Directors may authorise certificates to be issued with the authorised signature(s) affixed by mechanical process. All certificates for Shares shall be consecutively numbered or otherwise identified and shall specify the Shares to which they relate. All certificates surrendered to the Company for transfer shall be cancelled and subject to the Articles no new certificate shall be issued until the former certificate representing a like number of relevant Shares shall have been surrendered and cancelled.

- 6.2 The Company shall not be bound to issue more than one certificate for Shares held jointly by more than one person and delivery of a certificate to one joint holder shall be a sufficient delivery to all of them.
- If a share certificate is defaced, worn out, lost or destroyed, it may be renewed on such terms (if any) as to evidence and
   indemnity and on the payment of such expenses reasonably incurred by the Company in investigating evidence, as the Directors may prescribe, and (in the case of defacement or wearing out) upon delivery of the old certificate.
- 6.4 Every share certificate sent in accordance with the Articles will be sent at the risk of the Member or other person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

# 7 Transfer of Shares

- Subject to Article 3.1, Shares are transferable subject to the approval of the Directors by resolution who may, in their absolute
  discretion, decline to register any transfer of Shares without giving any reason. If the Directors refuse to register a transfer they shall notify the transferee within two months of such refusal.
- The instrument of transfer of any Share shall be in writing and shall be executed by or on behalf of the transferor (and if the
   Directors so require, signed by or on behalf of the transferee). The transferor shall be deemed to remain the holder of a Share until the name of the transferee is entered in the Register of Members.

Ex G-5

#### 8 Redemption, Repurchase and Surrender of Shares

Subject to the provisions of the Statute the Company may issue Shares that are to be redeemed or are liable to be redeemed at
the option of the Member or the Company. The redemption of such Shares shall be effected in such manner and upon such other terms as the Company may, by Special Resolution, determine before the issue of the Shares.

- 8.2 Subject to the provisions of the Statute, the Company may purchase its own Shares (including any redeemable Shares) in such manner and on such other terms as the Directors may agree with the relevant Member.
- 8.3 The Company may make a payment in respect of the redemption or purchase of its own Shares in any manner permitted by the Statute, including out of capital.
- 8.4 The Directors may accept the surrender for no consideration of any fully paid Share.

# 9 Treasury Shares

- 9.1 The Directors may, prior to the purchase, redemption or surrender of any Share, determine that such Share shall be held as a Treasury Share.
- 9.2 The Directors may determine to cancel a Treasury Share or transfer a Treasury Share on such terms as they think proper (including, without limitation, for nil consideration).

## 10 Variation of Rights of Shares

If at any time the share capital of the Company is divided into different classes of Shares, all or any of the rights attached to any class (unless otherwise provided by the terms of issue of the Shares of that class) may, whether or not the Company is being wound up, be varied without the consent of the holders of the issued Shares of that class where such variation is considered by the Directors not to have a material adverse effect upon such rights; otherwise, any such variation shall be made only with the consent in writing of the holders of not less than two thirds of the issued Shares of that class, or with the approval of a resolution

- 10.1 passed by a majority of not less than two thirds of the votes cast at a separate meeting of the holders of the Shares of that class. For the avoidance of doubt, the Directors reserve the right, notwithstanding that any such variation may not have a material adverse effect, to obtain consent from the holders of Shares of the relevant class. To any such meeting all the provisions of the Articles relating to general meetings shall apply *mutatis mutandis*, except that the necessary quorum shall be one person holding or representing by proxy at least one third of the issued Shares of the class and that any holder of Shares of the class present in person or by proxy may demand a poll.
- For the purposes of a separate class meeting, the Directors may treat two or more or all the classes of Shares as forming one class of Shares if the Directors consider that such class of Shares would be affected in the same way by the proposals under consideration, but in any other case shall treat them as separate classes of Shares.
- 10.3 The rights conferred upon the holders of the Shares of any class issued with preferred or other rights shall not, unless otherwise expressly provided by the terms of issue of the Shares of that class, be deemed to be varied by the creation or issue of further Shares ranking pari passu therewith.

#### Ex G-6

#### 11 Commission on Sale of Shares

The Company may, in so far as the Statute permits, pay a commission to any person in consideration of his subscribing or agreeing to subscribe (whether absolutely or conditionally) or procuring or agreeing to procure subscriptions (whether absolutely or conditionally) for any Shares. Such commissions may be satisfied by the payment of cash and/or the issue of fully or partly paid-up Shares. The Company may also on any issue of Shares pay such brokerage as may be lawful.

# 12 Non Recognition of Trusts

The Company shall not be bound by or compelled to recognise in any way (even when notified) any equitable, contingent, future or partial interest in any Share, or (except only as is otherwise provided by the Articles or the Statute) any other rights in respect of any Share other than an absolute right to the entirety thereof in the holder.

## 13 Lien on Shares

13.1 The Company shall have a first and paramount lien on all Shares (whether fully paid-up or not) registered in the name of a Member (whether solely or jointly with others) for all debts, liabilities or engagements to or with the Company (whether presently payable or not) by such Member or his estate, either alone or jointly with any other person, whether a Member or not, but the Directors may at any time declare any Share to be wholly or in part exempt from the provisions of this Article. The registration of a transfer of any such Share shall operate as a waiver of the Company's lien thereon. The Company's lien on a Share shall also extend to any amount payable in respect of that Share.

13.2 The Company may sell, in such manner as the Directors think fit, any Shares on which the Company has a lien, if a sum in respect of which the lien exists is presently payable, and is not paid within fourteen clear days after notice has been received or deemed to have been received by the holder of the Shares, or to the person entitled to it in consequence of the death or bankruptcy of the holder, demanding payment and stating that if the notice is not complied with the Shares may be sold.

To give effect to any such sale the Directors may authorise any person to execute an instrument of transfer of the Shares sold to, or in accordance with the directions of, the purchaser. The purchaser or his nominee shall be registered as the holder of the Shares comprised in any such transfer, and he shall not be bound to see to the application of the purchase money, nor shall his title to the Shares be affected by any irregularity or invalidity in the sale or the exercise of the Company's power of sale under the Articles.

The net proceeds of such sale after payment of costs, shall be applied in payment of such part of the amount in respect of which the lien exists as is presently payable and any balance shall (subject to a like lien for sums not presently payable as existed upon the Shares before the sale) be paid to the person entitled to the Shares at the date of the sale.

# 14 Call on Shares

14.1 Subject to the terms of the allotment and issue of any Shares, the Directors may make calls upon the Members in respect of any monies unpaid on their Shares (whether in respect of par value or premium), and each Member shall (subject to receiving at least fourteen clear days' notice specifying the time or times of payment) pay to the Company at the time or times so specified the amount called on the Shares. A call may be revoked or postponed, in whole or in part, as the Directors may determine. A call may be required to be paid by instalments. A person upon whom a call is made shall remain liable for calls made upon him notwithstanding the subsequent transfer of the Shares in respect of which the call was made.

- 14.2 A call shall be deemed to have been made at the time when the resolution of the Directors authorising such call was passed.
- 14.3 The joint holders of a Share shall be jointly and severally liable to pay all calls in respect thereof.

14.4 If a call remains unpaid after it has become due and payable, the person from whom it is due shall pay interest on the amount unpaid from the day it became due and payable until it is paid at such rate as the Directors may determine (and in addition all expenses that have been incurred by the Company by reason of such non-payment), but the Directors may waive payment of the interest or expenses wholly or in part.

An amount payable in respect of a Share on issue or allotment or at any fixed date, whether on account of the par value of the 14.5 Share or premium or otherwise, shall be deemed to be a call and if it is not paid all the provisions of the Articles shall apply as if that amount had become due and payable by virtue of a call.

14.6 The Directors may issue Shares with different terms as to the amount and times of payment of calls, or the interest to be paid.

14.7 The Directors may, if they think fit, receive an amount from any Member willing to advance all or any part of the monies uncalled and unpaid upon any Shares held by him, and may (until the amount would otherwise become payable) pay interest at such rate as may be agreed upon between the Directors and the Member paying such amount in advance. No such amount paid in advance of calls shall entitle the Member paying such amount to any portion of a Dividend or other distribution payable in respect of any period prior to the date upon which such amount would, but for such payment, become payable.

## 15 Forfeiture of Shares

If a call or instalment of a call remains unpaid after it has become due and payable the Directors may give to the person from whom it is due not less than fourteen clear days' notice requiring payment of the amount unpaid together with any interest which 15.1 may have accrued and any expenses incurred by the Company by reason of such non-payment. The notice shall specify where payment is to be made and shall state that if the notice is not complied with the Shares in respect of which the call was made will be liable to be forfeited.

If the notice is not complied with, any Share in respect of which it was given may, before the payment required by the notice has been made, be forfeited by a resolution of the Directors. Such forfeiture shall include all Dividends, other distributions or other monies payable in respect of the forfeited Share and not paid before the forfeiture.

A forfeited Share may be sold, re-allotted or otherwise disposed of on such terms and in such manner as the Directors think fit and at any time before a sale, re-allotment or disposition the forfeiture may be cancelled on such terms as the Directors think fit. Where for the purposes of its disposal a forfeited Share is to be transferred to any person the Directors may authorise some person to execute an instrument of transfer of the Share in favour of that person.

A person any of whose Shares have been forfeited shall cease to be a Member in respect of them and shall surrender to the Company for cancellation the certificate for the Shares forfeited and shall remain liable to pay to the Company all monies which 15.4 at the date of forfeiture were payable by him to the Company in respect of those Shares together with interest at such rate as the Directors may determine, but his liability shall cease if and when the Company shall have received payment in full of all monies due and payable by him in respect of those Shares.

A certificate in writing under the hand of one Director or officer of the Company that a Share has been forfeited on a specified date shall be conclusive evidence of the facts stated in it as against all persons claiming to be entitled to the Share. The certificate shall (subject to the execution of an instrument of transfer) constitute a good title to the Share and the person to whom the Share is sold or otherwise disposed of shall not be bound to see to the application of the purchase money, if any, nor shall his title to the Share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the Share.

The provisions of the Articles as to forfeiture shall apply in the case of non payment of any sum which, by the terms of issue of a Share, becomes payable at a fixed time, whether on account of the par value of the Share or by way of premium as if it had been payable by virtue of a call duly made and notified.

# 16 Transmission of Shares

If a Member dies the survivor or survivors (where he was a joint holder) or his legal personal representatives (where he was a sole holder), shall be the only persons recognised by the Company as having any title to his Shares. The estate of a deceased Member is not thereby released from any liability in respect of any Share, for which he was a joint or sole holder.

Any person becoming entitled to a Share in consequence of the death or bankruptcy or liquidation or dissolution of a Member (or in any other way than by transfer) may, upon such evidence being produced as may be required by the Directors, elect, by a notice in writing sent by him to the Company, either to become the holder of such Share or to have some person nominated

- 16.2 by him registered as the holder of such Share. If he elects to have another person registered as the holder of such Share he shall sign an instrument of transfer of that Share to that person. The Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution, as the case may be.
- 16.3 A person becoming entitled to a Share by reason of the death or bankruptcy or liquidation or dissolution of a Member (or in any other case than by transfer) shall be entitled to the same Dividends, other distributions and other advantages to which he

would be entitled if he were the holder of such Share. However, he shall not, before becoming a Member in respect of a Share, be entitled in respect of it to exercise any right conferred by membership in relation to general meetings of the Company and the Directors may at any time give notice requiring any such person to elect either to be registered himself or to have some person nominated by him be registered as the holder of the Share (but the Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution or any other case than by transfer, as the case may be). If the notice is not complied with within ninety days of being received or deemed to be received (as determined pursuant to the Articles) the Directors may thereafter withhold payment of all Dividends, other distributions, bonuses or other monies payable in respect of the Share until the requirements of the notice have been complied with.

Ex G-8

# 17 Amendments of Memorandum and Articles of Association and Alteration of Capital

- 17.1 The Company may by Ordinary Resolution:
  - (a) increase its share capital by such sum as the Ordinary Resolution shall prescribe and with such rights, priorities and privileges annexed thereto, as the Company in general meeting may determine;
  - (b) consolidate and divide all or any of its share capital into Shares of larger amount than its existing Shares;
  - (c) convert all or any of its paid-up Shares into stock, and reconvert that stock into paid-up Shares of any denomination;
  - (d) by subdivision of its existing Shares or any of them divide the whole or any part of its share capital into Shares of smaller amount than is fixed by the Memorandum or into Shares without par value; and
  - (e) cancel any Shares that at the date of the passing of the Ordinary Resolution have not been taken or agreed to be taken by any person and diminish the amount of its share capital by the amount of the Shares so cancelled.
- All new Shares created in accordance with the provisions of the preceding Article shall be subject to the same provisions of the Articles with reference to the payment of calls, liens, transfer, transmission, forfeiture and otherwise as the Shares in the original share capital.
- 17.3 Subject to the provisions of the Statute and the provisions of the Articles as regards the matters to be dealt with by Ordinary Resolution, the Company may by Special Resolution:
  - (a) change its name;
  - (b) alter or add to the Articles;
  - (c) alter or add to the Memorandum with respect to any objects, powers or other matters specified therein; and
  - (d) reduce its share capital or any capital redemption reserve fund.

# 18 Offices and Places of Business

Subject to the provisions of the Statute, the Company may by resolution of the Directors change the location of its Registered Office. The Company may, in addition to its Registered Office, maintain such other offices or places of business as the Directors determine.

# 19 General Meetings

19.1 All general meetings other than annual general meetings shall be called extraordinary general meetings.

The Company may, but shall not (unless required by the Statute) be obliged to, in each year hold a general meeting as its annual general meeting, and shall specify the meeting as such in the notices calling it. Any annual general meeting shall be held at such time and place as the Directors shall appoint and if no other time and place is prescribed by them, it shall be held at the Registered Office on the second Wednesday in December of each year at ten o'clock in the morning. At these meetings the report of the Directors (if any) shall be presented.

- 19.3 The Directors may call general meetings, and they shall on a Members' requisition forthwith proceed to convene an extraordinary general meeting of the Company.
- 19.4 A Members' requisition is a requisition of Members holding at the date of deposit of the requisition not less than ten per cent. in par value of the issued Shares which as at that date carry the right to vote at general meetings of the Company.
- 19.5 The Members' requisition must state the objects of the meeting and must be signed by the requisitionists and deposited at the Registered Office, and may consist of several documents in like form each signed by one or more requisitionists. If there are no Directors as at the date of the deposit of the Members' requisition or if the Directors do not within twenty-one
- days from the date of the deposit of the Members' requisition duly proceed to convene a general meeting to be held within a further twenty-one days, the requisitionists, or any of them representing more than one-half of the total voting rights of all of the requisitionists, may themselves convene a general meeting, but any meeting so convened shall be held no later than the day which falls three months after the expiration of the said twenty-one day period.
- 19.7 A general meeting convened as aforesaid by requisitionists shall be convened in the same manner as nearly as possible as that in which general meetings are to be convened by Directors.

Ex G-9

# 20 Notice of General Meetings

At least five clear days' notice shall be given of any general meeting. Every notice shall specify the place, the day and the hour of the meeting and the general nature of the business to be conducted at the general meeting and shall be given in the manner hereinafter mentioned or in such other manner if any as may be prescribed by the Company, provided that a general meeting of the Company shall, whether or not the notice specified in this Article has been given and whether or not the provisions of the Articles regarding general meetings have been complied with, be deemed to have been duly convened if it is so agreed:

- (a) in the case of an annual general meeting, by all of the Members entitled to attend and vote thereat; and
- (b) in the case of an extraordinary general meeting, by a majority in number of the Members having a right to attend and vote at the meeting, together holding not less than ninety five per cent. in par value of the Shares giving that right.
- 20.2 The accidental omission to give notice of a general meeting to, or the non receipt of notice of a general meeting by, any person entitled to receive such notice shall not invalidate the proceedings of that general meeting.

# 21 Proceedings at General Meetings

No business shall be transacted at any general meeting unless a quorum is present. Two Members being individuals present in person or by proxy or if a corporation or other non-natural person by its duly authorised representative or proxy shall be a quorum unless the Company has only one Member entitled to vote at such general meeting in which case the quorum shall be that one Member present in person or by proxy or (in the case of a corporation or other non-natural person) by its duly authorised representative or proxy.

- A person may participate at a general meeting by conference telephone or other communications equipment by means of which all the persons participating in the meeting can communicate with each other. Participation by a person in a general meeting in this manner is treated as presence in person at that meeting.
- 21.3 A resolution (including a Special Resolution) in writing (in one or more counterparts) signed by or on behalf of all of the Members for the time being entitled to receive notice of and to attend and vote at general meetings (or, being corporations or

other non-natural persons, signed by their duly authorised representatives) shall be as valid and effective as if the resolution had been passed at a general meeting of the Company duly convened and held.

If a quorum is not present within half an hour from the time appointed for the meeting to commence or if during such a meeting a quorum ceases to be present, the meeting, if convened upon a Members' requisition, shall be dissolved and in any other case it shall stand adjourned to the same day in the next week at the same time and/or place or to such other day, time and/or place as the Directors may determine, and if at the adjourned meeting a quorum is not present within half an hour from the time appointed for the meeting to commence, the Members present shall be a quorum.

The Directors may, at any time prior to the time appointed for the meeting to commence, appoint any person to act as chairman of a general meeting of the Company or, if the Directors do not make any such appointment, the chairman, if any, of the board of Directors shall preside as chairman at such general meeting. If there is no such chairman, or if he shall not be present within fifteen minutes after the time appointed for the meeting to commence, or is unwilling to act, the Directors present shall elect one of their number to be chairman of the meeting.

21.6 If no Director is willing to act as chairman or if no Director is present within fifteen minutes after the time appointed for the meeting to commence, the Members present shall choose one of their number to be chairman of the meeting.

The chairman may, with the consent of a meeting at which a quorum is present (and shall if so directed by the meeting) adjourn the meeting from time to time and from place to place, but no business shall be transacted at any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place.

21.8 When a general meeting is adjourned for thirty days or more, notice of the adjourned meeting shall be given as in the case of an original meeting. Otherwise it shall not be necessary to give any such notice of an adjourned meeting.

21.9 A resolution put to the vote of the meeting shall be decided on a show of hands unless before, or on the declaration of the result of, the show of hands, the chairman demands a poll, or any other Member or Members collectively present in person or by proxy (or in the case of a corporation or other non-natural person, by its duly authorised representative or proxy) and holding at least ten per cent. in par value of the Shares giving a right to attend and vote at the meeting demand a poll.

21.10 Unless a poll is duly demanded and the demand is not withdrawn a declaration by the chairman that a resolution has been carried or carried unanimously, or by a particular majority, or lost or not carried by a particular majority, an entry to that effect in the minutes of the proceedings of the meeting shall be conclusive evidence of that fact without proof of the number or proportion of the votes recorded in favour of or against such resolution.

## Ex G-10

- 21.11 The demand for a poll may be withdrawn.
- 21.12 Except on a poll demanded on the election of a chairman or on a question of adjournment, a poll shall be taken as the chairman directs, and the result of the poll shall be deemed to be the resolution of the general meeting at which the poll was demanded.

A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith. A poll demanded on any other question shall be taken at such date, time and place as the chairman of the general meeting directs, and any business other than that upon which a poll has been demanded or is contingent thereon may proceed pending the taking of the poll.

21.14 In the case of an equality of votes, whether on a show of hands or on a poll, the chairman shall be entitled to a second or casting vote.

## 22 Votes of Members

22.1 Subject to any rights or restrictions attached to any Shares, on a show of hands every Member who (being an individual) is present in person or by proxy or, if a corporation or other non-natural person is present by its duly authorised representative or by proxy, shall have one vote and on a poll every Member present in any such manner shall have one vote for every Share of which he is the holder.

22.2 In the case of joint holders the vote of the senior holder who tenders a vote, whether in person or by proxy (or, in the case of a corporation or other non-natural person, by its duly authorised representative or proxy), shall be accepted to the exclusion of the votes of the other joint holders, and seniority shall be determined by the order in which the names of the holders stand in the Register of Members.

A Member of unsound mind, or in respect of whom an order has been made by any court, having jurisdiction in lunacy, may vote, whether on a show of hands or on a poll, by his committee, receiver, curator bonis, or other person on such Member's behalf appointed by that court, and any such committee, receiver, curator bonis or other person may vote by proxy.

22.4 No person shall be entitled to vote at any general meeting unless he is registered as a Member on the record date for such meeting nor unless all calls or other monies then payable by him in respect of Shares have been paid.

No objection shall be raised as to the qualification of any voter except at the general meeting or adjourned general meeting at which the vote objected to is given or tendered and every vote not disallowed at the meeting shall be valid. Any objection made in due time in accordance with this Article shall be referred to the chairman whose decision shall be final and conclusive.

On a poll or on a show of hands votes may be cast either personally or by proxy (or in the case of a corporation or other nonnatural person by its duly authorised representative or proxy). A Member may appoint more than one proxy or the same proxy 22.6 under one or more instruments to attend and vote at a meeting. Where a Member appoints more than one proxy the instrument of proxy shall state which proxy is entitled to vote on a show of hands and shall specify the number of Shares in respect of which each proxy is entitled to exercise the related votes.

Ex G-11

On a poll, a Member holding more than one Share need not cast the votes in respect of his Shares in the same way on any resolution and therefore may vote a Share or some or all such Shares either for or against a resolution and/or abstain from voting a Share or some or all of the Shares and, subject to the terms of the instrument appointing him, a proxy appointed under one or more instruments may vote a Share or some or all of the Shares in respect of which he is appointed either for or against a resolution and/or abstain from voting a Share or some or all of the Shares in respect of which he is appointed.

# 23 Proxies

The instrument appointing a proxy shall be in writing and shall be executed under the hand of the appointor or of his attorney duly authorised in writing, or, if the appointor is a corporation or other non natural person, under the hand of its duly authorised representative. A proxy need not be a Member.

The Directors may, in the notice convening any meeting or adjourned meeting, or in an instrument of proxy sent out by the Company, specify the manner by which the instrument appointing a proxy shall be deposited and the place and the time (being not later than the time appointed for the commencement of the meeting or adjourned meeting to which the proxy relates) at which the instrument appointing a proxy shall be deposited. In the absence of any such direction from the Directors in the notice convening any meeting or adjourned meeting or in an instrument of proxy sent out by the Company, the instrument appointing a proxy shall be deposited Office not less than 48 hours before the time appointed for the meeting or adjourned meeting or adjourned meeting or the person named in the instrument proposes to vote.

The chairman may in any event at his discretion declare that an instrument of proxy shall be deemed to have been duly deposited.
 An instrument of proxy that is not deposited in the manner permitted, or which has not been declared to have been duly deposited by the chairman, shall be invalid.

The instrument appointing a proxy may be in any usual or common form (or such other form as the Directors may approve) and may be expressed to be for a particular meeting or any adjournment thereof or generally until revoked. An instrument appointing a proxy shall be deemed to include the power to demand or join or concur in demanding a poll.

23.5 Votes given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous death or insanity of the principal or revocation of the proxy or of the authority under which the proxy was executed, or the transfer of the Share

in respect of which the proxy is given unless notice in writing of such death, insanity, revocation or transfer was received by the Company at the Registered Office before the commencement of the general meeting, or adjourned meeting at which it is sought to use the proxy.

# 24 Corporate Members

Any corporation or other non-natural person which is a Member may in accordance with its constitutional documents, or in the absence of such provision by resolution of its directors or other governing body, authorise such person as it thinks fit to act as its representative at any meeting of the Company or of any class of Members, and the person so authorised shall be entitled to exercise the same powers on behalf of the corporation which he represents as the corporation could exercise if it were an individual Member.

## 25 Shares that May Not be Voted

Shares in the Company that are beneficially owned by the Company shall not be voted, directly or indirectly, at any meeting and shall not be counted in determining the total number of outstanding Shares at any given time.

## 26 Directors

There shall be a board of Directors consisting of not less than one person (exclusive of alternate Directors) provided however that the Company may by Ordinary Resolution increase or reduce the limits in the number of Directors. The first Directors of the Company shall be determined in writing by, or appointed by a resolution of, the Subscriber.

Ex G-12

## 27 Powers of Directors

Subject to the provisions of the Statute, the Memorandum and the Articles and to any directions given by Special Resolution, the business of the Company shall be managed by the Directors who may exercise all the powers of the Company. No alteration of the Memorandum or Articles and no such direction shall invalidate any prior act of the Directors which would have been valid if that alteration had not been made or that direction had not been given. A duly convened meeting of Directors at which a quorum is present may exercise all powers exercisable by the Directors.

All cheques, promissory notes, drafts, bills of exchange and other negotiable or transferable instruments and all receipts for 27.2 monies paid to the Company shall be signed, drawn, accepted, endorsed or otherwise executed as the case may be in such manner as the Directors shall determine by resolution.

The Directors on behalf of the Company may pay a gratuity or pension or allowance on retirement to any Director who has held any other salaried office or place of profit with the Company or to his widow or dependants and may make contributions to any fund and pay premiums for the purchase or provision of any such gratuity, pension or allowance.

27.4 The Directors may exercise all the powers of the Company to borrow money and to mortgage or charge its undertaking, property and assets (present and future) and uncalled capital or any part thereof and to issue debentures, debenture stock, mortgages, bonds and other such securities whether outright or as security for any debt, liability or obligation of the Company or of any third party.

#### 28 Appointment and Removal of Directors

28.1 The Company may by Ordinary Resolution appoint any person to be a Director or may by Ordinary Resolution remove any Director.

The Directors may appoint any person to be a Director, either to fill a vacancy or as an additional Director provided that the appointment does not cause the number of Directors to exceed any number fixed by or in accordance with the Articles as the maximum number of Directors.

#### 29 Vacation of Office of Director

The office of a Director shall be vacated if:

- (a) the Director gives notice in writing to the Company that he resigns the office of Director; or
- the Director absents himself (for the avoidance of doubt, without being represented by proxy or an alternate Director (b) appointed by him) from three consecutive meetings of the board of Directors without special leave of absence from the Directors, and the Directors pass a resolution that he has by reason of such absence vacated office; or
- (c) the Director dies, becomes bankrupt or makes any arrangement or composition with his creditors generally; or
- (d) the Director is found to be or becomes of unsound mind; or
- all of the other Directors (being not less than two in number) determine that he should be removed as a Director, either by a resolution passed by all of the other Directors at a meeting of the Directors duly convened and held in accordance (e) with the Articles or by a resolution in writing signed by all of the other Directors.

#### 30 **Proceedings of Directors**

The quorum for the transaction of the business of the Directors may be fixed by the Directors, and unless so fixed shall be two if there are two or more Directors, and shall be one if there is only one Director. A person who holds office as an alternate Director 30.1 shall, if his appointor is not present, be counted in the quorum. A Director who also acts as an alternate Director shall, if his appointor is not present, count twice towards the quorum.

Subject to the provisions of the Articles, the Directors may regulate their proceedings as they think fit. Questions arising at any meeting shall be decided by a majority of votes. In the case of an equality of votes, the chairman shall have a second or casting 30.2 vote. A Director who is also an alternate Director shall be entitled in the absence of his appointor to a separate vote on behalf of his appointor in addition to his own vote.

#### Ex G-13

A person may participate in a meeting of the Directors or any committee of Directors by conference telephone or other communications equipment by means of which all the persons participating in the meeting can communicate with each other 30.3 at the same time. Participation by a person in a meeting in this manner is treated as presence in person at that meeting. Unless otherwise determined by the Directors the meeting shall be deemed to be held at the place where the chairman is located at the start of the meeting.

A resolution in writing (in one or more counterparts) signed by all the Directors or all the members of a committee of the Directors or, in the case of a resolution in writing relating to the removal of any Director or the vacation of office by any Director, all of the Directors other than the Director who is the subject of such resolution (an alternate Director being entitled to sign such 30.4 a resolution on behalf of his appointor and if such alternate Director is also a Director, being entitled to sign such resolution both on behalf of his appointer and in his capacity as a Director) shall be as valid and effectual as if it had been passed at a meeting of the Directors, or committee of Directors as the case may be, duly convened and held.

A Director or alternate Director may, or other officer of the Company on the direction of a Director or alternate Director shall, call a meeting of the Directors by at least two days' notice in writing to every Director and alternate Director which notice shall 30.5 set forth the general nature of the business to be considered unless notice is waived by all the Directors (or their alternates) either at, before or after the meeting is held. To any such notice of a meeting of the Directors all the provisions of the Articles relating to the giving of notices by the Company to the Members shall apply *mutatis mutandis*.

The continuing Directors (or a sole continuing Director, as the case may be) may act notwithstanding any vacancy in their body, but if and so long as their number is reduced below the number fixed by or pursuant to the Articles as the necessary quorum of Directors the continuing Directors or Director may act for the purpose of increasing the number of Directors to be equal to such fixed number, or of summoning a general meeting of the Company, but for no other purpose.

30.6

The Directors may elect a chairman of their board and determine the period for which he is to hold office; but if no such 30.7 chairman is elected, or if at any meeting the chairman is not present within five minutes after the time appointed for the meeting to commence, the Directors present may choose one of their number to be chairman of the meeting.

All acts done by any meeting of the Directors or of a committee of the Directors (including any person acting as an alternate Director) shall, notwithstanding that it is afterwards discovered that there was some defect in the appointment of any Director or alternate Director, and/or that they or any of them were disqualified, and/or had vacated their office and/or were not entitled to vote, be as valid as if every such person had been duly appointed and/or not disqualified to be a Director or alternate Director and/or had not vacated their office and/or had been entitled to vote, as the case may be.

A Director but not an alternate Director may be represented at any meetings of the board of Directors by a proxy appointed in writing by him. The proxy shall count towards the quorum and the vote of the proxy shall for all purposes be deemed to be that of the appointing Director.

## 31 Presumption of Assent

A Director or alternate Director who is present at a meeting of the board of Directors at which action on any Company matter is taken shall be presumed to have assented to the action taken unless his dissent shall be entered in the minutes of the meeting or unless he shall file his written dissent from such action with the person acting as the chairman or secretary of the meeting before the adjournment thereof or shall forward such dissent by registered post to such person immediately after the adjournment of the meeting. Such right to dissent shall not apply to a Director or alternate Director who voted in favour of such action.

## **32** Directors' Interests

32.4

A Director or alternate Director may hold any other office or place of profit under the Company (other than the office of Auditor) 32.1 in conjunction with his office of Director for such period and on such terms as to remuneration and otherwise as the Directors may determine.

A Director or alternate Director may act by himself or by, through or on behalf of his firm in a professional capacity for the 32.2 Company and he or his firm shall be entitled to remuneration for professional services as if he were not a Director or alternate Director.

Ex G-14

32.3 A Director or alternate Director may be or become a director or other officer of or otherwise interested in any company promoted by the Company or in which the Company may be interested as a shareholder, a contracting party or otherwise, and no such Director or alternate Director shall be accountable to the Company for any remuneration or other benefits received by him as a director or officer of, or from his interest in, such other company.

No person shall be disqualified from the office of Director or alternate Director or prevented by such office from contracting with the Company, either as vendor, purchaser or otherwise, nor shall any such contract or any contract or transaction entered into by or on behalf of the Company in which any Director or alternate Director shall be in any way interested be or be liable to be avoided, nor shall any Director or alternate Director so contracting or being so interested be liable to account to the Company for any profit realised by or arising in connection with any such contract or transaction by reason of such Director or alternate Director holding office or of the fiduciary relationship thereby established. A Director (or his alternate Director in his absence) shall be at liberty to vote in respect of any contract or transaction in which he is interested provided that the nature of the interest of any Director or alternate Director or transaction shall be disclosed by him at or prior to its consideration and any vote thereon.

32.5 A general notice that a Director or alternate Director is a shareholder, director, officer or employee of any specified firm or company and is to be regarded as interested in any transaction with such firm or company shall be sufficient disclosure for the purposes of voting on a resolution in respect of a contract or transaction in which he has an interest, and after such general notice it shall not be necessary to give special notice relating to any particular transaction.

## 33 Minutes

The Directors shall cause minutes to be made in books kept for the purpose of recording all appointments of officers made by the Directors, all proceedings at meetings of the Company or the holders of any class of Shares and of the Directors, and of committees of the Directors, including the names of the Directors or alternate Directors present at each meeting.

#### 34 Delegation of Directors' Powers

34.1 The Directors may delegate any of their powers, authorities and discretions, including the power to sub-delegate, to any committee consisting of one or more Directors. They may also delegate to any managing director or any Director holding any other executive office such of their powers, authorities and discretions as they consider desirable to be exercised by him provided that an alternate Director may not act as managing director and the appointment of a managing director shall be revoked forthwith if he ceases to be a Director. Any such delegation may be made subject to any conditions the Directors may impose and either collaterally with or to the exclusion of their own powers and any such delegation may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of a committee of Directors shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

34.2 The Directors may establish any committees, local boards or agencies or appoint any person to be a manager or agent for managing the affairs of the Company and may appoint any person to be a member of such committees, local boards or agencies. Any such appointment may be made subject to any conditions the Directors may impose, and either collaterally with or to the exclusion of their own powers and any such appointment may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of any such committee, local board or agency shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

The Directors may by power of attorney or otherwise appoint any person to be the agent of the Company on such conditions as 34.3 the Directors may determine, provided that the delegation is not to the exclusion of their own powers and may be revoked by the Directors at any time.

The Directors may by power of attorney or otherwise appoint any company, firm, person or body of persons, whether nominated directly or indirectly by the Directors, to be the attorney or authorised signatory of the Company for such purpose and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under the Articles) and for such period and subject to such conditions as they may think fit, and any such powers of attorney or other appointment may contain such provisions for the protection and convenience of persons dealing with any such attorneys or authorised signatories as the

Directors may think fit and may also authorise any such attorney or authorised signatory to delegate all or any of the powers, authorities and discretions vested in him.

#### Ex G-15

The Directors may appoint such officers of the Company (including, for the avoidance of doubt and without limitation, any secretary) as they consider necessary on such terms, at such remuneration and to perform such duties, and subject to such provisions as to disqualification and removal as the Directors may think fit. Unless otherwise specified in the terms of his appointment an officer of the Company may be removed by resolution of the Directors or Members. An officer of the Company may vacate his office at any time if he gives notice in writing to the Company that he resigns his office.

# 35 Alternate Directors

34.4

35.1 Any Director (but not an alternate Director) may by writing appoint any other Director, or any other person willing to act, to be an alternate Director and by writing may remove from office an alternate Director so appointed by him.

35.2 An alternate Director shall be entitled to receive notice of all meetings of Directors and of all meetings of committees of Directors of which his appointor is a member, to attend and vote at every such meeting at which the Director appointing him is not personally present, to sign any written resolution of the Directors, and generally to perform all the functions of his appointor as a Director in his absence.

35.3 An alternate Director shall cease to be an alternate Director if his appointor ceases to be a Director.

- 35.4 Any appointment or removal of an alternate Director shall be by notice to the Company signed by the Director making or revoking the appointment or in any other manner approved by the Directors.
- 35.5 Subject to the provisions of the Articles, an alternate Director shall be deemed for all purposes to be a Director and shall alone be responsible for his own acts and defaults and shall not be deemed to be the agent of the Director appointing him.

#### 36 No Minimum Shareholding

The Company in general meeting may fix a minimum shareholding required to be held by a Director, but unless and until such a shareholding qualification is fixed a Director is not required to hold Shares.

## **37** Remuneration of Directors

37.1 The remuneration to be paid to the Directors, if any, shall be such remuneration as the Directors shall determine. The Directors shall also be entitled to be paid all travelling, hotel and other expenses properly incurred by them in connection with their attendance at meetings of Directors or committees of Directors, or general meetings of the Company, or separate meetings of the holders of any class of Shares or debentures of the Company, or otherwise in connection with the business of the Company or the discharge of their duties as a Director, or to receive a fixed allowance in respect thereof as may be determined by the Directors, or a combination partly of one such method and partly the other.

The Directors may by resolution approve additional remuneration to any Director for any services which in the opinion of the Directors go beyond his ordinary routine work as a Director. Any fees paid to a Director who is also counsel, attorney or solicitor to the Company, or otherwise serves it in a professional capacity shall be in addition to his remuneration as a Director.

#### 38 Seal

38.1 The Company may, if the Directors so determine, have a Seal. The Seal shall only be used by the authority of the Directors or of a committee of the Directors authorised by the Directors. Every instrument to which the Seal has been affixed shall be signed by at least one person who shall be either a Director or some officer of the Company or other person appointed by the Directors for the purpose.

The Company may have for use in any place or places outside the Cayman Islands a duplicate Seal or Seals each of which shall be a facsimile of the common Seal of the Company and, if the Directors so determine, with the addition on its face of the name of every place where it is to be used.

A Director or officer, representative or attorney of the Company may without further authority of the Directors affix the Seal over his signature alone to any document of the Company required to be authenticated by him under seal or to be filed with the Registrar of Companies in the Cayman Islands or elsewhere wheresoever.

#### Ex G-16

#### **39** Dividends, Distributions and Reserve

39.1 Subject to the Statute and this Article and except as otherwise provided by the rights attached to any Shares, the Directors may resolve to pay Dividends and other distributions on Shares in issue and authorise payment of the Dividends or other distributions out of the funds of the Company lawfully available therefor. A Dividend shall be deemed to be an interim Dividend unless the terms of the resolution pursuant to which the Directors resolve to pay such Dividend specifically state that such Dividend shall be a final Dividend. No Dividend or other distribution shall be paid except out of the realised or unrealised profits of the Company, out of the share premium account or as otherwise permitted by law.

Except as otherwise provided by the rights attached to any Shares, all Dividends and other distributions shall be paid according to the par value of the Shares that a Member holds. If any Share is issued on terms providing that it shall rank for Dividend as from a particular date, that Share shall rank for Dividend accordingly. 39.3 The Directors may deduct from any Dividend or other distribution payable to any Member all sums of money (if any) then payable by him to the Company on account of calls or otherwise.

The Directors may resolve that any Dividend or other distribution be paid wholly or partly by the distribution of specific assets and in particular (but without limitation) by the distribution of shares, debentures, or securities of any other company or in any one or more of such ways and where any difficulty arises in regard to such distribution, the Directors may settle the same as they

39.4 think expedient and in particular may issue fractional Shares and may fix the value for distribution of such specific assets or any part thereof and may determine that cash payments shall be made to any Members upon the basis of the value so fixed in order to adjust the rights of all Members and may vest any such specific assets in trustees in such manner as may seem expedient to the Directors.

Except as otherwise provided by the rights attached to any Shares, Dividends and other distributions may be paid in any currency. 39.5 The Directors may determine the basis of conversion for any currency conversions that may be required and how any costs involved are to be met.

The Directors may, before resolving to pay any Dividend or other distribution, set aside such sums as they think proper as a reserve or reserves which shall, at the discretion of the Directors, be applicable for any purpose of the Company and pending such application may, at the discretion of the Directors, be employed in the business of the Company.

Any Dividend, other distribution, interest or other monies payable in cash in respect of Shares may be paid by wire transfer to the holder or by cheque or warrant sent through the post directed to the registered address of the holder or, in the case of joint holders, to the registered address of the holder who is first named on the Register of Members or to such person and to such address as such holder or joint holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the person to whom it is sent. Any one of two or more joint holders may give effectual receipts for any Dividends, other distributions, bonuses, or other monies payable in respect of the Share held by them as joint holders.

39.8 No Dividend or other distribution shall bear interest against the Company.

Any Dividend or other distribution which cannot be paid to a Member and/or which remains unclaimed after six months from the date on which such Dividend or other distribution becomes payable may, in the discretion of the Directors, be paid into a separate account in the Company's name, provided that the Company shall not be constituted as a trustee in respect of that account and the Dividend or other distribution shall remain as a debt due to the Member. Any Dividend or other distribution which remains unclaimed after a period of six years from the date on which such Dividend or other distribution becomes payable shall be forfeited and shall revert to the Company.

#### 40 Capitalisation

The Directors may at any time capitalise any sum standing to the credit of any of the Company's reserve accounts or funds (including the share premium account and capital redemption reserve fund) or any sum standing to the credit of the profit and loss account or otherwise available for distribution; appropriate such sum to Members in the proportions in which such sum would have been divisible amongst such Members had the same been a distribution of profits by way of Dividend or other distribution; and apply such sum on their behalf in paying up in full unissued Shares for allotment and distribution credited as fully paid-up to and amongst them in the proportion aforesaid. In such event the Directors shall do all acts and things required to give effect to such capitalisation, with full power given to the Directors to make such provisions as they think fit in the case of Shares becoming distributable in fractions (including provisions whereby the benefit of fractional entitlements accrue to the Company rather than to the Members concerned). The Directors may authorise any person to enter on behalf of all of the Members interested into an agreement with the Company providing for such capitalisation and matters incidental or relating thereto and any agreement made under such authority shall be effective and binding on all such Members and the Company.

Ex G-17

#### 41 Books of Account

41.1 The Directors shall cause proper books of account (including, where applicable, material underlying documentation including contracts and invoices) to be kept with respect to all sums of money received and expended by the Company and the matters

in respect of which the receipt or expenditure takes place, all sales and purchases of goods by the Company and the assets and liabilities of the Company. Such books of account must be retained for a minimum period of five years from the date on which they are prepared. Proper books shall not be deemed to be kept if there are not kept such books of account as are necessary to give a true and fair view of the state of the Company's affairs and to explain its transactions.

41.2 The Directors shall determine whether and to what extent and at what times and places and under what conditions or regulations the accounts and books of the Company or any of them shall be open to the inspection of Members not being Directors and no Member (not being a Director) shall have any right of inspecting any account or book or document of the Company except as conferred by Statute or authorised by the Directors or by the Company in general meeting.

41.3 The Directors may cause to be prepared and to be laid before the Company in general meeting profit and loss accounts, balance sheets, group accounts (if any) and such other reports and accounts as may be required by law.

#### 42 Audit

42.1 The Directors may appoint an Auditor of the Company who shall hold office on such terms as the Directors determine.

Every Auditor of the Company shall have a right of access at all times to the books and accounts and vouchers of the Company 42.2 and shall be entitled to require from the Directors and officers of the Company such information and explanation as may be necessary for the performance of the duties of the Auditor.

Auditors shall, if so required by the Directors, make a report on the accounts of the Company during their tenure of office at the next annual general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an ordinary company, and at the next extraordinary general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an exempted company, and at any other time during their term of office, upon request of the Directors or any general meeting of the Members.

#### 43 Notices

43.1 Notices shall be in writing and may be given by the Company to any Member either personally or by sending it by courier, post, cable, telex, fax or e-mail to him or to his address as shown in the Register of Members (or where the notice is given by e-mail by sending it to the e-mail address provided by such Member). Any notice, if posted from one country to another, is to be sent by airmail.

Where a notice is sent by courier, service of the notice shall be deemed to be effected by delivery of the notice to a courier company, and shall be deemed to have been received on the third day (not including Saturdays or Sundays or public holidays) following the day on which the notice was delivered to the courier. Where a notice is sent by post, service of the notice shall be deemed to be effected by properly addressing, pre paying and posting a letter containing the notice, and shall be deemed to have been received on the fifth day (not including Saturdays or Sundays or public holidays) following the day on which the notice was posted. Where a notice is sent by cable, telex or fax, service of the notice shall be deemed to be effected by properly addressing and sending such notice and shall be deemed to have been received on the same day that it was transmitted. Where a notice is given by e-mail service shall be deemed to be effected by transmitting the e-mail to the e-mail address provided by the intended recipient and shall be deemed to have been received on the same day that it was sent, and it shall not be necessary for the receipt of the e-mail to be acknowledged by the recipient.

Ex G-18

43.3

A notice may be given by the Company to the person or persons which the Company has been advised are entitled to a Share or Shares in consequence of the death or bankruptcy of a Member in the same manner as other notices which are required to be given under the Articles and shall be addressed to them by name, or by the title of representatives of the deceased, or trustee of the bankrupt, or by any like description at the address supplied for that purpose by the persons claiming to be so entitled, or at the option of the Company by giving the notice in any manner in which the same might have been given if the death or bankruptcy had not occurred.

Notice of every general meeting shall be given in any manner authorised by the Articles to every holder of Shares carrying an entitlement to receive such notice on the record date for such meeting except that in the case of joint holders the notice shall be sufficient if given to the joint holder first named in the Register of Members and every person upon whom the ownership of a Share devolves by reason of his being a legal personal representative or a trustee in bankruptcy of a Member where the Member but for his death or bankruptcy would be entitled to receive notice of the meeting, and no other person shall be entitled to receive

44 Winding Up

notices of general meetings.

43.4

44.2

45.2

44.1 If the Company shall be wound up the liquidator shall apply the assets of the Company in satisfaction of creditors' claims in such manner and order as such liquidator thinks fit. Subject to the rights attaching to any Shares, in a winding up:

if the assets available for distribution amongst the Members shall be insufficient to repay the whole of the Company's

(a) issued share capital, such assets shall be distributed so that, as nearly as may be, the losses shall be borne by the Members in proportion to the par value of the Shares held by them; or

if the assets available for distribution amongst the Members shall be more than sufficient to repay the whole of the Company's issued share capital at the commencement of the winding up, the surplus shall be distributed amongst the

(b) Members in proportion to the par value of the Shares held by them at the commencement of the winding up subject to a deduction from those Shares in respect of which there are monies due, of all monies payable to the Company for unpaid calls or otherwise.

If the Company shall be wound up the liquidator may, subject to the rights attaching to any Shares and with the approval of a Special Resolution of the Company and any other approval required by the Statute, divide amongst the Members in kind the whole or any part of the assets of the Company (whether such assets shall consist of property of the same kind or not) and may for that purpose value any assets and determine how the division shall be carried out as between the Members or different classes of Members. The liquidator may, with the like approval, vest the whole or any part of such assets in trustees upon such trusts for the benefit of the Members as the liquidator, with the like approval, shall think fit, but so that no Member shall be compelled to accept any asset upon which there is a liability.

#### 45 Indemnity and Insurance

Every Director and officer of the Company (which for the avoidance of doubt, shall not include auditors of the Company), together with every former Director and former officer of the Company (each an "Indemnified Person") shall be indemnified out of the assets of the Company against any liability, action, proceeding, claim, demand, costs, damages or expenses, including legal expenses, whatsoever which they or any of them may incur as a result of any act or failure to act in carrying out their functions other than such liability (if any) that they may incur by reason of their own actual fraud or wilful default. No Indemnified Person shall be liable to the Company for any loss or damage incurred by the Company as a result (whether direct or indirect) of the carrying out of their functions unless that liability arises through the actual fraud or wilful default of such Indemnified Person. No person shall be found to have committed actual fraud or wilful default under this Article unless or until a court of competent jurisdiction shall have made a finding to that effect.

Ex G-19

The Company shall advance to each Indemnified Person reasonable attorneys' fees and other costs and expenses incurred in connection with the defence of any action, suit, proceeding or investigation involving such Indemnified Person for which indemnity will or could be sought. In connection with any advance of any expenses hereunder, the Indemnified Person shall execute an undertaking to repay the advanced amount to the Company if it shall be determined by final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification pursuant to this Article. If it shall be determined by a final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification with respect to such judgment, costs or expenses, then such party shall not be indemnified with respect to such judgment, costs or expenses and any advancement shall be returned to the Company (without interest) by the Indemnified Person. 45.3 The Directors, on behalf of the Company, may purchase and maintain insurance for the benefit of any Director or other officer of the Company against any liability which, by virtue of any rule of law, would otherwise attach to such person in respect of any negligence, default, breach of duty or breach of trust of which such person may be guilty in relation to the Company.

#### 46 Financial Year

Unless the Directors otherwise prescribe, the financial year of the Company shall end on 31st December in each year and, following the year of incorporation, shall begin on 1st January in each year.

#### 47 Transfer by Way of Continuation

If the Company is exempted as defined in the Statute, it shall, subject to the provisions of the Statute and with the approval of a Special Resolution, have the power to register by way of continuation as a body corporate under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.

#### 48 Mergers and Consolidations

The Company shall have the power to merge or consolidate with one or more other constituent companies (as defined in the Statute) upon such terms as the Directors may determine and (to the extent required by the Statute) with the approval of a Special Resolution.

Ex G-20

Dated this [\_\_] day of [\_\_\_] 2021.

ICS Corporate Services (Cayman) Limited

3-212 Governors Square, 23 Lime Tree Bay Avenue

P.O. Box 30746, Seven Mile Beach

Grand Cayman KY1-1203, Cayman Islands

acting by:

[ ]

[]

Witness to the above signature

Ex G-21

#### <u>Exhibit H</u>

#### Form of Amended Purchaser Charter

**Final Form** 

#### THE COMPANIES ACT (AS REVISED)

#### OF THE CAYMAN ISLANDS

#### **COMPANY LIMITED BY SHARES**

#### AMENDED AND RESTATED

#### MEMORANDUM AND ARTICLES OF ASSOCIATION

OF

#### **SAI.TECH Global Corporation**

#### (ADOPTED BY SPECIAL RESOLUTION DATED [•], 2021 AND EFFECTIVE ON [•], 2021 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

#### THE COMPANIES ACT (AS REVISED)

#### OF THE CAYMAN ISLANDS

#### **COMPANY LIMITED BY SHARES**

#### AMENDED AND RESTATED

#### MEMORANDUM OF ASSOCIATION

#### OF

#### **SAI.TECH Global Corporation**

#### (ADOPTED BY SPECIAL RESOLUTION DATED [•], 2021 AND EFFECTIVE ON [•], 2021 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

#### 1 The name of the Company is **SAI.TECH Gloabl Corporation**.

The Registered Office of the Company shall be at the offices of ICS Corporate Services (Cayman) Limited, 3-212 Governors Square, 23 Lime Tree Bay Avenue, P.O. Box 30746, Seven Mile Beach, Grand Cayman KY1-1203, Cayman Islands, or at such other place within the Cayman Islands as the Directors may decide.

The objects for which the Company is established are unrestricted and except as prohibited or limited by the laws of the Cayman Islands, the Company shall have full power and authority to carry out any object and shall have and be capable of from time to time and at all times exercising any and all of the powers at any time or from time to time exercisable by a natural person or body corporate in any part of the world whether as principal, agent, contractor or otherwise.

Ex H-1

4

2

The Company will not trade in the Cayman Islands with any Person, firm or corporation except in furtherance of the business of the Company carried on outside the Cayman Islands; provided, that nothing in this section shall be construed as to prevent the Company effecting and concluding contracts in the Cayman Islands, and exercising in the Cayman Islands all of its powers necessary for the carrying on of its business outside the Cayman Islands

- 5 The liability of each Member is limited to the amount unpaid on such Member's shares.
- <sup>6</sup> The authorized share capital of the Company is US\$50,000 divided into 350,000,000 shares of a par value of US\$0.0001 each, of which  $[\bullet]^1$  shall be designated as Class A ordinary shares,  $[\bullet]^2$  shall be designated as convertible Class B ordinary shares, and

10,000,000 preference shares]. Subject to the provisions of the Statue and the Articles, the Company has the power to redeem or purchase any of its shares and to increase or reduce the share capital and to issue any part of its capital, whether original, redeemed or increased with or without any preference, priority or special privilege or subject to any postponement of rights or to any conditions or restrictions and so that unless the conditions of issue shall otherwise expressly declare every issue of shares whether declared to be preference or otherwise shall be subject to the powers hereinbefore contained.

- 7 The Company has power to register by way of continuation as a body corporate limited by shares under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.
- 8 Capitalised terms that are not defined in this Memorandum bear the respective meanings given to them in the Articles.
- 1 Class B ordinary shares will be limited to the number of shares issuable as Merger Consideration at Closing. Balance of undesignated shares will be allocated as authorized Class A ordinary shares.
- 2 Class B ordinary shares will be limited to the number of shares issuable as Merger Consideration at Closing. Balance of undesignated shares will be allocated as authorized Class A ordinary shares.

Ex H-2

#### THE COMPANIES ACT (AS REVISED)

### OF THE CAYMAN ISLANDS

## **COMPANY LIMITED BY SHARES**

#### AMENDED AND RESTATED

### ARTICLES OF ASSOCIATION

OF

#### SAI.TECH Global Corporation

# (ADOPTED BY SPECIAL RESOLUTION DATED [•], 2021 AND EFFECTIVE ON [•], 2021 (BEING THE MERGER EFFECTIVE DATE IMMEDIATELY AFTER CONSUMMATION OF THE MERGER))

#### 1 Interpretation

1.1 In the Articles Table A in the First Schedule to the Statute does not apply and, unless there is something in the subject or context inconsistent therewith, the following words shall have the following meaning:

"Affiliate"

"Applicable Law"

in respect of a Person, means any other Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such Person, and (a) in the case of a natural Person, shall include, without limitation, such Person's spouse, parents, children, siblings, mother-in-law and father-in-law and brothers and sisters-in-law, whether by blood, marriage or adoption or anyone residing in such Person's home, a trust for the benefit of any of the foregoing, a company, partnership or any natural Person or entity wholly or jointly owned by any of the foregoing and (b) in the case of an entity, shall include a partnership, a corporation or any natural Person or entity which directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, such entity.

means, with respect to any Person, all provisions of laws, statutes, ordinances, rules, regulations, permits, certificates, judgments, decisions, decrees or orders of any governmental authority applicable to such Person.

"Articles"	means these amended and restated articles of association of the Company.
"Audit Committee"	means the audit committee of the board of directors of the Company established pursuant to the Articles, or any successor committee.
"Auditor"	means the Person for the time being performing the duties of auditor of the Company (if any).
<b>"Business Combination Agreement"</b>	means that certain Business Combination Agreement among the Company (f/k/a TradeUp Global Corporation), TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability, and SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability, dated as of September 27, 2021.
"Cause"	has the meaning ascribed to such term in Founder's employment agreement with the Company as in effect from time to time.
"Chairman"	means the chairman of the board of Directors, if any, appointed by the Directors pursuant to Article 31.7.
"Clearing House"	means a clearing house recognised by the laws of the jurisdiction in which the Shares (or depositary receipts therefor) are listed or quoted on a stock exchange or interdealer quotation system in such jurisdiction.
"Class A Share"	means a class A ordinary share of a par value of US\$0.0001 in the share capital of the Company.
"Class B Share"	means a convertible class B ordinary share of a par value of US\$0.0001 in the share capital of the Company.

Ex H-3

"Company"means the above named company."Company's Website"means the website of the Company and/or its web-address or domain name (if any)."Compensation Committee"means the compensation committee of the board of directors of the Company established"Designated Stock Exchange"means any United States national securities exchange on which the securities of the Company are listed for trading, including the Nasdaq Capital Market. means the directors for the time being of the Company."Dividend"means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the Articles.
"Compensation Committee"means the compensation committee of the board of directors of the Company established pursuant to the Articles, or any successor committee."Designated Stock Exchange"means any United States national securities exchange on which the securities of the Company are listed for trading, including the Nasdaq Capital Market. means the directors for the time being of the Company."Dividend"means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the
<ul> <li>"Compensation Committee" pursuant to the Articles, or any successor committee.</li> <li>"Designated Stock Exchange" means any United States national securities exchange on which the securities of the Company are listed for trading, including the Nasdaq Capital Market. means the directors for the time being of the Company.</li> <li>"Dividend" means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the</li> </ul>
"Designated Stock Exchange" are listed for trading, including the Nasdaq Capital Market. "Directors" means the directors for the time being of the Company. "Dividend" means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the
"Dividend"
Alticles.
<b>"Electronic Communication"</b> means a communication sent by electronic means, including electronic posting to the Company's Website, transmission to any number, address or internet website (including the website of the Securities and Exchange Commission) or other electronic delivery methods as otherwise decided and approved by the Directors.
<b>"Electronic Record"</b> has the same meaning as in the Electronic Transactions Act.

Copyright © 2022 <u>www.secdatabase.com</u>. All Rights Reserved. Please Consider the Environment Before Printing This Document

"Electronic Transactions Act"	means the Electronic Transactions Act (As Revised) of the Cayman Islands.	
"Exchange Act"	means the United States Securities Exchange Act of 1934, as amended, or any similar U.S. federal statute and the rules and regulations of the Securities and Exchange Commission thereunder, all as the same shall be in effect at the time.	
"Family Members"	means and includes only the following individuals:	
	(a) the applicable individual;	
	(b) the spouse of the applicable individual (including former spouses);	
	(c) the parents of the applicable individual;	
	(d) the lineal descendants of the applicable individual, the siblings of the applicable individual; and	
	(e) the lineal descendants of a sibling of the applicable individual.	
	For purposes of the (a) to (e), the descendants of any individual shall include adopted individuals and their issue but only if the adopted individual was adopted prior to attaining age eighteen.	
"Founder"	means Li Risheng.	
"Incapacity"	means, with respect to an individual, the permanent and total disability of such individual so that such individual is unable to engage in any substantial gainful activity by reason of any medically determinable mental impairment which can be expected to result in death or which has lasted or can be expected to last for a continuous period of not less than twelve (12) months as determined by a licensed medical practitioner. In the event of a dispute regarding whether an individual has suffered an Incapacity, no Incapacity of such individual will be deemed to have occurred unless and until an affirmative ruling regarding such Incapacity has been made by a court or arbitral panel of competent jurisdiction, and such ruling has become final and non-appealable.	
Ex H-4		
"Independent Director"	has the same meaning as in the rules and regulations of the Designated Stock Exchange or in Rule 10A-3 under the Exchange Act, as the case may be.	
"Member"	has the same meaning as in the Statute.	
"Memorandum"	means the amended and restated memorandum of association of the Company.	
"Merger"	has the meaning ascribed to such term in the Business Combination Agreement.	

means a Person appointed to hold an office in the Company.

has the meaning ascribed to such term in the Business Combination Agreement.

Company established pursuant to the Articles, or any successor committee.

means the nominating and corporate governance committee of the board of directors of the

"Merger Effective Date"

**Governance Committee**"

"Officer"

"Nominating and Corporate

"Ordinary Resolution"	means a resolution passed by a simple majority of the Members as, being entitled to do so, vote in Person or, where proxies are allowed, by proxy at a general meeting, and includes a unanimous written resolution. In computing the majority when a poll is demanded regard shall be had to the number of votes to which each Member is entitled by the Articles.
"Ordinary Shares"	means, collectively, the Class A Shares and the Class B Shares.
"Permitted Entity"	means with respect to Founder:
	(a) any Person in respect of which Founder has, directly or indirectly:
	(i) control with respect to the voting of all the Class B Shares held or to be transferred to such Person;
	<ul> <li>the ability to direct or cause the direction of management and policies of such Person</li> <li>(ii) or any other Person having the authority referred to in the proceeding clause (a)(i)</li> <li>(whether by contract, as executor, trustee, trust protector or otherwise); or</li> </ul>
	the operational or practical control of such Person, including the right to appoint, (iii) designate, remove or replace the Person having the authority referred to in the preceding clauses (a)(i) or (ii) above;
	(b) any trust the beneficiaries of which consist primarily of Founder, his Family Members, and/or any Persons controlled directly or indirectly by such a trust;
	(c) any Person Controlled by a trust described in the immediately preceding clause (b) above; and
	(d) for the avoidance of doubt, Energy Science Artist Holding Limited.

### Ex H-5

"Permitted Transferee"	means, with respect to the Class B Ordinary Shareholders, any or all of the following:
	(a) Founder;
	(b) any Founder Permitted Entities;
	(c) any Family Members of Founder;
	(d) any other relative of Founder approved by the Board;
	(e) any transferee by operation of law, including in connection with divorce proceedings;
	(f) any charitable organization, foundation or similar entity;
	in connection with a transfer as a result of, or in connection with the death or Incapacity of (g) Founder, any Family Member of Founder or other designee, in each case, approved by the Directors; and
	(h) the Company;
	provided that, (1) as a condition of any transfer to a Permitted Transferee under clauses (c) through (f) above, such Permitted Transferee shall have executed an original, a counterpart of, or a deed of adherence with respect to, a Proxy and Voting Deed; and (ii) in the case of any

	transfer of Class B Shares pursuant to clauses (b) through (f) above to a Person who at a later time ceases to qualify as a Permitted Transferee under the relevant clause, the Company shall be able to refuse registration of any subsequent transfer of such Class B Shares except back to the Founder or his Permitted Entities, and in the absence of such transfer back to the Founder or his Permitted Entities, the applicable Class B Shares shall automatically convert in accordance with Article 17.4 applied <i>mutatis mutandis</i> .
"Person"	means any individual, corporation, partnership, joint venture, trust, unincorporated organization, limited liability company, estate, association, joint stock company, unincorporated organization, company or other form of business or legal entity or government authority.
"Preference Share"	means a preference share of a par value of US\$0.0001 in the share capital of the Company.
"Proxy and Voting Deed"	means a deed or agreement conveying to Founder voting rights of outstanding Class B Shares.
"Register of Members"	means the register of Members maintained in accordance with the Statute and includes (except where otherwise stated) any branch or duplicate register of Members.
"Registered Office"	means the registered office for the time being of the Company.
"Seal"	means the common seal of the Company and includes every duplicate seal.
"Securities Act"	means the U.S. Securities Act of 1933.
"Securities and Exchange Commission"	means the United States Securities and Exchange Commission.
"Share"	means a Class A Share, a Class B Share, or a Preference Share and includes a fraction of a share in the Company.
"Special Resolution"	subject to Article 29.4, has the same meaning as in the Statute, and includes a unanimous written resolution.
"Statute"	means the Companies Act (As Revised) of the Cayman Islands.
"Treasury Share"	means a Share held in the name of the Company as a treasury share in accordance with the Statute.

Ex H-6

#### 1.2 In the Articles:

- (a) words importing the singular number include the plural number and vice versa;
- (b) words importing the masculine gender include the feminine gender;
- (c) references to any agreement, deed or other instrument (including to these Articles and the Memorandum) is a reference to that agreement or instrument as amended, restated, or novated from time to time;
- (d) "written" and "in writing" include all modes of representing or reproducing words in visible form, including in the form of an Electronic Record;
- (e) "shall" shall be construed as imperative and "may" shall be construed as permissive;

- (f) references to provisions of any law or regulation shall be construed as references to those provisions as amended, modified, re-enacted or replaced;
- (g) any phrase introduced by the terms "including", "include", "in particular" or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms;

the term "and/or" is used herein to mean both "and" as well as "or." The use of "and/or" in certain contexts in no respects qualifies or modifies the use of the terms "and" or "or" in others. The term "or" shall not be interpreted to be exclusive and the term "and" shall not be interpreted to require the conjunctive (in each case, unless the context otherwise requires);

- (i) headings are inserted for reference only and shall be ignored in construing the Articles;
- (j) any requirements as to delivery under the Articles include delivery in the form of an Electronic Record;
- (k) any requirements as to execution or signature under the Articles including the execution of the Articles themselves can be satisfied in the form of an electronic signature as defined in the Electronic Transactions Act;
- (l) sections 8 and 19(3) of the Electronic Transactions Act shall not apply;
- (m) the term "clear days" in relation to the period of a notice means that period excluding the day when the notice is received or deemed to be received and the day for which it is given or on which it is to take effect;
- (n) a reference to "\$", "US\$", "USD", or dollar(s) is a reference to the lawful currency of the United States of America;
- "directly or indirectly" means, except where the context requires otherwise, directly or indirectly through one or more intermediate Persons or through contractual or other arrangements, and "direct or indirect" has the correlative meaning; and
- (p) the term "holder" in relation to a Share means a Person whose name is entered in the Register of Members as the holder of such Share.

#### 2 Commencement of Business

(h)

- 2.1 The business of the Company may be commenced as soon after incorporation of the Company as the Directors shall see fit.
- 2.2 The Directors may pay, out of the capital or any other monies of the Company, all expenses incurred in or about the formation and establishment of the Company, including the expenses of registration.

Ex H-7

#### **3** Issue of Shares and other Securities

Subject to the provisions, if any, in the Memorandum (and to any direction that may be given by the Company in general meeting) and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, and without prejudice to any rights attached to any existing Shares, the Directors may allot, issue, grant options over or otherwise dispose of Shares (including

3.1 fractions of a Share) with or without preferred, deferred or other rights or restrictions, whether in regard to Dividends or other distributions, voting, return of capital or otherwise and to such Persons, at such times and on such other terms as they think proper, and may also (subject to the Statute and the Articles) vary such rights, save that the Directors shall not allot, issue, grant options over or otherwise dispose of Shares (including fractions of a Share) to the extent that it may affect the ability of the Company to carry out a conversion of the Class B Shares set out in the Articles. No holder of Ordinary Shares shall have pre-emptive rights.

- The Company may issue rights, options, warrants or convertible securities or securities of similar nature conferring the right
   upon the holders thereof to subscribe for, purchase or receive any class of Shares or other securities in the Company on such terms as the Directors may from time to time determine.
- Except as set forth otherwise in Article 10(c)(iv), the Directors may provide, out of the unissued shares, for a series of Preference
   Shares. Before any Preference Shares of any such series are issued, the Directors shall fix, by resolution or resolutions, the following provisions of the Preference Shares thereof, if applicable:
  - (a) the designation of such series, the number of Preference Shares to constitute such series and the subscription price thereof if different from the par value thereof;
  - (b) whether the shares of such series shall have voting rights, in addition to any voting rights provided by law, and, if so, the terms of such voting rights, which may be general or limited;
  - (c) the dividends, if any, payable on such series, whether any such dividends shall be cumulative, and, if so, from what dates, the conditions and dates upon which such dividends shall be payable, the preference or relation which such dividends shall bear to the dividends payable on any shares of any other class or any other series of Preference Shares;
  - (d) whether the Preference Shares of such series shall be subject to redemption by the Company, and, if so, the times, prices and other conditions of such redemption;
  - (e) the amount or amounts payable upon Preference Shares of such series upon, and the rights of the holders of such series in, a voluntary or involuntary liquidation, dissolution or winding up, or upon any distribution of the assets, of the Company;
  - (f) whether the Preference Shares of such series shall be subject to the operation of a retirement or sinking fund and, if so, the extent to and manner in which any such retirement or sinking fund shall be applied to the purchase or redemption of the Preference Shares of such series for retirement or other corporate purposes and the terms and provisions relative to the operation thereof;
  - (g) whether the Preference Shares of such series shall be convertible into, or exchangeable for, shares of any other class or any other series of Preference Shares or any other securities and, if so, the price or prices or the rate or rates of conversion or exchange and the method, if any, of adjusting the same, and any other terms and conditions of conversion or exchange;

#### Ex H-8

(h)

the limitations and restrictions, if any, to be effective while any Preference Shares of such series are outstanding upon the payment of dividends or the making of other distributions on, and upon the purchase, redemption or other acquisition by the Company of, the existing shares or shares of any other class of shares or any other series of Preference Shares;

- the conditions or restrictions, if any, upon the creation of indebtedness of the Company or upon the issue of any additional shares, including additional shares of such series or of any other class of shares or any other series of Preference Shares; and
- (j) any other powers, preferences and relative, participating, optional and other special rights, and any qualifications, limitations and restrictions thereof.

Without limiting the foregoing and subject to <u>Articles 17.3 and 23.1</u>, the voting powers of any series of Preference Shares may include the right, in the circumstances specified in the resolution or resolutions providing for the issuance of such Preference Shares, to elect one or more Directors who shall serve for such term and have such voting powers as shall be stated in the resolution or resolutions providing for the issuance of such Preference Shares. The term of office and voting powers of any Director elected in the manner provided in the immediately preceding sentence of this Article 3.3 may be greater than or less than those of any other Director or class of Directors.

The powers, preferences and relative, participating, optional and other special rights of each series of Preference Shares, and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. All shares of any one series of Preference Shares shall be identical in all respects with all other shares of such series, except that

shares of any one series issued at different times may differ as to the dates from which dividends thereon shall be cumulative.

## 3.5 The Company shall not issue Shares to bearer.

#### 4 Register of Members

3.4

5.1

6.1

4.1 The Company shall maintain or cause to be maintained the Register of Members in accordance with the Statute.

The Directors may determine that the Company shall maintain one or more branch Registers of Members in accordance with
 the Statute. The Directors may also determine which Register of Members shall constitute the principal register and which shall constitute the branch register or registers, and to vary such determination from time to time.

#### 5 Closing Register of Members or Fixing Record Date

For the purpose of determining Members entitled to notice of, or to vote at any meeting of Members or any adjournment thereof, or Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose, the Directors may, after notice has been given by advertisement in an appointed newspaper or any other newspaper or by any other means in accordance with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, provide that the Register of Members shall be closed for transfers for a stated period which shall not in any case exceed forty days.

5.2 In lieu of, or apart from, closing the Register of Members, the Directors may fix in advance or arrears a date as the record date for any such determination of Members entitled to notice of, or to vote at any meeting of the Members or any adjournment thereof, or for the purpose of determining the Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose.

5.3 If the Register of Members is not so closed and no record date is fixed for the determination of Members entitled to notice of, or to vote at, a meeting of Members or Members entitled to receive payment of a Dividend or other distribution, the date on which notice of the meeting is sent or the date on which the resolution of the Directors resolving to pay such Dividend or other distribution is passed, as the case may be, shall be the record date for such determination of Members. When a determination of Members entitled to vote at any meeting of Members has been made as provided in this Article, such determination shall apply to any adjournment thereof.

Ex H-9

#### 6 Certificates for Shares

A Member shall only be entitled to a share certificate if the Directors resolve that share certificates shall be issued. Share certificates representing Shares, if any, shall be in such form as the Directors may determine. Share certificates shall be signed by one or more Directors or other Person authorised by the Directors. The Directors may authorise certificates to be issued with the authorised signature(s) affixed by mechanical process. All certificates for Shares shall be consecutively numbered or otherwise identified and shall specify the Shares to which they relate. All share certificates shall be cancelled and, subject to the Articles, no new certificate shall be issued until the former certificate representing a like number of relevant Shares shall have been surrendered and cancelled.

6.2 The Company shall not be bound to issue more than one certificate for Shares held jointly by more than one Person and delivery of a certificate to one joint holder shall be a sufficient delivery to all of them.

- If a share certificate is defaced, worn out, lost or destroyed, it may be renewed on such terms (if any) as to evidence and
   indemnity and on the payment of such expenses reasonably incurred by the Company in investigating evidence, as the Directors may prescribe, and (in the case of defacement or wearing out) upon delivery of the old certificate.
- 6.4 Every share certificate sent in accordance with the Articles will be sent at the risk of the Member or other Person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

Share certificates shall be issued within the relevant time limit as prescribed by the Statute, if applicable, or as the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law may from time to time determine, whichever is shorter, after the allotment or, except in the case of a Share transfer which the Company is for the time being entitled to refuse to register and does not register, after lodgement of a Share transfer with the Company.

#### 7 Transfer of Shares

6.5

7.1

8.2

Subject to the terms of the Articles, any Member may transfer all or any of his Class A Shares by an instrument of transfer provided that such transfer complies with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law. If the Class A Shares in question were issued in conjunction with rights, options, warrants or units issued pursuant to the Articles on terms that one cannot be transferred without the other, the Directors shall refuse to register the transfer of any such Share without evidence satisfactory to them of the like transfer of such right, option, warrant or unit.

The instrument of transfer of any Class A Share shall be in writing in the usual or common form or in a form prescribed by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law or in any other form approved by the Directors and shall be executed by

7.2 or on behalf of the transferor (and if the Directors so require, signed by or on behalf of the transferee) and may be under hand or, if the transferor or transferee is a Clearing House or its nominee(s), by hand or by machine imprinted signature or by such other manner of execution as the Directors may approve from time to time. The transferor shall be deemed to remain the holder of a Share until the name of the transferee is entered in the Register of Members.

#### Ex H-10

#### 8 Redemption, Repurchase and Surrender of Shares

Subject to the provisions of the Statute, and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, the
Company may issue Shares that are to be redeemed or are liable to be redeemed at the option of the Member or the Company. The redemption of such Shares shall be effected in such manner and upon such other terms as the Company may, by Special Resolution, determine before the issue of such Shares.

Subject to the provisions of the Statute, and, where applicable, the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, the Company may purchase its own Shares (including any redeemable Shares) in such manner and on such other terms as the Directors may agree with the relevant Member. For the avoidance of doubt, redemptions, repurchases and surrenders of Shares in the circumstances described in the Article above shall not require further approval of the Members.

- 8.3 The Company may make a payment in respect of the redemption or purchase of its own Shares in any manner permitted by the Statute, including out of capital.
- 8.4 The Directors may accept the surrender for no consideration of any fully paid Share.
- 9 Treasury Shares

- 9.1 The Directors may, prior to the purchase, redemption or surrender of any Share, determine that such Share shall be held as a Treasury Share.
- 9.2 The Directors may determine to cancel a Treasury Share or transfer a Treasury Share on such terms as they think proper (including, without limitation, for nil consideration).

#### 10 Variation of Rights of Shares

Subject to Articles 3.1 and 17.3, if at any time the share capital of the Company is divided into different classes of Shares, all or any of the rights attached to any class (unless otherwise provided by the terms of issue of the Shares of that class) may, whether or not the Company is being wound up, be varied without the consent of the holders of the issued Shares of that class where such variation is considered by the Directors not to have a material adverse effect upon such rights; otherwise, any such variation shall be made only with the consent in writing of the holders of not less than two thirds of the issued Shares of that class or with

- 10.1 the approval of a resolution passed by a majority of not less than two thirds of the votes cast at a separate meeting of the holders of the Shares of that class. For the avoidance of doubt, the Directors reserve the right, notwithstanding that any such variation may not have a material adverse effect, to obtain consent from the holders of Shares of the relevant class. To any such meeting all the provisions of the Articles relating to general meetings shall apply *mutatis mutandis*, except that the necessary quorum shall be one Person holding or representing by proxy at least one third of the issued Shares of the class and that any holder of Shares of the class present in Person or by proxy may demand a poll.
- 10.2 The provisions of these Articles relating to general meetings shall apply to every such general meeting of the holders of one class or series of shares, except the following:
  - (a) separate general meetings of the holders of a call or series of shares may be called only by:
    - (i) the Chairman, if any;
    - (ii) a majority of all of the Directors (unless otherwise specifically provided by the terms of issue of the shares of such class or series); or
    - (iii) with respect to general meetings of the holders of Class B Shares, the Founder.
  - (b) except as set forth in clause (a) above, nothing in this Article 10 shall be deemed to give any Member the right to call a class or series meeting; and
  - (c) the necessary quorum shall be one or more Persons holding or representing by proxy at least one-third of the issues shares of the class or series and that any holder of shares of the class or series present in person or by proxy may demand a poll.

10.3 The rights conferred upon the holders of the Shares of any class issued with preferred or other rights shall not, unless otherwise expressly provided by the terms of issue of the Shares of that class, be deemed to be varied by the creation or issue of further Shares ranking *pari passu* therewith or Shares issued with preferred or other rights.

#### Ex H-11

#### 11 Commission on Sale of Shares

The Company may, in so far as the Statute permits, pay a commission to any Person in consideration of his subscribing or agreeing to subscribe (whether absolutely or conditionally) or procuring or agreeing to procure subscriptions (whether absolutely or conditionally) for any Shares. Such commissions may be satisfied by the payment of cash and/or the issue of fully or partly paid-up Shares. The Company may also on any issue of Shares pay such brokerage as may be lawful.

#### 12 Non Recognition of Trusts

No Person shall be recognised by the Company as holding any share upon any trust (other than any trust recognized as a Permitted Entity or Permitted Transferee). The Company shall not be bound by or compelled to recognise in any way (even

when notified) any equitable, contingent, future or partial interest in any Share, or (except only as is otherwise provided by the Articles or the Statute) any other rights in respect of any Share other than an absolute right to the entirety thereof in the holder. It is understood that a Proxy and Voting Deed shall not be the holding of any share upon a trust.

#### 13 Lien on Shares

13.1 The Company shall have a first and paramount lien on all Shares (whether fully paid-up or not) registered in the name of a Member (whether solely or jointly with others) for all debts, liabilities or engagements to or with the Company (whether presently payable or not) by such Member or his estate, either alone or jointly with any other Person, whether a Member or not, but the Directors may at any time declare any Share to be wholly or in part exempt from the provisions of this Article. The registration of a transfer of any such Share shall operate as a waiver of the Company's lien thereon. The Company's lien on a Share shall also extend to any amount payable in respect of that Share.

13.2 The Company may sell, in such manner as the Directors think fit, any Shares on which the Company has a lien, if a sum in respect of which the lien exists is presently payable, and is not paid within fourteen clear days after notice has been received or deemed to have been received by the holder of the Shares, or to the Person entitled to it in consequence of the death or bankruptcy of the holder, demanding payment and stating that if the notice is not complied with the Shares may be sold.

To give effect to any such sale the Directors may authorise any Person to execute an instrument of transfer of the Shares sold to, or in accordance with the directions of, the purchaser. The purchaser or his nominee shall be registered as the holder of the 13.3 Shares comprised in any such transfer, and he shall not be bound to see to the application of the purchase money, nor shall his title to the Shares be affected by any irregularity or invalidity in the sale or the exercise of the Company's power of sale under the Articles.

The net proceeds of such sale after payment of costs, shall be applied in payment of such part of the amount in respect of which the lien exists as is presently payable and any balance shall (subject to a like lien for sums not presently payable as existed upon the Shares before the sale) be paid to the Person entitled to the Shares at the date of the sale.

#### 14 Call on Shares

14.1 Subject to the terms of the allotment and issue of any Shares, the Directors may make calls upon the Members in respect of any monies unpaid on their Shares (whether in respect of par value or premium), and each Member shall (subject to receiving at least fourteen clear days' notice specifying the time or times of payment) pay to the Company at the time or times so specified the amount called on the Shares. A call may be revoked or postponed, in whole or in part, as the Directors may determine. A call may be required to be paid by instalments. A Person upon whom a call is made shall remain liable for calls made upon him notwithstanding the subsequent transfer of the Shares in respect of which the call was made.

Ex H-12

- 14.2 A call shall be deemed to have been made at the time when the resolution of the Directors authorising such call was passed.
- 14.3 The joint holders of a Share shall be jointly and severally liable to pay all calls in respect thereof.

14.4 If a call remains unpaid after it has become due and payable, the Person from whom it is due shall pay interest on the amount unpaid from the day it became due and payable until it is paid at such rate as the Directors may determine (and in addition all expenses that have been incurred by the Company by reason of such non-payment), but the Directors may waive payment of the interest or expenses wholly or in part.

An amount payable in respect of a Share on issue or allotment or at any fixed date, whether on account of the par value of the 14.5 Share or premium or otherwise, shall be deemed to be a call and if it is not paid all the provisions of the Articles shall apply as if that amount had become due and payable by virtue of a call.

14.6 The Directors may issue Shares with different terms as to the amount and times of payment of calls, or the interest to be paid.

The Directors may, if they think fit, receive an amount from any Member willing to advance all or any part of the monies uncalled and unpaid upon any Shares held by him, and may (until the amount would otherwise become payable) pay interest at such rate as may be agreed upon between the Directors and the Member paying such amount in advance.

No such amount paid in advance of calls shall entitle the Member paying such amount to any portion of a Dividend or other distribution payable in respect of any period prior to the date upon which such amount would, but for such payment, become payable.

#### 15 Forfeiture of Shares

If a call or instalment of a call remains unpaid after it has become due and payable the Directors may give to the Person from whom it is due not less than fourteen clear days' notice requiring payment of the amount unpaid together with any interest which 15.1 may have accrued and any expenses incurred by the Company by reason of such non-payment. The notice shall specify where payment is to be made and shall state that if the notice is not complied with the Shares in respect of which the call was made will be liable to be forfeited.

If the notice is not complied with, any Share in respect of which it was given may, before the payment required by the notice has been made, be forfeited by a resolution of the Directors. Such forfeiture shall include all Dividends, other distributions or other monies payable in respect of the forfeited Share and not paid before the forfeiture.

15.3 A forfeited Share (other than a Class B Share) may be sold, re-allotted or otherwise disposed of on such terms and in such manner as the Directors think fit and at any time before a sale, re-allotment or disposition the forfeiture may be cancelled on such terms as the Directors think fit. Where for the purposes of its disposal a forfeited Share is to be transferred to any Person the Directors may authorise some Person to execute an instrument of transfer of the Share in favour of that Person.

A Person any of whose Shares have been forfeited shall cease to be a Member in respect of them and shall surrender to the Company for cancellation the certificate for the Shares forfeited and shall remain liable to pay to the Company all monies which 15.4 at the date of forfeiture were payable by him to the Company in respect of those Shares together with interest at such rate as the Directors may determine, but his liability shall cease if and when the Company shall have received payment in full of all monies due and payable by him in respect of those Shares.

A certificate in writing under the hand of one Director or Officer that a Share has been forfeited on a specified date shall be conclusive evidence of the facts stated in it as against all Persons claiming to be entitled to the Share. The certificate shall (subject to the execution of an instrument of transfer) constitute a good title to the Share and the Person to whom the Share is sold or otherwise disposed of shall not be bound to see to the application of the purchase money, if any, nor shall his title to the Share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the Share.

The provisions of the Articles as to forfeiture shall apply in the case of non payment of any sum which, by the terms of issue of a Share, becomes payable at a fixed time, whether on account of the par value of the Share or by way of premium as if it had been payable by virtue of a call duly made and notified.

#### Ex H-13

#### 16 Transmission of Shares

If a Member dies, the survivor or survivors (where he was a joint holder), or his legal personal representatives (where he was a sole holder), shall be the only Persons recognised by the Company as having any title to his Shares. The estate of a deceased Member is not thereby released from any liability in respect of any Share, for which he was a joint or sole holder.

Any Person becoming entitled to a Share in consequence of the death or bankruptcy or liquidation or dissolution of a Member (or in any other way than by transfer) may, upon such evidence being produced as may be required by the Directors, elect, by a notice in writing sent by him to the Company, either to become the holder of such Share or to have some Person nominated by him registered as the holder of such Share. If he elects to have another Person registered as the holder of such Share he shall sign an instrument of transfer of that Share to that Person. The Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution, as the case may be.

A Person becoming entitled to a Share by reason of the death or bankruptcy or liquidation or dissolution of a Member (or in any other case than by transfer) shall be entitled to the same Dividends, other distributions and other advantages to which he would be entitled if he were the holder of such Share. However, he shall not, before becoming a Member in respect of a Share, be entitled in respect of it to exercise any right conferred by membership in relation to general meetings of the Company and the Directors may at any time give notice requiring any such Person to elect either to be registered himself or to have some Person

16.3 nominated by him be registered as the holder of the Share (but the Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution or any other case than by transfer, as the case may be). If the notice is not complied with within ninety days of being received or deemed to be received (as determined pursuant to the Articles), the Directors may thereafter withhold payment of all Dividends, other distributions, bonuses or other monies payable in respect of the Share until the requirements of the notice have been complied with.

#### 17 Rights and Restrictions Attaching to Ordinary Shares

Except as otherwise included in these Articles (including Articles 17.3 and 23.1), the rights attaching to the Class A Shares and Class B Shares shall have the same rights and powers, and rank *pari passu* in all respects (including as to dividends, distributions, and upon the occurrence of any liquidation or winding up of the Company), share ratably and be identical in all respects as to all matters, unless different treatment of the shares of each such class is approved by the affirmative vote of the holders of the Class A Shares and the Class B Shares, each voting exclusively and as a separate class.

Holders of Ordinary Shares have the right to receive notice of, attend, speak and vote at general meetings of the Company. Except as set forth in these Articles (including Articles 17.3, 17.4 and 23.1), holders of Class A Shares and Class B Shares shall at all times vote together as one class on all matters submitted to a vote for Members' consent. On all matters subject to a vote of the Members, Ordinary Shares shall be entitled to the voting rights as set forth in Article 23.

17.3 In addition to any rights provided by applicable law or otherwise set forth in these Articles, the Company shall not, without the approval by vote or written consent of the holders of a majority of the voting power of the Class B Shares, voting exclusively and as a separate class, directly or indirectly, or whether by amendment or through merger, recapitalization, consolidation or otherwise:

(a) increase the number of authorized Class B Shares;

issue any Class B Shares or securities convertible into or exchangeable for Class B Shares, other than to Founder or his(b) Affiliates (provided, that his Affiliate shall have executed an original, a counterpart of, or a deed of adherence with respect to, a Proxy and Voting Deed in respect of such Class B Shares);

(c) create, authorize, issue, or reclassify into, any Preference Shares or any shares in the capital of the Company that carry more than one (1) vote per share;

#### Ex H-14

(d) reclassify any Class B Shares into any other class of shares or consolidate or combine any Class B Shares without proportionately increasing the number of votes per Class B Share; or

amend, restate, waive, adopt any provision inconsistent with or otherwise vary or alter any provision of the Memorandum (e) or these Articles relating to the voting, conversion or other rights, powers, preferences, privileges or restrictions of the Class

- (e) or these Articles relating to the voting, conversion or other rights, powers, preferences, privileges or restrictions of the Class B Shares.
- 17.4 Optional and Automatic Conversion of Class B Shares:

each Class B Share is convertible into one (1) Class A Share (as adjusted for share splits, share combinations and similar(a) transactions occurring after the Merger Effective Date) at any time at the option of the holder thereof. In no event shall any Class A Share be convertible into any Class B Shares;

each Class B Share will automatically convert into one (1) Class A Share (as adjusted for share splits, share combinations(b) and similar transactions occurring after the Merger Effective Date) on the earliest to occur of 5:00 p.m., Cayman Islands time:

(i) on the first anniversary of Founder's death or Incapacity; or

on a date determined by the Board during the period commencing 90 days after, and ending 180 days after, the date (ii) on which Founder is terminated for Cause (and in the event of a dispute regarding whether there was Cause, Cause will be deemed not to exist unless and until an affirmative ruling regarding such Cause has been made by a court

- or arbitral panel of competent jurisdiction, and such ruling has become final and non-appealable);
- (c) no Class B Shares shall be issued by the Company after conversion of all Class B Shares into Class A Shares;

upon any sale, pledge, transfer, assignment or other disposition of Class B Shares by a holder thereof to any Person which is not a Permitted Transferee of such holder, each such Class B Share shall be automatically and immediately converted into one (1) Class A Share (as adjusted for share splits, share combinations and similar transactions occurring after the Acquisition Effective Time); provided that, notwithstanding anything to the contrary in these Articles, any pledge of Class B Shares by a holder thereof that creates a security interest in such Class B Shares pursuant to a bona fide loan or indebtedness

- (d) transaction shall be permitted (and not result in any such conversion) for so long as Founder or his Affiliates continue to control, directly or indirectly, the exercise of the voting rights of such pledged Class B Shares; provided, further, however, that a foreclosure on such Class B Shares or other similar action by the pledgee will result in automatic and immediate conversion of such Class B Shares into Class A Shares unless the transferee in such foreclosure or similar action qualifies as a Permitted Transferee at such time. For the avoidance of doubt, any sale, pledge, transfer, assignment or disposition of Class B Shares to a Permitted Transferee does not result in automatic conversion into Class A Shares.
- 17.5 The Company may, from time to time, establish such policies and procedures relating to the conversion of the Class B Shares to Class A Shares, including the issuance of share certificates with respect thereto, as it may deem necessary or advisable.

The Company shall at all times reserve and keep available out of its authorized but unissued Class A Shares, solely for the purpose of effecting the conversion of the Class B Shares, such number of its Class A Shares as shall from time to time be sufficient to effect the conversion of all outstanding Class B Shares; and if at any time the number of authorized but unissued Class A Shares shall not be sufficient to effect the conversion of all then-outstanding Class B Shares, the Company will take such corporate action as may, in the opinion of its counsel, be necessary to increase its authorized but unissued Class A Shares to such numbers of shares as shall be sufficient for such purpose.

#### 18 Amendments of Memorandum and Articles of Association and Alteration of Capital

17.6

- 18.1 The Company may by Ordinary Resolution, subject to the rights of the Class B Shares, including under Article 17.3:
  - (a) increase its share capital by such sum as the Ordinary Resolution shall prescribe and with such rights, priorities and privileges annexed thereto, as the Company in general meeting may determine;
  - (b) consolidate and divide all or any of its share capital into Shares of larger amount than its existing Shares;

by subdivision of its existing Shares or any of them divide the whole or any part of its share capital into Shares of smaller(c) amount; provided, that in the subdivision the amount paid and the amount, if any, unpaid on each reduced share shall be the same as it was in case of the share from which the reduced share is derived; or

(d) cancel any Shares that at the date of the passing of the Ordinary Resolution have not been taken or agreed to be taken by any Person and diminish the amount of its share capital by the amount of the Shares so cancelled.

- All new Shares created in accordance with the provisions of the preceding Article shall be subject to the same provisions of the 18.2 Articles with reference to the payment of calls, liens, transfer, transmission, forfeiture and otherwise as the Shares in the original share capital.
- 18.3 Subject to the rights of the Class B Shares, including under Article 17.3(e), the provisions of the Statute, the provisions of the Articles as regards the matters to be dealt with by Ordinary Resolution, the Company may by Special Resolution:
  - (a) change its name;
  - (b) alter or add to the Articles;
  - (c) alter or add to the Memorandum with respect to any objects, powers or other matters specified therein; and
  - (d) reduce its share capital or any capital redemption reserve fund.

#### 19 Offices and Places of Business

Subject to the provisions of the Statute, the Company may by resolution of the Directors change the location of its Registered Office. The Company may, in addition to its Registered Office, maintain such other offices or places of business as the Directors determine.

#### 20 General Meetings

20.4

20.1 All general meetings other than annual general meetings shall be called extraordinary general meetings.

The Company shall in each year hold a general meeting as its annual general meeting, and shall specify the meeting as such in 20.2 the notices calling it. Any annual general meeting shall be held at such time and place as the Directors shall appoint. At these meetings the report of the Directors (if any) shall be presented.

20.3 The Directors, the chief executive officer or the Chairman may call general meetings, and, for the avoidance of doubt, Members shall not have the ability to call general meetings.

Members seeking to bring business before the annual general meeting or to nominate candidates for appointment as Directors at the annual general meeting must deliver notice to the principal executive offices of the Company not less than 120 calendar days before the date of the Company's proxy statement released to Members in connection with the previous year's annual general meeting or, if the Company did not hold an annual general meeting the previous year, or if the date of the current year's annual general meeting has been changed by more than 30 days from the date of the previous year's annual general meeting, then the deadline shall be set by the board of Directors with such deadline being a reasonable time before the Company begins to print

Ex H-16

#### 21 Notice of General Meetings

and send its related proxy materials.

At least five clear days' notice shall be given of any general meeting. Every notice shall specify the place, the day and the hour of the meeting and the general nature of the business to be conducted at the general meeting and shall be given in the manner hereinafter mentioned or in such other manner if any as may be prescribed by the Company, provided that a general meeting of the Company shall, whether or not the notice specified in this Article has been given and whether or not the provisions of the Articles regarding general meetings have been complied with, be deemed to have been duly convened if it is so agreed:

(a) in the case of an annual general meeting, by all of the Members (or their proxies) entitled to attend and vote thereat; and

in the case of an extraordinary general meeting, by Members (or their proxies) having the right to attend and vote at the

- (b) meeting, together holding shares entitling the holders thereof to not less than two-thirds of the votes entitled to be cast at such extraordinary meeting.
- 21.2 The accidental omission to give notice of a general meeting to, or the non-receipt of notice of a general meeting by, any Person entitled to receive such notice shall not invalidate the proceedings of that general meeting.

#### 22 Proceedings at General Meetings

22.1 No business shall be transacted at any general meeting unless a quorum of Members is present. The holders of a majority of the Shares being individuals present in Person or by proxy or if a corporation or other non-natural Person by its duly authorised representative or proxy shall be a quorum for all purposes; provided, that the presence in person or by proxy of holders of a majority of the Class B Shares shall be required in any event.

If provided by the Company, a Person may participate at a general meeting by conference telephone or other communications equipment by means of which all the Persons participating in the meeting can communicate with each other. Participation by a Person in a general meeting in this manner is treated as presence in Person at that meeting.

22.3 A resolution (including a Special Resolution) in writing (in one or more counterparts) signed by or on behalf of all of the Members for the time being entitled to receive notice of and to attend and vote at general meetings (or, being corporations or other non-natural Persons, signed by their duly authorised representatives) shall be as valid and effective as if the resolution had been passed at a general meeting of the Company duly convened and held.

22.4 If a quorum is not present within half an hour from the time appointed for the meeting to commence, the meeting shall stand adjourned to the same day in the next week at the same time and/or place or to such other day, time and/or place as the Directors may determine, and if at the adjourned meeting a quorum is not present within half an hour from the time appointed for the meeting to commence, the Members present shall be a quorum.

#### Ex H-17

22.5 The Directors may, at the time prior to the time appointed for the meeting to commence, appoint any Person to act as a chairman of a general meeting or, if the Directors do not make any such appointment, the chairman, if any, of the board of Directors (as elected by a majority of the Directors in accordance with Article 31.7) shall preside as chairman at such general meeting. If there is no chairman present, or if the chairman is unwilling to act, the Directors present shall elect one of their number to be chairman of the meeting. The voting rights of the Chairman, if a Director, as to the matters to be decided by the Board of Directors shall be the same as other Directors.

22.6 The chairman may, with the consent of a meeting at which a quorum is present (and shall if so directed by the meeting) adjourn the meeting from time to time and from place to place (provided, that no special meeting called by the holders of Class B Shares may be adjourned unless a quorum does not exist), but no business shall be transacted at any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place.

22.7 When a general meeting is adjourned for thirty days or more, notice of the adjourned meeting shall be given as in the case of an original meeting. Otherwise, it shall not be necessary to give any such notice of an adjourned meeting.

- 22.8 A resolution put to the vote of the meeting shall be decided on a poll.
- 22.9 A poll shall be taken as the chairman directs, and the result of the poll shall be deemed to be the resolution of the general meeting at which the poll was demanded.
- 22.10 A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith (provided, that no special meeting called by a holder of Class B Shares may be adjourned unless a quorum does not exist). A poll demanded on any other question shall be taken at such date, time and place as the chairman of the general meeting directs, and any business other than that upon which a poll has been demanded or is contingent thereon may proceed pending the taking of the poll.

22.11 In the case of an equality of votes the chairman shall be entitled to a second or casting vote.

#### 23 Votes of Members

Subject to any rights or restrictions attached to any Shares, including in Articles 17.3 and 17.4, each Class A Share shall be entitled to one (1) vote on all matters subject to a vote of Members, and each Class B Share shall be entitled to ten (10) votes on all matters subject to a vote of Members.

23.2 In the case of joint holders the vote of the senior holder who tenders a vote, whether in Person or by proxy (or, in the case of a corporation or other non-natural Person, by its duly authorised representative or proxy), shall be accepted to the exclusion of the votes of the other joint holders, and seniority shall be determined by the order in which the names of the holders stand in the Register of Members.

A Member of unsound mind, or in respect of whom an order has been made by any court, having jurisdiction in lunacy, may vote by his committee, receiver, curator bonis, or other Person on such Member's behalf appointed by that court, and any such committee, receiver, curator bonis or other Person may vote by proxy.

23.4 No Person shall be entitled to vote at any general meeting unless he is registered as a Member on the record date for such meeting nor unless all calls or other monies then payable by him in respect of Shares have been paid.

No objection shall be raised as to the qualification of any voter except at the general meeting or adjourned general meeting at which the vote objected to is given or tendered and every vote not disallowed at the meeting shall be valid. Any objection made in due time in accordance with this Article shall be referred to the chairman whose decision shall be final and conclusive.

23.6 Votes may be cast either personally or by proxy (or in the case of a corporation or other non-natural Person by its duly authorised representative or proxy). A Member may appoint more than one proxy or the same proxy under one or more instruments to attend and vote at a meeting. Where a Member appoints more than one proxy the instrument of proxy shall specify the number of Shares in respect of which each proxy is entitled to exercise the related votes.

A Member holding more than one Share need not cast the votes in respect of his Shares in the same way on any resolution and therefore may vote a Share or some or all such Shares either for or against a resolution and/or abstain from voting a Share or some or all of the Shares and, subject to the terms of the instrument appointing him, a proxy appointed under one or more instruments may vote a Share or some or all of the Shares in respect of which he is appointed either for or against a resolution and/or abstain from voting a Share or some or all of the Shares in respect of which he is appointed.

#### Ex H-18

#### 24 Proxies

24.1 The instrument appointing a proxy shall be in writing and shall be executed under the hand of the appointor or of his attorney duly authorised in writing, or, if the appointor is a corporation or other non natural Person, under the hand of its duly authorised representative. A proxy need not be a Member.

The Directors may, in the notice convening any meeting or adjourned meeting, or in an instrument of proxy sent out by the Company, specify the manner by which the instrument appointing a proxy shall be deposited and the place and the time (being not later than the time appointed for the commencement of the meeting or adjourned meeting to which the proxy relates) at

24.2 which the instrument appointed for the commencement of the incenting of adjourned meeting to which the proxy relates) at which the instrument appointing a proxy shall be deposited. In the absence of any such direction from the Directors in the notice convening any meeting or adjourned meeting or in an instrument of proxy sent out by the Company, the instrument appointing a proxy shall be deposited physically at the Registered Office not less than 48 hours before the time appointed for the meeting or adjourned meeting to commence at which the Person named in the instrument proposes to vote.

The chairman may in any event at his discretion declare that an instrument of proxy shall be deemed to have been duly deposited.
 An instrument of proxy that is not deposited in the manner permitted, or which has not been declared to have been duly deposited by the chairman, shall be invalid.

24.4 The instrument appointing a proxy may be in any usual or common form (or such other form as the Directors may approve) and a proxy shall be deemed to include the power to demand or join or concur in demanding a poll.

Votes given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous death or insanity of the principal or revocation of the proxy or of the authority under which the proxy was executed, or the transfer of the Share
 in respect of which the proxy is given unless notice in writing of such death, insanity, revocation or transfer was received by the Company at the Registered Office before the commencement of the general meeting, or adjourned meeting at which it is sought to use the proxy.

#### 25 Corporate Members

Any corporation or other non-natural Person which is a Member may in accordance with its constitutional documents, or in the absence of such provision by resolution of its directors or other governing body, authorise such Person as it thinks fit to act as its representative at any meeting of the Company or of any class of Members, and the Person so authorised shall be entitled to exercise the same powers on behalf of the corporation which he represents as the corporation could exercise if it were an individual Member.

If a Clearing House (or its nominee(s)), being a corporation, is a Member, it may authorise such Persons as it sees fit to act as its representative at any meeting of the Company or at any meeting of any class of Members provided that the authorisation shall specify the number and class of Shares in respect of which each such representative is so authorised. Each Person so authorised under the provisions of this Article shall be deemed to have been duly authorised without further evidence of the facts and be entitled to exercise the same rights and powers on behalf of the Clearing House (or its nominee(s)) as if such Person was the registered holder of such Shares held by the Clearing House (or its nominee(s)).

#### 26 Shares that May Not be Voted

Treasury Shares and other shares in the Company that are beneficially owned by the Company shall not be voted, directly or indirectly, at any meeting and shall not be counted in determining the total number of outstanding Shares at any given time.

#### 27 Directors

27.2

27.1 There shall be a board of Directors consisting of not less than one Person provided however that the Company may by Ordinary Resolution increase or reduce the limits in the number of Directors.

The Directors shall be divided into three classes: Class I, Class II and Class III. The number of Directors in each class shall be as nearly equal as possible. Upon the adoption of the Articles, the existing Directors shall by resolution classify themselves as Class I, Class II or Class III Directors. The Class I Directors shall stand appointed for a term expiring at the Company's first annual general meeting, the Class II Directors shall stand appointed for a term expiring at the Company's second annual general meeting and the Class III Directors shall stand appointed for a term expiring at the Company's third annual general meeting. Commencing at the Company's first annual general meeting, and at each annual general meeting thereafter, Directors appointed to succeed those Directors whose terms expire shall be appointed for a term of office to expire at the third succeeding annual general meeting after their appointment. Except as the Statute or other Applicable Law may otherwise require, in the interim between annual general meetings or extraordinary general meetings called for the appointment of Directors and/or the removal of one or more Directors and the filling of any vacancy in that connection, additional Directors and any vacancies in the board of Directors, including unfilled vacancies resulting from the removal of Directors for cause, may be filled by the vote of a majority of the remaining Directors then in office, although less than a quorum (as defined in the Articles), or by the sole remaining Director. All Directors shall hold office until the expiration of their respective terms of office and until their successors shall have been appointed and qualified. A Director appointed to fill a vacancy resulting from the death, resignation or removal of a Director shall serve for the remainder of the full term of the Director whose death, resignation or removal shall have created such vacancy and until his successor shall have been appointed and qualified.

#### 28 **Powers of Directors**

Subject to the provisions of the Statute, the Memorandum and the Articles and to any directions given by Special Resolution, the business of the Company shall be managed by the Directors who may exercise all the powers of the Company. No alteration of the Memorandum or Articles and no such direction shall invalidate any prior act of the Directors which would have been valid if that alteration had not been made or that direction had not been given. A duly convened meeting of Directors at which a quorum is present may exercise all powers exercisable by the Directors.

All cheques, promissory notes, drafts, bills of exchange and other negotiable or transferable instruments and all receipts for 28.2 monies paid to the Company shall be signed, drawn, accepted, endorsed or otherwise executed as the case may be in such manner as the Directors shall determine by resolution.

The Directors on behalf of the Company may pay a gratuity or pension or allowance on retirement to any Director who has held any other salaried office or place of profit with the Company or to his widow or dependants and may make contributions to any fund and pay premiums for the purchase or provision of any such gratuity, pension or allowance.

28.4 The Directors may exercise all the powers of the Company to borrow money and to mortgage or charge its undertaking, property and assets (present and future) and uncalled capital or any part thereof and to issue debentures, debenture stock, mortgages, bonds and other such securities whether outright or as security for any debt, liability or obligation of the Company or of any third party.

#### 29 Appointment and Removal of Directors

29.1 The Directors may appoint any Person to be a Director, either to fill a vacancy or as an additional Director provided that the appointment does not cause the number of Directors to exceed any number fixed by or in accordance with the Articles as the maximum number of Directors.

29.2 The Company may by Ordinary Resolution appoint any Person to be a Director or may by Ordinary Resolution remove any Director.

#### **30** Vacation of Office of Director

Notwithstanding anything in these Articles, the office of a Director shall be vacated if:

(a) the Director gives notice in writing to the Company that he resigns the office of Director; or

the Director absents himself (for the avoidance of doubt, without being represented by proxy) from three consecutive(b) meetings of the board of Directors without special leave of absence from the Directors, and the Directors pass a resolution that he has by reason of such absence vacated office; or

(c) the Director dies, becomes bankrupt or makes any arrangement or composition with his creditors generally; or

(i) with respect to any Director (other than Founder), a licensed medical practitioner who has evaluated that Director gives

(d) a written opinion to the Company stating he or she has become physically or mentally incapable of acting as a Director (with reasonable accommodations) and may remain so for more than three (3) months, or (ii) with respect to Founder his Incapacity shall have been determined.

#### 31 Proceedings of Directors

31.1 The quorum for the transaction of the business of the Directors may be fixed by the Directors, and unless so fixed shall be a majority of the Directors then in office.

Subject to the provisions of the Articles, the Directors may regulate their proceedings as they think fit. Questions arising at any meeting shall be decided by a majority of votes. In the case of an equality of votes, the Chairman shall have a second or casting vote.

A Person may participate in a meeting of the Directors or any committee of Directors by conference telephone or other communications equipment by means of which all the Persons participating in the meeting can communicate with each other at the same time. Participation by a Person in a meeting in this manner is treated as presence in Person at that meeting. Unless otherwise determined by the Directors, the meeting shall be deemed to be held at the place where the Chairman is located at the start of the meeting.

31.4 A resolution in writing (in one or more counterparts) signed by all the Directors or all the members of a committee of the Directors or, in the case of a resolution in writing relating to the removal of any Director or the vacation of office by any Director, all of the Directors other than the Director who is the subject of such resolution shall be as valid and effectual as if it had been passed at a meeting of the Directors, or committee of Directors as the case may be, duly convened and held.

31.5 A Director may, or other Officer on the direction of a Director shall, call a meeting of the Directors by at least two days' notice in writing to every Director which notice shall set forth the general nature of the business to be considered unless notice is waived by all the Directors either at, before or after the meeting is held. To any such notice of a meeting of the Directors all the provisions of the Articles relating to the giving of notices by the Company to the Members shall apply *mutatis mutandis*.

Ex H-20

31.6 The continuing Directors (or a sole continuing Director, as the case may be) may act notwithstanding any vacancy in their body, but if and so long as their number is reduced below the number fixed by or pursuant to the Articles as the necessary quorum of Directors the continuing Directors or Director may act for the purpose of increasing the number of Directors to be equal to such fixed number, or of summoning a general meeting of the Company, but for no other purpose.

The Directors may elect a chairman of their board and determine the period for which he is to hold office; but if no such Chairman is elected, or if at any meeting the Chairman is not present within five minutes after the time appointed for the meeting to commence, the Directors present may choose one of their number to be chairman of the meeting.

31.8 All acts done by any meeting of the Directors or of a committee of the Directors shall, notwithstanding that it is afterwards discovered that there was some defect in the appointment of any Director, and/or that they or any of them were disqualified, and/ or had vacated their office and/or were not entitled to vote, be as valid as if every such Person had been duly appointed and/or not disqualified to be a Director and/or had not vacated their office and/or had been entitled to vote, as the case may be.

31.9 A Director may be represented at any meetings of the board of Directors by a proxy appointed in writing by him. The proxy shall count towards the quorum and the vote of the proxy shall for all purposes be deemed to be that of the appointing Director.

#### 32 Presumption of Assent

31.3

A Director who is present at a meeting of the board of Directors at which action on any Company matter is taken shall be presumed to have assented to the action taken unless his dissent shall be entered in the minutes of the meeting or unless he shall file his written dissent from such action with the Person acting as the chairman or secretary of the meeting before the adjournment thereof or shall forward such dissent by registered post to such Person immediately after the adjournment of the meeting. Such right to dissent shall not apply to a Director who voted in favour of such action.

### 33 Directors' Interests

- 33.1 A Director may hold any other office or place of profit under the Company (other than the office of Auditor) in conjunction with his office of Director for such period and on such terms as to remuneration and otherwise as the Directors may determine.
- 33.2 A Director may act by himself or by, through or on behalf of his firm in a professional capacity for the Company and he or his firm shall be entitled to remuneration for professional services as if he were not a Director.

33.3 A Director may be or become a director or other officer of or otherwise interested in any company promoted by the Company or in which the Company may be interested as a shareholder, a contracting party or otherwise, and no such Director shall be accountable to the Company for any remuneration or other benefits received by him as a director or officer of, or from his interest in, such other company.

No Person shall be disqualified from the office of Director or prevented by such office from contracting with the Company, either as vendor, purchaser or otherwise, nor shall any such contract or any contract or transaction entered into by or on behalf of the Company in which any Director shall be in any way interested be or be liable to be avoided, nor shall any Director so contracting or being so interested be liable to account to the Company for any profit realised by or arising in connection with any such contract or transaction by reason of such Director holding office or of the fiduciary relationship thereby established. A Director shall be at liberty to vote in respect of any contract or transaction in which he is interested provided that the nature of the interest of any Director in any such contract or transaction shall be disclosed by him at or prior to its consideration and any vote thereon.

33.5 A general notice that a Director is a shareholder, director, officer or employee of any specified firm or company and is to be regarded as interested in any transaction with such firm or company shall be sufficient disclosure for the purposes of voting on a resolution in respect of a contract or transaction in which he has an interest, and after such general notice it shall not be necessary to give special notice relating to any particular transaction.

#### 34 Minutes

The Directors shall cause minutes to be made in books kept for the purpose of recording all appointments of Officers made by the Directors, all proceedings at meetings of the Company or the holders of any class of Shares and of the Directors, and of committees of the Directors, including the names of the Directors present at each meeting.

#### **35 Delegation of Directors' Powers**

35.1 The Directors may delegate any of their powers, authorities and discretions, including the power to sub-delegate, to any committee consisting of one or more Directors (including, without limitation, the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee). Any such delegation may be made subject to any conditions the Directors may impose and either collaterally with or to the exclusion of their own powers and any such delegation may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of a committee of Directors shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

35.2 The Directors may establish any committees, local boards or agencies or appoint any Person to be a manager or agent for managing the affairs of the Company and may appoint any Person to be a member of such committees, local boards or agencies. Any such appointment may be made subject to any conditions the Directors may impose, and either collaterally with or to the exclusion of their own powers and any such appointment may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of any such committee, local board or agency shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

#### Ex H-22

35.3

necessary to exercise the rights of such committee set forth in the Articles and shall have such powers as the Directors may delegate pursuant to the Articles and as required by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law. Each of the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee, if established, shall consist of such number of Directors as the Directors shall from time to time determine (or such minimum number as may be required from time to time by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law). For so long as any class of Shares is listed on the Designated Stock Exchange, the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee and the Nominating and Corporate Governance Committee and the Nominating and Corporate Governance as is required from time to time by the rules and regulatory authority or otherwise under Applicable Law). For so long as any class of Shares is listed on the Designated Stock Exchange, the Audit Committee, the Compensation Committee and the Nominating and Corporate Governance Committee shall be made up of such number of Independent Directors as is required from time to

The Directors may adopt formal written charters for committees. Each of these committees shall be empowered to do all things

time by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law.

The Directors may by power of attorney or otherwise appoint any Person to be the agent of the Company on such conditions as the Directors may determine, provided that the delegation is not to the exclusion of their own powers and may be revoked by the Directors at any time.

The Directors may by power of attorney or otherwise appoint any company, firm, Person or body of Persons, whether nominated directly or indirectly by the Directors, to be the attorney or authorised signatory of the Company for such purpose and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under the Articles) and for such

35.5 period and subject to such conditions as they may think fit, and any such powers of attorney or other appointment may contain such provisions for the protection and convenience of Persons dealing with any such attorneys or authorised signatories as the Directors may think fit and may also authorise any such attorney or authorised signatory to delegate all or any of the powers, authorities and discretions vested in him.

35.6 The Directors may appoint such Officers as they consider necessary on such terms, at such remuneration and to perform such duties, and subject to such provisions as to disqualification and removal as the Directors may think fit. Unless otherwise specified in the terms of his appointment an Officer may be removed by resolution of the Directors or Members. An Officer may vacate his office at any time if he gives notice in writing to the Company that he resigns his office.

#### 36 No Minimum Shareholding

The Company in general meeting may fix a minimum shareholding required to be held by a Director, but unless and until such a shareholding qualification is fixed a Director is not required to hold Shares.

#### **37** Remuneration of Directors

37.1 The remuneration to be paid to the Directors, if any, shall be such remuneration as the Directors shall determine. The Directors shall also be entitled to be paid all travelling, hotel and other expenses properly and reasonably incurred by them in connection with their attendance at meetings of Directors or committees of Directors, or general meetings of the Company, or separate meetings of the holders of any class of Shares or debentures of the Company, or otherwise in connection with the business of the Company or the discharge of their duties as a Director, or to receive a fixed allowance in respect thereof as may be determined by the Directors, or a combination partly of one such method and partly the other.

The Directors may by resolution approve additional remuneration to any Director for any services which in the opinion of the Directors go beyond his ordinary routine work as a Director. Any fees paid to a Director who is also counsel, attorney or solicitor to the Company, or otherwise serves it in a professional capacity shall be in addition to his remuneration as a Director.

#### 38 Seal

The Company may, if the Directors so determine, have a Seal. The Seal shall only be used by the authority of the Directors or 38.1 of a committee of the Directors authorised by the Directors. Every instrument to which the Seal has been affixed shall be signed by at least one Person who shall be either a Director or some Officer or other Person appointed by the Directors for the purpose.

The Company may have for use in any place or places outside the Cayman Islands a duplicate Seal or Seals each of which shall be a facsimile of the common Seal of the Company and, if the Directors so determine, with the addition on its face of the name of every place where it is to be used.

A Director or Officer, representative or attorney of the Company may without further authority of the Directors affix the Seal over his signature alone to any document of the Company required to be authenticated by him under seal or to be filed with the Registrar of Companies in the Cayman Islands or elsewhere wheresoever.

Subject to the Statute and this Article and except as otherwise provided by the rights attached to any Shares, the Directors may resolve to pay Dividends and other distributions on Shares in issue and authorise payment of the Dividends or other distributions out of the funds of the Company lawfully available therefor. A Dividend shall be deemed to be an interim Dividend unless the

- 39.1 terms of the resolution pursuant to which the Directors resolve to pay such Dividend specifically state that such Dividend shall be a final Dividend. No Dividend or other distribution shall be paid except out of the realised or unrealised profits of the Company, out of the share premium account or as otherwise permitted by law. Notwithstanding anything contained in these Articles, no Dividend shall be made on any Class A Shares unless a Dividend in equal proportion is made on the Class B Shares.
- Except as otherwise provided by the rights attached to any Shares, all Dividends and other distributions shall be paid according to the par value of the Shares that a Member holds. If any Share is issued on terms providing that it shall rank for Dividend as from a particular date, that Share shall rank for Dividend accordingly.
- 39.3 The Directors may deduct from any Dividend or other distribution payable to any Member all sums of money (if any) then payable by him to the Company on account of calls or otherwise.

The Directors may resolve that any Dividend or other distribution be paid wholly or partly by the distribution of specific assets and in particular (but without limitation) by the distribution of shares, debentures, or securities of any other company or in any one or more of such ways and where any difficulty arises in regard to such distribution, the Directors may settle the same as they

- 39.4 think expedient and in particular may issue fractional Shares and may fix the value for distribution of such specific assets or any part thereof and may determine that cash payments shall be made to any Members upon the basis of the value so fixed in order to adjust the rights of all Members and may vest any such specific assets in trustees in such manner as may seem expedient to the Directors.
- Except as otherwise provided by the rights attached to any Shares, Dividends and other distributions may be paid in any currency. 39.5 The Directors may determine the basis of conversion for any currency conversions that may be required and how any costs involved are to be met.
- The Directors may, before resolving to pay any Dividend or other distribution, set aside such sums as they think proper as a reserve or reserves which shall, at the discretion of the Directors, be applicable for any purpose of the Company and pending such application may, at the discretion of the Directors, be employed in the business of the Company.

Any Dividend, other distribution, interest or other monies payable in cash in respect of Shares may be paid by wire transfer to the holder or by cheque or warrant sent through the post directed to the registered address of the holder or, in the case of joint holders, to the registered address of the holder who is first named on the Register of Members or to such Person and to such address as such holder or joint holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the Person to whom it is sent. Any one of two or more joint holders may give effectual receipts for any Dividends, other distributions, bonuses, or other monies payable in respect of the Share held by them as joint holders.

Ex H-24

39.8 No Dividend or other distribution shall bear interest against the Company.

Any Dividend or other distribution which cannot be paid to a Member and/or which remains unclaimed after six months from the date on which such Dividend or other distribution becomes payable may, in the discretion of the Directors, be paid into a separate account in the Company's name, provided that the Company shall not be constituted as a trustee in respect of that account and the Dividend or other distribution shall remain as a debt due to the Member. Any Dividend or other distribution which remains unclaimed after a period of six years from the date on which such Dividend or other distribution becomes payable shall be forfeited and shall revert to the Company.

#### 40 Capitalisation

The Directors may at any time capitalise any sum standing to the credit of any of the Company's reserve accounts or funds (including the share premium account and capital redemption reserve fund) or any sum standing to the credit of the profit and loss account or otherwise available for distribution; appropriate such sum to Members in the proportions in which such sum

would have been divisible amongst such Members had the same been a distribution of profits by way of Dividend or other distribution; and apply such sum on their behalf in paying up in full unissued Shares for allotment and distribution credited as fully paid-up to and amongst them in the proportion aforesaid. In such event the Directors shall do all acts and things required to give effect to such capitalisation, with full power given to the Directors to make such provisions as they think fit in the case of Shares becoming distributable in fractions (including provisions whereby the benefit of fractional entitlements accrue to the Company rather than to the Members concerned). The Directors may authorise any Person to enter on behalf of all of the Members interested into an agreement with the Company providing for such capitalisation and matters incidental or relating thereto and any agreement made under such authority shall be effective and binding on all such Members and the Company.

#### 41 Books of Account

41.1

The Directors shall cause proper books of account (including, where applicable, material underlying documentation including contracts and invoices) to be kept with respect to all sums of money received and expended by the Company and the matters in respect of which the receipt or expenditure takes place, all sales and purchases of goods by the Company and the assets and liabilities of the Company. Such books of account must be retained for a minimum period of five years from the date on which they are prepared. Proper books shall not be deemed to be kept if there are not kept such books of account as are necessary to give a true and fair view of the state of the Company's affairs and to explain its transactions.

41.2 The Directors shall determine whether and to what extent and at what times and places and under what conditions or regulations the accounts and books of the Company or any of them shall be open to the inspection of Members not being Directors and no Member (not being a Director) shall have any right of inspecting any account or book or document of the Company except as conferred by Statute or authorised by the Directors or by the Company in general meeting.

#### Ex H-25

41.3 The Directors may cause to be prepared and to be laid before the Company in general meeting profit and loss accounts, balance sheets, group accounts (if any) and such other reports and accounts as may be required by law.

#### 42 Audit

Applicable Law.

42.2

42.1 The Directors may appoint an Auditor of the Company who shall hold office on such terms as the Directors determine.

Without prejudice to the freedom of the Directors to establish any other committee, if the Shares (or depositary receipts therefor) are listed or quoted on the Designated Stock Exchange, and if required by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under Applicable Law, the Directors shall establish and maintain an Audit Committee as a committee of the Directors and shall adopt a formal written Audit Committee charter and review and assess the adequacy of the formal written charter on an annual basis. The composition and responsibilities of the Audit Committee shall comply with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory authority or otherwise under

- 42.3 If the Shares (or depositary receipts therefor) are listed or quoted on the Designated Stock Exchange, the Company shall conduct
   42.3 an appropriate review of all related party transactions on an ongoing basis and shall utilise the Audit Committee for the review and approval of potential conflicts of interest.
- 42.4 The remuneration of the Auditor shall be fixed by the Audit Committee (if one exists).

If the office of Auditor becomes vacant by resignation or death of the Auditor, or by his becoming incapable of acting by reason of illness or other disability at a time when his services are required, the Directors shall fill the vacancy and determine the remuneration of such Auditor.

Every Auditor of the Company shall have a right of access at all times to the books and accounts and vouchers of the Company 42.6 and shall be entitled to require from the Directors and Officers such information and explanation as may be necessary for the performance of the duties of the Auditor. Auditors shall, if so required by the Directors, make a report on the accounts of the Company during their tenure of office at the next annual general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an ordinary company, and at the next extraordinary general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an exempted company, and at any other time during their term of office, upon request of the Directors or any general meeting of the Members.

At least one member of the Audit Committee shall be an "audit committee financial expert" as determined by the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/or any other competent regulatory 42.8 authority or otherwise under Applicable Law. The "audit committee financial expert" shall have such past employment experience in finance or accounting, requisite professional certification in accounting, or any other comparable experience or background which results in the individual's financial sophistication.

#### Ex H-26

#### 43 Notices

42.7

Notices shall be in writing and may be given by the Company to any Member either personally or by sending it by courier, post, cable, telex, fax or e-mail to him or to his address as shown in the Register of Members (or where the notice is given by 43.1 e-mail by sending it to the e-mail address provided by such Member). Notice may also be served by Electronic Communication in accordance with the rules and regulations of the Designated Stock Exchange, the Securities and Exchange Commission and/ or any other competent regulatory authority or by placing it on the Company's Website.

43.2 Where a notice is sent by:

courier; service of the notice shall be deemed to be effected by delivery of the notice to a courier company, and shall be

(a) deemed to have been received on the third day (not including Saturdays or Sundays or public holidays) following the day on which the notice was delivered to the courier;

post; service of the notice shall be deemed to be effected by properly addressing, pre paying and posting a letter containing (b) the notice, and shall be deemed to have been received on the fifth day (not including Saturdays or Sundays or public holidays in the Cayman Islands) following the day on which the notice was posted;

cable, telex or fax; service of the notice shall be deemed to be effected by properly addressing and sending such notice and (c) shall be deemed to have been received on the same day that it was transmitted;

e-mail or other Electronic Communication; service of the notice shall be deemed to be effected by transmitting the e-mail (d) to the e-mail address provided by the intended recipient and shall be deemed to have been received on the same day that it was sent, and it shall not be necessary for the receipt of the e-mail to be acknowledged by the recipient; and

(e) placing it on the Company's Website; service of the notice shall be deemed to have been effected one hour after the notice or document was placed on the Company's Website.

A notice may be given by the Company to the Person or Persons which the Company has been advised are entitled to a Share or Shares in consequence of the death or bankruptcy of a Member in the same manner as other notices which are required to be given under the Articles and shall be addressed to them by name, or by the title of representatives of the deceased, or trustee 43.3 of the bankrupt, or by any like description at the address supplied for that purpose by the Persons claiming to be so entitled, or at the option of the Company by giving the notice in any manner in which the same might have been given if the death or bankruptcy had not occurred.

Notice of every general meeting shall be given in any manner authorised by the Articles to every holder of Shares carrying an entitlement to receive such notice on the record date for such meeting except that in the case of joint holders the notice shall be sufficient if given to the joint holder first named in the Register of Members and every Person upon whom the ownership of a 43.4 Share devolves by reason of his being a legal personal representative or a trustee in bankruptcy of a Member where the Member but for his death or bankruptcy would be entitled to receive notice of the meeting, and no other Person shall be entitled to receive notices of general meetings.

#### 44 Winding Up

- If the Company shall be wound up, the liquidator shall apply the assets of the Company in satisfaction of creditors' claims in 44.1 such manner and order as such liquidator thinks fit. Subject to the rights attaching to any Shares, in a winding up:
  - if the assets available for distribution amongst the Members shall be insufficient to repay the whole of the Company's issued (a) share capital, such assets shall be distributed so that, as nearly as may be, the losses shall be borne by the Members in proportion to the par value of the Shares held by them; or

if the assets available for distribution amongst the Members shall be more than sufficient to repay the whole of the Company's issued share capital at the commencement of the winding up, the surplus shall be distributed amongst the

(b) Members in proportion to the par value of the Shares held by them at the commencement of the winding up subject to a deduction from those Shares in respect of which there are monies due, of all monies payable to the Company for unpaid calls or otherwise.

If the Company shall be wound up the liquidator may, subject to the rights attaching to any Shares and with the approval of a Special Resolution of the Company and any other approval required by the Statute, divide amongst the Members in kind the whole or any part of the assets of the Company (whether such assets shall consist of property of the same kind or not) and may for that purpose value any assets and determine how the division shall be carried out as between the Members or different classes of Members. The liquidator may, with the like approval, yest the whole or any part of such assets in trustees upon such trusts for the benefit of the Members as the liquidator, with the like approval, shall think fit, but so that no Member shall be compelled to accept any asset upon which there is a liability.

#### 45 **Indemnity and Insurance**

44.2

Every Director and Officer (which for the avoidance of doubt, shall not include auditors of the Company), together with every former Director and former Officer (each an "Indemnified Person") shall be indemnified out of the assets of the Company against any liability, action, proceeding, claim, demand, costs, damages or expenses, including legal expenses, whatsoever which they or any of them may incur as a result of any act or failure to act in carrying out their functions other than such liability (if 45.1 any) that they may incur by reason of their own actual fraud, wilful neglect or wilful default. No Indemnified Person shall be liable to the Company for any loss or damage incurred by the Company as a result (whether direct or indirect) of the carrying out of their functions unless that liability arises through the actual fraud, wilful neglect or wilful default of such Indemnified Person. No Person shall be found to have committed actual fraud, wilful neglect or wilful default under this Article unless or until a court of competent jurisdiction shall have made a finding to that effect.

The Company shall advance to each Indemnified Person reasonable attorneys' fees and other costs and expenses incurred in connection with the defence of any action, suit, proceeding or investigation involving such Indemnified Person for which indemnity will or could be sought. In connection with any advance of any expenses hereunder, the Indemnified Person shall execute an undertaking to repay the advanced amount to the Company if it shall be determined by final judgment or other final 45.2 adjudication that such Indemnified Person was not entitled to indemnification pursuant to this Article. If it shall be determined by a final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification with respect to such judgment, costs or expenses, then such party shall not be indemnified with respect to such judgment, costs or expenses and any advancement shall be returned to the Company (without interest) by the Indemnified Person.

45.3

The Directors, on behalf of the Company, may purchase and maintain insurance for the benefit of any Director or Officer against any liability which, by virtue of any rule of law, would otherwise attach to such Person in respect of any negligence, default, breach of duty or breach of trust of which such Person may be guilty in relation to the Company.

#### 46 Financial Year

Unless the Directors otherwise prescribe, the financial year of the Company shall end on 31st December in each year and, following the year of incorporation, shall begin on 1st January in each year.

#### 47 Transfer by Way of Continuation

If the Company is exempted as defined in the Statute, it shall, subject to the provisions of the Statute and with the approval of a Special Resolution, have the power to register by way of continuation as a body corporate under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.

#### 48 Mergers and Consolidations

The Company shall have the power to merge or consolidate with one or more other constituent companies (as defined in the Statute) upon such terms as the Directors may determine and (to the extent required by the Statute) with the approval of a Special Resolution.

#### 49 Business Opportunities

49.1

To the fullest extent permitted by Applicable Law, no individual serving as a Director or an Officer ("**Management**") shall have any duty, except and to the extent expressly assumed by contract, to refrain from engaging directly or indirectly in the same or similar business activities or lines of business as the Company. To the fullest extent permitted by Applicable Law, the Company renounces any interest or expectancy of the Company in, or in being offered an opportunity to participate in, any potential transaction or matter which may be a corporate opportunity for Management, on the one hand, and the Company, on the other. Except to the extent expressly assumed by contract, to the fullest extent permitted by Applicable Law, Management shall have no duty to communicate or offer any such corporate opportunity to the Company and shall not be liable to the Company or its Members for breach of any fiduciary duty as a Member, Director and/or Officer solely by reason of the fact that such party pursues or acquires such corporate opportunity for itself, himself or herself, directs such corporate opportunity to another Person, or does not communicate information regarding such corporate opportunity to the Company.

49.2 Except as provided elsewhere in this Article, the Company hereby renounces any interest or expectancy of the Company in, or in being offered an opportunity to participate in, any potential transaction or matter which may be a corporate opportunity for both the Company and Management, about which a Director and/or Officer who is also a member of Management acquires knowledge.

To the extent a court might hold that the conduct of any activity related to a corporate opportunity that is renounced in this Article to be a breach of duty to the Company or its Members, the Company hereby waives, to the fullest extent permitted by 49.3 Applicable Law, any and all claims and causes of action that the Company may have for such activities. To the fullest extent permitted by Applicable Law, the provisions of this Article apply equally to activities conducted in the future and that have been conducted in the past.

Ex H-29

### <u>Exhibit I</u>

#### Form of Waiver and Release

**Final Form** 

#### WAIVER AND RELEASE

This WAIVER AND RELEASE (this "*Release*") is made and executed as of the [\_\_], 2021, by and among SAITECH Limited ("*Company*") and those Persons identified on the signature pages hereto as a "*Shareholder*" representing all of the shareholders of Company (each a "*Shareholder*", and collectively the "*Shareholder*"). As set forth in <u>Section 11</u> below, this Release shall only become

effective as of the Closing of the Purchase Agreement (as defined below), and if the Purchase Agreement is terminated prior to any such Closing, this Release shall terminate and be null and void *ab initio* and shall have no further force and effect.

#### BACKGROUND

A. Company, TradeUp Global Corporation, a Cayman Islands exempted company incorporated with limited liability with company number 370735 (together with its successors, "*Purchaser*"), and TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability with company number 380624 and a direct wholly owned subsidiary of Purchaser, have entered into a Business Combination Agreement dated September 27, 2021, (together with the Exhibits and Schedules thereto, the "*Purchase Agreement*"), pursuant to which, among other things, (i) Merger Sub would be merged with and into the Company, with the Company being the surviving company and becoming a wholly owned subsidiary of Purchaser and (ii) the all of the issued and outstanding shares of capital stock of the Company held by the Shareholders would automatically convert into shares of capital stock of the Purchaser, all on the terms and conditions set forth in the Purchase Agreement.

B. The Purchase Agreement provides, among other things, that the closing of the transactions contemplated thereby is conditioned upon, among other things, the execution and delivery by the Company and the Shareholders of a waiver and release as it relates to potential claims by any Shareholder under (i) that certain Series A Preferred Share and Warrant Purchase Agreement dated April 22, 2021, among the Company, certain subsidiaries of the Company, the Founder Parties (as defined therein) and the Series A Investors (as defined therein) (the "*Series A SPA*"), and (ii) that certain Amended and Restated Shareholders Agreement," and together with the Series A SPA, the "*Subject Agreements*"), which waiver and release is to only become effective upon a closing of the Purchase Agreement.

#### AGREEMENT

NOW, THEREFORE, the parties hereto, each intending to be legally bound, hereby covenant and agree as follows:

1. <u>Background; Defined Terms</u>. The Background provisions set forth above (including, without limitation, the defined terms set forth therein) are hereby incorporated by reference into this Release and made a part hereof as if set forth in their entirety in this <u>Section 1</u>. Capitalized terms used in this Release which are not otherwise defined herein shall have the respective meanings assigned to them in the Purchase Agreement.

2. <u>Acknowledgments; Purchaser as Third Party Beneficiary</u>. Each of the Shareholders, on the one hand, and Company, on the other hand, acknowledges that the execution and delivery of this Release by each of them is a condition to and in consideration of Purchaser's obligations under the Purchase Agreement, and that Purchaser is relying on this Release in consummating such transactions. This Release shall inure to the benefit of and be enforceable by Purchaser as an intended third party beneficiary hereof.

Ex I-1

3. Release of Company Group. Each Shareholder, for and on behalf of itself, himself or herself and each of their respective heirs, executors, administrators, trustees, Representatives, equity holders, successors and assigns, as applicable, and in consideration for and other good and valuable consideration received by such Shareholder (including the consideration payable under the Purchase Agreement), the receipt and sufficiency of which are hereby acknowledged by the Shareholders, do hereby irrevocably remise, waive, release and forever discharge each of the Company and each of its direct and indirect existing or former Subsidiaries and their respective present and former Representatives, equity holders, successors and assigns (collectively, the "Company Group Released Persons"), of and from any and all, and any and all manner of, claims, controversies, suits, rights, interests, losses, debts, actions, causes of action, claims, cross-claims, counter-claims, demands, charges, debts, obligations, compensatory damages, liquidated damages, punitive or exemplary damages, other damages, claims for costs and attorneys' fees, or liabilities of any nature whatsoever in law and in equity, and whether known or unknown, suspected or unsuspected, anticipated or unanticipated (collectively, "Claims"), which any of the Shareholders or any of their respective heirs, executors, administrators, trustees, representatives, stockholders, members, equity holders, successors or assigns now have, have ever had or may hereafter have against any of the Company Group Released Parties arising under the Subject Agreements, including with respect to any matter, event, cause, omission or failure under the Subject Agreements occurring prior to, on or after the Closing, such as, any breach or violation by the Company (or any of its present or former Subsidiaries) of any representation, warranty, covenant or agreement set forth in either of the Subject Agreements; provided, however, that nothing contained herein shall operate to remise, release or discharge (a) any Claim arising from or related to any obligations or liabilities of any Acquired Group Released Person or the Purchaser to any Shareholder not arising under, or otherwise attributable to, the Subject Agreements, including any Claim arising from or related to the Purchase Agreement or any of the other agreements (other than the Subject Agreements) to which any of the Target Companies is a party or by which any of them is bound operation and (b) any Claims to the extent such Claims cannot be released as a matter of Law.

4. <u>Waiver of Claims</u>. Each Shareholder, for and on behalf of itself, himself or herself and each of their respective heirs, executors, administrators, trustees, Representatives, equity holders, successors and assigns, as applicable, hereby irrevocably covenants to refrain from, directly or indirectly, asserting any claim or demand, or commencing, instituting or causing to be commenced, any Action of any kind against any Company Group Related Released Persons, based upon any matter purported to be released hereby.

#### 5. Representations and Warranties.

(a) Each Shareholder hereto represents and warrants to each of the Company Group Released Persons that it (i) is aware of no other parties having any interest in any Claim or Claims that are the subject of the releases and waivers by such party set forth in Sections 3 and 4 of this Release, (ii) has not sold, transferred, pledged, collateralized, hypothecated, or assigned or attempted to sell, transfer, pledge, collateralize, hypothecate, or assign any right, title, or interest in any such Claims to any Person not a party to this Release, and (iii) has not, in executing this Release, and does not rely upon any representation or statement not contained in this Release made by any other party to this Release or others released herein, or by any of their Representatives with regard to the subject matter, basis or effect of this Release, or otherwise.

(b) Each of the parties hereto represents and warrants to the other parties that they have full authority to enter into this Release and by their signatures bind the party on whose behalf they have signed the Release.

(c) Each of the Shareholders have had the opportunity to consult with counsel and fully understand the terms of this Release and enter into it voluntarily.

6. <u>Scope of Release</u>. Each of the Shareholders acknowledges that it is aware that the laws of certain jurisdictions may provide in substance as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS WHICH THE CREDITOR DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE, WHICH IF KNOWN BY IT, HIM OR HER MUST HAVE MATERIALLY AFFECTED ITS, HIS OR HER SETTLEMENT WITH THE DEBTOR."

Each of Shareholders further acknowledges that the releases given herein are certain releases of unknown claims. Accordingly, to the extent that the provisions of any law substantially similar to that which is quoted above is applicable to this Release, the parties knowingly and voluntarily waive such provisions and acknowledge and agree that this Release is an essential and material term of the releases and waivers set forth herein and should be given full force and effect. This waiver includes, but is not limited to, a waiver of the right to a claim of fraudulent inducement to enter into this Release as to such unknown facts and such unknown and unsuspected claims, except that it shall not include a waiver of the right to a claim of misrepresentation as to any of the representations and warranties contained herein.

Ex I-2

7. <u>Severability</u>. Whenever possible, each provision of this Release shall be interpreted in such manner as to be valid, binding and enforceable under applicable law, but if any provision of this Release is held to be invalid, void (or voidable) or unenforceable under applicable law, such provision shall be ineffective only to the extent held to be invalid, void (or voidable) or unenforceable, without affecting the remainder of such provision or the remaining provisions of this Release and this Release will be interpreted, construed and enforced as if such invalid, void (or voidable) or unenforceable provision had never been contained herein.

8. <u>Waiver; Remedies Cumulative</u>. The rights and remedies of the parties to this Release are cumulative and not alternative. Neither any failure nor any delay by any party in exercising any right, power or privilege under this Release or any of the documents referred to in this Release will operate as a waiver of such right, power or privilege, and no single or partial exercise of any such right, power or privilege will preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. To the maximum extent permitted by applicable law, (a) no claim or right arising out of this Release or any of

the documents referred to in this Release can be discharged by one party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by each party to be charged thereby and by Purchaser; (b) no waiver that may be given by a party will be applicable except in the specific instance for which it is given; and (c) no notice to or demand on one party will be deemed to be a waiver of any obligation of that party or of the right of the party giving such notice or demand to take further action without notice or demand as provided in this Release or the documents referred to in this Release.

9. <u>No Admission</u>. Each of the Shareholders acknowledge and agree that neither this Release nor the furnishing the consideration for this Release shall be deemed or construed at any time to be an admission by any of the Company Group Released Persons of any breach or violation of either of the Subject Agreements.

10. <u>Governing Law</u>. All matters relating to the interpretation, construction, validity and enforcement of this Release shall be governed by and construed in accordance with the domestic laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.

11. <u>Contingent upon Closing of Purchase Agreement</u>. This Release shall only become effective upon a Closing of the Purchase Agreement, and if the Purchaser Agreement is terminated prior to the occurrence of a Closing, this Release shall terminate and be null and void *ab initio* and shall have no further force and effect.

12. <u>Entire Agreement and Modification</u>. This Release supersedes all prior agreements, whether written or oral, between the parties with respect to its subject matter and constitutes a complete and exclusive statement of the terms of the agreement between the parties with respect to its subject matter. This Release may not be amended, supplemented, or otherwise modified except by a written agreement executed by each party to be charged under such amendment, supplement or modification, and by Purchaser.

13. <u>Counterparts; Electronic Mail and Facsimile Signatures</u>. This Release may be executed in multiple counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument. This Agreement, any and all agreements and instruments executed and delivered in accordance herewith, along with any amendments hereto or thereto, to the extent signed and delivered by means of E-mail, a facsimile machine or other means of electronic transmission, shall be treated in all manner and respects and for all purposes as an original signature, agreement or instrument and shall be considered to have the same binding legal effect as if it were the original signed version thereof delivered in person.

[Signature Page Follows.]

Ex I-3

IN WITNESS WHEREOF, the undersigned have executed and delivered this Release as of date first set forth above.

### **COMPANY**:

### SAITECH LIMITED

By:

Name: Title:

Signature Page to Waiver and Release

Ex I-4

#### **SHAREHOLDERS**:

[TO COME]

Signature Page to Waiver and Release

#### Ex I-5

#### <u>Exhibit J</u>

## Form of Purchaser Equity Plan

**Final Form** 

#### Form of SAI.TECH Global Corporation Incentive Equity Plan

#### **SAI.TECH Global Corporation**

#### 2021 Equity Incentive Plan

#### Adopted by the Board of Directors of SAI.TECH Global Corporation: [], 2021

#### Approved by the Shareholders of SAI.TECH Global Corporation: [], 2021

#### 1. General.

(a) Establishment. The SAI.TECH Global Corporation 2021 Equity Incentive Plan (the "*Plan*") is hereby established effective as of [ ], 2021, which is the date of the closing of the transactions contemplated by the Business Combination Agreement (the "*Effective Date*").

(b) Purpose. The Plan, through the granting of Awards, is intended to help the Company to secure and retain the services of eligible award recipients, provide incentives for such persons to exert maximum efforts for the success of the Company and any Affiliate and provide means by which the eligible recipients may benefit from increases in value of the Ordinary Shares.

(c) Available Awards. The Plan provides for the grant of the following types of Awards: (i) Options, (ii) Share Appreciation Rights, (iii) Restricted Share Awards, (iv) Restricted Share Unit Awards, and (v) Other Awards.

#### 2. Administration.

(a) Administration by Board. The Board will administer the Plan unless and until the Board delegates administration of the Plan to a Committee or Committees, as provided in Section 2(c).

(b) Powers of Board. The Board will have the power, subject to, and within the limitations of, the express provisions of the Plan:

(i) To determine from time to time (A) which of the persons eligible under the Plan will be granted Awards; (B) when and how each Award will be granted; (C) what type or combination of types of Award will be granted; (D) the provisions of each Award granted (which need not be identical or comparable), including the time or times when a person will be permitted to exercise or otherwise receive an issuance of Ordinary Shares or other payment pursuant to an Award; (E) the number of Ordinary Shares or cash equivalent with respect to which an Award will be granted to each such person; and (F) the Fair Market Value applicable to an Award.

(ii) To construe and interpret the Plan and Awards granted under it, and to establish, amend and revoke rules and regulations for administration of the Plan and Awards. The Board, in the exercise of these powers, may correct any defect, omission or inconsistency in the Plan or in any Award Agreement, in a manner and to the extent it will deem necessary or expedient to make the Plan or Award fully effective.

(iii) To settle all controversies regarding the Plan and Awards granted under it.

(iv) To accelerate, in whole or in part, the time at which an Award may be exercised or vest (or at which cash or Ordinary Shares may be issued).

(v) To prohibit the exercise of any Option, SAR or other exercisable Award during a period of up to 30 days prior to the consummation of any pending stock dividend, stock split, combination or exchange of shares, merger, consolidation or other distribution (other than normal cash dividends) of Company assets to shareholders, or any other change affecting the Ordinary Shares or the share price of Ordinary Shares, for reasons of administrative convenience.

(vi) To suspend or terminate the Plan at any time. Except as otherwise provided in the Plan or an Award Agreement, suspension or termination of the Plan will not impair a Participant's rights under his or her then-outstanding Award without his or her written consent.

(vii) To amend the Plan in any respect the Board deems necessary or advisable, subject to the limitations, if any, of applicable law; *provided, however* that shareholder approval will be required for any amendment to the extent required by applicable law. Except as provided in the Plan or an Award Agreement, no amendment of the Plan will materially impair a Participant's rights under an outstanding Award unless (A) the Company requests the consent of the affected Participant, and (B) such Participant consents in writing.

Ex J-1

(viii) To submit any amendment to the Plan for shareholder approval, including, but not limited to, amendments to the Plan intended to satisfy the requirements of Section 422 of the Code regarding Incentive Share Options.

(ix) To approve forms of Award Agreements for use under the Plan and to amend the terms of any one (1) or more Awards, including, but not limited to, amendments to provide terms more favorable to the Participant than previously provided in the Award Agreement, subject to any specified limits in the Plan that are not subject to Board discretion; *provided however*; that a Participant's rights under any Award will not be impaired by any such amendment unless (A) the Company requests the consent of the affected Participant, and (B) such Participant consents in writing. Notwithstanding the foregoing, (1) a Participant's rights will not be deemed to have been impaired by any such amendment if the Board, in its sole discretion, determines that the amendment, taken as a whole, does not materially impair the Participant's rights, and (2) subject to the limitations of applicable law, if any, the Board may amend the terms of any one (1) or more Awards without the affected Participant's consent (X) to maintain the tax qualified status of the Award, (Y) to clarify the manner of exemption from, or to bring the Award into compliance with, Section 409A of the Code or Section 457A of the Code; or (Z) to comply with other applicable laws.

(x) Generally, to exercise such powers and to perform such acts as the Board deems necessary or expedient to promote the best interests of the Company and that are not in conflict with the provisions of the Plan or Awards.

(xi) To adopt such procedures and sub-plans as are necessary or appropriate to permit participation in the Plan by Employees, Directors or Consultants who are foreign nationals or employed outside the United States (provided that Board approval will not be necessary for immaterial modifications to the Plan or any Award Agreement that are required for compliance with the laws of the relevant foreign jurisdiction). Without limiting the generality of the foregoing, the Board specifically is authorized to adopt rules, procedures and sub-plans, regarding, without limitation, conversion of local currency, obligations to pay payroll tax, determination of beneficiary designation requirements, withholding procedures and handling of share issuances, which may vary according to local requirements.

(xii) To effect, at any time and from time to time, (A) the reduction of the exercise, purchase or strike price of any outstanding Award (including without limitation any Option or SAR); (B) the cancellation of any outstanding Award and the grant in substitution therefor of a new (1) Option, Share Appreciation Right, Restricted Share Award, Restricted Share Unit Award, or Other Award under the Plan or another equity plan of the Company, covering the same or a different number of Ordinary Shares, (2) cash and/or (3) other valuable consideration determined by the Board, in its sole discretion; or (C) any other action that is treated as a repricing under generally accepted accounting principles; *provided*, that any repricing that the Board effectuates shall not require approval of the Company's shareholders.

(c) Delegation to Committee.

(i) General. The Board may delegate some or all of the administration of the Plan to a Committee or Committees. If administration of the Plan is delegated to a Committee, the Committee will have, in connection with the administration of the Plan, the powers theretofore possessed by the Board that have been delegated to the Committee, including the power to delegate to a subcommittee of the Committee

any of the administrative powers the Committee is authorized to exercise (and references in this Plan to the Board will thereafter be to the Committee or subcommittee). Any delegation of administrative powers will be reflected in resolutions, not inconsistent with the provisions of the Plan, adopted from time to time by the Board or Committee (as applicable). The Committee may, at any time, abolish the subcommittee and/or revest in the Committee any powers delegated to the subcommittee. The Board may retain the authority to concurrently administer the Plan with the Committee and may, at any time, revest in the Board some or all of the powers previously delegated.

(ii) Rule 16b-3 Compliance. To the extent an Award is intended to qualify of the exemption from Section 16(b) of the Exchange Act that is available under Rule 16b-3 of the Exchange Act, the Plan shall be administered by the Board or the Committee (subject to the Board's discretion) in accordance with the requirements of Rule 16b-3 under the Exchange Act to the extent necessary for such exemption to remain available. The failure of a member of the Board or Committee (as applicable) to qualify as a "non-employee director," as determined under Rule 16b-3(b)(3) of the Exchange Act, shall not invalidate any Award granted or action taken by the Board or Committee (as applicable) that is otherwise validly granted or taken under the terms of the Plan.

(d) Delegation to an Officer. The Board or any Committee may delegate to one (1) or more Officers the authority to do one (1) or both of the following: (i) designate Employees who are not Officers to be recipients of Options and SARs (and, to the extent permitted by applicable law, other Awards) and, to the extent permitted by applicable law, the terms of such Awards and (ii) determine the number of Ordinary Shares to be subject to such Awards granted to such Employees; *provided, however*, that the resolutions or charter adopted by the Board or any Committee evidencing such delegation will specify the total number of Ordinary Shares that may be subject to the Awards granted by such Officer and that such Officer may not grant an Award to himself or herself. Any such Awards will be granted on substantially the form of Award Agreement most recently approved for use by the Committee or the Board, unless otherwise provided in the resolutions approving the delegation authority. Notwithstanding anything to the contrary herein, neither the Board nor any Committee may delegate authority to an Officer who is acting solely in the capacity of an Officer (and not also as a Director) to determine the Fair Market Value (pursuant to Section 13(x) below).

Ex J-2

(e) Effect of Board's Decision. All determinations, interpretations and constructions made by the Board or any Committee in good faith will not be subject to review by any person and will be final, binding and conclusive on all persons. The Board's or any Committee's decisions and determinations need not be uniform and may be made selectively among Participants in the Board's or any Committee's sole discretion. The Board's or any Committee's decisions and determinations will be afforded the maximum deference provided by applicable law.

# 3. Ordinary Shares Subject to the Plan.

(a) Share Reserve. Subject to adjustment in accordance with <u>Section 3(c)</u> and to any adjustment as necessary to implement any Capitalization Adjustments, the Share Reserve on the Effective Date shall be 1,812,663 Ordinary Shares.

In addition, subject to any adjustment as necessary to implement any Capitalization Adjustments, such Share Reserve will automatically increase on January 1st of each year for a period of ten (10) years commencing on January 1, 2022 and ending on (and including) January 1, 2031, in an amount equal to five percent (5%) of the total number of Capital Shares (on a fully-diluted basis) outstanding on December 31st of the preceding year; provided, however that the Board or any Committee may act prior to January 1st of a given year to provide that the increase for such year will be a lesser number of Ordinary Shares.

(b) Aggregate Incentive Share Option Limit. Notwithstanding anything to the contrary in <u>Section 3(a)</u> and subject to any adjustments as necessary to implement any Capitalization Adjustment, the aggregate number of Ordinary Shares that may be issued pursuant to the exercise of Incentive Share Options is 1,812,663 Ordinary Shares.

(c) Reversion of Ordinary Shares to the Share Reserve. If an Award or any portion thereof (i) expires or otherwise terminates without all of the Ordinary Shares covered by such Award having been issued or (ii) is settled in cash (i.e., the Participant receives cash rather than Ordinary Shares), such expiration, termination or cash settlement will not reduce (or otherwise offset) the number of Ordinary Shares that may be available for issuance under the Plan. If any Ordinary Shares issued pursuant to an Award are forfeited back to or repurchased by the Company because of the failure to meet a contingency or condition required to vest such Ordinary Shares in the Participant, then the shares that are forfeited or repurchased will revert to and again become available for issuance under the Plan. Any Ordinary Shares

reacquired by the Company in satisfaction of tax withholding obligations on an Award or as consideration for the exercise or purchase price of an Award will again become available for issuance under the Plan.

(d) Source of Ordinary Shares. The Ordinary Shares issuable under the Plan will be authorized but unissued or reacquired Ordinary Shares, including Ordinary Shares repurchased by the Company on an open market or otherwise.

(e) Substitute Awards. In connection with an entity's merger or consolidation with the Company or the Company's acquisition of an entity's property or stock, including without limitation pursuant to the Business Combination Agreement, the Board may grant Awards in substitution for any options or other share or share-based awards granted before such merger or consolidation by such entity or its affiliate. Substitute Awards may be granted on such terms as the Board deems appropriate, notwithstanding limitations on Awards in the Plan. Substitute Awards will not count against the Share Reserve set forth in Section 3(a) above (nor shall Ordinary Shares subject to a Substitute Award be added to the Ordinary Shares available for Awards under the Plan as provided in Section 3(c) above), except that Ordinary Shares acquired by exercise of substitute Incentive Share Options will count against the maximum number of Ordinary Shares that may be issued pursuant to the exercise of Incentive Share Options under Section 3(b) of the Plan.

(f) Class of Ordinary Shares. Substitute Awards and other Awards, in either case, granted under the Plan shall be for Class A Ordinary Shares.

## 4. Eligibility and Non-Employee Director Limitation.

(a) Eligible Award Recipients. Subject to the terms of the Plan and applicable law, Awards may be granted to Employees, Directors and Consultants.

(b) Service Recipient Stock. Notwithstanding anything herein to the contrary, no Award under which a Participant may receive Ordinary Shares may be granted to an Employee, Director, or Consultant of any Affiliate of the Company if such Ordinary Shares do not constitute "service recipient stock" for purposes of Section 409A of the Code with respect to such Employee, Director or Consultant and such Ordinary Shares are required to constitute "service recipient stock" for such Award to comply with, or be exempt from, Section 409A of the Code.

Ex J-3

(c) Non-Employee Director Compensation Limitation. The aggregate value of all compensation granted or paid, as applicable, to any individual for service as a Non-Employee Director with respect to any calendar year, including Awards granted and cash fees paid by the Company to such Non-Employee Director, will not exceed (i) US\$750,000 total in value or (ii) in the event such Non-Employee Director is first appointed or elected to the Board during such calendar year, US\$1,000,000 in total value, in each case calculating the value of any equity awards based on the grant date fair value of such equity awards for financial reporting purposes. The limitations in this <u>Section 4(c)</u> shall apply commencing with the first calendar year that begins following the Effective Date. For avoidance of doubt, compensation will count towards this limit for the calendar year it was granted or earned, and not later when distributed, in the event it is deferred.

## 5. Provisions Relating to Options and Share Appreciation Rights.

Each Option or SAR will be in such form and will contain such terms and conditions as the Board deems appropriate. The provisions of separate Options or SARs need not be identical or comparable; *provided, however*, that each Award Agreement for Options or SARs will conform to (through incorporation of provisions hereof by reference in the applicable Award Agreement or otherwise) the substance of each of the following provisions:

(a) Term. No Option or SAR will be exercisable after the expiration of ten (10) years from the date of its grant or such shorter period specified in the Award Agreement.

(b) Exercise Price. The exercise or strike price of each Option or SAR shall be determined by the Board and set forth in the Award Agreement which, unless otherwise determined by the Board, may be a fixed or variable price determined by reference to the Fair Market Value of the Ordinary Shares over which such Award is granted; *provided, however*; that (i) no Option or SAR may be granted to a U.S. Participant with an exercise or strike price per Ordinary Share which is less than one hundred percent (100%) of the Fair Market Value of an Ordinary Share subject to the Option or SAR on the date of grant, without compliance with Section 409A of the Code or the Participant's consent, (ii) the exercise or strike price of each Option or SAR granted to a Participant that is not a U.S. Participant shall

comply with applicable law, and (iii) an Option or SAR may be granted with an exercise or strike price lower than that set forth herein if such Option or SAR is granted pursuant to an assumption or substitution for an option or share appreciation right granted by another company, whether in connection with an acquisition of such company or otherwise, and in a manner consistent with the provisions of Section 409A of the Code and other applicable law. Notwithstanding the foregoing, no Option or SAR may be granted with an exercise or strike price lower than the par value of the Ordinary Shares. Each SAR will be denominated in Ordinary Share equivalents.

(c) Exercise Procedure and Payment of Exercise Price for Options. In order to exercise an Option, the Participant must provide notice of exercise to the Company in accordance with the procedures specified in the Option Agreement or otherwise provided by the Company. The purchase price of Ordinary Shares acquired pursuant to the exercise of an Option may be paid, to the extent permitted by applicable law and as determined by the Board in its sole discretion, by any combination of the methods of payment set forth below. Any Ordinary Shares that are not fully paid will be subject to the forfeiture provisions in the Company's Memorandum and Articles of Association (as amended and/or restated from time to time). The Board will have the authority to grant Options that do not permit all of the following methods of payment (or otherwise restrict the ability to use certain methods) and to grant Options that require the consent of the Company to use a particular method of payment. Subject to applicable law, the permitted methods of payment are as follows:

(i) by cash, check, bank draft or money order payable to the Company;

(ii) pursuant to a "cashless exercise" program (developed under Regulation T as promulgated by the U.S. Federal Reserve Board or similar regulations in other applicable jurisdictions, if required for compliance with the laws of the relevant jurisdiction) that, prior to the issuance of the Ordinary Share, is subject to the Option results in either the receipt of cash (or check) by the Company or the receipt of irrevocable instructions to pay the aggregate exercise price to the Company from the sales proceeds;

(iii) by delivery to the Company (either by actual delivery or attestation) of Ordinary Shares that are already owned by the Participant free and clear of any liens, claims, encumbrances or security interests, with a Fair Market Value on the date of exercise that does not exceed the exercise price, *provided that* (A) at the time of exercise the Ordinary Shares are publicly traded, (B) any remaining balance of the exercise price not satisfied by such delivery is paid by the Participant in cash or other permitted form of payment, (C) such delivery would not violate any applicable law or agreement restricting the redemption of the Ordinary Shares, (D) any certificated shares are endorsed or accompanied by an executed assignment separate from the certificate, and (E) such Ordinary Shares have been held by the Participant for any minimum period necessary to avoid adverse accounting treatment as a result of such delivery as may be required by the Company;

(iv) if an Option is a Nonstatutory Share Option, by a "net exercise" arrangement pursuant to which the Company will reduce the number of Ordinary Shares issuable upon exercise by the largest whole number of Ordinary Shares with a Fair Market Value that does not exceed the aggregate exercise price; *provided, however*, that the Company will accept a cash or other payment from the Participant to the extent of any remaining balance of the aggregate exercise price not satisfied by such reduction in the number of whole Ordinary Shares to be issued. Ordinary Shares will no longer be subject to an Option and will not be exercisely thereafter to the extent that (A) Ordinary Shares issuable upon exercise are used to pay the exercise price pursuant to the "net exercise," (B) Ordinary Shares are delivered to the Participant as a result of such exercise, and (C) Ordinary Shares are withheld to satisfy tax withholding obligations; or

(v) in any other form of legal consideration that may be acceptable to the Board and permissible under applicable law.

(d) Exercise and Payment of a SAR. To exercise any outstanding SAR, the Participant must provide written notice of exercise to the Company in compliance with the provisions of the Award Agreement evidencing such SAR or otherwise provided by the Company. The appreciation distribution payable on the exercise of a SAR will be not greater than an amount equal to the excess of (i) the aggregate Fair Market Value (on the date of the exercise of the SAR) of a number of Ordinary Shares equal to the number of Ordinary Shares equivalents in which the Participant is vested under such SAR, and with respect to which the Participant is exercising the SAR on such date, over (ii) the aggregate strike price of the number of Ordinary Shares equivalents with respect to which the Participant is exercising the SAR on such date. The appreciation distribution may be paid in Ordinary Shares, in cash, in any combination of the two or in any other form of consideration, as determined by the Board and contained in the Award Agreement evidencing such SAR and subject to applicable law.

(e) Transferability of Options and SARs. The Board may, in its sole discretion, impose such limitations on the transferability of Options and SARs as the Board will determine. In the absence of such a determination by the Board to the contrary, the following restrictions on the transferability of Options and SARs will apply:

(i) Restrictions on Transfer. An Option or SAR will not be transferable except by will or by the laws of descent and distribution (and pursuant to subsections (ii) and (iii) below), and will be exercisable during the lifetime of the Participant only by the Participant. The Board may permit transfer of the Option or SAR in a manner that is not prohibited by applicable tax and securities laws. Except as explicitly provided herein, neither an Option nor a SAR may be transferred for consideration.

(ii) **Domestic Relations Orders**. Subject to the approval of the Board or a duly authorized Officer, an Option or SAR may be transferred pursuant to the terms of a domestic relations order, official marital settlement agreement or other divorce or separation instrument as permitted by Treasury Regulation 1.421-1(b)(2) or regulations in other applicable jurisdictions.

(iii) Beneficiary Designation. Subject to the approval of the Board or a duly authorized Officer, a Participant may, by delivering written notice to the Company, in a form approved by the Company (or the designated broker), designate a third party who, upon the death of the Participant, will thereafter be entitled to exercise the Option or SAR and receive the Ordinary Shares or other consideration resulting from such exercise. In the absence of such a designation, upon the death of the Participant, the executor or administrator of the Participant's estate will be entitled to exercise the Option or SAR and receive the Ordinary Shares or other consideration resulting from such exercise. However, the Company may prohibit designation of a beneficiary at any time, including due to any conclusion by the Company that such designation would be inconsistent with the provisions of applicable laws.

(f) Vesting Generally. The total number of Ordinary Shares subject to an Option or SAR may vest and become exercisable in periodic installments that may or may not be equal. The Option or SAR may be subject to such other terms and conditions on the time or times when it may or may not be exercised (which may be based on the satisfaction of performance goals or other criteria) as the Board may deem appropriate. The vesting provisions of individual Options or SARs may vary. The provisions of this <u>Section 5(f)</u> are subject to any Option or SAR provisions governing the minimum number of Ordinary Shares as to which an Option or SAR may be exercised.

# (g) Termination of Continuous Service.

(i) Termination of Continuous Service for Cause. Except as explicitly provided otherwise in a Participant's Award Agreement or other written agreement between a Participant and the Company or an Affiliate, if a Participant's Continuous Service is terminated for Cause, the Participant's Options and SARs (whether vested or unvested) will terminate and be forfeited immediately upon such termination of Continuous Service, and the Participant will be prohibited from exercising any portion (including any vested portion) of such Awards on and after the date of such termination of Continuous Service and the Participant will have no further right, title or interest in such forfeited Award, the Ordinary Shares subject to the forfeited Award, or any consideration in respect of the forfeited Award.

Ex J-5

(ii) Post-Termination Exercise Period Following Termination of Continuous Service for Reasons Other than Cause. Subject to <u>Section 5(h)</u>, if a Participant's Continuous Service terminates for any reason other than for Cause, the Participant may exercise his or her Option or SAR to the extent vested, but only within the following period of time or, if applicable, such other period of time provided in the Award Agreement or other written agreement between a Participant and the Company or an Affiliate; *provided, however*, that in no event may such Award be exercised after the expiration of its maximum term (as set forth in <u>Section 5(a)</u>):

(1) Three (3) months following the date of such termination if the Option is an Incentive Share Option (other than any termination due to the Participant's Disability or death);

(2) Six (6) months following the date of such termination if such termination is a termination without Cause (other than any termination due to the Participant's Disability or death) or due to the Participant's resignation from employment;

(3) Twelve (12) months following the date of such termination if such termination is due to the Participant's Disability;

(4) Twelve (12) months following the date of such termination if such termination is due to the Participant's death; or

(5) Twelve (12) months following the date of the Participant's death if such death occurs following the date of such termination but during the period such Award is otherwise exercisable (as provided in (1) or (2) above).

Following the date of such termination, to the extent the Participant does not exercise such Award within the applicable post-termination exercise period (or, if earlier, prior to the expiration of the maximum term of such Award), such unexercised portion of the Award will

terminate, and the Participant will have no further right, title or interest in the terminated Award, the Ordinary Shares subject to the terminated Award, or any consideration in respect of the terminated Award.

(iii) Termination of Continuous Service and Unvested Portion of Award. Except as otherwise provided in the applicable Award Agreement or other written agreement between the Participant and the Company, immediately upon termination of Continuous Service, all unvested portions of any outstanding Options or SARs of such Participant shall be forfeited without consideration as of the termination date.

(h) Restrictions on Exercise; Extension of Exercisability. A Participant may not exercise an Option or SAR at any time that the issuance of Ordinary Shares upon such exercise would violate applicable law. Except as otherwise provided in the applicable Award Agreement or other written agreement between the Participant and the Company, if the exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause and other than upon the Participant's death or Disability) would be prohibited at any time solely because the issuance of Ordinary Shares would violate applicable law, then the Option or SAR will terminate on the earlier of (i) the expiration of a total period of time (that need not be consecutive) equal to the applicable post termination exercise period after the termination requirements, and (ii) the expiration of the term of the Option or SAR as set forth in the applicable Award Agreement. In addition, unless otherwise provided in a Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR will terminate on the earlier of (i) the applicable post-termination of the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR will terminate on the earlier of (i) the expiration of a the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR will terminate on the earlier of (i) the expiration of a period of time (that need not be consecutive) equal to the applicable post-termination exercise of an Option or SAR following the termination of the Participant's Continuous Service (other than for Cause) would violate the Company's insider trading policy, then the Option or SAR will terminate on the earlier of (i) the expiration of a period of time (that need not be consecutive) equal to the applicable post-termination exercise period after the termina

(i) Modification or Assumption of Options. Except as otherwise provided in the Plan, the Board may modify, extend or assume outstanding Options or may accept the cancellation of outstanding stock options (whether granted by the Company or by another issuer) in return for the grant of new Options for the same or a different number of shares and at the same or a different exercise price. No modification of an Option shall, without the consent of the Participant, materially impair his or her rights or increase his or her obligations under such Option.

# 6. Provisions of Awards Other than Options and SARs.

(a) Restricted Share Awards. Each Restricted Share Award Agreement will be in such form and will contain such terms and conditions as the Board deems appropriate. To the extent consistent with the Company's Memorandum and Articles of Association (as amended and/ or restated from time to time) and other constitutional and governance documents, at the Board's election, Ordinary Shares underlying a Restricted Share Award may be held in book entry form subject to the Company's instructions until any restrictions relating to the Restricted Share Award lapse; and may be evidenced by a certificate, which certificate will be held in such form and manner as determined by the Board. The Company may require that any share certificates relating to Restricted Shares be held by the Company in escrow for the participant until all restrictions on such Restricted Shares have been removed. The terms and conditions of Restricted Share Award Agreements may change from time to time, and the terms and conditions of separate Restricted Share Award Agreements need not be identical or comparable. Each Restricted Share Award Agreement will conform to (through incorporation of the provisions hereof by reference in the agreement or otherwise) the substance of each of the following provisions:

Ex J-6

(i) Consideration. A Restricted Share Award may be awarded in consideration for (A) cash, check, bank draft or money order payable to the Company, (B) past services to the Company or an Affiliate, or (C) any other form of legal consideration (including future services) that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.

(ii) Vesting. Ordinary Shares awarded under the Restricted Share Award Agreement may be subject to forfeiture to the Company in accordance with a vesting schedule to be determined by the Board. Except as otherwise provided in an Award Agreement or other written agreement between a Participant and the Company or an Affiliate, vesting of Restricted Share Awards will cease upon termination of a Participant's Continuous Service.

(iii) Termination of Participant's Continuous Service. If a Participant's Continuous Service terminates, the Company may receive through a forfeiture condition or a repurchase right, any or all of the Ordinary Shares held by the Participant as of the date of termination of Continuous Service under the terms of the Restricted Share Award Agreement.

(iv) **Transferability**. Rights to acquire Ordinary Shares under the Restricted Share Award Agreement will be transferable by the Participant only upon such terms and conditions as are set forth in the Restricted Share Award Agreement, as the Board will determine in its sole discretion, so long as Ordinary Shares awarded under the Restricted Share Award Agreement remains subject to the terms of the Restricted Share Award Agreement.

(v) Dividends. A Restricted Share Award Agreement may provide that any dividends paid on Restricted Shares will be subject to the same vesting and forfeiture restrictions as apply to the Ordinary Shares subject to the Restricted Share Award to which they relate; provided, any performance-based Restricted Share Award Agreement shall provide that any dividends paid on Restricted Shares will be subject to the same vesting and forfeiture restrictions as apply to the Ordinary Shares subject to the Restricted Share Award to which they relate.

(vi) Shareholder Rights. Unless otherwise determined by the Board, a Participant will have voting and other rights as a shareholder of the Company with respect to any Ordinary Shares subject to a Restricted Share Award.

(b) Restricted Share Unit Awards. Each Restricted Share Unit Award Agreement will be in such form and will contain such terms and conditions as the Board deems appropriate. The terms and conditions of Restricted Share Unit Award Agreements may change from time to time, and the terms and conditions of separate Restricted Share Unit Award Agreements need not be identical or comparable. Each Restricted Share Unit Award Agreement will conform to (through incorporation of the provisions hereof by reference in the Agreement or otherwise) the substance of each of the following provisions:

(i) Consideration. At the time of grant of a Restricted Share Unit Award, the Board will determine the consideration, if any, to be paid by the Participant upon delivery of each Ordinary Share subject to the Restricted Share Unit Award. The consideration to be paid (if any) by the Participant for each Ordinary Share subject to a Restricted Share Unit Award may be paid in any form of legal consideration that may be acceptable to the Board, in its sole discretion, and permissible under applicable law.

(ii) Vesting. At the time of the grant of a Restricted Share Unit Award, the Board may impose such restrictions on or conditions to the vesting of the Restricted Share Unit Award as it, in its sole discretion, deems appropriate.

(iii) Settlement. A Restricted Share Unit Award may be settled by the delivery of Ordinary Shares, their cash equivalent, any combination thereof or in any other form of consideration, as determined by the Board and contained in the Restricted Share Unit Award Agreement. At the time of the grant of a Restricted Share Unit Award, the Board, as it deems appropriate, may impose such restrictions or conditions that delay the delivery of the Ordinary Shares (or their cash equivalent) subject to a Restricted Share Unit Award to a time after the vesting of such Restricted Share Unit Award.

(iv) Dividend Equivalents. Dividend equivalents may be credited in respect of Ordinary Shares covered by a Restricted Share Unit Award, as determined by the Board and contained in the Restricted Share Unit Award Agreement. At the sole discretion of the Board, such dividend equivalents may be converted into additional Ordinary Shares covered by the Restricted Share Unit Award in such manner as determined by the Board. Any additional Ordinary Shares covered by the Restricted Share Unit Award credited by reason of such dividend equivalents will be subject to all of the same terms and conditions of the underlying Restricted Share Unit Award Agreement to which they relate. Any dividend equivalents distributed under the Plan shall not be counted against the Share Reserve.

Ex J-7

(v) Termination of Participant's Continuous Service. Except as otherwise provided in the applicable Restricted Share Unit Award Agreement, such portion of the Restricted Share Unit Award that has not vested will be forfeited upon the Participant's termination of Continuous Service, and the Participant will have no further right, title or interest in the Restricted Share Unit Award, the Ordinary Shares issuable pursuant to the Restricted Share Unit Award, or any consideration in respect of the Restricted Share Unit Award.

(vi) Creditors' Rights. The Plan is intended to constitute an "unfunded" plan for incentive compensation. The Company shall not be required to establish any special or separate fund or to segregate any assets to assure the performance of its obligations under the Plan. With respect to any payments not yet made or Ordinary Shares not yet transferred to a Participant by the Company, nothing contained in the Plan shall give any Participant any rights that are greater than those of a general creditor of the Company.

(c) Other Awards. Other forms of Awards valued in whole or in part by reference to, or otherwise based on, Ordinary Shares, including the appreciation in value thereof (e.g., options or share rights with an exercise price or strike price less than one hundred percent (100%) of the Fair Market Value of the Ordinary Shares at the time of grant) may be granted either alone or in addition to Awards provided for under <u>Section 5</u> and the preceding provisions of this <u>Section 6</u>. Such Other Awards may include (without limitation) Awards that may vest or may be exercised or a cash Awards that may vest or become earned and paid contingent on the attainment during a performance period of performance goals or other criteria as the Board may determine. Subject to the provisions of the Plan, the Board will have sole and complete authority to determine the persons to whom and the time or times at which such Other Awards will be granted, the number of Ordinary Shares (or the cash equivalent thereof) to be granted pursuant to such Other Awards, and all other terms and conditions of such Other Awards (including without limitation, with respect to any performance Awards, the length of the performance period, the performance goals to be achieved during the performance period, and the measure of whether and to what degree such performance goals have been obtained).

## 7. Covenants of the Company.

(a) Availability of Ordinary Shares. The Company will keep available at all times the number of Ordinary Shares reasonably required to satisfy then-outstanding Awards.

(b) Securities Law Compliance. The Company will use commercially reasonable efforts to seek to obtain from each regulatory commission or agency having jurisdiction over the Plan such authority as may be required to grant Awards and to issue and sell Ordinary Shares upon exercise of the Awards; *provided, however*, that this undertaking will not require the Company to register the Plan, any Award or any Ordinary Shares issued or issuable pursuant to any such Award under the Securities Act or other applicable securities regulatory scheme. If, after reasonable efforts and at a reasonable cost, the Company is unable to obtain from any such regulatory commission or agency the authority that counsel for the Company deems necessary for the lawful issuance and sale of Ordinary Shares under the Plan, the Company will be relieved from any liability for failure to issue and sell Ordinary Shares upon exercise of such Awards unless and until such authority is obtained. A Participant will not be eligible for the grant of an Award or the subsequent issuance of cash or Ordinary Shares pursuant to the Award if such grant or issuance would be in violation of any applicable securities law or any other applicable law or regulation.

(c) No Obligation to Notify or Minimize Taxes. The Company will have no duty or obligation to any Participant to advise such holder as to the time or manner of exercising such Award. Furthermore, the Company will have no duty or obligation to warn or otherwise advise such holder of a pending termination or expiration of an Award or a possible period in which the Award may not be exercised. The Company has no duty or obligation to minimize the tax consequences of an Award to the holder of such Award.

## 8. Miscellaneous.

(a) Use of Proceeds from Sales of Ordinary Share. Proceeds from the sale of Ordinary Shares pursuant to Awards will constitute general funds of the Company.

Ex J-8

(b) Corporate Action Constituting Grant of Awards. Corporate action constituting a grant by the Company of an Award to any Participant will be deemed completed as of the date of such corporate action, unless otherwise determined by the Board, regardless of when the instrument, certificate, or letter evidencing the Award is communicated to, or actually received or accepted by, the Participant. In the event that the corporate records (e.g., Board consents, resolutions or minutes) documenting the corporate action constituting the grant contain terms (e.g., exercise price, vesting schedule or number of Ordinary Shares) that are inconsistent with those in the Award Agreement as a result of a clerical error in the papering of the Award Agreement, the corporate records will control and the Participant will have no legally binding right to the incorrect term in the Award Agreement.

(c) Shareholder Rights. No Participant will be deemed to be the holder of, or to have any of the rights of a holder with respect to, any Ordinary Shares subject to an Award unless and until (i) such Participant has satisfied all requirements for exercise of, or the issuance of Ordinary Shares under, the Award pursuant to its terms, including but not limited to, any applicable withholding or tax obligations relating to the Award, and (ii) the issuance of the Ordinary Shares subject to the Award has been entered into the books and records of the Company and the register of members of the Company has been accordingly updated. No adjustment shall be made for cash or stock

dividends or other rights for which the record date is prior to the date when such Ordinary Share is issued, except as expressly provided in this Plan.

(d) No Employment or Other Service Rights. Nothing in the Plan, any Award Agreement or any other instrument executed thereunder or in connection with any Award granted pursuant thereto will confer upon any Participant any right to continue to serve the Company or an Affiliate in the capacity in effect at the time the Award was granted or will affect the right of the Company or an Affiliate to terminate (i) the employment of an Employee with or without notice and with or without cause, (ii) the service of a Consultant pursuant to the terms of such Consultant's agreement with the Company or an Affiliate, or (iii) the service of a Director pursuant to the Company's Memorandum and Articles of Association (as amended and/or restated from time to time) and other constitutional and governance documents of the Company or an Affiliate, and any provisions of the applicable laws of the jurisdiction in which the Company or the Affiliate is incorporated, as the case may be.

(e) Government and Other Regulations. The obligation of the Company to make payment of awards in Ordinary Shares or otherwise shall be subject to all applicable laws, and to such approvals by government agencies as may be required. The Company shall be under no obligation to register any of the Ordinary Shares paid pursuant to the Plan under the Securities Act or any other similar laws in any applicable jurisdiction. If the Ordinary Shares paid pursuant to the Plan may in certain circumstance be exempt from registration pursuant to the Securities Act or other applicable laws, the Company may restrict the transfer of such Ordinary Shares in such manner as it deems advisable to ensure the availability of such exemption. The Company may, upon advice of counsel to the Company, place legends on share certificates issued under the Plan as such counsel deems necessary or appropriate in order to comply with applicable securities laws or other applicable laws, including, but not limited to, legends restricting the transfer of the Ordinary Shares.

(f) Withholding Obligations. Unless prohibited by the terms of an Award Agreement, the Company may, in its sole discretion, satisfy any tax withholding obligation relating to an Award by any of the following means or by a combination of such means: (i) causing the Participant to tender a cash payment; (ii) withholding Ordinary Shares from the Ordinary Shares issued or otherwise issuable to the Participant in connection with the Award; (iii) withholding cash from an Award settled in cash; (iv) withholding payment from payroll and/or any other amounts otherwise payable to the Participant; (vi) by allowing a Participant to effectuate a "cashless exercise"; or (vi) by such other method as may be set forth in the Award Agreement.

(g) Electronic Delivery. Any reference herein to a "written" agreement or document will include any agreement or document delivered electronically or posted on the Company's intranet (or other shared electronic medium controlled by the Company to which the Participant has access).

(h) Deferrals. To the extent permitted by applicable law, the Board, in its sole discretion, may determine that the delivery of Ordinary Shares or the payment of cash, upon the exercise, vesting or settlement of all or a portion of any Award may be deferred and may establish programs and procedures for deferral elections to be made by Participants. The Board is authorized to make deferrals of Awards and determine when, and in what annual percentages, Participants may receive payments, including lump sum payments, following the Participant's termination of Continuous Service, and implement such other terms and conditions consistent with the provisions of the Plan and in accordance with applicable law.

(i) Additional Information. The Company shall request the Participant to provide any information necessary to comply with applicable laws and regulations, including without limitation the Cayman Islands Tax Information Authority Law (2013 Revision), Tax Information Authority (International Tax Compliance) (United Kingdom) Regulations, 2014, and Tax Information Authority (International Tax Compliance) (United States of America) Regulations, 2014, and any anti-money laundering or anti-terrorist laws or regulations (the *"Relevant Regulations"*) and may delay updating the Company's books and records and register of members until the relevant Participant has provided satisfactory information to the Company. The Company may disclose any information concerning the Participant necessary to comply with the Relevant Regulations.

Ex J-9

(j) Buyout of Awards. The Board may at any time offer to buy out, and the Participants shall accept such offer, for a payment in cash or cash equivalents (including without limitation Ordinary Shares issued at Fair Market Value that may or may not be issued under this Plan), an Award previously granted based upon such terms and conditions as the Board shall establish.

(k) Clawback Policy. The Company may (i) cause the cancellation of any Award, (ii) require reimbursement of any Award by a Participant, and (iii) effect any other right of recoupment of equity or other compensation provided under this Plan or otherwise in

accordance with Company policies in effect on the date of grant of the applicable Award and/or applicable law (each, a "*Clawback Policy*"). In addition, a Participant may be required to repay to the Company certain previously paid compensation, whether provided under this Plan or an Award Agreement or otherwise, in accordance with the Clawback Policy.

(1) Foreign Currency. A Participant may be required to provide evidence that any currency used to pay the exercise or strike price of any Award was acquired and taken out of the jurisdiction in which the Participant resides in accordance with applicable laws, including foreign exchange control laws and regulations.

## 9. Adjustments upon Changes in Ordinary Share; Other Corporate Events.

(a) Capitalization Adjustments. In the event of a Capitalization Adjustment, the Board will appropriately and proportionately adjust: (i) the class(es) and maximum number of securities subject to the Plan pursuant to <u>Section 3(a)</u>, (ii) the class(es) and maximum number of securities that may be issued pursuant to the exercise of Incentive Share Options pursuant to <u>Section 3(b)</u>, (iii) the class(es) and number of securities and price per share of Ordinary Shares subject to outstanding Awards, or (iv) the issuer of the Awards. The Board will make such adjustments, and its determination will be final, binding and conclusive.

(b) In the event of any change in the capitalization of the Company or corporate change other than those specifically referred to in Section 9(a), including without limitation, any extraordinary cash dividend, spin-off, split-off, sale of a Subsidiary or business unit, public listing of a Subsidiary, or other similar transaction, the Board may make such adjustments in the issuer, number and class of shares subject to Awards outstanding on the date on which such change occurs, such as, for example, a rollover of Awards, as the Board may consider appropriate.

(c) Dissolution or Liquidation. Except as otherwise provided in the Award Agreement, in the event of a dissolution or liquidation of the Company, all outstanding Awards (other than Awards consisting of vested and outstanding Ordinary Shares not subject to a forfeiture condition or the Company's right of repurchase) will terminate immediately prior to the completion of such dissolution or liquidation, and the Ordinary Shares subject to the Company's repurchase rights or subject to a forfeiture condition may be repurchased or reacquired by the Company notwithstanding the fact that the holder of such Award is providing Continuous Service, *provided, however*, that the Board may, in its sole discretion, cause some or all Awards to become fully vested, exercisable and/or no longer subject to repurchase or forfeiture (to the extent such Awards have not previously expired or terminated) before the dissolution or liquidation is completed but contingent on its completion.

(d) Corporate Transactions. The following provisions will apply to Awards in the event of a Transaction unless otherwise provided in the Award Agreement or any other written agreement between the Company or any Affiliate and the Participant or unless otherwise expressly provided by the Board at the time of grant of an Award. In the event of a Transaction, then, notwithstanding any other provision of the Plan, the Board may take one (1) or more of the following actions with respect to Awards, contingent upon the closing or completion of the Transaction:

(i) arrange for the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company) to assume or continue the Award or to substitute a similar award for the Award (including, but not limited to, an award to acquire the same consideration (including property with the same value as the consideration) paid to the shareholders of the Company pursuant to the Transaction);

(ii) arrange for the assignment of any reacquisition or repurchase rights held by the Company in respect of Ordinary Shares issued pursuant to the Award to the surviving corporation or acquiring corporation (or the surviving or acquiring corporation's parent company);

(iii) accelerate the vesting, in whole or in part, of the Award (and, if applicable, the time at which the Award may be exercised) to a date prior to the effective time of such Transaction as the Board determines (or, if the Board does not determine such a date, to the date that is five (5) days prior to the effective date of the Transaction), with such Award terminating if not exercised (if applicable) at or prior to the effective time of the Transaction; provided, however, that the Board may require Participants to complete and deliver to the Company a notice of exercise before the effective date of a Transaction, which exercise is contingent upon the effectiveness of such Transaction;

(iv) arrange for the lapse, in whole or in part, of any reacquisition or repurchase rights held by the Company with respect to the Award;

(v) cancel or arrange for the cancellation of the Award, to the extent not vested or not exercised prior to the effective time of the Transaction, in exchange for such cash consideration, if any, as the Board, in its sole discretion, may consider appropriate; and

(vi) make a payment, in such form as may be determined by the Board equal to the excess, if any, of (A) the value of the property the Participant would have received upon the exercise of the Award immediately prior to the effective time of the Transaction, over (B) any exercise price payable by such holder in connection with such exercise. For clarity, this payment may be zero (\$0) and the Award may be cancelled with no consideration if the cash, value of the property or a combination of both that the Participant would be scheduled to receive at the consummation of the Transaction is equal to or less than the exercise price. Payments under this provision may be delayed or forfeited to the same extent that payment of consideration to the holders of the Company's Ordinary Shares in connection with the Transaction is delayed or forfeited as a result of escrows, earn outs, holdbacks or any other contingencies.

The Board need not take the same action or actions with respect to all Awards or portions thereof or with respect to all Participants. The Board may take different actions with respect to the vested and unvested portions of an Award.

(e) Change in Control. An Award may be subject to additional acceleration of vesting and exercisability upon or after a Change in Control as may be provided in the Award Agreement for such Award or as may be provided in any other written agreement between the Company or any Affiliate and the Participant, but in the absence of such provision, no such acceleration will occur without Board action.

#### 10. Plan Term; Earlier Termination or Suspension of the Plan

(a) Plan Term. The Board may suspend or terminate the Plan at any time. Unless terminated sooner by the Board, the Plan will automatically terminate on the day before the tenth anniversary of the Effective Date; provided, no Incentive Share Options shall be granted on or after the earlier of (i) the tenth anniversary of the date the Plan is adopted by the Board, or (ii) the tenth anniversary of the date the Plan is approved by the shareholders of the Company.No Awards may be granted under the Plan while the Plan is suspended or after it is terminated.

(b) No Impairment of Rights. Suspension or termination of the Plan will not impair rights and obligations under any Award granted while the Plan is in effect except with the written consent of the affected Participant or as otherwise permitted in the Plan.

# 11. Additional Provisions Applicable to U.S. Participants.

## (a) Incentive Share Options.

(i) Incentive Share Options may be granted only to employees of the Company or a "parent corporation" or "subsidiary corporation" thereof (as such terms are defined in Sections 424(e) and (f) of the Code, respectively).

(ii) A Ten Percent Shareholder shall not be granted an Incentive Share Option unless the exercise price of such Option is at least one hundred ten percent (110%) of the Fair Market Value on the date of grant and the Option is not exercisable after the expiration of five (5) years from the date of grant or such shorter period specified in the Award Agreement. "*Ten Percent Shareholder*" means a person who Owns (or is deemed to Own pursuant to Section 424(d) of the Code) Capital Shares possessing more than ten percent (10%) of the total combined voting power of all classes of shares of the Company or any "parent corporation" or "subsidiary corporation" of the Company, as such terms are defined in Sections 424(e) and (f) of the Code, respectively.

(iii) To the extent that the aggregate Fair Market Value (determined at the time of grant) of Ordinary Shares with respect to which Incentive Share Options are exercisable for the first time by any Optionholder during any calendar year (under all plans of the Company and any Affiliates) exceeds one hundred thousand dollars (\$100,000) (or such other limit established in the Code) or otherwise does not comply with the rules governing Incentive Share Options, the Options or portions thereof that exceed such limit (according to the order in which they were granted) or otherwise do not comply with such rules will be treated as Nonstatutory Share Options, notwithstanding any contrary provision of the applicable Option Agreement(s).

Ex J-11

(b) Compliance with Section 409A of the Code. To the extent that the Board determines that any Award granted hereunder is subject to Section 409A of the Code, the Award Agreement evidencing such Award shall incorporate the terms and conditions necessary to avoid the consequences specified in Section 409A(a)(1) of the Code. To the extent applicable, the Plan and Award Agreements shall

be interpreted in accordance with Section 409A of the Code. In the event that any provision of the Plan or an Award agreement is determined by the Board to not comply with the applicable requirements of Section 409A of the Code or the Treasury Regulations or other guidance issued thereunder, the Board shall have the authority to take such actions and to make such changes to the Plan or an Award Agreement as the Board deems necessary to comply with such requirements (including without limitation, after the grant date of an Award, increasing the exercise price to equal what was the Fair Market Value on the grant date of the Award). Each payment to a Participant made pursuant to this Plan shall be considered a separate payment and not one of a series of payments for purposes of Section 409A of the Code. Notwithstanding the foregoing or anything elsewhere in the Plan or an Award Agreement to the contrary, if upon a Participant's Separation From Service (as defined in Section 409A of the Code), he/she is then a "specified employee" (as defined in Section 409A of the Code), then solely to the extent necessary to comply with Section 409A of the Code and avoid the imposition of taxes under Section 409A of the Code, the Company shall defer payment of "nonqualified deferred compensation" subject to Section 409A of the Code payable as a result of and within six (6) months following such Separation From Service under this Plan until the earlier of (i) the first business day of the seventh month following the Participant's Separation From Service, or (ii) ten (10) days after the Company receives written confirmation of the Participant's death. Any such delayed payments shall be made without interest. While it is intended that all payments and benefits provided under this Plan will be exempt from or comply with Section 409A of the Code, the Company makes no representation or covenant to ensure that the Awards and payments under this Plan are exempt from or compliant with Section 409A of the Code. The Company will have no liability to any Participant or any other party if a payment or benefit under this Plan or any Award is challenged by any taxing authority or is ultimately determined not to be exempt or compliant. Each Participant further understands and agrees that each Participant will be entirely responsible for any and all taxes on any benefits payable to the Participant as a result of this Plan or any Award. In no event whatsoever shall the Company be liable for any additional tax, interest or penalties that may be imposed on a Participant by Section 409A of the Code or for any damages for failing to comply with Section 409A of the Code.

(c) Compliance with Section 457A of the Code. Notwithstanding any provision of the Plan or an Award Agreement to the contrary, to the extent that the Board determines in its sole discretion that the amount payable or Ordinary Shares issuable under any Award would be taxable to the Participant under Section 457A of the Code in the year such Award is no longer subject to a substantial risk of forfeiture, then the amount payable or Ordinary Shares issuable under such Award shall be paid or issued to the Participant as soon as reasonably practicable after such substantial risk of forfeiture lapses (or, for Awards that are not considered nonqualified deferred compensation subject to Section 409A of the Code, no later than the end of the short-term deferral period permitted by Section 457A of the Code).

(d) Section 280G of the Code. Notwithstanding any provision of the Plan or an Award Agreement to the contrary, if any payment or benefit that a U.S. Participant would receive pursuant to the Plan or an Award Agreement or any other agreement and/or arrangement with the Company or any of its Affiliates (a "*Payment*") would (i) constitute a "parachute payment" within the meaning of Section 280G of the Code, and (ii) but for this sentence, be subject to the excise tax imposed by Section 4999 of the Code (the "*Excise Tax*"), then such Payment will be equal to the Reduced Amount. The "*Reduced Amount*" will be either (x) the largest portion of the Payment that would result in no portion of the Payment being subject to the Excise Tax or (y) the largest portion, up to and including the total, of the Payment, whichever amount, after taking into account all applicable federal, state and local employment taxes, income taxes, and the Excise Tax (all computed at the highest applicable marginal rate), results in the U.S. Participant's receipt, on an after-tax basis, of the greater economic benefit notwithstanding that all or some portion of the Payment may be subject to the Excise Tax. If a reduction in payments or benefits constituting "parachute payments" is necessary so that the Payment equals the Reduced Amount, reduction will occur in the manner that results in the greatest economic benefit for the U.S. Participant, generally with such reduction made first from amounts that are not subject to Section 409A of the Code and, if necessary, from amounts that are subject to Section 409A of the Code and, if necessary, from amounts that are subject to Section 409A of the Code and, if necessary, from amounts that are subject to Section 409A of the Code and, if necessary, from amounts that are subject to Section 409A of the Code and, if necessary, from amounts that are subject to Section 409A of the Code and, if necessary for an as economic benefit, the items so reduced will be reduced pro rata.

# 12. Choice of Law; Arbitration.

(a) Governing Law. The laws of the Cayman Islands will govern all questions concerning the construction, validity and interpretation of this Plan, without regard to that state's conflict of laws rules.

(b) Dispute Resolution. All and any of the disputes arising from and in connection with this Agreement shall be referred to and finally resolved by arbitration in Singapore in accordance with the Arbitration Rules of the Singapore International Arbitration Centre for the time being in force, which rules are deemed to be incorporated by reference in this clause. For the avoidance of doubt, the law of the arbitration shall be governed by the International Arbitration Act (Chapter 143A, 2002 Ed, Statutes of the Republic of Singapore) or its modification or re-enactment thereof.

**13. Definitions**. As used in the Plan, the following definitions will apply to the capitalized terms indicated below:

(a) "Affiliate" means, at the time of determination, (i) any Subsidiary and any "parent corporation" or "subsidiary corporation" of the Company, as such terms are defined in Sections 424(e) and (f) of the Code, respectively, and (ii) any other entity that, directly or indirectly, controls, is controlled by or is under common control with the Company and/or one (1) or more Subsidiaries. For the purposes of this definition, "control" of a given entity means possessing the power or authority, whether exercised or not, to direct the business, management and policies of such entity, directly or indirectly, whether through the ownership of voting securities, by contract, arrangement, understanding, relationship or otherwise, which power or authority shall conclusively be presumed to exist upon possession of beneficial ownership or power to direct the vote of more than thirty percent (30%) of the votes entitled to be cast at a meeting of the members or shareholders of such entity or power to control the composition of at least thirty percent (30%) a majority of the board of directors of such entity; the term "controlled" has the meaning correlative to the foregoing. The Board will have the authority to determine the time or times at which "parent corporation" or "subsidiary corporation" or "control" status is determined within the foregoing definition.

(b) "Award" means any right to receive Ordinary Shares granted under the Plan, including an Option, a Restricted Share Award, a Restricted Share Unit Award, a Share Appreciation Right or any Other Award.

(c) "*Award Agreement*" means a written agreement between the Company and a Participant evidencing the terms and conditions of an Award grant. Each Award Agreement will be subject to the terms and conditions of the Plan.

(d) "Board" means the board of directors of the Company.

(e) "*Business Combination Agreement*" means the Business Combination Agreement, dated as of September 27, 2021 by and among (i) TradeUp Global Corporation (n/k/a as the Company), (ii) TGC Merger Sub, a Cayman Islands exempted company incorporated with limited liability and a direct wholly owned subsidiary of the Company, and (iii) SAITECH Limited, a Cayman Islands exempted company incorporated with limited liability.

(f) "Capital Shares" means each and every class of ordinary shares of the Company, regardless of the number of votes per share.

(g) "*Capitalization Adjustment*" means any change that is made in, or other events that occur with respect to, the Ordinary Shares subject to the Plan or subject to any Award after the Effective Date without the receipt of consideration by the Company through merger, consolidation, reorganization, recapitalization, reincorporation, share dividend, dividend in property other than cash, large nonrecurring cash dividend, share split, reverse share split, liquidating dividend, combination of shares, exchange of shares, change in corporate structure, or any similar equity restructuring transaction. Notwithstanding the foregoing, the conversion of any convertible securities of the Company will not be treated as a Capitalization Adjustment.

(h) "*Cause*" will have the meaning ascribed to such term in any written agreement between the Participant and the Company or an Affiliate defining such term and which is in effect on the applicable grant date of the Award and, in the absence of such agreement, such term means, with respect to a Participant, the occurrence of any of the following events: (i) such Participant's commission of any felony or any crime involving fraud, dishonesty or moral turpitude under the laws of the applicable jurisdiction; (ii) such Participant's attempted commission of, or participation in, a fraud or act of dishonesty against the Company or an Affiliate; (iii) such Participant's intentional, material violation of any material Company or Affiliate policy applicable to the Participant or any contract or agreement between the Participant and the Company or an Affiliate or of any statutory duty owed to the Company or an Affiliate; (iv) such Participant's unauthorized use or disclosure of the Company's or an Affiliate's confidential information or trade secrets (subject to any whistleblower, defense of trade secrets and similar limitations therein); or (v) such Participant's gross misconduct. The determination that a termination of the Participant's Continuous Service is either for Cause or without Cause will be made by the Company, in its sole discretion. Any determination by the Company that the Continuous Service of a Participant was terminated with or without Cause for the purposes of outstanding Awards held by such Participant will have no effect upon any determination of the rights or obligations of the Company or such a termination of the rights or obligations of the Company or such Participant for any other purpose.

Ex J-13

(i) "*Change in Control*" means the occurrence, in a single transaction or in a series of related transactions, of any one (1) or more of the following events:

(i) any Exchange Act Person becomes the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities other than by virtue of a merger, consolidation or similar transaction. Notwithstanding the foregoing, a Change in Control shall not be deemed to occur (A) on account of the acquisition of securities of the Company directly from the Company, (B) on account of the acquisition of securities of the Company by an investor, any affiliate thereof or any other Exchange Act Person that acquires the Company's securities in a transaction or series of related transactions the primary purpose of which is to obtain financing for the Company through the issuance of equity securities, or (C) solely because the level of Ownership held by any Exchange Act Person (the "*Subject Person*") exceeds the designated percentage threshold of the outstanding, provided that if a Change in Control would occur (but for the operation of this sentence) as a result of the acquisition of voting securities by the Company, and after such share acquisition, the Subject Person becomes the Owner of any additional voting securities that, assuming the repurchase or other acquisition had not occurred, increases the percentage of the then outstanding voting securities Owned by the Subject Person over the designated percentage threshold, then a Change in Control shall be deemed to occur;

(ii) there is consummated a merger, consolidation or similar transaction involving (directly or indirectly) the Company and, immediately after the consummation of such merger, consolidation or similar transaction, the shareholders of the Company immediately prior thereto do not Own, directly or indirectly, either (A) outstanding voting securities representing more than fifty percent (50%) of the combined outstanding voting power of the surviving Entity in such merger, consolidation or similar transaction or (B) more than fifty percent (50%) of the combined outstanding voting power of the parent of the surviving Entity in such merger, consolidation or similar transaction, in each case in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such transaction;

(iii) the shareholders of the Company approve or the Board approves a plan of complete dissolution or liquidation of the Company, or a complete dissolution or liquidation of the Company will otherwise occur, except for a liquidation into a parent corporation; or

(iv) there is consummated a sale, lease, exclusive license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries, other than a sale, lease, license or other disposition of all or substantially all of the consolidated assets of the Company and its Subsidiaries to an Entity, more than fifty percent (50%) of the combined voting power of the voting securities of which are Owned by shareholders of the Company in substantially the same proportions as their Ownership of the outstanding voting securities of the Company immediately prior to such sale, lease, license or other disposition.

Notwithstanding the foregoing definition or any other provision of this Plan, the term Change in Control will not include a sale of assets, merger or other transaction effected exclusively for the purpose of changing the domicile of the Company.

If required for compliance with Section 409A of the Code, in no event will a Change in Control be deemed to have occurred if such transaction is not also a "change in the ownership or effective control of" the Company or "a change in the ownership of a substantial portion of the assets of" the Company as determined under Treasury Regulation Section 1.409A-3(i)(5) (without regard to any alternative definition thereunder).

(j) "Class A Ordinary Shares" means Class A ordinary shares of the Company, par value \$0.0001 per share.

(k) "Class B Ordinary Shares" means Class B ordinary shares of the Company, par value \$0.0001 per share

(I) "Code" means the U.S. Internal Revenue Code of 1986, as amended, including any applicable regulations and guidance thereunder.

(m) "*Committee*" means a committee of one (1) or more Directors to whom authority has been delegated by the Board in accordance with <u>Section 2(c)</u>.

(n) "*Company*" means SAI.TECH Global Corporation (formerly named TradeUP Global Corporation), a Cayman Islands exempted company incorporated with limited liability with company number 370735.

(o) "*Consultant*" means any person, including an advisor, who is (i) engaged by the Company or an Affiliate to render consulting or advisory services and is compensated for such services, or (ii) serving as a member of the board of directors of an Affiliate and is compensated for such services. However, service solely as a Director, or payment of a fee for such service, will not cause a Director to be considered a "Consultant" for purposes of the Plan. Notwithstanding the foregoing, a person is treated as a Consultant under this Plan only if a Form S-8 Registration Statement under the Securities Act is available to register either the offer or the sale of the Company's securities to such person.

(p) "Continuous Service" means that the Participant's service with the Company or an Affiliate, whether as an Employee, Director or Consultant, is not interrupted or terminated. A change in the capacity in which the Participant renders service to the Company or an Affiliate as an Employee, Director or Consultant or a change in the Entity for which the Participant renders such service, provided that there is no interruption or termination of the Participant's service with the Company or an Affiliate, will not terminate a Participant's Continuous Service; provided, however, that if the Entity for which a Participant is rendering services ceases to qualify as an Affiliate, as determined by the Board in its sole discretion, such Participant's Continuous Service will be considered to have terminated on the date such Entity ceases to qualify as an Affiliate. For example, a change in status from an Employee of the Company to a Consultant of an Affiliate or to a Director will not constitute an interruption of Continuous Service. To the extent permitted by law, the Board or the chief executive officer of the Company, in that party's sole discretion, may determine whether Continuous Service will be considered interrupted in the case of (i) any leave of absence approved by the Board or chief executive officer, including sick leave, military leave or any other personal leave, or (ii) transfers between the Company, an Affiliate, or their successors. Notwithstanding the foregoing, a leave of absence will be treated as Continuous Service for purposes of vesting in an Award only to such extent as may be provided in the Company's leave of absence policy, in the written terms of any leave of absence agreement or policy applicable to the Participant, or as otherwise required by law.

(q) "Director" means a member of the Board.

(r) "*Disability*" means, with respect to a Participant, the inability of such Participant to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment that can be expected to result in death or that has lasted or can be expected to last for a continuous period of not less than twelve (12) months, and will be determined by the Board on the basis of such medical evidence as the Board deems warranted under the circumstances.

(s) "Effective Date" has the meaning set forth in Section 1(a).

(t) "*Employee*" means any person employed by the Company or an Affiliate. However, service solely as a Director, or payment of a fee for such services, will not cause a Director to be considered an "Employee" for purposes of the Plan.

(u) "Entity" means a corporation, partnership, limited liability company or other entity.

(v) "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

(w) "*Exchange Act Person*" means any natural person, entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act), except that "Exchange Act Person" will not include (i) the Company or any Subsidiary of the Company, (ii) any employee benefit plan of the Company or any Subsidiary of the Company or any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary of the Company, (iii) an underwriter temporarily holding securities pursuant to a registered public offering of such securities, (iv) an entity Owned, directly or indirectly, by the shareholders of the Company in substantially the same proportions as their Ownership of stock of the Company; (v) any natural person, Entity or "group" (within the meaning of Section 13(d) or 14(d) of the Exchange Act) that, as of the Effective Date, is the Owner, directly or indirectly, of securities of the Company representing more than 50% of the combined voting power of the Company's then outstanding securities; or (vi) any holder of Class B Ordinary Shares as of the Effective Date.

(x) "*Fair Market Value*" means a price that is based on the opening, closing, actual, high, low, or average selling prices of an Ordinary Share reported on NASDAQ or such other established stock exchange on which the Ordinary Shares are principally traded on the applicable date, the preceding trading day, the next succeeding trading day, or an average of trading days, as determined by the Committee in its discretion. Unless the Committee determines otherwise, Fair Market Value shall be deemed to be equal to the reported closing price of an Ordinary Share on the date as of which such value is being determined or, if there shall be no reported transactions for such date, on the preceding date for which transactions were reported; provided, however, that if the Ordinary Shares are not publicly traded at the time a determination of their value is required to be made hereunder, the determination of their Fair Market Value shall be made by the Committee in such manner as it deems appropriate and in accordance with Section 409A of the Code.

(y) "*Incentive Share Option*" means an Option that is intended to be, and that qualifies as, an "incentive stock option" within the meaning of Section 422 of the Code.

(z) "Non-Employee Director" means a Director who is not an Employee.

(aa) "Nonstatutory Share Option" means any Option that is not an Incentive Stock Option.

(bb) "Officer" means any person designated by the Company as an officer.

(cc) "Option" means an option to purchase Ordinary Shares granted pursuant to the Plan.

Ex J-15

(dd) "Option Agreement" means a written agreement between the Company and an Optionholder evidencing the terms and conditions of an Option grant. Each Option Agreement will be subject to the terms and conditions of the Plan.

(ee) "Optionholder" means a person to whom an Option is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Option.

(ff) "Ordinary Shares" means Class A Ordinary Shares.

(gg) "Other Award" means an award based in whole or in part by reference to the Ordinary Shares which is granted pursuant to the terms and conditions of <u>Section 6(c)</u>.

(hh) "Other Award Agreement" means a written agreement between the Company and a holder of an Other Award evidencing the terms and conditions of an Other Award grant. Each Other Award Agreement will be subject to the terms and conditions of the Plan.

(ii) "*Own*," "*Owned*," "*Owner*," "*Ownership*" means a person or Entity will be deemed to "Own," to have "Owned," to be the "Owner" of, or to have acquired "Ownership" of securities if such person or Entity, directly or indirectly, through any contract, arrangement, understanding, relationship or otherwise, has or shares voting power, which includes the power to vote or to direct the voting, with respect to such securities.

(jj) "Participant" means a person to whom an Award is granted pursuant to the Plan or, if applicable, such other person who holds an outstanding Award.

(kk) "*Plan*" means this SAI.TECH Global Corporation 2021 Equity Incentive Plan, as may be amended and/or amended and restated from time to time.

(II) "Restricted Share Award" means an award of Ordinary Shares which is granted pursuant to the terms and conditions of Section 6(a).

(mm) "Restricted Shares" means shares issued in accordance with a Restricted Share Award.

(nn) "*Restricted Share Award Agreement*" means a written agreement between the Company and a holder of a Restricted Share Award evidencing the terms and conditions of a Restricted Share Award grant. Each Restricted Share Award Agreement will be subject to the terms and conditions of the Plan.

(oo) "*Restricted Share Unit Award*" means a right to receive Ordinary Shares which is granted pursuant to the terms and conditions of <u>Section 6(b)</u>.

(**pp**) "*Restricted Share Unit Award Agreement*" means a written agreement between the Company and a holder of a Restricted Share Unit Award evidencing the terms and conditions of a Restricted Share Unit Award grant. Each Restricted Share Unit Award Agreement will be subject to the terms and conditions of the Plan.

(qq) "Securities Act" means the U.S. Securities Act of 1933, as amended.

(rr) "Share Appreciation Right" or "SAR" means a right to receive the appreciation on Ordinary Shares that is granted pursuant to the terms and conditions of <u>Section 5</u>.

(ss) "Share Appreciation Right Agreement" means a written agreement between the Company and a holder of a Share Appreciation Right evidencing the terms and conditions of a Share Appreciation Right grant. Each Share Appreciation Right Agreement will be subject to the terms and conditions of the Plan.

(tt) "Share Reserve" means the aggregate number of Ordinary Shares available for issuance pursuant to Awards from and after the Effective Date under the Plan as set forth in Section 3(a).

(uu) "Subsidiary" means, with respect to the Company, (i) any corporation of which more than fifty percent (50%) of the outstanding capital shares having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether, at the time, share of any other class or classes of such corporation will have or might have voting power by reason of the happening of any contingency) is at the time, directly or indirectly, Owned by the Company, and (ii) any partnership, limited liability company or other entity in which the Company has a direct or indirect interest (whether in the form of voting or participation in profits or capital contribution) of more than fifty percent (50%).

(vv) "*Substitute Awards*" means Awards granted or Ordinary Shares issued by the Company in assumption of, or in substitution or exchange for, awards previously granted, or the right or obligation to make future awards, in each case by a company or other entity acquired by the Company or any Subsidiary or with which the Company or any Subsidiary combines.

(ww) "Transaction" means a Corporate Transaction or a Change in Control.

(xx) "U.S." means the United States.

(yy) "U.S. Participant" means a Participant that is either a U.S. resident or a U.S. taxpayer.

Ex J-17

#### **INDEMNITY AGREEMENT**

**THIS INDEMNITY AGREEMENT** (this "*Agreement*") is made as of April 29, 2022, by and between SAI.TECH Global Corporation, a Cayman Islands exempted company (the "Company"), and the undersigned parties listed as "Indemnitees" on the signature page hereto (each, an "Indemnitee" and collectively, the "Indemnitees")

WHEREAS, highly competent persons have become more reluctant to serve publicly-held corporations as directors, officers or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of such corporations;

WHEREAS, the Board of Directors of the Company (the "*Board*") has determined that, in order to attract and retain qualified individuals, the Company will attempt to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. The amended and restated memorandum and articles of association of the Company (the "*Articles*") provide for the indemnification of the officers and directors of the Company. Indemnitee may also be entitled to indemnification pursuant to applicable Cayman Islands law. The Articles provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the Board, officers and other persons with respect to indemnification, hold harmless, exoneration, advancement and reimbursement rights;

WHEREAS, the uncertainties relating to such insurance and to indemnification have increased the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company's shareholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, hold harmless, exonerate and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so protected against liabilities;

WHEREAS, this Agreement is a supplement to and in furtherance of the Articles and any resolutions adopted pursuant thereto, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder; and

WHEREAS, Indemnitee may not be willing to serve as an officer or director without adequate protection, and the Company desires Indemnitee to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that he or she be so indemnified.

**NOW, THEREFORE,** in consideration of the premises and the covenants contained herein, the Company and Indemnitee do hereby covenant and agree as follows:

## 1. SERVICES TO THE COMPANY

In consideration of the Company's covenants and obligations hereunder, Indemnitee will serve or continue to serve as an officer, director, advisor, key employee or in any other capacity of the Company, as applicable, for so long as Indemnitee is duly elected or appointed or retained or until Indemnitee tenders his or her resignation or until Indemnitee is removed. The foregoing notwithstanding, this Agreement shall continue in full force and effect after Indemnitee has ceased to serve as a director, officer, advisor, key employee or in any other capacity of the Company, as provided in Section 17. This Agreement, however, shall not impose any obligation on Indemnitee or the Company to continue Indemnitee's service to the Company beyond any period otherwise required by law or by other agreements or commitments of the parties, if any.

#### As used in this Agreement:

(a) References to "*agent*" shall mean any person who is or was a director, officer or employee of the Company or a subsidiary of the Company or other person authorized by the Company to act for the Company, to include such person serving in such capacity as a director, officer, employee, advisor, fiduciary or other official of another corporation, partnership, limited liability company, joint venture, trust or other enterprise at the request of, for the convenience of, or to represent the interests of the Company or a subsidiary of the Company.

(b) The terms "*Beneficial Owner*" and "*Beneficial Ownership*" shall have the meanings set forth in Rule 13d-3 promulgated under the Exchange Act as in effect on the date hereof.

(c) "*Cayman Court*" shall mean the courts of the Cayman Islands.

(d) A "*Change in Control*" shall be deemed to occur upon the earliest to occur after the date of this Agreement of any of the following events:

(i) <u>Acquisition of Shares by Third Party</u>. Other than an affiliate of TradeUP Global Sponsor LLC (the "*Sponsor*") or Risheng Li ("*Li*"), including Energy Science Artist Holding Limited, any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company representing fifteen percent (15%) or more of the combined voting power of the Company's then outstanding securities entitled to vote generally in the election of directors, unless (1) the change in the relative Beneficial Ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors, or (2) such acquisition was approved in advance by the Continuing Directors and such acquisition would not constitute a Change in Control under part (iii) of this definition;

(ii) <u>Change in Board of Directors</u>. Individuals who, as of the date hereof, constitute the Board, and any new director whose election by the Board or nomination for election by the Company's shareholders was approved by a vote of at least two thirds of the directors then still in office who were directors on the date hereof or whose election or nomination for election was previously so approved (collectively, the "*Continuing Directors*"), cease for any reason to constitute at least a majority of the members of the Board;

(iii) <u>Corporate Transactions</u>. The effective date of a merger, share exchange, asset acquisition, share purchase, reorganization or similar business combination, involving the Company and one or more businesses (a "*Business Combination*"), in each case, unless, following such Business Combination: (1) all or substantially all of the individuals and entities who were the Beneficial Owners of securities entitled to vote generally in the election of directors immediately prior to such Business Combination beneficially own, directly or indirectly, more than 51% of the combined voting power of the then outstanding securities of the Company entitled to vote generally in the election of directors resulting from such Business Combination (including, without limitation, a corporation which as a result of such transaction owns the Company or all or substantially all of the Company's assets either directly or through one or more Subsidiaries) in substantially the same proportions as their ownership immediately prior to such Business Combination, of the securities entitled to vote generally in the election of directors; (2) other than an affiliate of the Sponsor or Li, no Person (excluding any corporation resulting from such Business Combination) is the Beneficial Owner, directly or indirectly, of 15% or more of the combined voting power of the then outstanding securities entitled to vote generally in the election of directors; and (3) at least a majority of the board of directors of the corporation resulting from such Business Combination; and (3) at least a majority of the initial agreement, or of the action of the Board, providing for such Business Combination; and the time of the execution of the initial agreement, or of the action of the Board, providing for such Business Combination;

2

(iv) <u>Liquidation</u>. The approval by the shareholders of the Company of a complete liquidation of the Company or an agreement or series of agreements for the sale or disposition by the Company of all or substantially all of the Company's assets, other than factoring the Company's current receivables or escrows due (or, if such approval is not required, the decision by the Board to proceed with such a liquidation, sale, or disposition in one transaction or a series of related transactions); or

(v) <u>Other Events</u>. There occurs any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or a response to any similar item on any similar schedule or form) promulgated under the Exchange Act, whether or not the Company is then subject to such reporting requirement.

(e) "*Corporate Status*" describes the status of a person who is or was a director, officer, trustee, general partner, manager, managing member, fiduciary, employee or agent of the Company or of any other Enterprise which such person is or was serving at the request of the Company.

(f) "*Disinterested Director*" shall mean a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

(g) "*Enterprise*" shall mean the Company and any other corporation, constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger to which the Company (or any of its wholly owned subsidiaries) is a party, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise of which Indemnitee is or was serving at the request of the Company as a director, officer, trustee, general partner, manager, managing member, fiduciary, employee or agent.

(h) "*Exchange Act*" shall mean the Securities Exchange Act of 1934, as amended.

(i) "*Expenses*" shall include all direct and indirect costs, fees and expenses of any type or nature whatsoever, including, without limitation, all reasonable attorneys' fees and costs, retainers, court costs, transcript costs, fees of experts, witness fees, travel expenses, fees of private investigators and professional advisors, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, fax transmission charges, secretarial services and all other disbursements, obligations or expenses in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a witness in, settlement or appeal of, or otherwise participating in, a Proceeding, including reasonable compensation for time spent by the Indemnitee for which he or she is not otherwise compensated by the Company or any third party. Expenses also shall include Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the principal, premium, security for, and other costs relating to any cost bond, supersedeas bond, or other appeal bond or its equivalent. Expenses, however, shall not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.

(j) References to "fines" shall include any excise tax assessed on Indemnitee with respect to any employee benefit plan.

(k) References to "*serving at the request of the Company*" shall include any service as a director, officer, employee, agent or fiduciary of the Company which imposes duties on, or involves services by, such director, officer, employee, agent or fiduciary with respect to an employee benefit plan, its participants or beneficiaries; and if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan, Indemnitee shall be deemed to have acted in a manner "*not opposed to the best interests of the Company*" as referred to in this Agreement.

(1) "*Independent Counsel*" shall mean a law firm or a member of a law firm with significant experience in matters of corporate law and that neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements); or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "*Independent Counsel*" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine

Indemnitee's rights under this Agreement.

(m) The term "*Person*" shall have the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act as in effect on the date hereof; provided, however, that "*Person*" shall exclude: (i) the Company; (ii) any Subsidiaries of the Company; (iii) any employment benefit plan of the Company or of a Subsidiary of the Company or of any corporation owned, directly or indirectly, by the shareholders of the Company in substantially the same proportions as their ownership of shares of the Company or of a corporation owned directly or indirectly or indirectly or indirectly or indirectly or indirectly by the shareholders of the Company or of a Subsidiary of the Company or of a Subsidiary of the Company or of a corporation owned directly or indirectly by the shareholders of the Company in substantially the same proportions as their ownership of as their ownership of shares of the Company or of a corporation owned directly or indirectly by the shareholders of the Company in substantially the same proportions as their ownership of shares of the Company or of a corporation owned directly or indirectly by the shareholders of the Company in substantially the same proportions as their ownership of shares of the Company.

(n) The term "*Proceeding*" shall include any threatened, pending or completed action, suit, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil (including intentional or unintentional tort claims), criminal, administrative or investigative or related nature, in which Indemnitee was, is, will or might be involved as a party or otherwise

by reason of the fact that Indemnitee is or was a director or officer of the Company, by reason of any action (or failure to act) taken by him or her or of any action (or failure to act) on his or her part while acting as a director or officer of the Company, or by reason of the fact that he or she is or was serving at the request of the Company as a director, officer, trustee, general partner, manager, managing member, fiduciary, employee or agent of any other Enterprise, in each case whether or not serving in such capacity at the time any liability or expense is incurred for which indemnification, reimbursement, or advancement of expenses can be provided under this Agreement.

(o) The term "*Subsidiary*," with respect to any Person, shall mean any corporation, limited liability company, partnership, joint venture, trust or other entity of which a majority of the voting power of the voting equity securities or equity interest is owned, directly or indirectly, by that Person.

(p) The phrase "to the fullest extent permitted by applicable law" shall include, but not be limited to: (a) to the fullest extent authorized or permitted by the provision of applicable Cayman Islands law that authorizes or contemplates additional indemnification by agreement, or the corresponding provision of any amendment to or replacement of applicable Cayman Islands law, and (b) to the fullest extent authorized or permitted by any amendments to or replacements of applicable Cayman Islands law adopted after the date of this Agreement that increase the extent to which a corporation may indemnify its officers and directors.

#### 3. INDEMNITY IN THIRD-PARTY PROCEEDINGS

To the fullest extent permitted by applicable law, the Company shall indemnify, hold harmless and exonerate Indemnitee in accordance with the provisions of this Section 3 if Indemnitee was, is, or is threatened to be made, a party to or a participant (as a witness, deponent or otherwise) in any Proceeding, other than a Proceeding by or in the right of the Company to procure a judgment in its favor by reason of Indemnitee's Corporate Status. Pursuant to this Section 3, Indemnitee shall be indemnified, held harmless and exonerated against all Expenses, judgments, liabilities, fines, penalties and amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines, penalties and amounts paid in settlement) actually, and reasonably incurred by Indemnitee or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal Proceeding, had no reasonable cause to believe that his or her conduct was unlawful; provided, in no event shall Indemnitee be entitled to be indemnified, held harmless or advanced any amounts hereunder in respect of any Expenses, judgments, liabilities, fines, penalties and amounts paid in settlement (if any) that Indemnitee may incur by reason of his or her own actual fraud or intentional misconduct. Indemnitee shall not be found to have committed actual fraud or intentional misconduct for any purpose of this Agreement unless or until a court of competent jurisdiction shall have made a finding to that effect.

## 4. INDEMNITY IN PROCEEDINGS BY OR IN THE RIGHT OF THE COMPANY

To the fullest extent permitted by applicable law, the Company shall indemnify, hold harmless and exonerate Indemnitee in accordance with the provisions of this Section 4 if Indemnitee was, is, or is threatened to be made, a party to or a participant (as a witness, deponent or otherwise) in any Proceeding by or in the right of the Company to procure a judgment in its favor by reason of Indemnitee's Corporate Status. Pursuant to this Section 4, Indemnitee shall be indemnified, held harmless and exonerated against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Company. No indemnification, hold harmless or exoneration for Expenses shall be made under this Section 4 in respect of any claim, issue or matter as to which Indemnitee shall have been finally adjudged by a court of competent jurisdiction to be liable to the Company, unless and only to the extent that any court in which the Proceeding was brought or the Cayman Court shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification, to be held harmless or to exoneration.

4

# 5. INDEMNIFICATION FOR EXPENSES OF A PARTY WHO IS WHOLLY OR PARTLY SUCCESSFUL

Notwithstanding any other provisions of this Agreement, but subject to Section 27, to the extent that Indemnitee was or is, by reason of Indemnitee's Corporate Status, a party to (or a participant in) and is successful, on the merits or otherwise, in any Proceeding or in defense of any claim, issue or matter therein, in whole or in part, the Company shall, to the fullest extent permitted by applicable law, indemnify, hold harmless and exonerate Indemnitee against all Expenses actually and reasonably incurred by him or her in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but

less than all claims, issues or matters in such Proceeding, the Company shall, to the fullest extent permitted by applicable law, indemnify, hold harmless and exonerate Indemnitee against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection with each successfully resolved claim, issue or matter. If Indemnitee is not wholly successful in such Proceeding, the Company also shall, to the fullest extent permitted by applicable law, indemnify, hold harmless and exonerate Indemnitee against all Expenses reasonably incurred in connection with a claim, issue or matter related to any claim, issue, or matter on which Indemnitee was successful. For purposes of this Section 5 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

# 6. INDEMNIFICATION FOR EXPENSES OF A WITNESS

Notwithstanding any other provision of this Agreement, but subject to Section 27, to the extent that Indemnitee is, by reason of his or her Corporate Status, a witness or deponent in any Proceeding to which Indemnitee is not a party or threatened to be made a party, he or she shall, to the fullest extent permitted by applicable law, be indemnified, held harmless and exonerated against all Expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith.

# 7. ADDITIONAL INDEMNIFICATION, HOLD HARMLESS AND EXONERATION RIGHTS

Notwithstanding any limitation in Sections 3, 4 or 5, but subject to Section 27, the Company shall, to the fullest extent permitted by applicable law, indemnify, hold harmless and exonerate Indemnitee if Indemnitee is a party to or threatened to be made a party to any Proceeding (including a Proceeding by or in the right of the Company to procure a judgment in its favor) against all Expenses, judgments, fines, penalties and amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines, penalties and amounts paid in settlement, hold harmless or exoneration rights shall be available under this Section 7 on account of Indemnitee's conduct which constitutes a breach of Indemnitee's duty of loyalty to the Company or its shareholders or is an act or omission not in good faith or which involves intentional misconduct or a knowing violation of applicable law.

# 8. CONTRIBUTION IN THE EVENT OF JOINT LIABILITY

(a) To the fullest extent permissible under applicable law, if the indemnification, hold harmless and/or exoneration rights provided for in this Agreement are unavailable to Indemnitee in whole or in part for any reason whatsoever, the Company, in lieu of indemnifying, holding harmless or exonerating Indemnitee, shall pay, in the first instance, the entire amount incurred by Indemnitee, whether for judgments, liabilities, fines, penalties, amounts paid or to be paid in settlement and/or for Expenses, in connection with any Proceeding without requiring Indemnitee to contribute to such payment, and the Company hereby waives and relinquishes any right of contribution it may have at any time against Indemnitee.

(b) The Company shall not enter into any settlement of any Proceeding in which the Company is jointly liable with Indemnitee (or would be if joined in such Proceeding) unless such settlement provides for a full and final release of all claims asserted against Indemnitee.

(c) The Company hereby agrees to fully indemnify, hold harmless and exonerate Indemnitee from any claims for contribution which may be brought by officers, directors or employees of the Company other than Indemnitee who may be jointly liable with Indemnitee. Indemnitee shall seek payments or advances from the Company only to the extent that such payments or advances are unavailable from any insurance policy of the Company covering Indemnitee.

5

#### 9. EXCLUSIONS

Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnification, advance Expenses, hold harmless or exoneration payment in connection with any claim made against Indemnitee:

(a) for which payment has actually been received by or on behalf of Indemnitee under any insurance policy or other indemnity or advancement provision, except with respect to any excess beyond the amount actually received under any insurance policy, contract, agreement, other indemnity or advancement provision or otherwise;

(b) for an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act (or any successor rule) or similar provisions of state statutory law or common law; or

(c) except as otherwise provided in Sections 14(f) and (g) hereof, prior to a Change in Control, in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation or

(ii) the Company provides the indemnification, hold harmless or exoneration payment, in its sole discretion, pursuant to the powers vested in the Company under applicable law.

## 10. ADVANCES OF EXPENSES; DEFENSE OF CLAIM

(a) Notwithstanding any provision of this Agreement to the contrary, but subject to Section 27, and to the fullest extent not prohibited by applicable law, the Company shall pay the Expenses incurred by Indemnitee (or reasonably expected by Indemnitee to be incurred by Indemnitee within three months) in connection with any Proceeding within ten (10) days after the receipt by the Company of a statement or statements requesting such advances from time to time, prior to the final disposition of any Proceeding. Advances shall, to the fullest extent permitted by law, be unsecured and interest free. Advances shall, to the fullest extent permitted by law, be made without regard to Indemnitee's ability to repay the Expenses and without regard to Indemnitee's ultimate entitlement to be indemnified, held harmless or exonerated under the other provisions of this Agreement. Advances shall include any and all reasonable Expenses incurred pursuing a Proceeding to enforce this right of advancement, including Expenses incurred preparing and forwarding statements to the Company to support the advances claimed. To the fullest extent required by applicable law, such payments of Expenses in advance of the final disposition of the Proceeding shall be made only upon the Company's receipt of an undertaking, by or on behalf of Indemnitee, to repay the advanced amounts to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company under the provisions of this Agreement, the Articles, applicable law or otherwise. If it shall be determined by a final judgment or other final adjudication that Indemnitee was not so entitled to indemnification, any advancement shall be returned to the Company (without interest) by the Indemnitee. This Section 10(a) shall not apply to any claim made by Indemnitee for which an indemnification, hold harmless or exoneration payment is excluded pursuant to Section 9, but shall apply to any Proceeding referenced in Section 9(b) prior to a final determination that Indemnitee is liable therefor.

(b) The Company will be entitled to participate in the Proceeding at its own expense.

(c) The Company shall not settle any action, claim or Proceeding (in whole or in part) which would impose any Expense, judgment, fine, penalty or limitation on Indemnitee without Indemnitee's prior written consent.

# 11. PROCEDURE FOR NOTIFICATION AND APPLICATION FOR INDEMNIFICATION

(a) Indemnitee agrees to notify promptly the Company in writing upon being served with any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding, claim, issue or matter therein which may be subject to indemnification, hold harmless or exoneration rights, or advancement of Expenses covered hereunder. The failure of Indemnitee to so notify the Company shall not relieve the Company of any obligation which it may have to Indemnitee under this Agreement, or otherwise.

6

(b) Indemnitee may deliver to the Company a written application to indemnify, hold harmless or exonerate Indemnitee in accordance with this Agreement. Such application(s) may be delivered from time to time and at such time(s) as Indemnitee deems appropriate in his or her sole discretion. Following such a written application for indemnification by Indemnitee, Indemnitee's entitlement to indemnification shall be determined according to Section 12(a) of this Agreement.

## 12. PROCEDURE UPON APPLICATION FOR INDEMNIFICATION

(a) A determination, if required by applicable law, with respect to Indemnitee's entitlement to indemnification shall be made in the specific case by one of the following methods, which shall be at the election of Indemnitee: (i) by a majority vote of the Disinterested Directors, even though less than a quorum of the Board, (ii) by a committee of such directors designated by majority vote of such

directors, (iii) if there are no Disinterested Directors or if such directors so direct, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee, or (iv) by vote of the shareholders by ordinary resolution. The Company promptly will advise Indemnitee in writing with respect to any determination that Indemnitee is or is not entitled to indemnification, including a description of any reason or basis for which indemnification has been denied. If it is so determined that Indemnitee is entitled to indemnification, payment to Indemnitee shall be made within ten (10) days after such determination. Indemnitee shall reasonably cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. Any costs or Expenses (including reasonable attorneys' fees and disbursements) incurred by Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification) and the Company hereby agrees to indemnify and to hold Indemnitee harmless therefrom.

(b) In the event the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 12(a) hereof, the Independent Counsel shall be selected as provided in this Section 12(b). The Independent Counsel shall be selected by Indemnitee (unless Indemnitee shall request that such selection be made by the Board), and Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected and certifying that the Independent Counsel so selected meets the requirements of "Independent Counsel" as defined in Section 2 of this Agreement. If the Independent Counsel is selected by the Board, the Company shall give written notice to Indemnitee advising him or her of the identity of the Independent Counsel so selected and certifying that the Independent Counsel so selected meets the requirements of "Independent Counsel" as defined in Section 2 of this Agreement. In either event, Indemnitee or the Company, as the case may be, may, within ten (10) days after such written notice of selection shall have been received, deliver to the Company or to Indemnitee, as the case may be, a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 2 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or a court of competent jurisdiction has determined that such objection is without merit. If, within twenty (20) days after submission by Indemnitee of a written request for indemnification pursuant to Section 11(b) hereof, no Independent Counsel shall have been selected and not objected to, either the Company or Indemnitee may petition the Cayman Court for resolution of any objection which shall have been made by the Company or Indemnitee to the other's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by the Cayman Court, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 12(a) hereof. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 14(a) of this Agreement, Independent Counsel shall be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

(c) The Company agrees to pay the reasonable fees and expenses of Independent Counsel and to fully indemnify and hold harmless such Independent Counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.

7

## 13. PRESUMPTIONS AND EFFECT OF CERTAIN PROCEEDINGS

(a) In making a determination with respect to entitlement to indemnification hereunder, the person, persons or entity making such determination shall presume that Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 11(b) of this Agreement, and the Company shall have the burden of proof to overcome that presumption in connection with the making by any person, persons or entity of any determination contrary to that presumption. Neither the failure of the Company (including by the Disinterested Directors or Independent Counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by the Disinterested Directors or Independent Counsel) to the action or create a presumption that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.

(b) If the person, persons or entity empowered or selected under Section 12 of this Agreement to determine whether Indemnitee is entitled to indemnification shall not have made a determination within thirty (30) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall, to the fullest extent permitted by law, be deemed to have been

made and Indemnitee shall be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a final judicial determination that any or all such indemnification is expressly prohibited under applicable law; provided, however, that such 30-day period may be extended for a reasonable time, not to exceed an additional fifteen (15) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto.

(c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of nolo contendere or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which he or she reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that his or her conduct was unlawful.

(d) For purposes of any determination of good faith, Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise, including financial statements, or on information supplied to Indemnitee by the directors, manager, or officers of the Enterprise in the course of their duties, or on the advice of legal counsel for the Enterprise, its Board, any committee of the Board or any director, trustee, general partner, manager or managing member, or on information or records given or reports made to the Enterprise, its Board, any committee of the Board or any director, trustee, general partner, manager or other expert selected by the Enterprise, its Board, any committee of the Board or any director, trustee, general partner, manager or managing member, by an independent certified public accountant or by an appraiser or other expert selected by the Enterprise, its Board, any committee of the Board or any director, trustee, general partner, manager or managing member. The provisions of this Section 13(d) shall not be deemed to be exclusive or to limit in any way the other circumstances in which Indemnitee may be deemed or found to have met the applicable standard of conduct set forth in this Agreement.

(e) The knowledge and/or actions, or failure to act, of any other director, officer, trustee, partner, manager, managing member, fiduciary, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement.

#### 14. **REMEDIES OF INDEMNITEE**

(a) In the event that (i) a determination is made pursuant to Section 12 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses, to the fullest extent permitted by applicable law, is not timely made pursuant to Section 10 of this Agreement, (iii) no determination of entitlement to indemnification shall have been made pursuant to Section 12(a) of this Agreement within thirty (30) days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to Sections 5, 6, 7 or the last sentence of Section 12(a) of this Agreement within ten (10) days after receipt by the Company of a written request therefor, (v) a contribution payment is not made in a timely manner pursuant to Section 8 of this Agreement, (vi) payment of indemnification pursuant to Section 3 or 4 of this Agreement is not made within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification, or (vii) payment to Indemnitee pursuant to any hold harmless or exoneration rights under this Agreement or otherwise is not made in accordance with this Agreement within ten (10) days after receipt by the Company of a written request therefor, Indemnitee shall be entitled to an adjudication by the Cayman Court to such indemnification, hold harmless, exoneration, contribution or advancement rights. Alternatively, Indemnitee, at his or her option, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules and Mediation Procedures of the American Arbitration Association. Except as set forth herein, the Commercial Arbitration Rules and Mediation Procedures of the American Arbitration Association shall apply to any such arbitration. The Company shall not oppose Indemnitee's right to seek any such adjudication or award in arbitration.

8

(b) In the event that a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 14 shall be conducted in all respects as a de novo trial, or arbitration, on the merits and Indemnitee shall not be prejudiced by reason of that adverse determination.

(c) In any judicial proceeding or arbitration commenced pursuant to this Section 14, Indemnitee shall be presumed to be entitled to be indemnified, held harmless, exonerated to receive advancement of Expenses under this Agreement and the Company shall have the burden of proving Indemnitee is not entitled to be indemnified, held harmless, exonerated and to receive advancement of Expenses, as the case may be, and the Company may not refer to or introduce into evidence any determination pursuant to Section 12(a) of this

Agreement adverse to Indemnitee for any purpose. If Indemnitee commences a judicial proceeding or arbitration pursuant to this Section 14, Indemnitee shall not be required to reimburse the Company for any advances pursuant to Section 10 until a final determination is made with respect to Indemnitee's entitlement to indemnification (as to which all rights of appeal have been exhausted or lapsed).

(d) If a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 14, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.

(e) The Company shall be precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 14 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement.

(f) The Company shall indemnify and hold harmless Indemnitee to the fullest extent permitted by law against all Expenses and, if requested by Indemnitee, shall (within ten (10) days after the Company's receipt of such written request) pay to Indemnitee, to the fullest extent permitted by applicable law, such Expenses which are incurred by Indemnitee in connection with any judicial proceeding or arbitration brought by Indemnitee: (i) to enforce his or her rights under, or to recover damages for breach of, this Agreement or any other indemnification, hold harmless, exoneration, advancement or contribution agreement or provision of the Articles now or hereafter in effect; or (ii) for recovery or advances under any insurance policy maintained by any person for the benefit of Indemnitee, regardless of the outcome and whether Indemnitee ultimately is determined to be entitled to such indemnification, hold harmless or exoneration right, advancement, contribution or insurance recovery, as the case may be (unless such judicial proceeding or arbitration was not brought by Indemnitee in good faith).

(g) Interest shall be paid by the Company to Indemnitee at the legal rate under New York law for amounts which the Company indemnifies, holds harmless or exonerates, or advances, or is obliged to indemnify, hold harmless or exonerate or advance for the period commencing with the date on which Indemnitee requests indemnification, to be held harmless, exonerated, contribution, reimbursement or advancement of any Expenses and ending with the date on which such payment is made to Indemnitee by the Company.

#### 15. SECURITY

Notwithstanding anything herein to the contrary, but subject to Section 27, to the extent requested by Indemnitee and approved by the Board, the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral. Any such security, once provided to Indemnitee, may not be revoked or released without the prior written consent of Indemnitee.

9

## 16. NON-EXCLUSIVITY; SURVIVAL OF RIGHTS; INSURANCE; SUBROGATION; PRIORITY OF OBLIGATIONS

(a) The rights of Indemnitee as provided by this Agreement shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Articles, any agreement, a vote of shareholders or a resolution of directors, or otherwise. No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any Proceeding (regardless of when such Proceeding is first threatened, commenced or completed) or claim, issue or matter therein arising out of, or related to, any action taken or omitted by such Indemnitee in his or her Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in applicable law, whether by statute or judicial decision, permits greater indemnification, hold harmless or exoneration rights or advancement of Expenses than would be afforded currently under the Articles or this Agreement, then this Agreement (without any further action by the parties hereto) shall automatically be deemed to be amended to require that the Company indemnifies the Indemnitee to the fullest extent permitted by law. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

(b) The Articles permit the Company to purchase and maintain insurance or furnish similar protection or make other arrangements including, but not limited to, providing a trust fund, letter of credit, or surety bond ("*Indemnification Arrangements*") on behalf of Indemnitee against any liability asserted against him or her or incurred by or on behalf of him or in such capacity as a director, officer, employee or agent of the Company, or arising out of his or her status as such, whether or not the Company would have the power to indemnify him or her against such liability under the provisions of this Agreement and the Articles. The purchase, establishment, and maintenance of any such Indemnification Arrangement shall not in any way limit or affect the rights and obligations of the Company or of Indemnitee shall not in any way limit or affect the rights and obligations of the Company and Indemnitee shall not in any way limit or affect the rights and obligations of the return or parties thereto under any such Indemnification Arrangement.

(c) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, trustees, partners, managers, managing members, fiduciaries, employees, or agents of the Company or of any other Enterprise which such person serves at the request of the Company, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such director, officer, trustee, partner, manager, managing member, fiduciary, employee or agent under such policy or policies. If, at the time the Company receives notice from any source of a Proceeding as to which Indemnitee is a party or a participant (as a witness, deponent or otherwise), the Company has director and officer liability insurance in effect, the Company shall give prompt notice of such Proceeding to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter use commercially reasonable efforts to cause such insurers to pay, on behalf of Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies.

(d) In the event of any payment under this Agreement, the Company, to the fullest extent permitted by law, shall be subrogated to the extent of such payment to all of the rights of recovery of Indemnitee, who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights. No such payment by the Company shall be deemed to relieve any insurer of its obligations.

(e) The Company's obligation to indemnify, hold harmless, exonerate or advance Expenses hereunder to Indemnitee who is or was serving at the request of the Company as a director, officer, trustee, partner, manager, managing member, fiduciary, employee or agent of any other Enterprise shall be reduced by any amount Indemnitee has actually received as indemnification, hold harmless or exoneration payments or advancement of expenses from such Enterprise. Notwithstanding any other provision of this Agreement to the contrary, but subject to Section 27, (i) Indemnitee shall have no obligation to reduce, offset, allocate, pursue or apportion any indemnification, hold harmless, exoneration, advancement, contribution or insurance coverage among multiple parties possessing such duties to Indemnitee prior to the Company's satisfaction and performance of all its obligations under this Agreement, and (ii) the Company shall perform fully its obligations under this Agreement without regard to whether Indemnitee holds, may pursue or has pursued any indemnification, advancement, hold harmless, exoneration, contribution or insurance coverage rights against any person or entity other than the Company.

10

(f) Notwithstanding anything contained herein, the Company is the primary indemnitor, and any indemnification or advancement obligation of either the Sponsor or Li or their affiliates or members or any other Person is secondary.

## 17. DURATION OF AGREEMENT

All agreements and obligations of the Company contained herein shall continue during the period Indemnitee serves as a director or officer of the Company or as a director, officer, trustee, partner, manager, managing member, fiduciary, employee or agent of any other corporation, partnership, joint venture, trust, employee benefit plan or other Enterprise which Indemnitee serves at the request of the Company and shall continue thereafter so long as Indemnitee shall be subject to any possible Proceeding (including any rights of appeal thereto and any Proceeding commenced by Indemnitee pursuant to Section 14 of this Agreement) by reason of his or her Corporate Status, whether or not he or she is acting in any such capacity at the time any liability or expense is incurred for which indemnification or advancement can be provided under this Agreement.

## **18. SEVERABILITY**

If any provision or provisions of this Agreement shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this Agreement (including, without limitation, each portion of any Section, paragraph or sentence of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and shall remain enforceable to the fullest extent permitted by law; (b) such provision or provisions shall be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of any Section, paragraph or sentence of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested thereby.

#### **19. ENFORCEMENT AND BINDING EFFECT**

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director, officer or key employee of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as a director, officer or key employee of the Company.

(b) Without limiting any of the rights of Indemnitee under the Articles as they may be amended from time to time, this Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof.

(c) The indemnification, hold harmless, exoneration and advancement of expenses rights provided by or granted pursuant to this Agreement shall be binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business and/or assets of the Company), shall continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or a director, officer, trustee, general partner, manager, managing member, fiduciary, employee or agent of any other Enterprise at the Company's request, and shall inure to the benefit of Indemnitee and his or her spouse, assigns, heirs, devisees, executors and administrators and other legal representatives.

(d) The Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all, substantially all or a substantial part, of the business and/or assets of the Company, by written agreement in form and substance satisfactory to Indemnitee, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.

(e) The Company and Indemnitee agree herein that a monetary remedy for breach of this Agreement, at some later date, may be inadequate, impracticable and difficult of proof, and further agree that such breach may cause Indemnitee irreparable harm. Accordingly, the parties hereto agree that Indemnitee may, to the fullest extent permitted by law, enforce this Agreement by seeking, among other things, injunctive relief and/or specific performance hereof, without any necessity of showing actual damage or irreparable harm and that by seeking injunctive relief and/or specific performance, Indemnitee shall not be precluded from seeking or obtaining any other relief to which he or she may be entitled. The Company and Indemnitee further agree that Indemnitee shall, to the fullest extent permitted by law, be entitled to such specific performance and injunctive relief, including temporary restraining orders, preliminary injunctions and permanent injunctions, without the necessity of posting bonds or other undertaking in connection therewith. The Company acknowledges that in the absence of a waiver, a bond or undertaking may be required of Indemnitee by a court of competent jurisdiction, and the Company hereby waives any such requirement of such a bond or undertaking to the fullest extent permitted by law.

# 20. MODIFICATION AND WAIVER

No supplement, modification or amendment of this Agreement shall be binding unless executed in writing by the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provisions of this Agreement nor shall any waiver constitute a continuing waiver.

## 21. NOTICES

All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given (i) if delivered by hand and receipted for by the party to whom said notice or other communication shall have been directed, or (ii) if mailed by certified or registered mail with postage prepaid, on the third (3rd) business day after the date on which it is so mailed: (a) If to Indemnitee, at the address indicated on the signature page of this Agreement, or such other address as Indemnitee shall provide in writing to the Company.

(b) If to the Company, to:

SAI.TECH Global Corporation #01-05, 195 Pearl's Hill Terrace Singapore, 168976 Attention: Risheng Li

With a copy, which shall not constitute notice, to:

Winston & Strawn LLP 800 Capitol Street, Suite 2400 Houston, TX 77002-2925 Attn: Michael J. Blankenship

or to any other address as may have been furnished to Indemnitee in writing by the Company.

# 22. APPLICABLE LAW AND CONSENT TO JURISDICTION

This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of New York, without regard to its conflict of laws rules. Except with respect to any arbitration commenced by Indemnitee pursuant to Section 14(a) of this Agreement, to the fullest extent permitted by law, the Company and Indemnitee hereby irrevocably and unconditionally: (a) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Cayman Court and not in any other state or federal court in the United States of America or any court in any other country; (b) consent to submit to the exclusive jurisdiction of the Cayman Court for purposes of any action or proceeding arising out of or in connection with this Agreement; (c) waive any objection to the laying of venue of any such action or proceeding in the Cayman Court has been brought in an improper or inconvenient forum, or is subject (in whole or in part) to a jury trial. To the fullest extent permitted by law, the parties hereby agree that the mailing of process and other papers in connection with any such action or proceeding in the manner provided by Section 21 or in such other manner as may be permitted by law, shall be valid and sufficient service thereof.

12

# 23. IDENTICAL COUNTERPARTS

This Agreement may be executed in one or more counterparts, each of which shall for all purposes be deemed to be an original but all of which together shall constitute one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sought needs to be produced to evidence the existence of this Agreement.

# 24. MISCELLANEOUS

The headings of the paragraphs of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.

## 25. PERIOD OF LIMITATIONS

No legal action shall be brought and no cause of action shall be asserted by or in the right of the Company against Indemnitee, Indemnitee's spouse, heirs, executors or personal or legal representatives after the expiration of two years from the date of accrual of such cause of action, and any claim or cause of action of the Company shall be extinguished and deemed released unless asserted by the timely filing of a legal action within such two-year period; provided, however, that if any shorter period of limitations is otherwise applicable to any such cause of action such shorter period shall govern.

#### 26. ADDITIONAL ACTS

If for the validation of any of the provisions in this Agreement any act, resolution, approval or other procedure is required to the fullest extent permitted by law, the Company undertakes to cause such act, resolution, approval or other procedure to be affected or adopted in a manner that will enable the Company to fulfill its obligations under this Agreement.

#### 27. [RESERVED]

[RESERVED].

#### **28. MAINTENANCE OF INSURANCE**

The Company shall use commercially reasonable efforts to obtain and maintain in effect during the entire period for which the Company is obligated to indemnify the Indemnitee under this Agreement, one or more policies of insurance with reputable insurance companies to provide the officers/directors of the Company with coverage for losses from wrongful acts and omissions and to ensure the Company's performance of its indemnification obligations under this Agreement. The Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such director or officer under such policy or policies. In all such insurance policies, the Indemnitee shall be named as an insured in such a manner as to provide the Indemnitee with the same rights and benefits as are accorded to the most favorably insured of the Company's directors and officers.

(Signature Page Follows)

13

IN WITNESS WHEREOF, the parties hereto have caused this Indemnity Agreement to be signed as of the day and year first above written.

## SAI.TECH GLOBAL CORPORATION

By:	/s/ Li Risheng
Name:	Li Risheng
Title:	Director

#### **INDEMNITEES:**

Name: Zou Jian Address: Jian Zou Room 705, Tower C, Sino-Ocean Guanghua International, No.10 Jintong West Road, Hujialou Street, Chaoyang District, Beijing (北京市朝阳区呼家楼街道金桐西路10号 远洋光华国际c座 705)

/s/ Risheng Li

Name: Risheng Li Address: XUEYUAN ROAD NO.20, HAIDIAN DISTRICT, BEIJING, CHINA

/s/ Yao Shi

Name: Yao Shi Address: Room 204, Building 22, Lake Left Bank

Garden, Suzhou Industrial Park, Jiangsu Province (江苏省苏州工业园区湖左岸花园 22栋204室)

/s/ Yusen Chen

Name: Yusen Chen Address: 2101, Building 5, Aux shidaizhicheng, Cangqian Street, Yuhang District, Hangzhou, Zhejiang Province (浙江省杭州市余杭区仓 前街道奥克斯时代之城5号楼 2101)

/s/ Jinlong Zhu

Name: Jinlong Zhu Address: Southern University of Science and Technology, Huiyuan Building #1 R-504, 1088 Xueyuan Avenue, Shenzhen 518055 (深圳市 学院路<sup>1088</sup>号南方科技大学惠园1号楼R-504)

Signature Page to Indemnity Agreement

# SAI Announces Closing of Business Combination With TradeUP Global Corporation

#### The combined company to begin trading on the Nasdaq Stock Market on May 2, 2022 under the ticker symbols "SAI" and "SAITW"

SINGAPORE, April 29, 2022 (GLOBE NEWSWIRE) – SAI.TECH Global Corporation ("SAI.TECH" or "SAI"), a global energy-saving bitcoin mining operator and a clean-tech company that integrates Bitcoin mining, heating and power industries, announced today that it is closing today, April 29, 2022, the previously announced business combination of SAITECH Limited and TradeUP Global Corporation ("TradeUP").

The transaction was approved by TradeUP shareholders on April 22, 2022. In connection with the transaction, TradeUP was renamed to "SAI.TECH Global Corporation." The Class A ordinary shares and warrants of SAI will commence trading on the Nasdaq Stock Market on May 2, 2022, under the new ticker symbols, "SAI" and "SAITW," respectively. Such shares and warrants also reflect the continuation of the same shares and warrants of TradeUP Global Corporation under the company's new name following the completion of its business combination with SAITECH Limited, which has become a wholly owned subsidiary of SAI.

"Today marks an important milestone for SAI. I want to start off by thanking everyone who played a part in making SAI become a global leader in the Bitcoin mining industry and a listed company on Nasdaq." said Arthur Lee, Founder and Chief Executive Officer of SAI.TECH. "SAI is a game changer in the global Bitcoin mining industry because our innovation goes beyond Bitcoin mining. We believe our revolutionary liquid cooling and waste heat recovery technology will reshape the computing infrastructure for the Bitcoin network while simultaneously promoting the electrification and decarbonization of the world's traditional heating industry."

TradeUP Global Corporation's Chairman and Co-Chief Executive Officer and the controlling owner of its sponsor, Jianwei Li, is continuing as a director of the combined company. Mr. Li has a demonstrated track record of successful venture capital investments in disruptive technologies including Cloud / SaaS / AI / Robotics / Drone / Network security. He has served as the founding and managing partner of Zhencheng Capital, a venture capital firm specializing in early-stage technology investments since 2016. In his past career, Mr. Li occupied senior positions in renowned investment institutions and consulting groups including ZhenFund, Sequoia Capital China, Fidelity Growth Partners Asia and Boston Consulting Group. He also serves as Chairman & Co-Chief Executive Officer of other special purpose acquisition companies - TradeUP Acquisition Corp., which consummated its initial public offering in July 2021; and TradeUp 88 Corp., which filed a registration statement for its initial public offering in 2021 and remains in process.

"I am very delighted to be joining SAI as a director," said Mr. Jianwei Li. "We are firmly supportive to help SAI promote their innovative and high ESG criteria technology and operation to the global market. As one of the most attractive new asset classes, cryptocurrency, especially Bitcoin, has accelerated its institutional acceptance in recent years. However, its energy-intensive infrastructure has impeded its widespread adoption and public trust with this class of assets. SAI is the most unique Bitcoin mining operator we observed globally who has a set of mature energy-saving technology and solutions that can lead the transition of the industry to a green future."

"By entering the global market, we will continue to make revolutionary changes in the industry." added Arthur Lee. "We launched SAIHUB 3.0 – the next-generation Bitcoin mining integrated solutions featured by greater energy efficiency and an all-in-one cloud management APP earlier this year," he said. "I can't wait to see SAI's next step as we are working together with global industry partners to build a silicon-based world driven by sustainable high-performance computing, available clean power and innovative chip heating."

On March 28, 2022, SAI announced it has joined two international district heating associations, including Euroheat & Power (EHP) which is headquartered in Brussels, Belgium, and the International District Energy Association (IDEA) which is headquartered in Westborough, Massachusetts of the United States. SAI expects to collaborate with global leading energy and energy service companies, utilities operators, heating equipment manufacturers, universities and research institutes with support from the two organizations to promote clean heating and build a sustainable Bitcoin mining infrastructure with SAI's unique liquid cooling and waste heat recovery expertise.

#### About SAI

SAI is a global energy-saving Bitcoin mining operator and a clean-tech company that integrates the bitcoin mining, power and heating industries. SAI uses proprietary liquid cooling and waste heat recovery technology for its digital asset mining machines, which utilizes

waste heat to provide recycled energy and heating to potential customers while lowering mining operating costs. SAI strives to globally become the most energy-efficient digital asset mining operations company, while simultaneously promoting the clean transition of the bitcoin mining, power and heating industries. For more information on SAI, please visit https://sai.tech/.

#### **Forward-Looking Statements**

This Press Release includes "forward-looking statements" within the meaning of the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. SAI's actual results may differ from their expectations, estimates and projections and consequently, you should not rely on these forward looking statements as predictions of future events. Words such as "expect," "estimate," "project," "budget," "forecast," "anticipate," "intend," "plan," "may," "will," "could," "should," "believes," "predicts," "potential," "continue," and similar expressions are intended to identify such forward-looking statements. These forward-looking statements include, without limitation, SAI's expectations with respect to future performance and anticipated financial impacts of the business combination, the satisfaction of the closing conditions to the business combination and the timing of the closing. These forward-looking statements involve significant risks and uncertainties that could cause the actual results to differ materially from the expected results. SAI cautions readers not to place undue reliance upon any forward-looking statements, which speak only as of the date made. SAI does not undertake or accept any obligation or undertaking to release publicly any updates or revisions to any forward-looking statements to reflect any change in its expectations or any change in events, conditions or circumstances on which any such statement is based.

#### **Investor Relations Contact**

r@sai.tech